



FRANCHISE DISCLOSURE DOCUMENT

Sky Zone Franchise Group, LLC A Missouri limited liability company 86 N. University Avenue, Suite 350 Provo, Utah 84601

Phone: 385-482-1020 Facsimile: 310-734-0307 www.skyzone.com legal@skyzone.com

The franchisee will operate a Sky Zone Indoor Trampoline Park featuring trampoline attractions to be used for sports, fitness, and recreational activities, and other active entertainment attractions.

The total investment necessary to begin operation of a Sky Zone Indoor Trampoline Park is \$2,178,0002,330,960 to \$4,722,5005,175,460, depending on the size of the park (parks may range in size from 16,000 to 50,000 square feet). This includes \$530,393811,460 to \$978,3671,609,960 that must be paid to the franchisor or affiliateits affiliates. The total investment necessary to rebrand an existing trampoline park from one of our affiliated brands and begin operation of a Sky Zone Indoor Trampoline Park is \$141,750166,750 and \$428,250463,250. This includes \$29,000 to \$99,500 that must be paid to the franchisor or affiliateits affiliates. If you acquire rights under a Multi-Unit Development Agreement, the total investment necessary to begin operation of the first Sky Zone Indoor Trampoline Park with rights to establish two additional Parks is \$2,381,7002,433,060 to \$3,925,1004,290,060. This includes \$630,393911,460 to \$800,0001,374,960 that must be paid to the franchisor or affiliateits affiliates.

This disclosure document summarizes certain provisions of your franchise agreement and other information in plain English. Read this disclosure document and all accompanying agreements carefully. You must receive this disclosure document at least 14 calendar-days before you sign a binding agreement with, or make any payment to, the franchisor or an affiliate in connection with the proposed franchise sale. Note, however, that no governmental agency has verified the information contained in this document.

You may wish to receive your disclosure document in another format that is more convenient for you. To discuss the availability of disclosures in different formats, contact Mike Revak, <u>PresidentChief Operating Officer</u>, Sky Zone Franchise Group, LLC, at 86 N. University Avenue, Suite 350, Provo, Utah 84601, 385-482-1020.

The terms of your contract will govern your franchise relationship. Don't rely on the disclosure document to understand your contract. Read your contract carefully. Show your contract and this disclosure document to an advisor, like a lawyer or accountant.

Buying a franchise is a complex investment. The information in this disclosure document can help you make up your mind. More information on franchising, such as "A Consumer's Guide to Buying a Franchise," which can help you understand how to use this disclosure document, is available from the

Federal Trade Commission. You can contact the FTC at 1-877-FTC-HELP or by writing to the FTC at 600 Pennsylvania Avenue, NW, Washington, D.C. 20580. You can also visit the FTC's home page at www.ftc.gov for additional information. Call your state agency or visit your public library for other sources of information on franchising.

There may also be laws on franchising in your state. Ask your state agencies about them.

ISSUANCE DATE: April 1222, **20242025**

How to Use This Franchise Disclosure Document

Here are some questions you may be asking about buying a franchise and tips on how to find more information:

QUESTION	WHERE TO FIND INFORMATION
How much can I earn?	Item 19 may give you information about outlet sales, costs, profits or losses. You should also try to obtain this information from others, like current and former franchisees. You can find their names and contact information in Item 20 or Exhibit E.
How much will I need to invest?	Items 5 and 6 list fees you will be paying to the franchisor or at the franchisor's direction. Item 7 lists the initial investment to open. Item 8 describes the suppliers you must use.
Does the franchisor have the financial ability to provide support to my business?	Item 21 or Exhibit F includes financial statements. Review these statements carefully.
Is the franchise system stable, growing, or shrinking?	Item 20 summarizes the recent history of the number of company-owned and franchised outlets.
Will my business be the only Sky Zone business in my area?	Item 12 and the "territory" provisions in the franchise agreement describe whether the franchisor and other franchisees can compete with you.
Does the franchisor have a troubled legal history?	Items 3 and 4 tell you whether the franchisor or its management have been involved in material litigation or bankruptcy proceedings.
What's it like to be a Sky Zone franchisee?	Item 20 or Exhibit E lists current and former franchisees. You can contact them to ask about their experiences.
What else should I know?	These questions are only a few things you should look for. Review all 23 Items and all Exhibits in this disclosure document to better understand this franchise opportunity. See the table of contents.

What You Need To Know About Franchising Generally

<u>Continuing responsibility to pay fees</u>. You may have to pay royalties and other fees even if you are losing money.

<u>Business model can change</u>. The franchise agreement may allow the franchisor to change its manuals and business model without your consent. These changes may require you to make additional investments in your franchise business or may harm your franchise business.

<u>Supplier restrictions</u>. You may have to buy or lease items from the franchisor or a limited group of suppliers the franchisor designates. These items may be more expensive than similar items you could buy on your own.

<u>Operating restrictions</u>. The franchise agreement may prohibit you from operating a similar business during the term of the franchise. There are usually other restrictions. Some examples may include controlling your location, your access to customers, what you sell, how you market, and your hours of operation.

<u>Competition from franchisor</u>. Even if the franchise agreement grants you a territory, the franchisor may have the right to compete with you in your territory.

Renewal. Your franchise agreement may not permit you to renew. Even if it does, you may have to sign a new agreement with different terms and conditions in order to continue to operate your franchise business.

<u>When your franchise ends</u>. The franchise agreement may prohibit you from operating a similar business after your franchise ends even if you still have obligations to your landlord or other creditors.

Some States Require Registration

Your state may have a franchise law, or other law, that requires franchisors to register before offering or selling franchises in the state. Registration does not mean that the state recommends the franchise or has verified the information in this document. To find out if your state has a registration requirement, or to contact your state, use the agency information in Exhibit G.

Your state also may have laws that require special disclosures or amendments be made to your franchise agreement. If so, you should check the State Specific Addenda. See the Table of Contents for the location of the State Specific Addenda.

Special Risks to Consider About *This* Franchise

Certain states require that the following risk(s) be highlighted:

- 1. <u>Out-of-State Dispute Resolution</u>. The Franchise Agreement and Multi-Unit Development Agreement require you to resolve disputes with the franchisor by mediation, arbitration and/or litigation only in <u>MissouriUtah</u>. Out-of-state mediation, arbitration, or litigation may force you to accept a less favorable settlement for disputes. It may also cost more to mediate, arbitrate, or litigate with the franchisor in <u>MissouriUtah</u> than in your own state.
- 2. <u>Supplier Control</u>. You must purchase all or nearly all of the inventory or supplies that are necessary to operate the business from the franchisor, its affiliates, or suppliers that the franchisor designates, at prices the franchisor or they set. These prices may be higher than prices you could obtain elsewhere for the same or similar goods. This may reduce the anticipated profit of your franchise business.

Certain states may require other risks to be highlighted. Check the "State Specific Addenda" (if any) to see whether your state requires other risks to be highlighted.

THE FOLLOWING PROVISIONS APPLY ONLY TO TRANSACTIONS GOVERNED BY_THE MICHIGAN FRANCHISE INVESTMENT LAW

THE STATE OF MICHIGAN PROHIBITS CERTAIN UNFAIR PROVISIONS THAT ARE SOMETIMES IN FRANCHISE DOCUMENTS. IF ANY OF THE FOLLOWING PROVISIONS ARE IN THESE FRANCHISE DOCUMENTS, THE PROVISIONS ARE VOID AND CANNOT BE ENFORCED AGAINST YOU:

- (A) A prohibition on the right of a franchise to join an association of franchisees.
- (B) A requirement that a franchisee assent to a release, assignment, novation, waiver, or estoppel which deprives a franchisee of rights and protections provided in this act. This shall not preclude a franchisee, after entering into a franchise agreement, from settling any and all claims.
- (C) A provision that permits a franchisor to terminate a franchise prior to the expiration of its term except for good cause. Good cause shall include the failure of the franchisee to comply with any lawful provision of the franchise agreement and to cure such failure after being given written notice and a reasonable opportunity, which in no event need be more than 30 days, to cure such failure.
- (D) A provision that permits a franchisor to refuse to renew a franchise without fairly compensating the franchisee by repurchase or other means for the fair market value, at the time of expiration, of the franchisee's inventory, supplies, equipment, fixtures, and furnishings. Personalized materials which have no value to the franchisor and inventory, supplies, equipment, fixtures, and furnishings not reasonably required in the conduct of the franchise business are not subject to compensation. The subsection applies only if: (i) the term of the franchise is less than five years; and (ii) the franchisee is prohibited by the franchise or other agreement from continuing to conduct substantially the same business under another trademark, service mark, trade name, logotype, advertising, or other commercial symbol in the same area subsequent to the expiration of the franchise or the franchisee does not receive at least six months advance notice of the franchisor's intent not to renew the franchise.
- (E) A provision that permits the franchisor to refuse to renew a franchise on terms generally available to other franchisees of the same class or type under similar circumstances. This section does not require a renewal provision.
- (F) A provision requiring that arbitration or litigation be conducted outside this state. This shall not preclude the franchisee from entering into an agreement, at the time of arbitration at a location outside this state.
- (G) A provision which permits a franchisor to refuse to permit a transfer of ownership of a franchise, except for good cause. This subdivision does not prevent a franchisor from exercising a right of first refusal to purchase the franchise. Good cause shall include, but is not limited to:
- (i) The failure of the proposed transferee to meet the franchisor's then current reasonable qualifications or standards.

- (ii) The fact that the proposed transferee is a competitor of the franchisor or subfranchisor.
- (iii) The unwillingness of the proposed transferee to agree in writing to comply with all lawful obligations.
- (iv) The failure of the franchisee or proposed transferee to pay any sums owing to the franchisor or to cure any default in the franchise agreement existing at the time of the proposed transfer.
- (H) A provision that requires the franchisee to resell to the franchisor items that are not uniquely identified with the franchisor. This subdivision does not prohibit a provision that grants to a franchisor a right of first refusal to purchase the assets of a franchise on the same terms and conditions as a bona fide third party willing and able to purchase those assets, nor does this subdivision prohibit a provision that grants the franchisor the right to acquire the assets of a franchise for the market or appraised value of such assets if the franchisee has breached the lawful provisions of the franchise agreement and has failed to cure the breach in the manner provided in subdivision (C).
- (I) A provision which permits the franchisor to directly or indirectly convey, assign, or otherwise transfer its obligations to fulfill contractual obligations to the franchisee unless provision has been made for providing the required contractual services.

* * * *

THE FACT THAT THERE IS A NOTICE OF THIS OFFERING ON FILE WITH THE ATTORNEY GENERAL DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION, OR ENFORCEMENT BY THE ATTORNEY GENERAL.

* * * *

If the franchisor's most recent financial statements are unaudited and show a net worth of less than \$100,000.00, the franchisor must, at the request of the franchisee, arrange for the escrow of the initial investment and other funds paid by the franchisee until the obligations to provide real estate, improvements, equipment, inventory, training, or other items included in the franchise offering are fulfilled. At the option of the franchisor, a surety bond may be provided in place of escrow.

Any questions regarding the notice should be directed to the Department Michigan Dept. of the Attorney General, Consumer Protection Division, Attn: Franchise Section, G. Mennen Williams Building, 1st Floor, 525 West Ottawa St., Lansing, MI 48913 (517-335-7567).

NOTE: NOTWITHSTANDING SECTION (F) ABOVE, WE INTEND TO, AND YOU AGREE THAT WE AND YOU WILL, ENFORCE FULLY THE PROVISIONS OF THE ARBITRATION SECTION OF OUR AGREEMENTS. WE BELIEVE THAT SECTION (F) IS UNCONSTITUTIONAL AND CANNOT PRECLUDE US FROM ENFORCING THE ARBITRATION PROVISIONS.

SKY ZONE FRANCHISE GROUP, LLC FRANCHISE DISCLOSURE DOCUMENT TABLE OF CONTENTS

<u>Franchise Disclosure Document</u> <u>Table of Contents</u>

ITEM—NAME	PAGE
Hem 1 . THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFF	
Item <u>ITEM</u> 2 .BUSINESS EXPERIENCE	
Item ITEM 3 LITIGATION	
Item ITEM 4. BANKRUPTCY	
ItemITEM 5 INITIAL FEES	
HemITEM 6. OTHER FEES	0-
HemITEM 7. ESTIMATED INITIAL INVESTMENT	
HemITEM 8. RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES	
Item ITEM 9 . FRANCHISEE'S OBLIGATIONS	
Item ITEM 10. FINANCING	28 30
HemITEM 11 FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SY	YSTEMS, AND
TRAINING 2831	,
ItemITEM 12. TERRITORY	37 40
ItemITEM 13. TRADEMARKS	39 43
HemITEM 14. PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION	N 4145
HemITEM 15. OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERAT	TION OF THE
FRANCHISE BUSINESS	42 46
HemITEM 16. RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL	43 <mark>47</mark>
HemITEM 17 . RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESO	LUTION THE
FRANCHISE RELATIONSHIP	44 47
<u>ItemITEM</u> 18 <u>. PUBLIC FIGURES</u>	
Item ITEM 19.	10
FINANCIAL PERFORMANCE REPRESENTATIONS	
Item ITEM 20. OUTLETS AND FRANCHISEE INFORMATION	
Item 1 21 FINANCIAL STATEMENTS	
ItemITEM 22 . CONTRACTS	62 70
HtemITEM 23. RECEIPTS	63 70

EXHIBITS:—

- A. A. Franchise Agreement
- B. Multi-Unit Development Agreement
- $\overline{\mathbb{C}}$. C. State Law Addenda
- D. Operations Manual Table of Contents of Manual
- E. List of Franchisees
- F. Financial Statements
- $\overline{\overline{G}}$. G. List of State Agencies/Agents for Service of Process
- H. Form of Purchase Order

- I. Agreement and Conditional Consent to Transfer (including Sample Release of Claims)
- J. Form of Renewal Addendum (including Sample Release of Claims)
- K. Representations and Acknowledgments Statement
- T. State Effective Dates
- $\overline{\overline{M}}$. $\overline{\overline{L}$ -Receipts

Item 1

ITEM 1. THE FRANCHISOR, AND ANY PARENTS, PREDECESSORS, AND AFFILIATES

To simplify the language in this Disclosure Document, "Franchisor", "us", "our", "we" or "SFG" means Sky Zone Franchise Group, LLC, the franchisor. "You," and "your," or "Franchisee" means the person(s) or legal entity to whom we grant a franchise. If the franchise will be owned by (including a corporation, general or limited partnership, or limited liability company, "you" and "your" also means the association, cooperative or other legal entity, and its owners of the corporation, partners of the partnership, or members and manager of the limited liability company and their spouses, as applicable) to whom we grant a franchise. If you are a legal entity, your direct and indirect owners will be required to guarantee your obligations in the form set forth by us (a "Guarantee") and be bound by the provisions of the franchise agreement and other agreements as described in this Disclosure Document.

The Franchisor

We are a limited liability company organized under the laws of Missouri on November 18, 2008. Our principal business address is 86 N. University Avenue, Suite 350, Provo, Utah 84601. We do not do business under any name other than Sky Zone Franchise Group, LLC.

We offer franchises under the names "Sky Zone" and "Sky Zone Indoor Trampoline Park." We began offering franchises in January 2009. As of December 31, 20232024, we had a total of 151there were 120 franchised parks open and operating within the United States. We do not operate businesses of the type that we franchise and do not engage in business activities other than franchising Sky Zone parks and providing services to our franchisees; however, our affiliates own and operate Sky Zone parks as further detailed in Item 20 of this Disclosure Document.

Our agents for service of process are disclosed in Exhibit G to this Disclosure Document.

Our Parent, Predecessors and Affiliates

On or around December 27, 2017, we and our former principal(s) entered a transaction that resulted in CircusTrix Holdings, LLC becoming our parent company ("CTH"). Its principal business address is 86 N. University Avenue, Suite 350, Provo, Utah 84601. CTH is owned by Trampoline Acquisition Holdings, LLC, which is in turn owned by Trampoline Acquisition Parent Holdings, LLC, which is in turn majority-owned by Trampoline Acquisition Corp. (referred to as "TAC"). Each of the companies above CTH has the same principal business address as CTH, except for TAC, which has a principal business address of 1270 Avenue of the Americas, 31st Floor, New York, NY 10020. TAC is owned by Palladium Equity Partners IV LP, which has the same principal business address as TAC. None of these companies offer franchises in any line of business or provide products or services to our franchisees. We have no predecessors.

Our affiliate Sky Zone, LLC ("**Licensing**") owns the Intellectual Property (defined below) and has granted us a worldwide license to grant franchises using the Intellectual Property. Licensing is also the only approved supplier for Attractions, <u>JumpSocksSkySocks®</u>, stickers, wristbands, and balls. Licensing has never offered franchises in any line of business. Licensing's principal business address is 86 N. University Avenue, Suite 350, Provo, Utah 84601.

Our affiliate Rockin' Jump Franchise, LLC ("**RJF**") offered franchises for indoor trampoline parks and entertainment facilities under the ROCKIN' JUMP® mark from June 2014 to February 2018 and from September 2020 to March 1, 2023. As of December 31, 20232024, RJF had a total of 12there were three franchised facilities open and operating within the United States.

Our affiliate House of Trix, LLC ("HOT") began offering franchises for indoor trampoline and entertainment facilities under the HOUSE OF TRIX® mark from October 2017 until March 2018 and under the DEFY^{TM®} mark from October 2018 to March 1, 2023. As of December 31, 20232024, HOT had 5there were two franchised facilities and, through its HOT's affiliates, 32eight company-owned facilities, open and operating under the DEFY® mark within the United States. DefyDEFY parks were designed to be more extreme than Sky Zone parks and use different trade dress that is "edgier" in nature. DefyDEFY parks also tended to occupy larger premises as compared to Sky Zone parks.

As of January 1, 2023, Sky Zone Franchise Group, LLCRJF and HOT no longer offers House of Trix, LLC ("HOT" or "DEFY")offer Rockin' Jump and Defy franchises respectively.

Our affiliate Loscann Insurance Company ("**Loscann**") is a Cayman Islands company and a wholly-owned subsidiary of CTH formed in 2024 to provide general liability insurance coverage to franchisees under the self-insured retention component of the Master Insurance Program, effective as of September 1, 2023. Loscann has never offered franchises in any line of business.

The Business

Sky Zone Indoor Trampoline Parks offer—all trampoline and walled Attractions to be used for sports, fitness and recreational activities using the System and Intellectual Property. "Attractions" mean the trampoline playing field—fields (including the trampolinetrampolines, the padding, and nets), and other related recreational equipment, including but not limited to slides, ziplines, ball courts, and foam pits. "Sky Zone Indoor Trampoline Park" means any facility that is operated under the System and Intellectual Property.

Our "**System**" includes a specially developed method of operating a Sky Zone Indoor Trampoline Park using the Intellectual Property, as well as selling other services (such as food, beverages, and parties) and products (including merchandise bearing the Marks), using certain procedures and methods, site evaluation criteria, layouts, advertising, sales and promotional techniques, personnel training, trade secrets and any other matters relating to the operation and promotion of a Sky Zone Indoor Trampoline Park, as they may be periodically changed, improved, modified and further developed by us or our affiliates. The mandatory and suggested specifications, standards, operating procedures (including safety standards), and rules that we prescribe from time to time for the operation of Parks are sometimes referred to as our "**Methods of Operation**."

"Intellectual Property" means the Marks, Patents, Copyrights patents, Copyrighted Materials, and any of our trade secrets and know-how. "Marks" means the service marks, trademarks, trade dress, trade names and copyrights and all configurations and derivations, as may presently exist, or which may be modified, changed, or acquired by us or our affiliates, in connection with the operation of the Sky Zone Indoor Trampoline Parks. The Marks include "Sky Zone," "SZ and design," and "Skyrobics".

You will do business under the fictitious or assumed name of "Sky Zone" or "Sky Zone Indoor Trampoline Park," or any other name that we decide to use in the future. The Sky Zone Indoor Trampoline Park you will operate according to the terms of the Franchise Agreement is referred to as the

"Park." Sky Zone Indoor Trampoline Parks range in size but are typically between 16,000 to 50,000 square feet and feature two-(2) or more Attractions. Regardless of the number of Attractions, each Sky Zone Indoor Trampoline Park will contain party rooms and a concession area. All your employees must be thoroughly screened before hiring, including conducting annual criminal background checks-for erimes against minors.

This Disclosure Document describes our two franchise programs:

- 1. Single Unit Franchise Program. If we approve you as a franchisee, you must sign a Franchise Agreement, in the form attached as Exhibit A ("Franchise Agreement"), to operate a single Sky Zone Indoor Trampoline Park.
- 2. Development Program. If you elect to participate in, and are approved for, the Development Programdevelopment program, you will execute a Multi-Unit Development Agreement (the "MUDA") in the form attached as Exhibit B. Under the MUDA, we will assign you a territory (a "Development Area") within which you must open and operate a designated number of Sky Zone Indoor Trampoline Parks ("Businesses") within the specified periods of time as set forth in AttachmentAppendix A to the MUDA ("Development Schedule"). In no event will you sign a Franchise Agreement for any UnitPark until we have complied with any applicable waiting periods prescribed by law. You must sign a Franchise Agreement for your first UnitPark at the same time you sign the MUDA, and within the times specified in the Development Schedule, you must execute a Franchise Agreement for each additional Park in accordance with your Development Schedule.

Under the Franchise Agreement and MUDA, each of your shareholders, partners, or members (and their shareholders, partners, or members if they are an entity) who has a 10% or greater interest <u>or such other persons as we may designate (including owners' spouses)</u>, must sign a Guarantee of your monetary obligations and all other obligations under the respective Franchise Agreement and MUDA (as applicable). Additionally, your Responsible Person (as <u>further defined and described in Item 15</u> below) must execute the Guarantee.

If you currently operate a trampoline park in one of our affiliated franchise systems, Rockin' Jump or Defy, and we approve you as a franchisee, you will be required to sign a Franchise Agreement and rebrand your trampoline park as a Sky Zone Indoor Trampoline Park.

Market and Competition

The services offered by the Sky Zone Indoor Trampoline Parks are used by people of all ages and are not limited to any specific submarket. Your Park will have to compete with other recreation and entertainment facilities, including, other indoor centers featuring trampolines. However, other Other franchises will likely also operate businesses providing entertainment facilities that will compete with your guest base.

Industry Regulations

There may be regulations specific to the operation of a Sky Zone Indoor Trampoline Park in your state that, among other things, require you to maintain a certain ratio between <u>your</u> supervisory employees and the number of jumpers on the Attractions or in <u>theeach</u> party room. You must comply with all local, state, and federal health, safety, and sanitation laws and regulations.

You should consult with your attorney and local, state, and federal government agencies before investing in a Sky Zone franchise to determine all the legal requirements that you must comply with and consider their impact on you and the cost of compliance. Building codes and requirements vary in different jurisdictions and it is important for you and your architect to be aware of and comply with all local laws as well as the federal laws, including the Americans with Disabilities Act. You may also be subject to certain health and safety requirements as well as licensing requirements in teaching and supervising children. Many states and municipalities have laws and regulations regarding fitness center contracts with guests, operations and licenses. Some state and local laws may regulate the duration and terms of membership contracts length and terms, advertising and limitations on agreements and pre-opening sales and advertising. You may also have to obtain a bond to protect any pre-paid membership fees you collect and there may be buyer's remorse cancellation rights and other types of cancellation rights. There may be laws requiring you to have an employee at your Park who is certified in basic cardiopulmonary resuscitation or on the use of an automated external defibrillator. There may be laws requiring you to have employees at your Park who are certified or otherwise trained to prepare and serve food. There may be a requirement that you must have certain types of first aid equipment on the premises, such as an automated external defibrillator. You should additionally check with your local attorney for advice on complying with applicable law-before you purchase a franchise and during the operation of your Park. You must investigate and satisfy and stay current on all local, state, and federal laws and regulations since they vary from place to place and can change over time.

You must also comply with all Sky Zone required <u>American Society for Testing and Materials</u> (<u>"ASTM"</u>) and other standards as they relate to regulation and safety of indoor trampoline parks. We reserve the right to require additional ASTM upgrades if such standards change.

Item 2

ITEM 2. BUSINESS EXPERIENCE

Shawn Hassel: Chief Executive Officer, CTH

Mr. Hassel has served as Chief Executive Officer for CTH since January 2024September 2023. Before that, Mr. Hassel was Interim CEO of CTH from July 2023 to September 2023 to December 2023. From January 2023 to July 2023, Mr. Hassel served as Chief Financial Officer for us and CTH. Additionally, Mr. Hassel has been Co-Founder and Managing Partner of Areté Holdings LLC and The Areté Group since January 2020, and has splitsplits his time between New York, New York and Park City, Utah. Since August 2015, Mr. Hassel has also been Co-Founder, Managing Partner, and Board Member of Bestige Holdings, LLC in Park City, Utah. Mr. Hassel has also been a Director of the following: Growve (June 2023 to present; St. Petersburg, Florida); Ryze Renewables (August 2022 to present; Irvine, California); Camofire/Black Ovis (April 2022 to present; Salt Lake City, Utah); Infinity Trading & Solutions, LLC (November 2020 to February 2022; Chandler, Arizona); Consolidated Glass (January 2020 to January 2022; New Castle, Pennsylvania); Standard Furniture/Albany Furniture (October 2019 to June 2023; New Albany, Mississippi); National Auto Performance Warehouse (June 2019 to present; Hialeah, Florida); National Waste (February 2017 to present; Bay Shore, New York); FINCA Impact Finance (March 2013 to present; Washington, DC); and Kona Grill Inc. (March 2019 to February 2020; Scottsdale, Arizona).

Mike Revak: President, Chief Operating Officer, CTH and SFG

Mr. Revak has served as Chief Operating Officer for us and CTH since March 2025. Mr. Revak previously served as our President sincefrom February 2024. Mr. Revak has also served to March 2025 and as Chief Business Officer of CTH sincefrom July 2023 to March 2025. From October 2021 to July 2023, Mr. Revak served as Senior Vice President of Franchise and Business Development for CTH. From October 2012 to October 2021, Mr. Revak was the Chief Operating Officer of Rockin' Jump Franchise, LLC, in Dublin, California.

Karen Luey: Chief Financial Officer, CTH and SFG

Ms. Luey has served as Chief Financial Officer for us and CTH since May 2023. Ms. Luey previously served as Chief Financial Officer for The Mina Group, LLC in San Francisco, CA from April 2018 to May 2023. Ms. Luey was also on the Board of Directors for Del Taco Restaurants, Inc. in Lake Forest, California from July 2021 through March 2022.

Yogi Jashnani: Chief Revenue Officer, CTH and SFG

Mr. Jashnani has served as Chief Revenue Officer for us and CTH since November 2023. Before that, Mr. Jashnani served as Chief Commercial Officer at Ideal Image in Tampa, Florida from December 2019 to March 2023. Mr. Jashnani previously served as Senior Vice President, Marketing, Insights and Analytics at Advance Auto Parts in Raleigh-Durham, North Carolina from June 2017 to November 2019.

Eric Taylor: Chief Development Officer, CTH and SFG

Mr. Taylor has served as Chief Development Officer for us and CTH since March 2023. Mr. Taylor previously served as Vice President of Development at Tijuana Flats Restaurants, LLC in Orlando, Florida from December 2019 to March 2023, and the President of Baxter Wilhelm Franchise Consulting in Dallas, Texas, from March 2018 to March 2020.

Joe Tenczar: Chief Information Officer, CTH and SFG

Mr. Tenczar has served as Chief Information Officer for us and CTH since March 2023. Mr. Tenczar previously served as Chief Strategy Officer and Chief Information Officer for Sonny's Franchise Company in Winter Park, Florida, from January 2014 to March 2023, and Founder and President of Restaurant CIOs Consulting in Orlando, Florida, from July 2017 to December 2023.

Amanda Ellis: Franchise Sales Manager

Ms. Ellis has served as our Franchise Sales Manager since August 2022. Before joining us, she was Director of Onboarding and Franchise Coordinator at Premier Martial Arts in Knoxville, TN, from July 2013 to August 2022.

Item 3

ITEM 3. LITIGATION

Pending

None.

Concluded

Cherokee Gray Eagle IP, LLC and Rebounderz Franchise and Development, Inc. vs Sky Zone, LLC, Sky Zone Franchise Group, LLC, et al., United States District Court for the Middle District of Florida, Orlando Division, Case No. 6:18 cv 355 Orl 40EJK. On March 8, 2018, the Plaintiff filed a complaint against us, Sky Zone, LLF, RPSZ Construction, LLC, as well as certain Sky Zone franchisees, alleging infringement of United States Patent No. 8,764,575, a design patent for an angled wall unit for trampolines and seeking an injunction and damages and legal fees and costs. The parties reached a settlement without any admission of infringement and the cases have all been dismissed with prejudice on January 28, 2020.

Barclay Poole v. Jeffrey Platt and Sky Zone Franchise Group, LLC, Circuit Court for the State of Missouri, County of Cole, Case No. 16AC-CC00572. On December 29, 2016, Plaintiff Barclay Poole filed a petition against Defendants Jeffrey Platt and Sky Zone Franchise Group for breach of contract and breach of fiduciary duty seeking damages. Plaintiff alleged that: (i) Defendants breached the Sky Zone Franchise Group Operating Agreement by issuing certain equity interests in Sky Zone Franchise Group without the required consent of the members of Sky Zone Franchise Group; (ii) Defendant Jeffrey Platt breached his fiduciary duties owed to Plaintiff by preventing Rick Platt (another member of Sky Zone Franchise Group) from serving as manager, ceding all management authority to a board of managers, using company funds to pay a law firm that provided advice to remove Rick Platt as manager, using company funds to pay for the legal funds in defense of removing Rick Platt as manager, issuing equity interests in Sky Zone Franchise Group to non-members without member consent, inappropriately issuing interests to himself and his designees, using company funds to pay a settlement to Brent Platt (another member of Sky Zone Franchise Group), making decisions that were not in the best interests of Sky Zone Franchise Group, making expenditures in excess of \$25,000 without the consent of the members, and commencing a pre-emptive offering to attempt to correct the breaches of the Operating Agreement, which resulted in damages to Plaintiff. In May 2018, the parties reached a reached a settlement of all claims whereby we paid to Plaintiff the sum of \$828,426, in exchange for a dismissal of all claims with prejudice and mutual releases.

Ottober 11, 2018, Ottway II, LLC, a Sky Zone franchise Group, LLC, AAA Arbitration, Case No. 01-18-0003-7867. On October 11, 2018, Ottway II, LLC, a Sky Zone franchisee located in Daytona Beach, Florida, filed a demand for arbitration with the American Arbitration Association, in Chicago, Illinois. In its demand, Ottway alleged that (1) we unfairly competed with Ottway through one of our affiliate brands that also operates a trampoline park in Daytona Beach, Florida, (2) we used Ottway's confidential and proprietary information for the benefit of our affiliate that operates a trampoline park in Daytona Beach, Florida, and (3) we used or allowed others to use or duplicate our knowledge, know-how and expertise to directly compete with Ottway. Without admitting any liability, on February 14, 2020, we and Ottway entered into a Settlement Agreement and related agreements, under which Ottway released us from all claims, we terminated Ottway's franchise agreement, and one of our affiliate's acquired all Ottway's assets for \$1,112,500 and assumed the lease covering Ottway's Sky Zone Indoor Trampoline Park in Daytona Beach, Florida.

Disclosure Regarding the House of Trix, LLC Franchise Program

The Commissioner of Financial Protection and Innovation v. House of Trix, LLC, Consent Order Dated February 21, 2023. As a result of an inquiry by the California Commissioner of Financial Protection and Innovation ("Commissioner") into the franchise sales activities of our affiliate, House of Trix, LLC ("HOT"), the Commissioner found that in late 2021, during a period in which HOT was not registered to sell franchises, HOT sold an unregistered franchise in California. In relation to the Commissioner's inquiry, it also found that HOT made an untrue statement to the Commissioner when it told the California Corporations Code. In fact, HOT had sold an additional unregistered franchise in late 2021, which had previously been mutually terminated. Consequently, as part of the Consent Order, HOT agreed to pay an administrative penalty of \$5,000, deliver Notices of Violation to the two franchisees affected by its violations, and deliver a copy of the Consent Order to all California franchisees who purchased a franchise from HOT from April 22, 2020, to January 27, 2023.

Other than the 4 matters as set forth above, no litigation is required to be disclosed in this Item.

Item 4

ITEM 4.BANKRUPTCY

No bankruptcy information is required to be disclosed in this Item.

ITEM 5. INITIAL FEES

Franchise Agreement

Initial Franchise Fee You will be required to pay us an initial franchise fee of \$75,000 (the "**Initial Franchise Fee**") upon signing the Franchise Agreement. The Initial Franchise Fee is uniform and not non-refundable.

If you are rebranding an existing trampoline park from one of our affiliated brands, Rockin' Jump[®] or Defy[®], you will not be required to pay us an Initial Franchise Fee to rebrand.

Other Amounts Payable in Connection with a Single Park

Veterans' and Active-Duty Military Discount. We offer a 20% discount on the Initial Franchise Fee for veterans and active-duty members of the United States armed forces who will hold at least a 51% ownership interest in the Park to be developed and operated under a Franchise Agreement. We reserve the absolute right to determine whether you qualify for this discount and we may deny this discount or modify this discount at any time for any reason. This veterans' and active-duty military discount cannot

be combined with any other discounts, including the discounts incorporated into the Development Fee described below.

<u>Other Initial Payments</u>. Before you open your Park, you may be required to pay us or our affiliates the fees described below.

<u>Attractions, Inventory, and Insurance.</u> You must pay us or our affiliates for the Attractions equipment and(including their installation), all on-court sports equipment (e.g., balls), jump wristbands, <u>JumpSocksSkySocks®</u>, mobile device applications (SkyApp) and other items we designate for use or resale at Parks. We and our affiliates are the only approved suppliers for these products and services. <u>Additionally, you must participate in the Master Insurance Program developed by us and our affiliate (as further described in Items 7 and 8 of this Disclosure Document) pursuant to which you will pay for your allocated share of the total system-wide <u>Master Insurance Program costs.</u> The amounts for these items are nonrefundable and will range from \$454,893 to \$898,367, payable to SFG and its <u>affiliateaffiliates</u>, and are estimated to range from \$728,500 to \$1,527,000, depending on the size of your Park.</u>

<u>Technology Fees.</u> You must pay us the monthly Technology Fee (currently, \$1,480 per month) beginning two months prior to the anticipated opening of your Park.

<u>Site Development Fee.</u> No later than 10 days before you open your Park for business, you must pay us a non-refundable site development fee equal to \$5,000 in the form of a lump sum payment.

<u>Site Visits.</u> We may require you to reimburse us for our reasonable expenses, including the costs of travel, lodging and food incurred in site evaluation for each visit we make at your request. We estimate our reasonable expenses related to site selection will range from \$500 to \$5,000. If we require you to pay these expenses, the amounts will be <u>uniform and nonrefundable</u>non-refundable.

<u>Space Plan Fees.</u> We will provide three space floor plans to you at no additional cost. If additional space plans are necessary, you will be required to pay us \$350 for each additional space plan.

Multi-Unit Development Agreement

<u>Development Fee.</u> If you are approved and we grant you rights to develop multiple Parks within a designated Development Area as described more fully in <u>Item 1 of this Disclosure Document</u>, you will be required to pay us a lump-sum fee amounting to the sum of: (i) \$75,000 for the first Park you are granted the right to develop; and (ii) \$50,000 for the second Park and each additional Park we grant you the right to develop under the MUDA (collectively, the "**Development Fee**"). For example, the Development Fee for rights to develop three (3)—Parks will be \$175,000. The Development Fee is paid to us upon <u>the</u> execution of the MUDA. It is uniform and non-refundable.

Item 6

ITEM 6. OTHER FEES

TYPE OF FEE*	AMOUNT	DUE DATE	REMARKS
Royalty Fee (Notes 1 and 2)	6% of Gross Sales	Payable twiceonce per month on the 16 th and firstthird day after preceding month ends	Fees and sales report due. Payments must be made via electronic funds transfer (EFT). See Note 1 and 2
Advertising Ad Fee (Notes 1 and 2)	Currently, 2% of Gross Sales	Payable twiceonce per month on the 16 th and firstthird day after preceding month ends	Payments must be made via electronic funds transfer (EFT). We have the right to increase your Advertising Fee to an amount equal to up to 2.5% of Gross Sales. See Note 1 and 2
On line Marketing ToolLocal Advertising Fee (Note 3)Funding Requirement	Then current fee charged by our approved or designated supplier \$12,000 during first month of Park operations Currently, \$20 per month per user 4% of Gross Sales thereafter Amount of shortfall (as applicable; see Note)	Monthly	We have a mandatory marketing platform for all approved marketing resources. This fee is charged based on the number of users at a given Park. See Note 3
Extension Tech nology Fee (Note 4)	The then current initial franchise fee Currently, \$1,480 per month	Upon our approval of extension request Monthly	See Note 4
Transfer Fee Under [Franchise Agreement]	50% of the then-current Initial Franchise Fee \$2,500 for select transfers	Prior to transfer Transfer	Payable by transferee when you transfer your Franchise Agreement, the assets of your Park or your ownership. See Note 5
Transfer Fee (Multi-Unit Development Agreement)	\$2,500	Prior to Transfer	See Note 6
Transfer Fee Deposit	<u>\$15,000</u>	Prior to Transfer; refunded (less any due amounts) within 60 days following transfer closing date	See Note 7
Successor Fee Under the	25% of the then-current Initial Franchise Fee	At the time execution of the then	Payable when you renew the Franchise Agreement. See

TYPE OF FE	TYPE OF FEE* AMOUNT DUE I		DUE DATE	REMARKS
Franchise Agreement Fee		exec	rentUpon cuting successor chise agreement	Note 8
Extension Fee (Franchise Agreement)	100% of then-current Initial Franchise Fee		n our approval o nsion request	See Note 9
Extension Fee (Multi-Unit Development Agreement)	50% of Development Fee		n our approval o nsion request	See Note 10
Space Plan Fee	\$350 per space plan		n delivery of the	See Note 11
Relocation Expenses	Actual costs	<u>As in</u>	ncurred	See Note 12
Call Center Program Fee	Not currently charged by us, by estimated to be from \$1,000 to \$1,400 per month		<u>ithly</u>	See Note 13
Iditional Currently, we reserve the right to charge \$500 per day we requote 5)		Upon re we requ	ire	This is for additional training we may periodically provide or require.
	Currently, we reserve the right to charge \$500 per day	We requ	ire	This is for additional assistance that you need or request.
ote 6)	Our costs and expenses of the audit, including salaries, professional fees, travel, meals and lodging, plus any understated amount with interes	Upon de	: 4 :	Payable if audit or review shows an understatement of Gross Sales for the audited or reviewed period of 2% or more.
Interest on Late Payments (Note 6) Lesser of 18% annually or maximum legal rate we are able t charge (e.g., based on where the Park is located) subject topermitted by applicable lawslaw		ole to payı	all overdue ments <u>Upon</u> and	Payable on all overdue amounts. Interest accrues from the original due date until payment is received in full. See Note 14
Overdue Supplier Actual costs Payments		<u>Upo</u>	n demand	See Note 15
Non-Compliance Fee \$250 for each day you remain out of compliance. \$2,500 per violation for failure to obtain guest waivers		out Upo	n demand	If you are found in violatio of Franchise Agreement. Se
_	\$2,500 per violation for failure	e to		<u>Note 16</u>

TYPE OF FE	EE* A	MOUNT		DUE DATE		F	REMARKS
Alternate Supplier Approval	the test					oplier to us oroval.	for our
Product Purchases and Installation of Attractions (Note 7)	Varies based on the products ordered (e.g., Attractions (including installation), balls, stickers, wristbands and JumpSocks)		set forth in the		goo		include cost of shipping and
Inspection and Compliance Reimbursemen	meals and hou	ts, including trave ir wage<u>living</u>	l ,	Upon demand		Park we d	
Online POS- Related Fees	Then-current fees approved supplies sale (POS) system software. Currently, the PO are \$600 - \$900 n	r for point of and related	As	incurred	sur	yable to our pplier for Po tware.	
y Suite (Note 8)Trainin g and Assistance Fees week	nen-current chnology or cense fee we narge in onnection with e any echnology Suite e develop and/or esignate for use in onnection with e System	As arranged.	tec sk but rig fut lf v des req im tec us	chave not rently developed designated a chnology Suite to adle certain a POS related hnology within Y ZONE Parks, we reserve the ht to do so in the ure. We develop or signate such a chnology Suite in future, we may uire you to plement that hnology and pay or our approved oplier a recurring	or a	on request is we uire	See Note 18

TYPE OF F	EE*	AN	MOUNT		DUE DATE		R	REMARKS
	·			Tec acc Cui the \$50 trav	rently, we reserve right to charge on per day, plus vel and living benses			
Taxes (Note 9)	Actı	ual costs		Up	on demand			
Annual Convention <u>F</u>	<u>ees</u>	UpCurrently, uperson, plus trans relatedliving e			15 days before convention begins. I not paid by the required deadline, the fee will be automatically debite from your bank account	he	from each the conver the conver include tra	ne representative park must attend ntion. The cost of ntion does not evel, lodging, meal expenses for the nee Note 19
Maintenance <u>Costs</u>		Actual costs for failing or refusing to correct deficiencies at your Par			<u> </u>		See Note 20	0
Product Purchand Installations Attractions	on of	Varies based or ordered (e.g., A installation), ba wristbands and	Attractions (includable, stickers,	ling	As incurred and as a forth in the applicate Purchase Order (a forth of which is attached this Disclosure Document as Exhiber H)	ole orm I to	See Note 2	1
Alternate Supplier Approval Cos	<u>sts</u>	proposed suppl of testing prop	sts of inspecting lier and actual cosposed products or posed service or er	<u>ts</u>	Upon demand		See Note 22	2
Space Planning Fee			-	on delivery of the ace Plan	Spa Eac afte	nce Plans at th additiona or the 3 rd , if t you \$350	de the first 3 no cost to you. al floor plan necessary, will per Space	
Liquidated damages	<u> </u>		terr	mediately upon mination by us for se or you without se				

TYPE OF FEE*		AMOUNT	DUE DATE	REMARKS	
	rem Ter	naining months in the Initial			
Master Insurand Program Costs	ce	Total Cost of Risk (defined in Item 7), currently between 2.962.0% and 8.6910.0% of Gross Sales, but subject to recalculation every 6six months and could increase or decrease	Monthly	All franchisees are required to participate in the Master Insurance Program (defined in Item 7). You must pay us or our affiliate via electronic funds transfer (EFT) a percentage of Gross Sales monthly to participate in the Master Insurance Program. This amount will depend on several factors, including revenue, attendance, location, losses, compliance metrics (such as timely incident reporting, number of incidents reported, and results and resolutions of inspections), and other factors and risks we periodically consider. Your Total Cost of Risk may be higher or lower than this estimated range. This fee does not include the cost of purchasing other required insurance coverage from third parties, as described in Item 8. See Note 23	
Third-Party Insurance Coverages		<u>Actual costs</u>	<u>Upon demand</u>	See Note 24	
Quality Control Programs	1	Actual costs	<u>As incurred</u>	See Note 25	
Advertising Cooperatives & Special Promot Fees	_	As required	As incurred	See Note 26	
Audit		Understated amounts with interest Our costs and expenses of the audi including professional fees, travel, meals and lodging (as applicable)		See Note 27	
Management Fe	<u>ee</u>		<u>As incurred</u>	See Note 28	

TYPE OF FEE*	AMOUNT	DUE DATE	REMARKS
	50% of Gross Sales; travel and living expenses		
<u>Taxes</u>	Actual costs	<u>Upon demand</u>	See Note 29
Indemnification	Amount of our liabilities, fines, losses, damages, costs and expenses (including reasonable attorneys' fees)	Upon demand .	Payable if we incur losses due to your breach of the Franchise Agreement or any other action or inaction by you or any other person (including, employees, agents, officers) relating to your Park. See Note 30
Costs and Attorneys' Fees for Review of Park Waivers	Actual costs and expenses	As incurred. <u>Upon</u> <u>demand.</u>	If applicable to your state, all franchisees located within your state will share the costs for the attorney to review the Park waiver for your state. See Note 31
Arbitration and Proceeding Costs	Reasonable costs and expenses, including attorneys' fees	Upon conclusion of arbitration or other proceeding	See Note 32
<u>Liquidated</u> <u>Damages</u>	Amount equal to the greater of: (i) \$100,000; or (ii) the combined monthly average of Royalty Fees, Ad Fees, and other fees due pursuant to the Franchise Agreement paid by you during the 12 months preceding the date of termination multiplied by the number of remaining months in the Initial Term after termination (or multiplied by 12, if the Park had been open for less than 12 months upon the date of termination)	Immediately upon termination by us for cause or you without cause	See Note 33
Reimbursement for Other Obligations and Duties	Actual costs and expenses	<u>Upon demand</u>	See Note 34
Late Report Fee	\$100 per instance, plus \$100 per day or \$100 per month you remain out of compliance (as applicable)	As incurred	See Note 35

^{*} All Except as otherwise noted, all fees are imposed non-refundable, uniformly imposed, and collected by and payable to us or our affiliates, non-refundable, and uniformly imposed. Your costs for certain items listed above may differ depending on the suppliers used, local costs, and other factors.

You will be required to establish a designated bank account from which we or our authorized designee will be authorized to withdraw any amounts due to us or our affiliates, including wire transfers or via electronic transfer of funds ("EFT"). You will be responsible for all costs and expenses of establishing and maintaining your designated bank account, including any transaction fees and wire transfer fees. You must ensure that sufficient funds are available in your designated bank account at all times to cover our withdrawals.

Note 1: <u>Amount of Royalty Fee and AdvertisingAd Fee.</u> You must pay us a Royalty Fee equal to 6% of Gross Sales and you will pay us 2% of Gross Sales for an an Advertising Ad Fee. Payments must be made via electronic funds transfer (EFT). We have the right to increase your Ad Fees to an amount equal to up to 2.5% of Gross Sales.

Timing and Calculation of Royalty Fee and AdvertisingAd Fee. Royalty Fees and Note 2: Advertising Ad Fees are due and payable twiceonce per month on the 16th and the first third day after the preceding month ends. If the payment date falls on a holiday, the fees will be due on the next business day. Any payment or report not received by us by the date due will be deemed overdue. "Gross Sales" means the total amount of all sales of products, services, merchandise, programs sold from, through, or in connection with the Park, whether for cash, on credit, barter or otherwise, but not including applicable sales, use or service taxes. Gross Sales additionally exclude refunds that are provided to customers (not including chargebacks). You will comply with the procedures specified in the Operations Manual (further described in Item 11 of this Disclosure Document) or as otherwise communicated for any electronic funds transfer program and shall perform the acts and sign the documents, including authorization forms that we, your bank and our bank may require to conduct payment by electronic funds transfer, including authorizations for us to initiate debit entries and/or credit correction entries to a designated checking or savings account for payments of Royalty Fees and Advertising Ad Fees and other amounts, including interest payable to us. In addition, you will pay all costs associated with using an electronic funds transfer payment program. If you fail to timely report to us in accordance with the procedures set forth in the Franchise Agreement and in the Operations Manual, in addition to any applicable late charge, we have the right, but not the obligation, to debit from your account an estimated amount equal to the fees due and payable to us according to the most recent reports you sent to us.

Note 3: Online Marketing Tool Advertising Fee. This monthly fee is for the Online Marketing Tool, which is used to host Sky Zone marketing and advertising collateral such as brochures, flyers, direct mail, and print advertising pieces.

Note 4:

Note 3: Local Advertising Funding Requirement. In addition to Ad Fees and the Grand Opening Marketing Expense (as further described in Items 7 and 11 of this Disclosure Document), you must spend at least \$12,000 on local advertising and promotional activities in your Protected Territory (see Item 12 of this Disclosure Document) during the first month following the opening of your Park. Thereafter, you must spend a minimum of 4% of your Park's Gross Sales per month on local advertising. If you fail to spend (or prove that you spent) the Local Advertising Funding Requirement in any month, we may require you to pay us the shortfall as an additional Ad Fee or to pay us the shortfall for us to spend on local marketing for your Park. In addition, we may require you to pay the Local Advertising Funding Requirement to us to administer your local advertising if, in our business judgment, we determine that: (i) your Park is under-performing, (ii) our participation is appropriate or necessary, or (iii) you have failed to comply with the Local Advertising Funding Requirement. If we require you to pay the Local Advertising Funding Requirement to us, you must pay it in the same manner and at the same time as the Royalty Fees.

- Note 4: Technology Fee. We require you to pay a fee to us, or a service provider we designate (which may be one of our affiliates), for technology-related services, including, but not limited to, website or email hosting, help desk support, software or website development, enterprise solutions, point-of-sale systems, and other services associated with your Technology System (the "Technology Fee"). The Technology Fee is currently \$1,480 per month and payable in the same manner as the Royalty Fee, beginning two months prior to your Park's opening. The Technology Fee may be increased or decreased by us based upon changes in amounts billed to us from service providers upon 30 days' prior written notice to you; provided, however, the Technology Fee will not be more than our costs plus 20% during the term of your Franchise Agreement. Notwithstanding our collection of the Technology Fee, you may be required to purchase hardware, software, or other components of the Technology System from required vendors. The Technology Fee includes two email accounts; we reserve the right to charge our costs to you if you request, and we agree to provide, additional accounts.
- Note 5: Transfer Fee (Franchise Agreement). You must pay us a transfer fee if you sell or transfer ownership of your Park, the Franchise Agreement, or a controlling interest in you (if you are an entity). You must pay us a transfer fee equal to fifty percent (50%) of our then-current initial franchise fee for new franchises; provided, however, if the transfer is among your existing owners (if you are an entity) or does not involve transfers in the Franchise Agreement, the Park, or a controlling interest in you (if you are an entity), you must instead pay us \$2,500 for administrative costs and expenses we incur in connection with documenting and otherwise processing such transfer, including our reasonable legal fees (which includes in-house counsel fees). You must pay us this transfer fee in a lump sum by wire transfer at the time you sign the conditional consent to transfer (attached as Exhibit I to this Disclosure Document). If we terminate our conditional consent to the transfer or the transferee's franchise agreement for certain reasons (for instance, if the transfer does not occur or the transferring parties fail to meet the conditions to our consent), and the transferring parties sign a general release, then we will refund the transfer fee; however, if the transferee has already attended any portion of initial training, we will only refund 50% of the transfer fee.

You do not have to pay a transfer fee if you transfer your individual interest in the Franchise Agreement to a corporation, limited liability company, partnership or similar entity in which you maintain management control and of which you own and control 100% of the equity and voting power of all issued and outstanding capital stock or membership interests.

- Note 6: Transfer Fee (Multi-Unit Development Agreement). The development rights set forth in the Multi-Unit Development Agreement cannot be transferred. If you are an entity, we will not unreasonably withhold our consent to the transfer of any direct or indirect ownership interests in you which are less than or equal to 49% of your ownership interests. You must pay us a transfer fee equal to \$2,500 for administrative costs and expenses we incur in connection with documenting and otherwise processing such transfer, including our reasonable legal fees (which includes in-house counsel fees).
- Note 7: Transfer Fee Deposit. In the event of a transfer, you must pay us a fee deposit of \$15,000 when you sign the conditional consent to transfer. We will refund the deposit to you, less any amounts which may be due under the Franchise Agreement or any other agreement between you and us (or your affiliates and our affiliates), including but not limited to insurance deductibles or other amounts due pursuant to the Master Insurance Program or Captive SIR Program (as described in Items 7 and 8 of this Disclosure Document), within 60 days following the effective date of the transfer or the date on which you and the transferee have complied with all terms in our agreement and conditional consent to transfer, whichever is later.
- Note 8: Successor Franchise Fee. You must pay a non-refundable successor franchise fee equal to twenty-five percent (25%) of our then-current initial franchise fee for new franchises, in the form of a

<u>lump sum payment by wire transfer or via EFT from your designated bank account upon executing a successor franchise agreement.</u>

Note 9: Extension Fee (Franchise Agreement). You must lease, sublease, or purchase the premises for your Park within nine months after signing a Franchise Agreement. You may request a three (3)—month extension of this time frame to lease, sublease or purchase the Location by sending us a written request at least thirty (30) days before the deadline ("First Extension Period"). We may grant the First Extension Period, without your payment of any extension fee, if we believe you have engaged in a good faith effort to find a Location premises for your Park. If no less than 30 days prior to the expiration of the First Extension Period, you request an additional extension of time to lease, sublease or purchase the Location premises for your Park ("Additional Extension Period"), we shall have the right, in our sole discretion, to grant such Additional Extension Period provided you pay us an extension fee in an amount equal to theour then-current initial franchise fee.

Note 5

- Note 10: Extension Fee (Multi-Unit Development Agreement). As further described in Item 12 of this Disclosure Document, if you fail to comply with your Development Schedule, such failure will be a material breach of the Multi-Unit Development Agreement, which may result in us terminating the Multi-Unit Development Agreement or granting similar development or franchise rights to others within the Development Area. Alternatively, we have the right to refrain from exercising our termination rights in favor of granting you a written extension to the Development Schedule. Such an extension may, in our business judgment, be conditioned on several factors and, if granted, will additionally include the requirement for you to pay us an amount equal to half of your original Development Fee as an extension fee.
- Note 11: Space Plan Fee. We will provide the first three space plans for your Park at no cost to you. Each additional space plan after the third, if necessary, will cost you \$350 per space plan.
- Note 12: Relocation Expenses. If we consent to the Park's relocation, we have the right to charge you for the expenses we incur in connection with the relocation.
- Note 13: Call Center Program Fee. You may be required to participate in a call center program, as may be modified by us from time to time (the "Call Center Program"). Participation in the Call Center Program may include, without limitation, using and publishing a telephone number designated by us; engaging a designated service provider (which may be us, our Affiliate, or approved third-party service providers) to answer calls and handle customer service matters; and acquiring, installing, and using additional hardware, software, or other technology. You agree to pay all fees imposed by us or our approved third-party service providers for these services. We do not currently collect Call Center Program fees, which are instead payable to approved third-party service providers. If imposed, we may increase the Call Center Program fees up to 20% once per calendar year.
- Note 14: Interest on Late Payments. All amounts which you owe us and do not pay us when due will bear interest after their due date until payment is received in full at the lesser of: (a) the highest rate of interest permitted by applicable law; or (b) 18% per annum.
- Note 15: Overdue Supplier Payments. In the event we receive notice from any supplier that you are past due on any payment to such supplier, and you have not provided any notice to the supplier disputing such overdue amount prior to our receipt of notice from the supplier concerning any such past due amount, you hereby authorize us to make payment on your behalf of any such overdue amount to

such supplier. You acknowledge and agree we may pay any such overdue amount by withdrawing from your designated bank account an amount equal to the overdue amount owed to the supplier.

Non-Compliance Fee. In the event you receive a written notice of default for failing to comply with any terms of your Franchise Agreement and fail to cure such default within the specified cure period (if any), we may impose a fee of \$250 per day for each day you remain out of compliance. If there is no cure period, the \$250 per day non-compliance fee shall begin the day after we give you written notice. Notwithstanding the foregoing, if you do not obtain a waiver for any jumper before they use any Attractions at your Park, we have the right to charge you a non-compliance fee of \$2,500 per violation.

Note 17: Inspection and Compliance Reimbursement. You agree to reimburse us for our actual costs if, after an inspection of your Park, we determine that additional follow up inspections or assessments are required. Our actual costs may include (but are not necessarily limited to) the travel and living expenses of our employees or designees who perform such follow up inspections or assessments.

Note 18: Training and Assistance Fees. We reserve the right to charge you a reasonable amount, currently up to \$500 per day (which we may increase by up to 20% annually), for any training we provide to you or your managers or employees after we have provided the minimum required training described in Item 11. We will also make available continuing advisory assistance in a manner as we deem appropriate, and we can charge a reasonable fee for it. You must pay for any travel, meal, incidental, and lodging expenses incurred by persons conducting the training programs and (or providing additional guidance and assistance), and by persons attending the training program. We will make available continuing advisory assistance in a manner as we deem appropriate, and we can charge a reasonable fee for itprograms (including additional or special training programs requested by you or required by us). We may further require you, your Responsible Person (as further described in Item 15), and/or any of your employees to attend periodic refresher training courses at such times and locations that we designate. You may be required to pay fees for such refresher courses to third-parties or pay our then-current fees for such courses to us (as applicable).

Note 6: Interest, Audit Fees and Late Report Fees. If you under report your Gross Sales, in addition to paying us for unpaid Royalty Fees and Advertising Fees, you must pay interest on any unpaid amounts at a rate equal to the lesser of 18% per annum or the maximum legal rate in the jurisdiction where your Park is located. In addition, if the Gross Sales you report for any calendar year is understated by 2% or more of the actual Gross Sales for that period, you must reimburse us for all costs of the investigation or audit that uncovered the under reported sales, including salaries, professional fees, travel, meals, and lodging. If you fail to send us your weekly reports when due, we can charge you, to the extent permitted by law, a late report fee of \$100 plus \$100 for each day your report is late. If you fail to send us a report that is due on a quarterly basis or annual basis, we can charge you a late report fee of \$100 plus \$100 for each month your report is late.

Note 19: Annual Convention Fees. At least one representative from your Park must attend our annual convention at locations to be determined by us at our sole discretion. The annual convention fee may vary from year-to-year due to various economic factors and does not include travel and living expenses for your attendees. The annual convention fee will not be more than our costs plus 20%.

Note 20: Maintenance Costs. If at any time the general state of repair, appearance or cleanliness of your Park, or its Operating Assets, does not meet our standards, we may notify you and specify the action you must take to correct such deficiency. If, within ten (10) days after receiving such notice, you

fail or refuse to initiate efforts to complete such required maintenance, we have the right, but not the obligation, to enter the Park and do such maintenance on your behalf and at your expense. You must promptly reimburse us for such expenses.

Note 721: Product Purchases and Installation of Attractions. You must purchase certain items such as the Attractions (including the nets, padding, and installation), balls, wristbands, party-and event supplies, JumpSocksSkySocks®, certain promotional materials, merchandise, and other Sky Zone branded items from Sky Zone Franchise Group or our affiliates. You may decide to purchase certain other products (such as promotional materials and merchandise) from us or our affiliates. (See Item 8 for additional information). The prices, terms, and conditions for these purchases are contained on the price list that we, or our affiliates, will periodically supply to you ("Price List"). We reserve the right, for ourselves and our affiliates, to update the Price List and change the terms and conditions for these purchases at any time on 30 days'upon written notice. All the individuals who guarantee the Franchise Agreement must also personally guarantee all your purchases of products and equipment from us or our affiliates. You must pay for all taxes, shipping, and handling costs. If you fail to pay for any products or equipment when payment is due, we, or our affiliates, can and require you to pay for future products and/or equipment on a cash on demand basis or withhold shipment of products and/or equipment until payment is received, in addition to requiring you to pay late fees and interest.

Note 8: Technology Suite. We may, but are under no obligation to, introduce non POS technology products or services in the future for use in connection with the System and Parks that may use or incorporate custom applications and technologies. We intend for new franchisees, after the new system is launched, to use this technology suite. We expect there will be a monthly fee, the exact amount of which is to be determined by us. We expect that the costs associated will involve comparable, but perhaps higher, costs than franchisees are currently required to expend on the same or similar technology.

Note 22: Alternate Supplier Approval Costs. If you would like to offer products, services, or classes or use any supplies, Operating Assets (as defined in Item 8), suppliers or service providers that we have not approved, you must submit a written request for approval and provide us with any information that we request. We have the right to inspect the proposed supplier's facilities and test samples of the proposed products and to evaluate the proposed service provider and the proposed service offerings. We may require the proposed supplier or service provider to visit our headquarters to evaluate the proposed supplier or service provider in person. You agree to pay us a charge not to exceed the reasonable cost of the inspection and our actual costs of testing the proposed product or evaluating the proposed service or service provider, including travel and living expenses for our personnel, whether or not the product, service, supplier, or service provider is ultimately approved.

Note 23: Master Insurance Program Costs. All franchisees are required to participate in the Master Insurance Program (defined in Item 7). You must pay us or our affiliate Loscann via EFT a fixed amount billed monthly, and adjusted every six months, to participate in the Master Insurance Program. This amount will depend on several factors, including exposure data (attendance / revenue), loss history, compliance metrics (such as timely incident reporting, number of incidents reported, and results and resolutions of inspections), the Park's location, and other factors and risks we periodically consider. Your Total Cost of Risk may be higher or lower than the estimated range that we provide. This fee does not include the cost of purchasing other required insurance coverage from third parties, as further described in Item 8 of this Disclosure Document.

Note 24: Third-Party Insurance Coverages. Notwithstanding the Master Insurance Program, you must maintain in force, at your sole expense, other required insurance coverages from third-parties

for the Park in the amounts, and covering the risks, we periodically specify in the Operations Manual. We have the right to obtain insurance coverage for your Park at your expense if you fail to do so, in which case you must reimburse our costs in a timely fashion.

Note 25: Quality Control Programs. You shall participate in all quality control programs we designate, such as mystery or secret shopper programs, and we reserve the right to seek reimbursement of all associated costs and expenses.

Note 26: Advertising Cooperatives & Special Promotion Fees. We have the right to establish or approve local and/or regional advertising cooperatives for Parks, covering such geographical areas as we may designate from time to time (each a "Cooperative"). You and each other member of the Cooperative shall contribute to the Cooperative, using a collection structure selected and established by us, the amount determined in accordance with the Cooperative's by-laws. At our request, you shall furnish us with copies of such information and documentation evidencing your Cooperative contributions. Contributions to such Cooperatives will be credited towards your Local Advertising Funding Requirement; however, if we provide you and your Cooperative (if applicable) notice of any special promotion, you must participate in such promotion and pay to us any special promotion advertising fees assessed in connection therewith, beginning on the effective date of such notice and continuing until such special promotion is concluded. Any such special promotion advertising fees shall be in addition to, and not credited towards, the other advertising expenditures and commitments required under the Franchise Agreement.

Note 27: Audit. We have the right to inspect and audit, or cause to be inspected and audited, your and the Park's business, bookkeeping and accounting records, sales and income tax records and returns and other records. If any inspection or audit discloses an understatement of your Park's Gross Sales, you agree to pay us the Royalty Fees, Ad Fees, and any other fees understated, plus interest on the understated amounts from the date originally due until the date of payment, within 15 days after receiving the inspection or audit report. In the event any inspection or audit is made necessary by your failure to furnish reports, supporting records or other information required, or to furnish such items on a timely basis, or if our inspection or audit reveals understatement in any report of 2% or more, you must, in addition to the payment of monies owed with interest, reimburse us for all costs and expenses connected with the inspection or audit, including the professional fees for attorneys and independent accountants and the travel and living expenses of our employees.

Note 28: Management Fee. If we issue you a notice of default and you fail to cure such default within any applicable period, we have the right to assume the operation of the Park for such length of time as we determine. You authorize us to operate the Park for so long as we deem necessary and practical, and without waiver of any other rights or remedies we may have under the Franchise Agreement. All monies from the operation of the Park during such period of operation by us shall be accounted for separately and the expenses of the business, including the travel and living expenses of our representatives who operate the Park, shall be charged to such account. We shall be entitled to retain 50% of the Gross Sales of your Park as our management fee after operating expenses are paid.

Note 929: Taxes. You must pay us the amount of any state or local sales, use, gross receipts, or similar tax that the Stateany state or local government authority imposes on fees which you pay to us, without offset or deduction of any kind. This does not include income-type taxes which a Stateany state or local government imposes on our income.

Note 30: Indemnification. You must indemnify us and our affiliates for all losses, expenses, obligations, liabilities, damages (actual, consequential, or otherwise), and reasonable defense costs from all claims related to your Park, the Franchise Agreement, or your breach of the Franchise Agreement.

Note 31: Costs and Attorneys' Fees. If we incur expenses in connection with your failure to pay when due amounts owed to us or to submit when due any reports, information or supporting records or otherwise to comply with the Franchise Agreement, you agree to reimburse us for any of the costs and expenses which we incur, including reasonable accounting, attorneys', arbitrators', and related fees.

Note 32: Arbitration and Proceeding Costs. The prevailing party in any action or proceeding arising under, out of, in connection with, or in relation to the Franchise Agreement will be entitled to recover its reasonable costs and expenses (including attorneys' fees, arbitrator's fees and expert witness fees, costs of investigation and proof of facts, court costs, and other arbitration or litigation expenses) incurred in connection with the claims on which it prevailed.

Note 33: Liquidated Damages. Subject to applicable laws, you will be required to pay us the amount of damages that we would suffer due to the loss or interruption of the revenue stream we otherwise would have derived from your continued payment of Royalty Fees, Ad Fees, and Cooperative contributions, less any cost savings, through the remainder of the term of this Agreement (the "Liquidated Damages") if we terminate the Franchise Agreement for cause or if you terminate the Franchise Agreement without cause before its expiration. Liquidated Damages will be equal to the greater of: (x) \$100,000, or (y) the combined monthly average of Royalty Fees, Ad Fees, and any other fees payable to us or our affiliates under the Franchise Agreement (without regard to any fee waivers or other reductions) paid by you during the 12 months preceding the date of termination, multiplied by: (A) the number of remaining months in the term of this Agreement after termination, or (B) 12 if the Park had been open for less than 12 months upon the date of termination. The present value of the total calculated at a discount rate of 8%, assuming payment at the end of each month, will be our Liquidated Damages. In addition to Liquidated Damages, we reserve all other remedies we have under the Franchise Agreement.

Note 34: Reimbursement for Other Obligations and Duties. If we issue you a notice of default and you fail to cure such default within any applicable period, we have the right to undertake or perform on your behalf any obligation or duty that you are required to, but fail to, perform under the Franchise Agreement. You will reimburse us upon demand for all costs and expenses that we reasonably incur in performing any such obligation or duty.

Note 35: Late Report Fees. If you fail to send us your weekly reports when due, we can charge you, to the extent permitted by law, a late report fee of \$100 plus \$100 for each day your report is late. If you fail to send us a report that is due on a quarterly basis or annual basis, we can charge you a late report fee of \$100 plus \$100 for each month your report is late.

Item 7

ITEM 7. ESTIMATED INITIAL INVESTMENT

 $\overline{\mathbf{A.a.}}$ YOUR ESTIMATED INITIAL INVESTMENT $\overline{\mathbf{-FRANCHISE}}$ AGREEMENT

YOUR ESTIMATED INITIAL INVESTMENT 16,000 – 27,000 SQUARE FEET								
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMEN T	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE			
Initial Franchise Fee (including Deposit)	\$75	,000	Lump Sum	Upon signing the Franchise Agreement	Us			
	Low	High						
Lease & Security Deposits (Note 1)	\$0	\$ 150,000 <u>115,000</u>	Lump Sum	Upon signing the lease	Landlord			
Leasehold Improvements/ Architect (Note 2)	\$1,000,00 0	\$ 1,900,00 \$\frac{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\tin\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\texi{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\texi{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\ti}\tint{\text{\text{\text{\text{\text{\text{\texi}}\text{\text{\tin\text{\text{\texicl{\text{\text{\texi}\text{\text{\texi}\text{\text{\texit{\texi{\texi{\texi{\texi{\texi{\texi{\texi{\texi{\ti}\tint{\tiint{\texit{\texi{\texi{\tii}\tiint{\tii}\tiint{\tii}\t	Lump Sum	Before Opening	Landlord, architect, construction company			
Signage	\$20,000	\$54,000	Lump Sum	Before Opening	Vendors			
Attractions (Note 3)	\$700,000	\$1,100,00 0	Lump Sum	Before Opening	Our affiliate			
Furniture/Fixtures (Note 4)	\$30,000	\$65,000	Lump Sum	Before Opening	Vendors			
Computer Software License and Hardware Equipment (Note 5)	\$60,000	\$80,000	Lump Sum	Before Opening	Vendors			
Equipment and Supplies (Note 6)	\$125,000	\$165,000	Lump Sum	Before Opening	Vendors			
Licenses, Dues, and Utility Deposits	\$5,000	\$12,000	Lump Sum	Before Opening	Government Agencies and Organizations			
Inventory (Note 7)	\$20,000	\$40,000	Lump Sum	Before Opening	Vendors and us or our affiliates			
Travel Expenses/Pre-Opening Salaries	\$40,000	\$60,000	Lump Sum	Before Opening	Airlines, hotels, etc.			
Professional Fees	\$4,000	\$14,000	Lump Sum	Before Opening	Attorney, accountant			
Insurance Premiums, Allocations and Other Insurance for first quarter of operations (Note 8)	\$24,000	\$4 0,000 <u>65,000</u>	Lump Sum	Before Opening	Our affiliate; Insurance underwriterunderwri ters			
Additional Funds —3	\$50,000	\$120,000	As incurred	As incurred	Employees, suppliers,			

YOUR EST	YOUR ESTIMATED INITIAL INVESTMENT 16,000 – 27,000 SQUARE FEET								
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMEN T	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE				
months_ Three Months (Note 9)	<u>185,000</u>	<u>200,000</u>			utilities, etc.				
Grand Opening Advertising Marketing Expense (Note 10)	\$ 25,000 <u>35,000</u>	\$ 25,000 <u>35,000</u>	Lump Sum	Before opening and within the first month of operation Opening	Advertising company				
Technology Fees (Note 11)	<u>\$2,960</u>	<u>\$2,960</u>	As incurred	Beginning two months before opening	<u>Us</u>				
Site Development Fee (Note 12)	\$5,000	<u>\$5,000</u>	Lump Sum	No later than 10 days before opening	<u>Us</u>				
TOTAL (Note <u>1113</u>)	\$2,178,00 02,330,9 60	\$3,900,00 04,187,9 60							

YOUR ESTIMATED INITIAL INVESTMENT 27,000 – 39,000 SQUARE FEET									
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE				
Initial Franchise Fee-(including Deposit)	\$75,000		Lump Sum	Upon signing the Franchise Agreement	Us				
	Low	High							
Lease & Security Deposits (Note 1)	\$0	\$ 195,000 200,000	Lump Sum	Upon signing the lease	Landlord				
Leasehold Improvements/ Architect (Note 2)	\$1,250,000	\$ 2,000,00 \$\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\tinx{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\tinx{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\ti}\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\texi{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\texi}\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\text{\tin}\tint{\text{\text{\text{\tin}\text{\text{\text{\text{\text{\text{\texict{\text{\texicl{\ti}\text{\text{\text{\texi}\text{\text{\ti}\tint{\tiint{\text{\tinte\tint{\tiint{\text{\tii}\tiint{\tii}}\	Lump Sum	Before Opening	Landlord, architect, construction company				
Signage	\$30,000	\$65,000	Lump Sum	Before Opening	Vendors				
Attractions (Note 3)	\$700,000	\$1,200,00 0	Lump Sum	Before Opening	Our affiliate				
Furniture/Fixtures (Note 4)	\$50,000	\$70,000	Lump Sum	Before Opening	Vendors				
Computer Software License and	\$60,000	\$80,000	Lump Sum	Before Opening	Vendors				

YOUR ESTIMATED INITIAL INVESTMENT 27,000 – 39,000 SQUARE FEET

	1		1	<u> </u>	
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE
Hardware Equipm ent (Note 5)					
Equipment and Supplies (Note 6)	\$130,000	\$170,000	Lump Sum	Before Opening	Vendors
Licenses, Dues, and Utility Deposits	\$5,000	\$15,000	Lump Sum	Before Opening	Government Agencies and Organizations
Inventory (Note 7)	\$25,000	\$60,000	Lump Sum	Before Opening	Vendors and us or our affiliates
Travel Expenses/Pre-Openi ng Salaries	\$30,000	\$62,000	Lump Sum	Before Opening	Airlines, hotels,
Professional Fees	\$5,500	\$14,000	Lump Sum	Before Opening	Attorney, accountant
Additional Funds	\$50,000	\$130,000			
Insurance Premiums, Allocations and Other Insurance for first quarter of operations (Note 8)	\$ 30,000 <u>24,</u> <u>000</u>	\$ 60,000 <u>65,000</u>	Lump Sum	Before Opening	Our affiliate; Insurance underwriterunderwri ters
Additional Funds – Three Months (Note 9)	<u>\$260,000</u>	\$280,000	As incurred	As incurred	Employees, suppliers, utilities, etc.
Grand Opening Advertising Marke ting Expense (Note 10)	\$ 25,000 <u>35,</u> <u>000</u>	\$ 25,000 <u>35,000</u>	Lump Sum	Before opening and within the first month of operation Opening	Advertising company
Technology Fees (Note 11)	<u>\$2,960</u>	<u>\$2,960</u>	As incurred	Beginning two months before opening	<u>Us</u>
Site Development Fee (Note 12)	\$5,000	\$5,000	Lump Sum	No later than 10 days before opening	<u>Us</u>
Total (Note <u>1113</u>)	\$ 2,465,500 <u>2,687,460</u>	\$4,221,00 04,698,9 60			

YOUR ESTIMATED INITIAL INVESTMENT 39,000 – 50,000 SQUARE FEET								
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE			
Initial Franchise Fee (including Deposit)	\$75,000		Lump Sum	Upon signing the Franchise Agreement	Us			
	Low	High						
Lease & Security Deposits (Note 1)	\$0	\$ 225,000 <u>130,000</u>	Lump Sum	Upon signing the lease	Landlord			
Leasehold Improvements/ Architect (Note 2)	\$1,250,000	\$ 2,400,000 <u>2,600,00</u> <u>0</u>	Lump Sum	Before Opening	Landlord, architect, construction company			
Signage	\$30,000	\$65,000	Lump Sum	Before Opening	Vendors			
Attractions (Note 3)	\$700,000	\$ 1,200,000 <u>1,400,00</u> <u>0</u>	Lump Sum	Before Opening	Our affiliate			
Furniture/Fixtures (Note 4)	\$50,000	\$80,000	Lump Sum	Before Opening	Vendors			
Computer Software License and Hardware Equipment (Note 5)	\$60,000	\$80,000	Lump Sum	Before Opening	Vendors			
Equipment and Supplies (Note 6)	\$140,000	\$170,000	Lump Sum	Before Opening	Vendors			
Licenses, Dues, and Utility Deposits	\$5,000	\$17,500	Lump Sum	Before Opening	Government Agencies and Organizations			
Inventory (Note 7)	\$25,000	\$75,000	Lump Sum	Before Opening	Vendors and us or our affiliates			
Travel Expenses/Pre-Opening Salaries	\$40,000	\$65,000	Lump Sum	Before Opening	Airlines, hotels, etc.			
Professional Fees	\$5,500	\$20,000	Lump Sum	Before Opening	Attorney, accountant			
Insurance Premiums, Allocations and Other Insurance for first quarter of operations (Note 8)	\$30,000 <u>24</u> ,000	\$65,000	Lump Sum	Before Opening	Our affiliate; Insurance underwriterunderw riters			
Additional Funds —3 months — Three Months (Note 9)	\$ 85,000 <u>26</u> <u>5,000</u>	\$ 150,000 <u>290,000</u>	As incurred	As incurred	Employees, suppliers, utilities, etc.			
Grand Opening Advertising Marketing Expense (Note 10)	\$ 25,000 <u>35</u> , <u>000</u>	\$35,000	Lump Sum	Before opening and within the first month of operation opening	Advertising company			
Technology Fees (Note 11)	<u>\$2,960</u>	<u>\$2,960</u>	As incurred	Beginning two months before opening	<u>Us</u>			

YOUR ESTIMATED INITIAL INVESTMENT 39,000 – 50,000 SQUARE FEET									
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE				
Site Development Fee (Note 12)	<u>\$5,000</u>	<u>\$5,000</u>	<u>Lump Sum</u>	No later than 10 days before opening	<u>Us</u>				
ГОТАL (Note <u>4113</u>)	\$ 2,520,500 <u>2,712,460</u>	\$4, 722,500 <u>5,175,46</u> <u>0</u>							

YOUR ESTIMATED INITIAL INVESTMENT REBRAND - ALL PARK SIZES							
TYPE OF EXPENDITURE	AMOUNT		METHOD OF PAYMENT	WHEN DUE	TO WHOM PAYMENT IS TO BE MADE		
Initial Franchise Fee	\$0		N/A	N/A	N/A		
	Low	High					
Leasehold Improvements (Note 2)	\$40,000	\$122,500	Lump Sum	Before Opening	Landlord, architect, construction company		
Attractions (Note 3)	\$26,500	\$87,500	Lump Sum	Before Opening	Our affiliate		
Furniture/Fixtures (Note 4)	\$2,000	\$8,500	Lump Sum	Before Opening	Vendors		
Signage	\$30,000	\$72,000	Lump Sum	Before Opening	Vendors		
Graphics	\$25,000	\$35,000	Lump Sum	Before Opening	Vendors		
Inventory (Note 7)	\$2,500	\$12,000	Lump Sum	Before Opening	Vendors and us or our affiliates		
Professional Fees	\$750	\$750	Lump Sum	Before Opening	Attorney, accountant		
Additional Funds - 3 monthsThree Months (Note 9)	\$40,000	\$125,000	As incurred	As incurred	Employees, suppliers, utilities, etc.		
TOTAL	\$\frac{141,750}{6,750}\$	\$428,250 <u>46</u> 3,250					

Explanatory Notes

The estimated expenses in the tables above are estimates based upon expenses incurred by franchisees in the development of new Parks, our affiliate's development of Parks and, as applicable, Parks rebranded from one of our affiliated brands. You are encouraged to make an independent investigation and analysis of the potential expenses which you may incur to begin operating your Park.

Note 1: <u>Lease and Security Deposit.</u> The ideal size of a Park featuring two (2) or more Attractions should range between 25,000 and 40,000 square feet. Larger Parks may feature additional Attractions. In most cases, the landlord will require a security and/or rental deposit. Usually, the landlord

will require you to pay the equivalent of at least one month's rent. Rental rates or deposits on an unknown location cannot be predicted in advance. However; however, the deposit will most likely depend on the size and location of the Park. These costs will vary greatly depending on the area where the Park will be located. The item in each table represents one month's rent as a security deposit and your first month's rent. The low range in each table represents the least expensive deposit for a Park of the size described in the applicable Charttable above, and the high range is the most expensive deposit for a Park of that size. In most cases, franchisees rent rather than purchase property. The initial investment This estimated range assumes you will rent the premises for your Park and does not cover any debt service you incur or related fees such as loan interest, loan origination fees, and any and all other fees associated with the financing of the project. If you purchase the property, your initial expenses will dramatically increase.

Note 2: <u>Leasehold Improvements and Architect Fees.</u> When a site has been selected by you and approved by us in writing, you will select a licensed architect that meets our standards or has been approved by us who will provide you with a preliminary layout and suggestions for the design of a typical Park and then a full set of design plans. You will pay for the local architect's services directly.

The cost of construction, improvements or building varies widely by the size of the space, the existing improvements (including if you are rebranding from one of our affiliated brands) and local construction rates. For example, if the electrical outlets, bathrooms, or heating/cooling are not already completed in the site, your costs for leasehold improvements will be morehigher. Sometimes you may receive a construction allowance from the landlord and if so, the costs may be reduced accordingly. The cost of leasehold improvements in the above Tabletables do not include any of the items already listed in this Table entitledseparately as "Furniture/Fixtures", "Attractions" or "Equipment and Supplies." In some cases, costs may be significantly higher or lower depending on lease negotiations with local landlords. We provide the first 3three space plans to you at no additional cost. Each additional space plan you request will cost \$350.

- **Note 3:** <u>Attractions.</u> Attractions costs vary based on <u>the</u> selected attractions. You must purchase the Attractions and the installation from us or our affiliates, or through approved vendors. The cost of the Attractions does not include the entrance platforms.
- Note 4: <u>Furniture/Fixture.</u> This range includes furniture, storage racks, lockers, kiosk, displays, etc. You must purchase these items from approved suppliers. This The costs within this range assumes that these items being purchased assume you are purchasing new items and the costs include estimated installation costs. If you are rebranding your Park from one of our affiliated brands, this estimated range is lower because the fewer furniture and fixtures required will be less required.
- Note 5: Computer Equipment. The type of POS System you must purchase is described in Item 11. You are required to purchase and install the POS system approved by Sky Zone. Also included in this range is the cost of the register and the network. Thea technology package from our approved vendor which will include certain computer hardware, software, and other devices for use in operating your Park, as further described in the Franchise Agreement and in Item 11 of this Disclosure Document. The estimated costs in this range are inclusive of on-site installation and certain services described in Item 11. Your costs will depend on the number of computer stations you purchase and whether you purchase certain additional extra equipment will determine the amount of this Item. You must purchase the designated POS package from our approved vendor and all the necessary modules that come with the program. Additional information about the required POS System is included in Item 11. for your Park.

Note 6: <u>Equipment and Supplies</u>. These amounts represent This range includes an audio visual system, TVs, clocks, concession equipment and supplies, cleaning supplies and equipment, first aid equipment, filming and photography equipment, telephone system, <u>a</u> video surveillance camera system/security system, radios, and uniforms. <u>All the The</u> costs <u>within this range</u> assume these itemsyou are <u>purchasing new items</u> and include <u>anyestimated</u> installation <u>which is required costs</u>.

Your <u>securitysurveillance</u> camera system must monitor all <u>the-Attractions</u>, <u>including and areas of your Park and include</u> recording equipment to record and store video records of injuries for minimum of 365 days (ready access storage). Additionally, the <u>videosurveillance camera</u> system <u>specification</u>-must allow for us and our insurance carrier to <u>always-have direct and independent</u> access to your security cameras and video storage (365 days' history) at all times.

Note 7: <u>Inventory.</u> These amounts represent the cost of your inventory including party favors and supplies, wristbands, <u>JumpSocksSkySocks®</u>, merchandise, concessions, and uniforms.

Insurance and Captive SIR Program. The amounts represent the cost of the required Note 8: insurance for the first fiscal quarter of operation. In conjunction with our broker and underwriters, we and Sky Zone, LLCour affiliate Loscann have developed the mandatory insurance program, which provides our franchisees with primary general liability and follow form excess liability coverage tailored specifically to Parks, in addition to a risk management program (the "Master Insurance Program"). You must participate in the Master Insurance Program. Currently, under the program, coverage for each franchise Park is \$\frac{1}{2}\$ Million per occurrence and \$\frac{3}{4}\$ Million aggregate for the primary general liability with excess liability coverage of \$4 Million. The total liability limit is \$6 Million. You will pay for your allocated share of the total system-wide Master Insurance Program costs (the "Total Cost of Risk"). The Total Cost of Risk currently covers the following costs: (a) the Captive SIR Program (defined below) premiums; (b) the premiums for a designated commercial insurance carrier (or carriers) to provide primary general liability and follow form excess liability coverage; (c) the insurance broker fees paid by us or our affiliate(s); (d) our and our affiliates' administrative fees incurred in connection with the administration of the Captive SIR Program; and (e) an administrative fee retained by us for our administration of the Master Insurance Program. The Total Cost of Risk will depend on a number of factors, including revenue, attendance, location, losses, compliance metrics (such as timely incident reporting, number of incidents reported, and inspection results and resolutions), and other factors which we periodically determine reflect the risks generated by each franchisee. We or our affiliate may recalculate the Total Cost of Risk every 6six months (each 6-month period, a "Coverage Period"), at which time the Total Cost of Risk may increase or decrease. We will notify you of your Total Cost of Risk before the beginning of each Coverage Period. If the Master Insurance Program is discontinued, you must secure liability insurance from our approved supplier of insurance. Under the Master Insurance Program, each claim has a \$250,000 retention. We have established the Captive Self-Insured Retention (SIR) Program (the "Captive SIR Program"), which is part of the Master Insurance Program in which you must participate. The Captive SIR Program Terms and Conditions included in the Operations Manual will govern your participation in the Captive SIR Program. The Captive SIR Program will be used to fund claims that are above \$10,000 and up to the \$250,000 retention. The funds will be held in a separate account and managed by Sky Zone or one of our affiliates. All costs of administering the Captive SIR Program will come from amounts contributed to the Master Insurance Program by individual Parks through the Total Cost of Risk payments. Regular actuarial analysis of the amount of estimated losses covered by the \$250,000 retention, the estimated losses covered by each Park's \$10,000 deductible responsibility for the per-claim retention and administrative fees will be completed by our-insurance broker and their actuarial team. We may discontinue the Captive SIR Program at any time. You must also carry additional insurance such as commercial property insurance, crime, business auto liability, workers compensation, and other coverages as we may require from time to time. Property insurance

premiums <u>can</u> range from \$4,000 to \$14,000 annually depending on the Park location and size of the facility. Workers compensation insurance premiums can range from \$2,000 to \$25,000 annually depending on the state in which your park is located, job classification, prior claims history, and the number of employees.

Note 9: Additional Funds. This item estimates your initial startup expenses during the initial period of the operation of your Park, which we estimate is three months. These expenses include rent (for two months, as the "Lease and Security Deposit" range already includes one month of rent), payroll costs, benefits, utilities, additional inventory requirements, supplies, advertising, and marketing costs after the first month the Park is open, etc., but do not include Royalty Fees and Advertising, Ad Fees, or Technology Fees and do not include an owner's salary or draw or any expenses which are listed in the tables above chart. These figures are estimates, and you may have additional expenses to start the business. If you are rebranding a Park from one of our affiliated brands, these figures do not include your existing rent and payroll costs.

Note 10: <u>Grand Opening Advertising Marketing Expense.</u> As explained in <u>Item 11 of this Disclosure Document</u>, you must spend a minimum of \$25,00035,000 ("Grand Opening Advertising Amount Marketing Expense") for grand opening advertising and sales promotions for your Park. You must spend ¹/₂-of-this Grand Opening Advertising Amount before you open your Park, and the remainder in the first month your Park is in operation.

Note 11: <u>Technology Fees.</u> You must pay us monthly Technology Fees (currently, \$1,480 per month) beginning two months before opening.

Note 12: Site Development Fee. No later than 10 days before the date that you open your Park, you must pay to us a non-refundable site development fee equal to \$5,000, in the form of a lump sum payment.

Total. Costs and expenses can vary depending on factors like local real estate **Note 13:** values, cost of labor and supplies. These figures were based on the experience of our affiliates in operating Sky Zone Indoor Trampoline Parks since 2001, as well as certain figures/estimates we have compiled regarding certain franchisee openings since 2004. The expenses may differ in other parts of the country. None of the fees listed in this Item are refundable vary based on your geographic location. You should review these figures carefully with a business advisor before making any decision to purchase the franchise. We do not offer, either directly or indirectly, financing to you for any items Fees paid to us are not refundable under any circumstances, unless otherwise stated in the Franchise Agreement. Fees or costs due to any other entity are subject to the terms set under their agreements. Inflation, discretionary expenditures, fluctuating interest rates, freight and other shipping expenses, and other factors, such as how long it takes for you to secure an approved site for your Park and its particular size, may affect your actual costs to open your Park. You are responsible for all costs and variances from the estimated costs in this Item 7, or variances from any other estimates we may provide during any phase of the development of your Park. The availability and terms of financing depend on the availability of financing generally, your creditworthiness, your available collateral, and lending policies of financial institutions. The estimate does not include any finance charges, interest, or debt service obligation and does not include any state and local taxes and any shipping expenses. We do not currently finance any portion of the initial investment.

B.-B. YOUR ESTIMATED INITIAL INVESTMENT - DEVELOPER (3-PACK)¹

Type of		Method of		To Whom Payment Is to be
Expenditure	Amount	Payment	When Due	Made
Development Fee (Note 2)	\$175,000	Lump Sum	Upon Execution of Development Agreement	Us
Vehicle (3three months) (Note 3)	\$2,100		As arranged with this	rd-party
Estimated Investment to Open Initial Park Minus Initial Franchise Fee That Is Collected In The Development Fee (Note 4)	\$2,027,500 \$3,748,000 \$2,255,960 - \$4,112,960		<u>Table</u> of this Item 7 <u>f</u> <u>quare Feet</u> minus Init	
Total (Assuming the First Park is 16,000-27,000 Square Feet)	\$2,381,700 \$3,925,100 \$2,433,060 - \$4,290,060	MUDA for the right a total of three (3) commence operation	estimated initial invest at to ownacquire frame. Parks, as well as the ag your initial Park ed more fully in the t	chises to develop e costs to open and for the first three

Explanatory Notes

- **Note 1:** General. All fees and payments are non-refundable, unless otherwise stated or permitted by the payee. This table details the estimated initial investment associated with executing a Multi-Unit Development Agreement for the right to own and operate acquire franchises to develop three (3) Parks, as well as the initial investment to open the first Park under the Development Schedule. This table assumes that none of the Parks to be developed under the Multi-Unit Development Agreement will be rebranded from our affiliated brands.
- **Note 2:** Development Fee. The Development Fee is described in greater detail in Item 5 of this Disclosure Document, and thisthe Development Fee is forin the table above corresponds with the right to open and operate a total of acquire franchises to develop three (3) franchised Parks (provided you comply with your development obligations under the Development Agreement). If you choose, and we agree, to open develop more than three (3) Parks, your Development Fee will be higher.
- **Note 3:** <u>Vehicle.</u> We expect that you will need a vehicle to view potential <u>Sitessites</u>, oversee buildouts <u>of Parks</u>, supervise multiple Park locations, etc. We estimate that your vehicle costs will be \$2,100, including gas, maintenance, and monthly payments. The amount you expend may be less if you use a vehicle that you already own to perform these tasks.
- Note 4: Estimated Investment to Open Initial Park. This figure represents the total estimated initial investment required to open the initial Park you agreed to open and operated evelop under the MUDA. You must execute our current form of Franchise Agreement for the first Park we grant you the right to open within the Development Area concurrently with executing the MUDA. The range includes all the items outlined in the table above in this Item 7 for a Park that is 16,000-27,000 square feet, except

for the Initial Franchise Fee for the first Park-because you pay this Initial Franchise Fee once as part of Development Fee. This range does not include any of the initial investmentcosts you will incur when you establish additional Parks under a MUDA. Initial investments Your costs to develop additional Parks under the MUDA may be affected by factors including inflation, labor and materials costs, and other factors not within our control.

Item 8

ITEM 8. RESTRICTIONS ON SOURCES OF PRODUCTS AND SERVICES

In order to ensure a uniform image and uniform quality of products and services throughout the Sky Zone System, you must build and operate the Park according to our standards and specifications as we may establish from time to time, which includes purchasing approved products, services, supplies, inventory, equipment, and materials required for the operation of the Park from manufacturers, suppliers or distributors we approve. Although you are not required to purchase or lease real estate from us or our affiliates, we must accept the location of your Park (see Item 11 of this Disclosure Document). You must construct and equip your Park in accordance with our then current approved design, specifications and standards. In addition, it is your responsibility to ensure that your building plans comply with the Americans with Disabilities Act and all other federal, state and local laws. You also must use equipment (which includes hardware and software for a computerized record-keeping systemthe Technology System described in Item 11 of this Disclosure Document), signage, fixtures, furnishings (hereinafter sometimes referred to as the "Operating Assets"), products, supplies and marketing and sales promotion materials that meet our specifications and standards.

We reserve the right to designate a primary or single source of supply for certain products, services, and supplies, and we or an affiliate may be that single source. For example, as of the date of this Disclosure Document, you must purchase those items designated in the next section of this Item 8 from us, our affiliate or our designated approved third-party vendor. Other than any required products or services where we designate one or more-designated approved suppliers, you have no obligation to purchase or lease products, goods, services, supplies, fixtures, equipment, inventory, computer hardware and software (including the any Technology Suite we develop), real estate, or comparable items related to establishing or operating the Park from us, our affiliates or designated third parties.

Approved Suppliers

Suppliers must demonstrate their ability to supply franchisees' needs in a timely and reliable manner and their products must meet our specifications and standards as to quality and appearance. Our supplier specifications and standards are described in our <u>Confidential confidential</u> Operations Manual (the "Manual") (as <u>further</u> described in <u>Item 11</u> below) or on our <u>company</u> intranet site <u>for the System</u>, as we may update from time to time.

As of the date of this Disclosure Document, you must purchase the following designated products or services from approved suppliers we designate (each an "Approved Supplier"): (1) All Attractions (including installation) & replacement parts; (2) fixtures, furniture, equipment, signage (safety, interior & exterior), décor, architect services, paper products; (3) uniforms, shirts and all merchandised items intended for retail sale (whether or not bearing our—Proprietary Marks); (4) bags, packaging, party supplies, or general supplies (whether or not bearing our Proprietary—Marks); (5) Point—of Salepoint-of-sale equipment, software—and, technology solutions (IT equipment/software) identified by usand other components of the Technology System; (6) insurance policies—from—our Approved

Supplier(s); (7) general contractor services; (8) any equipment or apparel required for customers to use our service, including but not limited to socks, SkySocks and wristbands; (9) arcade & vending machines; (10) music licenses; (11) certain digital marketing Digital Marketing services (as defined and further described in Item 11 of this Disclosure Document); (1211) food & beverage; including but not limited to distributor, certain brands & products / menu items; (1312) park cleaning supplies; (1413) park scenting; (1514) gift cards; and (1615) certain support services & other products related to the operation of your park Park. For some of these Approved Suppliers, e.g. food distribution, you will be required to sign a participation agreement directly with the distributor prior to commencing services.

Except for products-or, services or supplies for which we have designated a single Approved Supplier, if you wish to purchase or lease any goods, products, equipment, or supplies, from a suppliers or service providers we have not yet approved by us as meeting our specifications, you must submit a written request for approval and provide us with any information that we request. We have the right to inspect the proposed supplier's facilities and test samples of the proposed products and to evaluate the proposed service provider and the proposed service offerings. We may require the proposed supplier or service provider to visit our headquarters to evaluate the proposed supplier or service provider in person first notify us in writing. You agree to pay us a charge not to exceed the reasonable cost of the inspection and our actual eostcosts of testing the proposed product or evaluating the proposed service or service provider, including personnel and our employees' travel eosts and living expenses, whether or not the itemproduct, service, supplier, or service provider is ultimately approved. We have the right to grant, deny, or revoke approval of products, services, suppliers, or service providers based solely on our judgment. We will notify you in writing of our decision as soon as practicable following our evaluation. If you do not receive our approval within 90 days after submitting all the information that we request, our failure to respond will be deemed a disapproval of the request. You acknowledge that the products and services that we approve for you to offer or use in your BusinessPark may differ from those that we permit or require to be offered or used in other businesses.

We and our affiliates reserve the right to receive discounts, rebates, commissions, promotional allowances and other benefits or other consideration from Approved Suppliers in connection with franchisee purchases of products and services as described in this Item 8, as well as in connection with any future purchase of any products or services. Most of these payments are calculated on an amount based on products or services sold. We willand our affiliates may retain and use such payments as all amounts received from suppliers and distributors, whether based on your or other franchisees' actual or prospective dealings with them, without restriction for any purposes we or our affiliates deem appropriate or as required by the pursuant to our agreements with Approved Suppliers. In certain instances, Approved Suppliers may charge our affiliate a lower amount for our affiliate operated Parks than available to a franchisee, based on volume purchases or other similar conditions for those Parks. We also may derive revenue from any items we or any affiliate sells directly to you by charging you more than our or the affiliate's cost. You will pay the then-current price in effect at the time for items you purchase from us or our affiliates.

During the fiscal year ending December 31, <u>20232024</u>, our affiliate Sky Zone, LLC received revenue of \$12,254,75011,562,280 related to franchisee purchases as described in this Item 8, according to its internal financial statements. Of this amount, \$10,007,1109,330,960 was paid by Sky Zone, LLC directly to third party suppliers for construction, retrofit and replacements expenses.

During our fiscal year ending December 31, $2023\underline{2024}$, we received revenue of $$2,247,640\underline{3},703,110$ related to franchisee purchases as described in this Item 8, which is approximately $10.9\underline{16.7}\%$ of total consolidated revenue of $$20,520,900\underline{22,140,100}$, based on the audited financial statements included as $\underline{\text{Exhibit F}}$ to the Disclosure Document. In addition, we received a total of $$558,943\underline{554,300}$ from suppliers to be applied to our convention costs and expenses which helps reduce franchisee convention fees.

The purchase of products from approved sources will represent approximately 90% to 95% of your overall product purchases in opening the franchise and 80 to 85% of your overall product purchases in operating the franchise.

As of the date of this Disclosure Document, none of our officers has any interest in any approved supplier.

There are currently no purchasing or distribution cooperatives.

We do not provide material benefits to you (for example, renewal or granting additional franchises) based on your purchase of products or services or use of particular suppliers.

Insurance

<u>Master Insurance Program.</u> In conjunction with our broker and underwriters, we and <u>Sky Zone, LLCour affiliate Loscann</u> have developed the Master Insurance Program. You must participate in the Master Insurance Program, which provides our franchisees with primary general liability and follow form excess liability coverage tailored specifically to Parks, in addition to a risk management program. The risk management program includes a software platform for the reporting of injuries, periodic compliance inspections and reports, a third-party administrator for claims, and various reserves for claims incurred but not reported.

You must pay, via electronic funds transfer (EFT), your Total Cost of Risk each month to participate in the Master Insurance Program. The Total Cost of Risk currently covers the following costs: (a) the Captive SIR Program premiums; (b) the premiums for a designated commercial insurance carrier (or carriers) to provide primary general liability and follow form excess liability coverage; (c) the insurance broker fees paid by us or our affiliate(s); (d) our and our affiliates' administrative fees incurred in connection with the administration of the Captive SIR Program; and (e) an administrative fee retained by us for our administration of the Master Insurance Program. The Total Cost of Risk will depend on a number of factors, including revenue, exposure (attendance, location, losses/revenue), loss history, compliance metrics (such as timely incident reporting, number of incidents reported, and inspection results and resolutions), your Park's location, and other factors which we periodically determine reflect the risks generated by each franchisee. We or our affiliate may recalculate the Total Cost of Risk every 6six months, at which time the Total Cost of Risk may increase or decrease. We will notify you of your Total Cost of Risk before the beginning of each Coverage Period. Failure to timely pay any Total Cost of Risk payment will constitute a breach of the Franchise Agreement. If the Master Insurance Program is discontinued, you must secure liability insurance from our approved supplier of insurance.

Captive SIR Program. Under the Master Insurance Program, each claim has a \$250,000 retention. We have established the Captive SIR Program, which is part of the Master Insurance Program in which you must participate. The Captive SIR Program Terms and Conditions included in the Operations Manual will govern your participation in the Captive SIR Program. The Captive SIR Program will be used to fund claims that are above \$10,000 and up to the \$250,000 retention. The funds are non-refundable and will be held in a separate account and managed by us or one of our affiliates. All costs of administering the Captive SIR Program will come from amounts contributed to the Master Insurance Program by individual Parks through the Total Cost of Risk payments. A regular actuarial analysis of the amount of estimated losses covered by the \$250,000 retention and the estimated losses covered by each Park's \$10,000 deductible responsibility for the per-claim retention will be completed by our-insurance broker and their actuarial team. Each Park must make monthly payments into the Captive SIR Program (as part of the Total Cost of Risk payments) and pay up to-the \$10,000 per claim costs as part of the deducible. All administrative costs and expenses associated with the Captive SIR Program will be paid out of the Total Cost of Risk payments.

The current coverage provided under the Master Insurance Program is based on an insurance policy that expires on March 1, 20252026. Upon the expiration of the policy, we intend to negotiate a new policy with aour broker and underwriting partners, but we cannot guarantee that the program coverage will remain the same or that the Total Cost of Risk or other fees related to the Master Insurance Program will not change at that point or in the future. We may discontinue or modify the Master Insurance Program at any time and may require you to participate in additional or replacement mandatory insurance programs. If we at any time discontinue, or reduce the scope of, the Master Insurance Program, you will be required to buy and maintain at your own expense, insurance with at least the same level of coverage as provided under the then current Master Insurance Program, from an approved supplier of general liability and excess insurance.

During the fiscal year ending December 31, 20232024, our affiliate received revenue of \$13,493,762 invoiced \$11,395,300 from the Master Insurance Program or other insurance related payments in effect during the year, according to its internal financial statements.

Listed below are the types and minimum coverage amounts that are included under the Master Insurance Program as of the date of this Disclosure Document. If your state requires greater coverage amounts for the categories listed below, you must obtain and maintain any additional coverage as required by your state. We may also designate different insurance coverage, endorsements, exclusions or amounts from time to time in our Operations Manual.

COMMERCIAL GENERAL LIABILITY:

Each Occurrence \$1,000,0002,000,000.

General Aggregate Limit: \$3,000,0004,000,000. Products/Completed Operations \$2,000,000.

Personal and Advertising Injury Limit: \$1,000,0002,000,0000. Hired and Non-owned auto coverage \$1,000,0002,000,0000. Damage to premises rented to you \$1,000,0002,000,000.

Employee Benefits Coverage \$1,000,000 ach employee/\$2,000,000 Aggregate – Claims made coverage with retro date of October 1, 2017.

Policy Aggregate Limit: \$25,000,000.

Sexual Misconduct Liability by endorsement to the policy Communicable disease excluded.

UMBRELLA LIABILITY

Each Occurrence \$5,000,0003,000,000.

Annual Aggregate per Location \$5,000,000.

Policy Aggregate Limit \$15,000,0006,000,000.

Communicable disease excluded.

ADDITIONAL REQUIRED AND OPTIONAL INSURANCE. In addition to your participation in the Master Insurance Program, you must procure and maintain in force from an approved insurance company with an "A" or better rating by AM Best and a Financial Size Rating of "VII" or better: (a) Special Form property insurance, including fire and extended coverage, vandalism and malicious mischief insurance, for 100% of the replacement value of your Park and its contents; and (b) any other insurance policies, including but not limited to business interruption insurance, hired and non-owned automobile insurance, unemployment insurance, excess umbrella insurance and worker's compensation insurance (with a broad form all-states endorsement) insurance, as we may determine periodically or as required by law. For any interruption in the operation of the Park for any other reason, you must continue to pay us, during the period of interruption, continuing royalty fees based on the average monthly royalty feesRoyalty Fees paid by you during the twelve (12) months immediately preceding the period of interruption. All insurance policies must: (1) contain the types and minimum amounts of coverage, exclusions, and maximum deductibles as we prescribe; (2) name us and our affiliates as additional insureds; (3) include such other provisions as we may require periodically. Excess and Surplus Lines insurance is not permitted for any of the types and amounts of coverages listed below, as may be modified by us from time to time. You shall agree to provide us with 30-days prior written notice of any material modification, cancellation, or expiration of any insurance policies we may require. Listed below are the types and minimum coverage amounts that are not currently included in the Master Insurance Program. If your state requires greater coverage amounts for the categories listed below, you must obtain and maintain any additional coverage as required by your state. We may also designate different insurance coverage, endorsements, exclusions or amounts from time to time in our Operations Manual or amendments thereto.

PROPERTY INSURANCE:

Business Personal Property Replacement value of personal property Tenant Improvements and Betterments Replacement value of tenant improvements Business Income Coverage 50% of Annual Gross Revenue (2 or more Locations – Coverage written on a blanket basis WORKERS' COMPENSATION:

Workers' Compensation STATUTORY (with All States Broad Form)

Employer's Liability: STATUTORY (with All States Broad Form)\$1M/\$1M/\$1M

<u>Item</u>

ITEM 9. FRANCHISEE'S OBLIGATIONS

This table lists your principal obligations under the Franchise Agreement (FA) and Multi-Unit Development Agreement (MUDA). It will help you find more detailed information about your obligations in the Franchise Agreement, MUDA, and in other items of this Disclosure Document.

Obligation	Section in Agreement	Disclosure Document Item
a. Site selection and acquisition/lease	FA: 4	7 and 11

	Obligation	Section in Agreement	Disclosure Document Item
b.	Pre-opening purchases / leases	FA: 4	8
c.	Site development and other pre- opening requirements	FA: 4 ; MUDA: 1 <u>.B.</u> , 2 <u>.C.</u>	6, 7 and 11
d.	Initial and ongoing training	FA: 6	11
e.	Opening	FA: 4 ; MUDA: 2 <u>.D., 2.E.</u>	11
f.	Fees	FA: <u>4.3</u> , 4.5, <u>4.6</u> , 4.9, 5, 6.2, 6.4 (7) <u>6.4</u> , 6.6, 9.3, <u>9.4</u> , 9.6, 9.14 <u>9.9</u> , 10.19.10, 9.12, 10.2, 10.3, 10.4, 10.6, 12.3, 13.3, 13.6, 14.1, <u>15.3</u> , 15.4, 16.1, 19.8; 16.4, 19.7, 19.11 MUDA: 6, 7 <u>B.</u> , 98, 1612.B., 19.412.D., 17.I.	5 and 6
g.	Compliance with standards and policies / Operations Manual	FA: 6, <u>7</u> , 8, 9, <u>10.2</u> , <u>10.5</u> , <u>15.2</u> , <u>15.3</u> , <u>15.4</u> <u>MUDA: 2.C., 2.D., 2.E.</u>	11
h.	Trademarks and proprietary information	FA: 6, 7, 8 <u>MUDA: 1.A.</u>	13 and 14
i.	Restrictions on products/services offered	FA: 6, <u>7.2,</u> 8, 9	16
j.	Warranty and customer service requirements	FA: 9.1 <u>, 9.7, 9.16, 9.17</u>	11
k.	Territorial development and sales quotas	MUDA: 1 <u>.B., 1.C.</u> , 2 <u>.C.</u> , 7 <u>2.D.</u> , 9 <u>2.E., 2.F.</u>	12
1.	Ongoing product/service purchases	FA: 9	8
m.	Maintenance, appearance, and remodeling requirements	FA: <u>9.1,</u> 9.3, 9.4, <u>13.3,</u> 14.1	11
n.	Insurance	FA: 4.3, 4.7, 9.9 <u>, 13.3</u>	6, 7 and 8
0.	Advertising	FA: 4.9, <u>7.2, 7.4, 9.11,</u> 10	6 and 11
p.	Indemnification	FA: <u>2.3, 15.3,</u> 18.4 . <u>MUDA: 11.B.</u>	6
q.	Owner's participation/	FA: 2.3, 2.4, <u>6.2, 8.2, 9.1</u> ;	11 and 15

Obligation	Section in Agreement	Disclosure Document Item
management/staffing	MUDA: <u>8</u> 1.A., 1.B., <u>1.D., 9</u>	
r. Records and reports	FA: 11 <u>MUDA: 1.E.</u>	6
s. Inspections and audits	FA: 12	6 and 11
t. Transfer	FA: 13 ; MUDA: 15, 16 <u>6.A.,</u> <u>6.B.</u>	17
u. Renewal	FA: 14.1, 14.2 <u>14</u>	17
v. Post-termination obligations	FA: 16 ; MUDA: 10 <u>.C.</u> , <u>1410.D.</u> , <u>10.E., 10.F.</u>	17
w. Non-competition covenants	FA: 2.2, <u>8.2,</u> 16.5, 16.6; <u>17</u> MUDA: 13 5.A., 5.B., 5.C., 5.D., 10.E.	17
x. Dispute resolution	FA: 19 ; MUDA: 19 <u>12, 13, 14,</u> <u>15, 16</u>	17
y. Other	Not Applicable	Not Applicable

Item 10

ITEM 10. FINANCING

Neither we nor any agent or affiliate of ours offers We do not offer direct or indirect financing to you, guarantees. We do not guarantee any of your notes, leases, or other obligations, or has any practice or intent to sell, assign or discount to a third party all or part of any of your financing arrangements.

Item 11

·

ITEM 11. FRANCHISOR'S ASSISTANCE, ADVERTISING, COMPUTER SYSTEMS, AND TRAINING

Except as listed below, we are not required to provide you with any assistance.

Franchise Agreement

Pre-Opening Assistance:

Before you open your Park, we will:

- 1. Instruct you in our Methods of Operation (Article 9);
- 2. Accept Evaluate and, if approved, accept your choice of a location (Article Section 4.1.) though you are responsible for locating your site. The factors we consider regarding your choice of a location for your Park include general location and neighborhood, demographics, zoning, traffic patterns, parking, overall interior and exterior size, physical characteristics of the existing building and lease terms. There is no time limit within which we must accept or not accept reject the suggested site or any contractual consequences if we cannot agree on a site;
- 3. You will typically lease, sublease, or purchase a location from a third party landlord. We will review Review and approve certain provisions of your lease for your Park's location (Article Section 4.2.); however, we will not help you negotiate your lease;
- 4. Provide you with general specifications and a general floor plan for your Sky Zone location (Article 4.4Park (Section 4.3) though you are responsible for conforming the premises to local ordinances and building codes and obtainobtaining any required permits, and construct, remodelconstructing, remodeling and decoratedecorating the premises. We will furnish guidance to you in developing the location for your Park as we deem appropriate. Note that you must work with your own architect, who must be approved by us, to ensure that the layout of your space conforms with local ordinances and building codes and that your contractors are performing work in conformance with any architectural engineering (or similar) plans. You must further consult with our development team throughout the construction process. We may review your plans, but our review is only for the purpose of ensuring that the space conforms with the Sky Zoneour System requirements, and we are not responsible for reviewing architectural or engineering plans (or similar plans) before or during construction. In consideration for the foregoing site development activities and guidance, no later than 10 days before the date that you open your Park, you must pay to us a one-time, non-refundable site development fee equal to \$5,000 in the form of a lump sum payment via EFT (Section 4.3(c));
- 5. Provide you with an initial training program (<u>ArticleSection</u> 6.1.). We do not <u>help you</u> <u>hireprovide any assistance to you with respect to the recruitment or hiring of</u> your employees or provide them with training aside from the initial training program; <u>and</u>
- 6. Provide you with lists of start-up inventory, furniture, fixtures, software, equipment, signs, and supplies; (ArticleSection 4.6.)—but, which you must purchase such items yourself—from approved or designated suppliers; and (which include us or our affiliates).

Ongoing Assistance:

During the operation of your business, we willmay:

- 1. Provide you with refresher training (Articleif required by us (Section 6.2.) or requested by you (Section 6.4), though you are responsible for all other training and the hiring of your employees;
- 2. Provide you with general guidance on operating issues concerning the location, system standardsPark, our Methods of Operation, marketing programs approvals, etc. (Article Section 6.3.);
 - 3. Provide you thewith Internet and telephone consultation (Article Section 6.4.);
- 4. Provide you with wholesaling services—occasionally where we may ourselves act as an approved or designated source for products, merchandise, accessories, fixtures, furnishings, equipment, signs, etc. (ArticleSection 6.4.);
- 5. Provide you with ongoing marketing programs (notexcluding collateral) (ArticleSection 6.4.);
- 6. Provide you, at your cost and expense, with mandatory and optional meetings—and, seminars, and conferences where we may get together meet with you and other Parkes franchisees for business or social purposes (Article Section 6.4.);
- 7. Provide you with research and development regarding our Methods of Operation (ArticleSection 6.4.);
- 8. Conduct <u>quality control programs</u>, <u>which may include</u> mystery shopping services, or inspections of your business (ArticlePark (Sections 9.10. and 12.1.); and
- 9. Provide you with access to the Master Insurance Program for so long as it is available to our franchisees (ArticleSection 9.9.).

Multi-Unit Development Agreement

The Multi-Unit Development Agreement grants you the right to acquire franchises to develop multiple Parks. However, for each Park; however, you will sign a separate Franchise Agreement to develop and operate each Park. Therefore, we have no ongoing obligations such as training or operational assistance to you under the Multi-Unit Development Agreement. All ongoing and future obligations to you in opening your locations Parks shall be provided in any Franchise Agreement between you and us; including our standard site selection criteria.

Franchise Site Selection

You select<u>will find and submit to us</u> the <u>proposed</u> site for your Park location. We will provide you with our standard site selection criteria or an on-site evaluation of your proposed sites, as we deem appropriate. You must verify to us that your site complies with our site selection criteria. If we do not accept a site you propose, you may propose another site. We do not select or endorse your site.

Before you lease or purchase any site for a Park location, you must submit to us certain information related to the site. We will review the site information and determine whether we approve or object to the

site you propose. Factors we deem appropriate include the general location and neighborhood, demographic information, traffic patterns, access, visibility, location of other competing facilities, size, configuration, appearance, and other physical characteristics of the site.

If you operate are rebranding an existing trampoline park from our affiliated brands, you will already have a site for your Park location, subject to our approval before you sign a Franchise Agreement with us.

Opening

We estimate the time from the date you sign the Franchise Agreement to the date you open your Park location to be between 12 and 18 months. However, this time estimate may vary depending on numerous factors including, entitlements and permits, construction schedules and financing. You must sign a lease for a location we approve within <u>6nine</u> months of your signing the Franchise Agreement. If you and we <u>can'tdo not</u> reasonably agree on a suitable location for your <u>franchisePark</u> and you do not have a signed lease or purchase a location within <u>6nine</u> months of your signing the Franchise Agreement, we may terminate the Franchise Agreement, unless we have agreed with you to extend the amount of time for you to open the <u>franchise locationPark</u>. Your Park-<u>location</u> must be open and operating within 18 months after you sign the Franchise Agreement or we may terminate the Franchise Agreement and retain your Initial Franchise Fee.

You may open a Park under a MUDA only by signing a Franchise Agreement for that Park. You will sign a Franchise Agreement for the first Park at the same time you sign the MUDA.

Operations Manual

During the term of the Franchise Agreement, we will allow you to access or borrow our then-current Operations Manual (the "Operations Manual" or "Manual"). In our discretion, we may provide you the Manual in paper or electronic form. The information provided Manual contains mandatory and suggested specifications, standards, operating procedures, mandatory and approved vendors, and rules that we prescribe from time to time for the operation of a Business Park and information relating to your other obligations under the Franchise Agreement and related agreements. We may modify the Manual from time to time to reflect changes in the law, the marketplace, or our Methods of Operation. The Manual and our Methods of Operations constitute confidential trade secrets and will remain our property-You agree to keep your copy of any bound operations manual and any information we share with vou that is a part of our Methods of Operation and/or s Manual confidential pursuant at all times. If there is a dispute over the Manual's contents, our most recently updated master copy of the Manual shall control. The Manual's contents are confidential and you cannot disclose the Manual to any person other than any employee of yours who needs to know its contents and who is subject to your restrictions on confidentiality obligations set forth in the Franchise Agreement. You further agree to keep all such information secure whether hard copies located at the Business or by keepingthat would prevent the employee from disclosing the contents of the Manual to unauthorized individuals. You may not at any time copy, duplicate, record, or otherwise reproduce any part of the Manual. You must keep the Manual current and in a secure location at your Park and keep all password -protected electronic information secure. YouBy signing the Franchise Agreement, you will agree that these requirements are reasonable and necessary to preserve the identity, reputation, value and goodwill of the system. System (Franchise Agreement - Article Section 6.5).

A table of contents of the Manual, including allocation of pages to each subject, <u>may beis</u> included as <u>Exhibit D</u> to this Disclosure Document. Currently, there are 312 pages in the Manual (excluding the Appendix).

Pricing

Subject to applicable laws, we may periodically set a maximum or minimum price that you may advertise for products and services offered by your Park (as set forth in the Operations Manual or otherwise in writing from time to time). If we impose a maximum advertised price for any product or service, you may not advertise a higher price for the product or service than the maximum advertised price we impose. If we impose a minimum advertised price for any product or service, you may not advertise a lower price for such product or service than the minimum advertised price we impose. Further, you must comply with any advertising policy we adopt which may prohibit you from advertising any price for a product or service that is different than our suggested retail price.

Advertising

National Advertising Fund

We have established <u>ana National</u> Advertising Fund (the "NAF") to which you must contribute <u>bi-weekly Advertising monthly Ad</u> Fees in an amount equal to 2% of <u>the Gross Sales of your Park</u>. The <u>Advertising FundNAF</u> may be used for <u>the creation and development of marketing, advertising on television, radio, direct marketing mailings, and related programs and materials, including electronic, <u>print and Internet media as well as</u> the <u>newspaperplanning</u> and <u>for any tournaments or events that we feel are appropriate purchasing of network advertising</u>. We have the right to determine the type of advertising and the media in which it will appear. We have the sole right to formulate and make policy decisions concerning every aspect of the advertising and franchise expansion program, consistent with applicable law. (Franchise Agreement – Article 10).</u>

We reserve the right to increase or decrease the Advertising Ad Fee upon at least thirty (30) days' notice to you, up to a maximum of 2.5%. We agree that all Advertising Ad Fees will be used for the payment of all costs associated with the creation, production, distribution, media placement and administration of local, state, regional or national advertising programs (wherever located) and for any taxes incurred on this Advertising Fundthe NAF. In addition, the Advertising Fund NAF can be used to pay administrative expenses, including, without limitation, material, services, reasonable salaries, administrative costs, travel expenses, and overhead incurred by us or our Affiliates, in connection with administering the Advertising FundNAF.

You acknowledge that the Advertising Fund The NAF may not benefit you proportionately to the Advertising Ad Fees you paid, that; advertising may benefit some geographic areas more than others; and that we are not obligated to spend any amount on advertising in your area or other territories separate from the Advertising Fund. We do not have to spend the Advertising Ad Fees during any specific time period. Advertising may be handled by the an outside advertising agency which we select or in-houseby our employees.

We will not use the Advertising FundNAF for the sale or attempted sale of new franchises.

We are under no obligation to continue selling or offering franchises. We have the sole right to formulate and make policy decisions concerning every aspect of the advertising and franchise expansion program,

consistent with applicable law. Unaudited financial statements of the Advertising FundNAF will be made available to franchisees annually on reasonableupon written request. If we do not use all of the funds in a particular fiscal year, the remaining funds will be carried over to the next fiscal year and be included in that year's advertising budget. We are entitled to reimbursement from the Advertising Fund to cover our administrative and overhead expenses associated with operating the Advertising Fund. AllAs of the date of this Disclosure Document, other franchisees and anymay be contributing to the NAF on different terms. Our affiliate-owned Sky Zone Indoor Trampoline Parks willmake contributions to the NAF, but may not contribute toon the Advertising Fund on an equal basis as you and our other franchisees.

In the calendar year ending December 31, 20232024, the Advertising Fund contribution was NAF contributions were used as follows: (i) 63.666.8% toward media, search engine, and social media marketing; (ii) 11.37.7% toward content, asset creation and production; (iii) 64.1% toward website maintenance and site creation; (iv) 11.211.9% toward non-media (e.g., research, analytics, advertising professional services support); (v) 7.19.3% toward administrative expenses; and (vi) 0.80.2% toward other expenses.

Advisory Council

We <u>currently</u> have a Franchise Advisory Council (<u>the "FAC"</u>) that works with us to improve the System, including <u>the productsits product</u> and <u>services offeredservice offerings</u>, advertising campaigns, and other matters of interest to us and our franchisees. The FAC is <u>currently</u> made up of seven members <u>from the franchise network</u>, <u>six of which are elected by the System and one who is appointed by us</u>. All <u>franchiseelected</u> members of the council are nominated by and voted for by the franchisees on a global basis (not based on geography). For further information about the FAC membership, please see <u>FAC bylaws The FAC does not have any decision-making authority and may be constituted, modified, or <u>dissolved by us in our sole discretion</u>. We will have the right to form, change, merge or dissolve any advisory council at any time.</u>

Grand Opening Advertising Marketing Expense

In addition to the other advertising expenditures you must make, you must conduct a grand opening advertising and sales promotion program for your Park and you must spend a minimum of \$25,000\[35,000\] for this program (the "Grand Opening Marketing Expense"). You must spend 50% of the Grand Opening Advertising Amount Marketing Expense before your Park opens for business—and the remainder during the first month of operations. You must also provide to us, within three months after the opening of the Park, proof of your grand opening advertising and sales promotion expenditures in the form, and including the details and copies of the advertising and materials and receipts, as we request. (Franchise Agreement – ArticleSection 4.9).

Your Own Advertising

You must spend certain amounts <u>fortoward</u> advertising and <u>promotion of promoting</u> your Park in addition to the Grand Opening <u>Advertising Marketing Expense</u> described above and the required <u>Advertising Fund contributions</u>. In the second, third and fourth month in which your Park is <u>open, you Ad Fees. You</u> must spend a minimum of \$12,000 per month on local advertising (<u>depending on the size and promotional activities during the first month following the opening of your Park and its <u>location</u>). After your <u>fourthfirst</u> month, you must spend a minimum of 4% of your Park's <u>monthly</u></u>

Gross Sales <u>per month</u> on local advertising on a quarterly basis. (the "Local Advertising Funding Requirement") (Franchise Agreement – Article 10.5 Section 10.2).

We currently have a third-party mandatory marketing platform for all approved marketing resources. You must have proof of your expenditures if weAt our request, you must submit documentation to review verify your books and records compliance with your Local Advertising Funding Requirement. If you develop wish to use any advertising for your own use and promotional materials which were not prepared or previously approved by us during the preceding 12 months, you must submit itsamples of your proposed materials to us for our prior consentapproval at least 2030 days before you plan prior to their intended use them. Unless If we notify you that do not provide our written approval within 15 days from the date we receive your proposed materials are disapproved, the materials are considered approvedrejected. You may not advertise or use the Marks in advertising or other form of promotion without the appropriate copyright, trademark, and service mark registration symbols for those Marks which are registered, nor may you use them in a manner which would misuse or dilute the Marks or damage the goodwill associated with the Marks. All advertising and promotional materials you use must be completely factual, comply with all applicable laws and conform to the highest ethical standards of ethics and quality the advertising and marketing policies that we periodically specify. We reserve the right to require you to include certain language in your local advertising materials, such as "Franchises Available" and our Website address and telephone number.

We or our Affiliates have the right to establish and operate websites, social media accounts (such as Facebook, X/Twitter, Bluesky, LinkedIn, Yelp, Instagram, Pinterest, YouTube, TikTok etc.), applications, keyword or adword purchasing programs, accounts with websites featuring gift certificates or discounted coupons (such as Groupon®, Living Social®, etc.), mobile applications, or other means of digital advertising on the Internet or any electronic communications network (collectively, "Digital Marketing") that are intended to promote the Marks, your BusinessPark, and the entire network of Businesses System. We will have the sole right to control all aspects of any Digital Marketing, including those related to your BusinessPark. Unless we consent otherwise in writing, you and your employees may not, directly or indirectly, conduct or be involved in any Digital Marketing that use the Marks or that relate to the BusinessPark or the networkSystem. You may not separately register any domain name, create any username, or operate any web site containing any of the Marks without our written approval. If we do permit you or your employees to conduct any Digital Marketing, you or your employees must comply with any policies, standards, guidelines, or content requirements, or vendor requirements that we establish periodically and must immediately modify or delete any Digital Marketing that we determine, in our sole discretion, is not compliant with such policies, standards, guidelines, or requirements. We may withdraw our approval for any Digital Marketing at any time. (Franchise Agreement – Article 10.8 Section 10.5).

Advertising Cooperatives

There are currently no advertising cooperatives.

We have the right to establish or approve local and/or regional advertising cooperatives for businesses in your local or regional areasParks, covering such geographical areas as we may designate from time to time (each a "Cooperative"). Each ecooperative Cooperative will be organized and governed in a form and manner, and begin operating on a date, we determine in advance. You must participate in any such ecooperative Cooperative and its programs and abide by its by-laws. If your businessPark is within the territorygeographic area of an existing ecooperative at the time your businessPark opens for business, you must immediately become a member of the cooperativeCooperative as will all

businessesParks, franchised or company-owned, which are located in such territorygeographic area. If a cooperativeCooperative applicable to your businessPark is established during the term of ouryour Franchise Agreement, you must become a member no later than 30 days after the date approved by us for the cooperativeCooperative to commence operation. The amounts you contribute to a Cooperative will count toward your Local Advertising Funding Requirement, excluding any fees for special promotions (as further described in Item 6 of this Disclosure Document).

Computer System

You must use in the development and operation of the Park the management system and computer hardware and software and related technology designated by us, including without limitation, features such as redundant, high-speed broadband connectivity, high-speed broadband monitoring, methods and means of encryption and access to our network resources, and other internet _based technology and peripheral devices that we specify from time to time (the "**Technology System**," which includes all the components described in this Computer System section).

Our required Technology System currently includes a web-hosted point-of-sale platform application (sometimes referred to as the "POS System").

You must use the Technology System, and the POS System in particular, to: (i) enter and track purchase orders and receipts, attendance, and customer information, (ii) update inventory, (iii) enter and manage your customers' contact information, (iv) create membership contracts; (v) generate sales reports and analysis relating to the BusinessPark, (vi) manage waivers, and (vii) provide other services relating to the operation of the BusinessPark. If we require you to use any proprietary software or to purchase any software from a designated vendor, you must execute and pay any fees associated with any software license agreements or any related software maintenance agreements that we or the licensor of the software require.

The standard package of both-hardware and software you must purchase (as described in Item 7 of this Disclosure Document) includes on-site installation, training, technical support for the first year and 3a three-year warranty on the hardware. The hardware components in this package and the initial costs associated with the required software ranges from \$60,000 to \$80,000. The hardware currently includes networking equipment; 2two back-office computers; 9-11 POS terminals, Party-Check-in & Cafe Stations; 4one Waiver Station; receipt printer; cash drawers; credit card terminals; and barcode scanner. The software currently includes a web-hosted POS platform application. All of the components described in this paragraph are required purchases. We estimate that the annual fees for maintenance and technical support will be between \$11,000 to \$15,000 and you will be responsible for these fees if necessary.

Neither we nor any of our affiliates will provide you with any ongoing maintenance, repairs, upgrades or updates for the Technology System. You are required to purchase maintenance from the POS SoftwareSystem's software provider and continue to keep this maintenance current.

You must also purchase and install a surveillance camera system for your building security, unless applicable law does not permit you to install a surveillance camera system. Prices are estimated to range from \$47,261 to \$54,435. Your surveillance system must be in good working order at all times and must cover all angles of your trampoline courts and the park (inside and out). Additionally, the surveillance camera system must contain 365 days of storage capacity. All recorded injuries must be downloaded and retained separately, and a copy of the video must be provided to us on our request. parkeveillance camera system for at least one

year. You must ensure that your surveillance camera system is always operational and that we have independent access to your surveillance camera system and video recordings at all times.

We may add, remove, or modify components of the Technology System periodically and we may designate approved suppliers or specifications for such items. As of the date of this Disclosure Document, you are required to use the following are included services and systems from Approved Vendors as part of the Technology System: (i) Roller—the POS_System; (ii) Ryze—digital AVA/V content; (iii) Control Play—streaming music; (iv) Go to Connect—phones; and (v) Communication Tools—Microsoft—@skyzone.com_email, Power_BI, and Sharepoint communication tools, including branded e-mail accounts and business reporting systems.

You must replace, upgrade, or update at your expense the Technology System at your expense as we may require periodically without limitation. You may be required to invest in and implement new technology initiatives at your own expense, which may include changes related to the acceptance of implementing new payment methods, monitors, music, Internet TV broadcast broadcasts, software management applications, surveillance systems, e-learning modules, and software applications designed to better manage business functions and control costs. We will establish reasonable deadlines for implementation of any changes to our Technology System requirements. There is no limitation in the Franchise Agreement on either our right to require you to obtain updates or upgrades or the cost of any updates or upgrades.

You must pay us a monthly Technology Fee (currently, \$1,480 per month and further described in Item 6 of this Disclosure Document) beginning two months prior to the opening of your Park. Notwithstanding our collection of the Technology Fee, you may be required to purchase additional hardware, software, or other components of the Technology System from required vendors.

We additionally reserve the right to require you to present music, movies, television and other media at the Park. You may be required to pay your pro-rata share of music, movie, and other media licensing fees to show or otherwise present media at your Park. If we have not identified any approved vendors for media content and the display of such media content is not otherwise prohibited by our Methods of Operation, you are solely responsible for ensuring that you have paid all necessary licensing fees to show or otherwise present such media content at your Park.

The types of business information which will be collected by the POS Technology System will be include customer information, sales reports, scheduling of events and programs, and labor functions, all on a daily, weekly and monthly basis. It will also include gathering guest information. In addition, the POS Technology System can provide customized reports if we request this information. We will have the right to have independent access to all information or data in the Technology System (and the POS System in particular) and the surveillance camera system, and there are no limitations on our rights to do so. We will also have the right to use and publish the information we collect from your POS Technology System, including for purposes of disclosure in our Franchise Disclosure Document. We are not obligated to provide or to assist you in obtaining the above item or services. In the future, you may be required to change, upgrade or modify the type of computer hardware and software you must use at your expense.

Training

All the The designated training for franchisees is mandatory and must be completed to our satisfaction after you have signed the Franchise Agreement and before you are approved by the Operations teamus

to open your Park. If your Management Team and/or Operating Principal Owner (as described below) fail to complete the training program to our satisfaction, we may elect to postpone the opening of the Park or terminate the Franchise Agreement and keep the entire Initial Franchise FeeYou are also required to participate in all other activities required by us to operate the Park. If you currently operate a Sky Zone Park, and have previously completed the initial training program—is, you may not be required for any additional Parks you commit to open and for new management hired for your existing parks (management turnover)to attend the initial training program depending on various factors, including but not limited to how long it has been since you completed the initial training program, your compliance with our Methods of Operations in operating your existing Park(s), and the extent of changes to the initial training program since you last completed such program. We may require that certain of your employees, including any new general manager, complete the initial training program.

We will provide an initial training program for the operation of the Park for your Operating Principal Owner and up to three members of the Parkto you (or if you are an entity, your "Responsible Person" as further described in Item 15 of this Disclosure Document) and your management team that consists of a minimum of three managers and one additional leader (collectively, "Management Team"). You must hire, train and maintain a sufficient number of managers so that there is a manager on dutyall shifts are supervised by at least one manager, assistant manager, or team lead during all Park operating hours. The initial training program will take place at one of our Sky Zone Training Parks and will last anywhere from between seven (7) to twenty-one (and 21) days. We do not charge you for the initial training program, but you must pay for the wages, travel, and living expenses forincurred by you and your employees. Training will be conducted for your Management team members and Operating Principal Owner after the Franchise Agreement has been signed and in attending such program. Your Management Team will complete the initial training program while your Park is being developed. In the event you are purchasing an existing, operating Park, the Operating Principal Owner is required toyour Responsible Person and Management Team must attend such training before the purchase is complete and the Park changes possession. As further described in Item 6 of this Disclosure Document, if the purchase is not completed and you attended any portion of the initial training program b, we may retain half of the transfer fee. Training is provided on a regular basis and a training schedule is provided on our intranet. Additional training requirements may be added to the schedule on an as-needed basis. You must open the Park with the minimum approved managers (as outlined above) that have completed and been certified as having completed our training program. If there are not the minimum certified managers in place you will not be allowed to, or approved to, open the Park

If your Management Team and/or Responsible Person fail to complete the initial training program to our satisfaction, we may elect to postpone the opening of the Park or terminate the Franchise Agreement and retain the entire Initial Franchise Fee.

We will, at our expense, also provide on-site, opening assistance, consisting of at least one (1) to two (2) personsperson, for a minimum period of five (5) days at your Park's location when it opens. Even if you currently operate a Sky Zone Park, the initial If additional training and the is required on-site opening assistance is mandatory, we will have the right to charge a reasonable fee for such training and our employees' travel and living expenses.

TRAINING PROGRAM

SUBJECT	HOURS OF CLASSROOM TRAINING	HOURS OF ON-THE-JOB TRAINING	LOCATION		
	eLearning Hours	Experience Hours			
Sky Zone Culture [®] <u>Mission</u> & Core Values	0.5	<u>5.0</u> <u>1.0</u>	1.0	Sky Zone University & In-	
Park Ops & Programming Overview	θ <u>1.5</u>	<u>1.54.0</u>	1.0	Trainir	g Park
Front line team memberCourt Monitors & the Guest Experience	<u> 1.0</u> <u>1.5</u>	4.0	4.0	Sky Zone University & Train	
Sales AssociateService Team, Memberships & the Guest Experience	<u>1.0</u> <u>1.5</u>	4.0	4.0	Sky Zone University & Train	
Events <u>Team</u> & the Guest Experience	1.5	<u>5.04.0</u>	4.0	Sky Zone University & Train	
MOD & the Guest Experience Manag er on Duty Routines & Responsibilities	θ <u>1.5</u>	4.0	2.0	Trainir	g Park
MOD Rotations	0	1.0	3.0	Trainir	g Park
Interpersonal Communication for Managers	θ	1.0	1.0	Trainir	g Park
Hiring Team Members	θ	3.0	1.0	Training Park	& Your Park
Training Team Members Fuel Zon e (Food Café)	θ <u>1.5</u>	<u>3.04.0</u>	1.0	Training Park	& Your Park
Retaining Training Team Members	<u> </u>	<u>1.5</u> 2.0	1.0	Training Park	& Your Park
Risk Management & Incident Response	<u>1.00.5</u>	<u>3.02.0</u>	1.0	Sky Zone University Training Park	

SUBJECT	HOURS OF CLASSROOM TRAINING	HOURS OF ON-THE-JOB TRAINING	LOCATION		
	eLearning Hours	Experience Hours			
Service Guest Recovery (H.E.A.R.T.)	<u>1.00.5</u>	1.0	1.0		/ (Online) <u>eLearning</u> , k & Your Park
Sales Basics	1.0	2.0	3.0	•	ty (Online), Training Your Park
TOTAL	7.0 <u>11.0</u>	4 0 + <u>30.0</u>	27+		

* Note that The hours presented above are merely estimates, which may be changed at any time in our discretion. Further, some subjects may be combined, and time periods and subject matter may be subject to change. The hours listed for On _the _Job <u>Training</u> will typically include more than one subject matter. It is the nature of this business that all aspects of training are integrated. The above are merely estimates.

The

<u>Our</u> training programs are <u>managed_conducted</u> and provided by our <u>professional</u> training team. The minimum experience of our training team and the instructors in the field that is relevant to the <u>subject taught and our operations will typically be two (2) years.</u> The <u>Director under the supervision of our Vice President of Operations Services, Ra'Chard Dennis, and our Senior Director of New Park Openings, Christine Stohr. Mr. Dennis has been our Vice President of Operations Services since June 2024 and has 18 years' experience in the service industry. Ms. Stohr has been our Senior Director of New Park Openings since October 2024 and has 20 years of learning and development experience in the service industry. Ms. Stohr will supervise the initial training program described above and the on-site training and support we provide for new Park openings.</u>

Other members of our training team include: Steve Dupuy, our Manager of Training & Development, who has a team of experienced trainers who have at least twomore than 10 years' experience in training and instructional design; Gordon Carver, our Manager of New Park Opening Training, who has more than five years of experience as a General Manager; and Allison Selg, our Operations Implementation Trainer, who has more than three years of training and development experience as well as operations experience.

We <u>Additional employees</u> may also provide refresher programs. These programs are not mandatory at this time but may be in the future. assist Mr. Dennis and Ms. Stohr with the development and administration of the initial training program and on-site training and support for new Parks.

If you hire a new manager, this manager must satisfactorily complete the training program before managing your Park. We may charge a reasonable fee for the such training, and you must pay for the employeeour employees's travel, lodging, meals, personal and living expenses, and salary.

We may elect to charge a reasonable fee <u>and our employees' travel and living expenses</u> for any <u>other</u> training provided after the opening of the Park. You or your employees must pay the compensation of the trainee as well as the trainee's travel, lodging and personal expenses or if you request (or if we require) additional or special training for your employees.

We may also provide refresher programs, which may include electronic training courses, trade shows, ongoing education or certification programs, franchisor-sponsored performance groups, and/or webinars, including courses and programs provided by third parties we designate. You may be required to pay fees for refresher programs to third-parties or pay our then-current fee for such courses to us. These programs are not mandatory at this time but may be in the future. The location, duration, and-content, and pricing of these programs have not been determined yet.

Item 12

ITEM 12. TERRITORY

Franchise Agreement

The Franchise Agreement grants to you the right to own and operate a single Park at a specific location. The Multi-Unit Development Agreement grants you the right to develop multiple Parks at specific locations under our then current site criteria. You may not conduct the business of your Park at any site other than the premises or relocate your Park without our prior written consent. The Franchise Agreement does not provide you with any options, rights of first refusal or similar rights to acquire additional franchises. The Multi-Unit Development Agreement does not provide you with any options, rights of first refusal or similar rights to acquire additional franchises. You will not receive an exclusive territory. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. You will receive a Protected Territory, except for the circumstances set forth below. Whether you are a single unit franchise or a Multi-Unit Developer, your territory You must select the site for your Park from within the "Protected Territory" identified in Appendix E to your Franchise Agreement. The Protected Territory will be delineated at the timeagreed upon by you signand us before your execution of the Franchise-Agreement or Multi-Unit Development Agreement and shallwill be based on zip codes or metes and bounds or other territory delineations we may utilize. Typically, but not in all cases, each Protected Territory granted under a Franchise Agreement will encompass a population of at least 150,000 people. If you sign a Multi Unit Development Agreement, you will receive multiple Protected Territories that typically each will encompass a population of at least 150,000 people. We use third party analytics tools (currently Buxton Analytics, but we have the right to use other sources or add additional analytics tools) for purposes of determining population density. We will mutually agree on If you are signing a Franchise Agreement in connection with a Multi-Unit Development Agreement, our then-current criteria will apply when determining the Protected Territory at the time you sign the under that particular Franchise Agreement or the Protected Territories.

If you have not designated the premises for your Park at the time you sign the Multi Unit Developmentyour Franchise Agreement and maintaining exclusivity does not depend on achieving a certain sales volume, market penetration, or other contingency or circumstances. We, we reserve the right to modify athe Protected Territory provided once such premises have been designated by you and accepted by us. Further, we may modify the modified Protected Territory encompasses a population of at least 100,000 people upon: (i) any transfer, redemption or issuance of a legal or beneficial ownership interest in the capital stock of, or a partnership interest in, you or of any interest convertible to or exchangeable for capital stock of, or a partnership interest in, you; (ii) any merger or consolidation between you and another entity, whether or not you are the surviving entity; (iii) any transfer in, or as a result of, a divorce, insolvency, corporate or partnership dissolution proceeding or otherwise by operation of law; (iv) any transfer upon your death or the death of any of your owners by

will, declaration or transfer in trust or under applicable laws of intestate succession; or (v) any foreclosure upon your Park or the transfer, surrender or loss by you of possession, control or management of your Park. Otherwise, maintaining your Protected Territory does not depend on achieving a certain sales volume, market penetration, or any other contingencies or circumstances.

You will not receive an exclusive territory under the Franchise Agreement. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control.

However, if you remain in compliance with the Franchise Agreement and all other agreements entered into with us (or any of our affiliates) and except as described below, we will not own and operate, or license a third party to own and operate, a Park within the Protected Territory while the Franchise Agreement is in effect.

You may not conduct the business of your Park at any site other than the premises or relocate your Park without our prior written consent. If you want to relocate your locationPark, we will evaluate your suggested new location in the same way as we evaluate requests for new locations. If you want to open another locationPark, you will have to apply to us to enter into anotheran additional Franchise Agreement with us and if you want to acquire rights to develop several new locationsParks in a designated Development Area (as defined below), you will have to apply to us to enter into a Multi-Unit Development Agreement.

Specifically, except for

Multi-Unit Development Agreement

The Multi-Unit Development Agreement grants you the right to acquire franchises to develop, own and operate Parks within the designated "Development Area" that will be described in Appendix A attached to your Multi-Unit Development Agreement. The Development Area is separate and distinct from the Protected Territories for your Parks which will be set forth in their corresponding Franchise Agreements.

The Development Area will be agreed upon by you and us before your execution of the Multi-Unit Development Agreement and will be based on zip codes or metes and bounds or other territory delineations we may utilize. Typically, but not in all cases, the Development Area may be comprised and represented by multiple maps featuring an estimated population density of at least 150,000 people in each map. As above, we use third party analytics tools for purposes of determining population density. Maintaining the Development Area's exclusivity does not depend on achieving a certain sales volume, market penetration, or other contingency or circumstances, but your failure to comply with your Development Schedule will be a material breach of the Multi-Unit Development Agreement, which may result in us terminating the Multi-Unit Development Agreement or granting similar development or franchise rights to others within the Development Area. Alternatively, we have the right to refrain from exercising our termination rights in favor of granting you a written extension to the Development Schedule. Such an extension may, in our business judgment, be conditioned on any or all of the following: (a) reducing the size of the Development Area; (b) modifying your Development Schedule (in terms of timing and/or number of Parks to be opened); (c) requiring you to sign our then-current form of general release; and (d) requiring you to sign our then-current form of Multi-Unit Development Agreement, the terms of which may differ substantially from the terms of the original Multi-Unit Development Agreement you signed. Further, if we grant you a written extension, you may be required to pay us an amount equal to half of your original Development Fee as an extension fee.

You will not receive an exclusive territory under the Multi-Unit Development Agreement. You may face competition from other franchisees, from outlets that we own, or from other channels of distribution or competitive brands that we control. However, if you remain in compliance with the Multi-Unit Development Agreement and all other agreements entered into with us (or any of our affiliates) and except as described below, we will not own and operate, or license a third party to own and operate, a Park within the Development Area while the Multi-Unit Development Agreement remains in effect.

The Development Area will not include the Protected Territories for any existing Parks within the Development Area as of the date you sign the Multi-Unit Development Agreement, regardless of whether we define, describe, or otherwise illustrate such territories in maps attached to your Multi-Unit Development Agreement. If a third-party's development area is adjacent to your Development Area, the Protected Territories for the third-party's Parks may extend into your Development Area depending on those Parks' final locations.

As an exception to the above, if multiple maps are attached to your Multi-Unit Development Agreement as your Development Area, once you develop a specified number of Parks within a specific map, we may establish or license others to establish new Parks within such map but not within the Protected Territories specified in the Franchise Agreements for your Parks.

Reservation of Rights

Other than your rights expressly granted to you under your Franchise Agreement or your Multi-Unitas described above with respect to your Protected Territory or Development AgreementArea, we (and our affiliates) retain all our rights with respect to the Marks, the System, and Parks anywhere in the world, and retain the right to engage in any business whatsoever and any other right not expressly granted to you in the Franchise Agreement or Multi-Unit Development Agreement, without compensation to you, including the rightrights to do the following:

- 1. Own and operate, and grant to others the right to own and operate, Competitive Businesses (as defined in Item 17 of this Disclosure Document) or other businesses offering similar or identical products, and services, and classes and using the System or elements of the System: (a) under the Marks anywhere outside of your Development Area or Protected Territory(ies), or (b) under names, symbols, or marks other than the Marks anywhere, including inside and outside of the Development Area and Protected Territory(ies) of your Park(s);
- 2. <u>offerOffer</u> to sell, or sell and distribute, any products or services under any trade names, trademarks, service marks or trade dress, including the Marks, anywhere through any distribution channels or methods, which may include including the Internet, retail stores, wholesale, and the Internet (gift cards, distribution or fulfillment centers, or any other existing or future form of electronic commerce); and regardless of the nature or location of the customers with whom such other businesses and distribution channels do business, even if such businesses sell products or services that are identical or similar to, and/or competitive with those that Parks customarily sell;
- 3. <u>acquire, be acquired by Acquire the assets or ownership interests of,</u> merge, affiliate with or engage in any transaction with other businesses (whether competitive or notincluding Competitive Businesses), with units located anywhere or business conducted anywhere, including ineven if such businesses are located in the

<u>Development Area or within</u> the Protected Territory; (ies) of your Park(s), and: (a) rebrand the other businesses to operate under the Sky Zone name, Marks, and System; (b) franchise, license, or create similar arrangements with respect to the other businesses and/or permit the other businesses to continue to operate under another name; and/or (bc) permit the businesses to operate under another name and convert existing Parks (including yours) to such other name;

- 4. Be acquired or become controlled (regardless of the form of transaction) by a business providing products or services similar to those provided at Parks, or by any other business, even if such business operates, franchises, and/or licenses Competitive Businesses; and
- 5. Operate, or grant any third party the right to operate, any Parks that we or our designees acquire as a result of the exercise of a right of first refusal or purchase right that we have under any agreements.

Neither the Franchise Agreement or the Multi-Unit Development Agreement provide you with any options, rights of first refusal or similar rights to acquire additional franchises.

You may not use the Internet, catalog sales or other direct marketing for the sale of Sky Zone branded merchandise or any products or services other than for the sale of jump times and parties. In addition, you may not use, reference, or promote the Sky Zone Marks or System in connection with any current or future form of social media networks or platforms, including, without limitation, Facebook®, X/Twitter®, Bluesky®, LinkedIn® Yelp®, Instagram®, Pinterest®, YouTube®, TikTok®, etc. without our prior approval and only in conformance with our Methods of Operation.

Rockin' Jump and Defy Parks

Our affiliate, Rockin' Jump Franchise, LLC, no longer offers Rockin' Jump franchises for indoor trampoline parks and entertainment facilities, and our affiliate House of Trix, LLC, no longer offers Defy franchises for indoor trampoline and active entertainment facilities. However; however, existing Rockin' Jump and Defy parks may exist within your Protected Territory. And, under or Development Area, if applicable.

<u>For</u> brands other than Sky Zone, we <u>and our affiliates</u> may open company-owned and operated <u>parks</u> ourselves or through affiliates, <u>facilities</u> anywhere, including in your Protected Territory <u>and/or</u> <u>Development Area (as applicable)</u>, and we may solicit or accept orders within your Protected Territory. The corporate address of Rockin' Jump Franchise, LLC, and House of Trix, LLC, is 86 N. University Avenue, Suite 350, Provo, Utah 84601.

Conflicts

There is no formal arrangement in place for resolving conflicts between trampoline parks, including Parks, that we and our affiliates operate and trampoline parks, including Parks, that our and our affiliates' franchisees operate, regarding territory, customers, and franchisor support. If conflicts arise, we will resolve them as we deem appropriate.

There is no restriction on us or any other party to solicit or accept orders from inside your Protected Territory, and we are not required to pay you any compensation for such solicitations. We may use other channels of distribution, such as the Internet, catalog sales and other direct marketing sales and such channels of distribution may be used by us in your Protected Territory, both using the Marks and other trade and service marks.

Sky Zone Franchise Group, LLC

April 2025 FDD

You may not use the Internet, catalog sales or other direct marketing for the sale of Sky Zone branded merchandise or any products or services other than for the sale of jump times and parties. In addition, you may not use, reference, or promote the Sky Zone Marks or System in connection with any current or future form of social media networks or platforms, including, without limitation, Facebook, Twitter, LinkedIn and so on without our prior approval and only in conformance with our Methods of Operation.

Item 13

ITEM 13. TRADEMARKS

We will grant you the right to operate your Park under the name of "Sky Zone". You may also use our other current or future Marks to operate your Park if we permit you to do so.

Our affiliate, Sky Zone, LLC <u>(or "Licensing," as previously defined in Item 1)</u>, has applied for, registered, or been assigned ownership of the following principal trademarks and service marks, which are registered on the Principal Register of the United States Patent and Trademark Office ("**USPTO**"):

Mark	Application Date	Serial Number	Registration Date	Registration No.
SKY ZONE	7/8/1996	75/132,259	3/31/1998	2,148,279
SKY ZONE	11/26/2013	86/130,005	8/5/2014	4,579,253
SKY ZONE	5/24/2018	87/908604 <u>87/90</u> 8,604	6/2/2020	6,070,008
SKYROBICS	1/31/2005	76/630,088	8/1/2006	3,122,435
	3/24/2008	77/430,248	10/21/2008	3,520,145
SKY CLIMBSKYCLIMB	6/24/2015	86/673575 <u>86/67</u> 3,575	5/9/2017	5,200,587
SKY CLIMBSKYCLIMB	2/15/2017	87/336789 6,789	9/19/2017	5,349,117
SKYMANIA	10/10/2005	78/729,809	10/17/2006	3,157,653
SKYSOCKS	7/3/2013	86/001,855	9/9/2014	4,602,835
<u>FUELZONE</u>	<u>3/8/2017</u>	87/363,272	9/5/2017	<u>5,281,388</u>
SKY ZONE TRAMPOLINE PARK &	5/20/2021	90/724,866	5/21/2022	6,743,821
	11/4/2024	98/835,646	<u>Pending</u>	<u>Pending</u>

Mark	Application Date	Serial Number	Registration Date	Registration No.
SKY ZONE	11/4/2024	<u>98/835,752</u>	Pending	<u>Pending</u>
SKY	11/4/2024	98/836,040	Pending	Pending

Neither we nor Licensing <u>hashave</u> any state registrations, nor have we or Licensing filed for any state registration for the above Marks or any other marks. Licensing intends to make all applicable filings to retain its rights in the Marks. Licensing has filed all required affidavits and renewed its registration for the SKY ZONE Mark. <u>Neither we nor Licensing know of any infringing uses or superior prior rights that could materially affect your right to use the Marks</u>.

We have entered into a license agreement effective January 1, 2009 (the "License Agreement"), with Licensing for a 50-year term. The License Agreement grants us the right to use, promote and license the Marks in connection with the grant of franchises to franchisees. If there is a default that is not cured under this license agreement License Agreement between Licensing and us, you will still be able to use the Marks and System in the operation of the Business until the end of the term and any renewal term of your Franchise Agreement. There are no other agreements currently in effect which significantly limit our right to use or license the use of the Marks in any manner material to the franchise.

You must follow our rules and our quality control standards when you use the Marks. You cannot use athe Marks, or any name or markthat is confusingly similar with any Marks: (i) as part of any corporate name or withor legal business name; (ii) as part of an Internet domain name or Internet email address, other than the email address we provide you; (iii) with any prefix, suffix, or other modifying words, terms, designs or symbols (except for those which we license to you; (iv) in any modified form; (v) in connection with the performance or sale of any unauthorized services or products; (vi) in any advertising concerning the transfer, sale or other disposition of your Park or an ownership interest in you; or (vii) in any other manner we have not expressly authorized in writing. You must also operate your franchise in a high-quality manner and adhere to all applicable laws and regulations. You may not use the Marks in connection with the sale of an unauthorized product or service or in a manner not authorized in writing by us. You cannot use the Marks (or any marks or names confusingly similar to the Marks) as an Internet domain name or in the content of any Worldwide Website. You will not register or attempt to register any of the Marks (or any marks or names confusingly similar to the Marks). You will only be permitted to use the Marks during the term of your Franchise Agreement.

You must notify us immediately of any known, actual, suspected, threatened, or apparent infringement or challenge to your use of any MarkMarks, or any claim by any person of any rights in any MarkMarks. We will have the sole discretion ight to actake such action as we deem appropriate and the right to control exclusively control—any litigation, USPTO proceeding or any other administrative proceedingsproceeding arising out of any such infringement, challenge or claim or otherwise relating to any Marks. You must sign all instruments and documents, provide any assistance, and take any action that may be necessary or advisable to protect and maintain our interest in any litigation or other proceeding or otherwise to protect and maintain our interest in the Marks. The Franchise Agreement does not require us to participate in your defense and/or indemnify you for expenses or damages if you are a party to an administrative or judicial proceeding involving a Mark or if the proceeding is resolved unfavorably to you

We agree to reimburse you for all damages and expenses that you incur in responding to any trademark infringement proceeding disputing your authorized use of any Marks under your Franchise Agreement if you have timely notified us of the proceeding, complied with our directions in responding to the proceeding, and have used the Marks in accordance with the terms of your Franchise Agreement and our Methods of Operation; provided, however, we will not pay any of your attorneys' costs or fees if you hire your own attorney. At our option, we may defend and control the defense of any proceeding arising from your use of any Mark under your Franchise Agreement and our Methods of Operation.

You agree that, neither during nor after the term of your Franchise Agreement, will you challenge any of our Marks or any applications or registrations relating thereto.

If we decide We may at any time require you to modify or discontinue the use of the Marks using any Mark and/or to-use one or more additional or substitute names or marks, you must make the changes at your own expense and without claim against us Marks. You will need agree to comply replace the Marks at your Park with the modified, additional, or substitute Marks we specify and comply with all other directions we give regarding the Marks at your Business within a reasonable time of the requestafter receiving notice from us. We doare not actually know of either superior prior rights or infringing uses that could materially affect a franchisee's use of the Marks in any state required to reimburse you for any costs or expenses associated with making such changes, for any loss of revenue due to any modified or discontinued Mark, or for your expenses of promoting any modified or substitute Mark.

Item

ITEM 14. PATENTS, COPYRIGHTS AND PROPRIETARY INFORMATION

We and our affiliates do not own any

Our Patents and Copyrights

There are no patents or copyrights that are material to the franchise. As of the date of this disclosure document, there are no Patents or Copyrightscurrently registered or pending, and no patent applications owned by us or our affiliates that are material to the franchise. "Patents" means any patents as may System.

There is no presently or in the future exist, or which may be modified, changed, or acquired by us or our affiliates, in connection with the operationeffective determination of the U.S. Copyright Office (Library of Congress), the Sky Zone Indoor Trampoline Parks. "Copyrights" means any work entitled to copyright protection we license USPTO, or any court affecting our copyrights. We are not obligated by the Franchise Agreement, or otherwise, to protect any rights you have to use the copyrights or to defend you for against claims arising from your use in connection withof the operation of the Sky Zone Indoor Trampoline Parks and for which we or our affiliates copyrights. We have no actual knowledge of any infringements that could materially affect the ownership, use or licensing of the copyrights.

We do claim copyright protection.

We and our affiliates have copyright rights in the Copyrights. Although neither we nor any of our affiliates have filed an application for a copyright registration for these materials, we and our affiliates have copyright rights and the information is proprietary. Item 11 describes limitations on the use of the Manual by you and your employees. Sky Zone Franchise Group, LLC

April 2025 FDD

There are currently no effective determinations for any material proceeding and no pending infringement, opposition or cancellation proceedings pending in the USPTO or the United States Copyright Office. We do not have any pending or concluded litigation involving any active patents or pending patent applications that are material to the Park or the franchise offering that must be disclosed in this Item.

We do not actually know of any infringement of the patents or copyrights that could materially affect a franchisee. We need not protect or defend our patents or copyrights, although we intend to do so if it is in the franchise system's best interest. We may control any action we choose to bring, even if you voluntarily bring the matter to our attention. We need not participate in your defense and/or indemnify for damage or expenses in a proceeding involving a patent or copyright.

You must notify us immediately of any apparent infringement or challenge to your use of the Intellectual Property. We will have the sole discretion to act as we deem appropriate and the right to exclusively control any litigation or administrative proceedings arising out of any infringement, challenge, or claim. You must sign all instruments and documents, provide any assistance, and take any action that may be necessary or advisable to protect and maintain our interest in any litigation or other proceeding or otherwise to protect and maintain our interest in any Intellectual Property. The Franchise Agreement does not require us to participate in your defense and/or indemnify you for expenses or damages if you are a party to an administrative or judicial proceeding involving any Intellectual Property or if the proceeding is resolved unfavorably to you.

If we decide to (a) modify or discontinue the use of certain Intellectual Property associated with the franchised business, and/or (b) use one or more additional or substitute Intellectual Property, you must make the changes at your own expense and without claim against us. You will need to comply within a reasonable time of the request.

You acknowledge that the System and the methods of operation licensed by us for the operation of a Sky Zone Indoor Trampoline Park are proprietary, confidential trade secrets of ours, and you will maintain the confidentiality of all materials and information lent or otherwise always furnished to you by us, including after the termination or expiration of the Franchise Agreement, for any reason.

In addition, you shall not, during the term of the Franchise Agreement (other than as necessary to operate the Park) or after its expiration or termination, for any reason, communicate or divulge to any others, any information or knowledge concerning the System and any trade secrets except those in the public domain. You also must exercise the highest degree of diligence and to make every effort to maintain the absolute confidentiality of all trade secrets and proprietary rights during and after the term of the Franchise Agreement. and proprietary rights in the original materials used in the operation of Parks, including the Operations Manual, bulletins, correspondence and communications with our franchisees, training, advertising and promotional materials, and other written materials relating to the operation of Parks and the System (the "Copyrighted Materials"). We treat some of this information as trade secrets and you must treat any of this information we communicate to you as Confidential Information (as defined below). You may not use any of our Copyrighted Materials without our prior written permission. This includes the display of the Copyrighted Materials in connection with any digital advertising or social media content.

Confidential Information

You will have access to proprietary and confidential information relating to the development and operation of Parks (the "Confidential Information").

During the term of your Franchise Agreement and thereafter, you and your owners will not acquire any interest in Confidential Information, other than the right to utilize Confidential Information disclosed to you in operating the Park during the term of your Franchise Agreement. The Confidential Information is proprietary to us and our Affiliates and includes our trade secrets. You and your owners: (1) may not use the Confidential Information in any other business or capacity, which would constitute an unfair method of competition; (2) must exert your best efforts to maintain the confidentiality of the Confidential Information; (3) may not make unauthorized copies of any portion of the Confidential Information disclosed in written, electronic or other form; (4) must implement all reasonable procedures we prescribe from time to time to prevent unauthorized use or disclosure of the Confidential Information, including the use of nondisclosure agreements with individuals who will have access to the Confidential Information; and (5) will not sell, trade or otherwise profit in any way from the Confidential Information, except using methods approved by us.

Works Made-For-Hire

The Franchise Agreement also provides that all ideas, concepts, techniques, or materials concerning, or processes related to your Park or the ParkSystem, whether or not protectable intellectual property and whether created by or for you or your owners or employees, must be promptly disclosed to us and will be deemed to be our sole and exclusive property, part of the System, and works made-for-hire for us. If any item does not qualify as a "work made-for-hire" for us, you must assign ownership of that item and all related rights to that item, to us and must take whatever action (including signing and causing your owners, employees and contractors to sign assignments or other documents) we request to show our ownership or help us obtain intellectual property rights in the item.

You agree that, neither during nor after the term, you will not challenge any of our Intellectual Property or any applications or registrations relating thereto.

<u>ITEM 15.</u> OBLIGATION TO PARTICIPATE IN THE ACTUAL OPERATION OF THE FRANCHISE BUSINESS

You (or your Responsible Person) agree to personally manage and operate the franchise your Park and will not, without our prior written consent, delegate your (or your Responsible Person's) authority and responsibility with respect to management and operation. You agree that you (or your Responsible Person) will always faithfully, honestly, and diligently perform the obligations hereunder and continuously exert best efforts to promote and enhance the franchise Park. Each of your Owners must jointly and severally be bound by the terms of the Franchise Agreement and personally guarantee your (or your Responsible Person's) performance.

If you are, or at any time become, a business corporation, partnership, limited liability company or other legal entity, you must designate <u>a "Responsible Person"</u> in "Appendix A" to <u>either</u> the Franchise Agreement or Exhibit "C" to the Multi Unit Development Agreement a "Responsible Person," MUDA (as applicable) who is an individual approved by us who: (a) has the authority to bind

you regarding all operational decisions with respect to your Park; and (b) has completed our training program to our satisfaction.

You (or your Responsible Person): (a) must exert best efforts to the development and operation of your Park and all other Parks you own; and (b) absent our prior approval, may not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitments or which may otherwise may conflict with your obligations under the Franchise Agreement or MUDA.

Your Park must be managed by you (or your Responsible Person) or by <u>an on-site general</u> manager or <u>shift supervisorassistant manager or an hourly team lead</u> who has completed <u>ourall appropriate</u> training <u>programprograms</u> to our satisfaction. Your Responsible Person or manager need not have an equity interest in the franchise.

As more fully described in the Franchise Agreement, you (or your Responsible Person) must implement all reasonable procedures we periodically prescribe to prevent unauthorized use or disclosure of confidential information. These procedures include the use of nondisclosure agreements with your owners, officers, directors, managers, assistant managers, and yourshift supervisors, Responsible Person, and any other individuals who will have access to our Confidential Information. You and your owners must deliver these agreements to us upon our request. Aside from requiring your Responsible Person, managers or shift supervisorssuch personnel to sign nondisclosure agreements, we do not currently require you to place any other restrictions on your Responsible Person, managers or shift supervisors. AtUpon the endexpiration or termination of the term of ayour Franchise Agreement, you must deliver to us all eonfidential informationConfidential Information in your possession.

If you or one of your affiliates have entered into a Multi-Unit Development Agreement MUDA with us and are entering into a Franchise Agreement under that Multi-Unit Development Agreement to such MUDA, and you are a business corporation, partnership, limited liability company or other legal entity, you must be at least 51% owned or controlled by a person or group of people that has at least a 51% ownership interest in and voting control of the entity that signed the Multi-Unit Development Agreement MUDA, except as we approved otherwise approve in writing in our business judgment. We have the right to approve in advance your ownership structure.

If you are a partnership, corporation, limited liability company or other legal entity, any person who has a 10% or greater than a 10% interest in you, or such other persons as we may designate (including owners' spouses), must undertake to be personally bound, jointly and severally, by your obligations under the Franchise Agreement and Multi-Unit Development Agreement (as applicable). Copies of these guarantees are contained in Appendix B of both the Franchise Agreement and Exhibit "B" of the Multi-Unit Development Agreement MUDA, which are attached to this Disclosure Document. We do not currently require the spouses of owners to sign a personal guarantee.

Item 16

ITEM 16. RESTRICTIONS ON WHAT THE FRANCHISEE MAY SELL

Authorized Products and Services Generally

You may offer only the products and services we have <u>authorized and</u> approved in writing. You must offer all services <u>and</u>, products <u>and merchandise</u> that we designate as required for franchisees. <u>You may sell products and services only in the varieties, forms, and packages that we have approved in accordance <u>with our Methods of Operation</u>. There are no limits on our right to make modifications to the approved products and services periodically listed in the Operations Manual, <u>and we may designate specific products and services as optional or mandatory</u>. Any failure to comply with our Methods of Operation or Operations Manual may result in termination of your Franchise Agreement. You may use only marketing and promotional materials that we have approved. You are not limited in the type of customers to whom you may sell approved products or services, <u>but we reserve the right to implement Methods of Operation regarding the solicitation of customers in the future.</u></u>

Memberships

You must offer and sell rights of access to your Park, referred to individually as a "Membership" or collectively as "Memberships." You will only offer Memberships for sale in strict compliance with our Methods of Operation. All Memberships must be evidenced by a Membership Agreement and may not be for a term that extends beyond the expiration of your Franchise Agreement. We may provide you a form of Membership Agreement, and if we do so, you will use the form of Membership Agreement that we provide to you, and you must make modifications to the form to ensure it complies with all applicable laws for your Park, subject to our final written approval. We own all information relating to customers and members of your Park. We may contact any member(s) associated with your Park at any time.

<u>Item 17</u>

ITEM 17. RENEWAL, TERMINATION, TRANSFER AND DISPUTE RESOLUTION

THE FRANCHISE RELATIONSHIP

This table lists important provisions of the Franchise Agreement, MUDA and related agreements. You should read these provisions in the agreements Franchise Agreement attached to this Disclosure Document.

	<u>Article</u> Secti	
	<u>on</u> in	
	Franchise	
Provision	Agreement	Summary
a. Length of the franchise term	3.1	10 years.
b. Renewal or extension of the term	14 14.1, 14.2	A successor franchise may be granted for
		a period of 10 years if you meet our
		conditions.
c. Requirements for franchisee to renew	14.1	Deficiencies are corrected within a
or extend		reasonable cure period we specify; you
		have substantially complied with the
		Franchise Agreement; you <u>are in</u>
		compliance with the Franchise
		Agreement upon providing notice to
		renew and on the renewal date; you
		remodel or relocate the Park; you provide
		us not less than <u>6six</u> months nor more
		than 12 months' notice prior to

	Article Secti	
	on in	
	Franchise	
Provision	Agreement	Summary
		expiration; you pay successor initial franchise fee; and may be required to sign a Franchise Agreement with materially different terms and conditions than your original Franchise Agreement alongside guarantees and general releases.
d. Termination by franchisee	15.1	Subject to state law, you may terminate the agreement <u>upon</u> 30 days' <u>written</u> <u>notice</u> after we materially fail to comply with <u>the</u> Franchise Agreement after <u>a</u> 60-day cure period <u>following our receipt of</u> written notice.
e. Termination by franchisor without	Not	
cause \	Applicable	
f. Termination by franchisor with cause	15.2 , 15.3	We can terminate only if you commit one of several violations.
g. "Cause" defined – curable defaults	15.3 <u>15.2</u>	Suit to foreclose any lien or mortgage not dismissed within 30 days; refusal or failure to comply with safety, cleanliness, or sanitation standard not corrected within 24 hours; failure to pay us or affiliates or suppliers within 30 days of due date; and or 10 days of written notice, whichever is shorter; two written notices of non-payment within preceding 24 months; failure to provide access to Technology System within 10 days of written notice; failure to comply with any offerother provision of Franchise Agreement within 30 days of notice of failure to comply.
h. "Cause" defined – non- curable defaults	15.2	We can terminate upon delivery of notice of termination for insolvency, if you are adjudicated bankrupt, if you file a petition for bankruptcy, have a receiver appointed, if you make a general assignment for the benefit of your creditors; abandonment of Parkor failure to operate the Park for three consecutive business days; unauthorized Transfertransfer; material misrepresentation in connection with purchase of franchise; termination of right to occupy premises; conviction of a felony or crime involving consumer fraud or moral turpitude; unauthorized disclosure of

	Article Secti	
	<u>on</u> in	
	Franchise	
Provision	Agreement	Summary
		Confidential Information; underreporting
		of Total-Gross RevenueSales; repeated
		violationsnon-compliance with Franchise
		Agreement; failure to pay
		taxes; failure to appoint a manager within
		15 days following death or disability or
		manager's failure to complete training
		within 60 days following appointment;
		failure to conduct business with regard to
		<u>public conventions</u> and <u>morals</u> ; violation
		of covenants not to compete; and failure
		to conduct background checks for
		employees you hire. We can further
		terminate the Franchise Agreement if we
		terminate any other franchise agreement,
		multi-unit development agreement, or
		any other agreement between you and us
		(or your affiliates and our affiliates).
i. Franchisee's obligations on	16	Pay us what you owe us, including
termination/ nonrenewal		liquidated damages, cease using the Marks
		and System, and follow our termination
		procedures, including de-branding your
		locationPark, cancelcancelling any
		assumed name registrations, and stop
		using ceasing to use and returning (or at
		our direction, destroying) copies of our
	10.1	Confidential Information.
j. Assignment of contract by franchisor	13.1	Fully transferable by us.
k. "Transfer" by franchisee - defined	1.2	A transfer includes the direct or indirect.
		voluntary or involuntary, sale, assignment,
		transfer, grant of a security, or other
		disposition of the Franchise Agreement,
		any form of ownership interest in you or
		the Park's assets, including any merger or
		transfer because of a divorce, insolvency,
		dissolution or otherwise by operation of
1 Eugrahiaan'a angustral of turn of the	12 2 12 4	law, by reason of death or foreclosure.
1. Franchisor's approval of transfer by	13.2, 13.4	All transfers require our <u>prior</u> approval,
you		except that you may transfer the Franchise
		Agreement to an entity wholly owned and controlled by you.
m Conditions for franchises ammayal of	13.3	The conditions include the character,
m. Conditions for franchisor approval of transfer	13.3	aptitude and experience of the transferee,
u ansici		you having paid; paying all amounts
		owed, and paying to us a \$15,000
		refundable fee deposit; transferee and
		returnuable fee deposit, transferee and

	Article Secti	
	<u>on</u> in	
	Franchise	
Provision	Agreement	Summary
		others completing initial training and
		accepting the terms of the Franchise
		Agreement or signing our then current form
		of franchise agreement, you pay
		(including personal guarantees); providing
		executed copies of any documents to
		effectuate the transfer; paying a transfer fee
		in the amount of 50% of the then-current
		franchise fee, (or \$2,500 for ownership
		interest transfers among your owners); you
		(and your Owners) signing a general
		release, we approve our approval for the
		general material terms and conditions of
		the transfer; the transferee's obligations
		under any promissory notes or similar
		agreements must be subordinate to
		obligations due to us under the Franchise
		Agreement; providing evidence of notice to
		or approval from the landlord in connection
		with your lease for the premises; correcting
		any deficiencies of the Park or the
		transferee's agreement to modernize, renovate, or upgrade the Park in
		accordance with our requirements;
		providing evidence of taking appropriate
		measures to effectuate the transfer
		(including transferring all necessary
		business licenses and material agreements);
		and executing an agreement to remain
		bound by certain post-termination
		covenants.
n. Franchisor's right of first refusal to	13.7, 13.8, 13.9	For 30 day right to exercise for all
acquire franchisee's business	<u>, , , , , , , , , , , , , , , , , , , </u>	third-party bona fide offers.
o. Franchisor's option to purchase	16.4	60- day option upon termination or
franchisee's business		expiration of the Franchise Agreement
p. Death or disability of franchisee	13.5	Treated as a transfer
q. Non-competition covenants during the	1.2, 17.1	A "Competitive Business" is any business
term of the franchise		(other than any franchise operated under a
		separate franchise agreement entered into
		with us or our affiliates) operating as, or
		granting franchises or licenses to others to
		operate any business operating as, any
		family recreational center including, but
		not limited to: (i) a trampoline park; (ii)
		centers offering inflatable indoor bouncing,
		rock climbing, go carts, or miniature golf;

	Article Secti	
	on in	
	Franchise	
Provision	Agreement	Summary
		(iii) active entertainment facilities; or (iv)
		similar family entertainment
		establishments; this definition is not
		intended to include any centers specializing
		in gymnastics, which may contain
		trampolines. Subject to state law, no direct
		or indirect involvement in the operation of
		any Competitive Business—or, diversion
		of business or customers or solicitation of
		employees to Competitive Businesses,
		engagement in any other activity which
		might adversely affect the goodwill of the
		Marks and System, solicitation,
		interference, or attempt to interfere with
		our or our affiliates' relationships with
		customers, vendors, consultants, or other
	15.0	franchisees.
r. Non-competition covenants after the	17.2	Subject to state law, no direct or indirect
franchise is terminated or expires		involvement in the operation of any
		Competitive Business for 2two years
		after termination, expiration, or transfer
		at the Location Park's premises, within
		15 miles of the Location Park's premises,
		or within 15 miles of any other Park in
		operation or under construction at time of
		termination. You cannot solicit, interfere,
		or attempt to interfere with our or our
		affiliates' relationships with customers,
		vendors, consultants, or other franchisees
		or engage in any other activity that might
		adversely affect the goodwill of the Marks
		and the System.
s. Modification of the agreement	19.24 <u>19.23</u>	Requires consent of both parties and must
		be in writing
t. Integration / merger clause	19.22 , 20.5	Oral statements not binding. Franchise
		Agreement is the entire agreement. Only
		the terms of the franchise agreement are
		binding (subject to state law). Any
		representations or promises outside the
		Disclosure Document and Franchise
		Agreement may not be enforceable.
		Notwithstanding the foregoing, nothing in
		any Franchise Agreement is intended to
		disclaim the express representations made
		in this Disclosure Document.
u. Dispute resolution by arbitration or	19.11	
mediation		by mediation informal resolution and
u. Dispute resolution by arbitration or	19.11	Subject to state law, all disputes resolved

	Article Secti				
	on in				
	Franchise	_			
Provision	Agreement	Summary			
		arbitration except for actions for declaratory or equitable relief—and; actions in ejectment or for possession of any interest in real or personal property; actions which cannot be arbitrated under applicable law; and our decisions in the first instance to issue notices of default or termination or undertake other conduct with respect to the franchise relationship.			
v. Choice of forum	19.11(b) <u>; 19.13</u>	City elosest to in which our then current headquartersprincipal place of business is located (currently, Provo, Utah) (subject to applicable state law).			
w. Choice of law	19.12	MissouriState in which our principal place of business is located (currently, Utah) (subject to applicable state law).			

MULTI-UNIT DEVELOPMENT AGREEMENT

This table lists certain important provisions of the $\frac{\text{Multi-Unit Development}}{\text{Agreement}\underline{\text{MUDA}}}$. You should read these provisions in the $\frac{\text{Multi-Unit Development}}{\text{Multi-Unit Development}}$

	Section in	
Provision	Agreement	Summary
a. Length of the term of the franchise	<u>34</u>	The openingearlier of the execution
	_	of the franchise agreement for the last
		Business listed under Park required to
		be developed pursuant to the
		Multi-Unit Development
		AgreementSchedule or the last day
		of the last development deadline set
		forth in Appendix A to the MUDA,
		unless sooner terminated.
b. Renewal or extension of the term	Not	You do not have the right to renew or
	Applicable.2.E.	extend the Multi-UnitMUDA. If you
		fail to comply with the Development
		AgreementSchedule, we have the
		right to refrain from exercising our
		termination rights in favor of granting
		you a written extension to the
		Development Schedule, which will
		require you to pay us an extension fee.
c. Requirement for franchisee to renew or	Not	You do not have the right to renew or
extend	Applicable.2.E.	extend the Multi-Unit Development
		Agreement. You do not have the right

	Provision	Agreement	Summary
	Provision	Section in Agreement	to renew or extend the MUDA. If you fail to comply with the Development Schedule, we have the right to refrain from exercising our termination rights in favor of granting you a written extension to the Development Schedule. Such an extension may, in our business judgment, be conditioned on any or all of the following: reducing the size of the Development Area; a modified Development Schedule (in terms of timing and/or number of Parks to be opened); your execution of our then-current form of general release; and your execution of our then-current form of MUDA, the terms of which may differ substantially from the terms contained herein. If we grant you a
ד ג	Formation by from abises	Not Applicable	written extension, you may be required to pay to us an extension fee equal to half of your Development Fee.
	Termination by franchisee Γermination by franchisor without cause	Not Applicable. Not	Not Applicable. We will not may terminate the
e. T	termination by tranchisor without cause	Applicable. <u>10.</u>	Multi-Unit Development
		<u>B.</u>	Agreement MUDA without cause if we terminate any other multi-unit development agreement, franchise agreement, or any other agreement between us and you (or your affiliates and our affiliates).
f. T	Termination by franchisor with cause	9, 10 <u>.A.</u>	We can terminate the Multi Unit Development Agreement MUDA if you default or fail to comply with your obligations.
	'Cause'' defined – curable defaults	9 <u>10.A.</u>	HWe can terminate upon delivery of notice to you if you fail to furnish reports or any other documentation within 10 days after receiving notice; or if you fail to comply with the Development Schedule or withany other terms of the Multi Unit Development Agreement, you will have (30) days to MUDA and do not cure within 10 days after receiving notice from us.
h. "	'Cause'' defined – non-curable defaults	9 <u>10.A.</u>	Termination of a Franchise Agreement. We can terminate upon delivery of notice of termination if you fail to pay the Development Fee; you

	Section in	
Provision	Agreement	Summary
		cease or threaten to cease to carry on the
		obligations granted to you under the
		MUDA, or take or threaten to take any
		action to liquidate your assets, or if you
		do not pay any debts or other amounts
		incurred by you in exercising the
		Development Rights hereunder when
		such debts or amounts are due and
		payable; you fail to comply with the
		Development Schedule (or any
		development deadline therein); if you
		are adjudicated bankrupt, if you file a
		petition for bankruptcy, have a receiver
		appointed, if you make a general
		assignment for the benefit of your
		creditors; material misrepresentation in
		connection with purchase of franchise;
		unauthorized transfer; conviction of a
		felony, crime, or other behavior likely to
		adversely affect the reputation and
		goodwill of the Marks and System; or if
		you fail on three or more occasions
		within a 12 consecutive month period to
		comply with the MUDA or two or more
		occasions within any six consecutive
		month period to comply with the same
		obligation under the MUDA.
i. Franchisee obligations on	14 <u>10.C., 10.D.,</u>	All development rights revert to us.
termination/non-renewal	10.E., 10.F.	Unless concurrently terminated, any
		franchise agreements entered into
		between you and us remain in effect.
		You must cease using all of our
		Confidential Information, pay any
		amounts owing to us, and abide by the
		post-termination covenants not to
		compete.
j. Assignment of contract by franchisor	15 6.A.	No restriction on our right to assign.
k. "Transfer" by franchisee – defined	16 6.B.	Transfer or conveyance of rights under
		the Agreement MUDA or in the
		ownership of the entity that is a party
		to the Multi-Unit Development
		Agreement MUDA with us.
Ll.Franchisor approval of transfer by you	16 6.B.	You must obtain our consent and
= 11		transfer all rights under all agreements
		with us. The development rights
		cannot be transferred. For transfers of
		direct or indirect ownership interests in
		you which are less than or equal to
		49% of your ownership interests, you
		TO OF YOUR OWNERSHIP HICIOSIS, YOU

	Provision	Section in Agreement	Summary
			must obtain our consent.
m.	Conditions for franchisor's approval of transfer	16 <u>6.B.</u>	The proposed transferee must provide information to us sufficient for us to assess the transferee's business experience and aptitude; you must be in compliance with the Development Schedule, the MUDA, and all other franchise agreements in effect between you (or your affiliates) and us; you must be in compliance with the MUDA for at least 60 days before requesting the transfer and between the request date and closing date; the transferee cannot have ownership interests in Competitive Businesses; you must pay us a \$2,500 transfer fee; your transferring owners must sign a consent to transferee owners' must sign personal guarantees; the transferee's obligations under any promissory notes or similar agreements must be subordinate to obligations due to us under the MUDA; and the your
			owners must abide by applicable
		NY . A 1' 11	post-termination obligations.
n.	Franchisor's right of first refusal to acquire franchisee's business	Not Applicable	Not Applicable
0.	Franchisor's option to purchase franchisee's business	Not Applicable	Not Applicable
p.	Death or disability of franchisee	Not Applicable	Not Applicable
q.	Non-competition covenants during the term of the franchise	13.1 <u>5.B.</u>	A "Competitive Business" is any business (other than any franchise operated under a separate franchise agreement entered into with us or our affiliates) operating as, or granting franchises or licenses to others to operate any business operating as, any family recreational center including, but not limited to: (i) a trampoline park; (ii) centers offering inflatable indoor bouncing, rock climbing, go carts, or miniature golf; (iii) active entertainment facilities; or (iv) similar

n	Section in	G.
Provision	Agreement	Summary
		family entertainment establishments;
		this definition is not intended to
		include any centers specializing in
		gymnastics, which may contain trampolines. No direct or indirect
		involvement in the operation of any
		Competitive Business other than the
		business authorized in the Multi-Unit
		Development Agreement, no diversion
		of <u>business or customers or solicitation</u>
		of employeesto Competitive
		Businesses, cannot engage in activities
		which might adversely affect the
		goodwill of the Marks and System.
r. Non-competition covenants after the	13.2 <u>10.E.</u>	No direct or indirect involvement in a
franchise is terminated or expires		Competitive Business for 2 years: (i)
		within the Development Area, (ii)
		within 15 miles of any Park developed
		by you, or (iii) within 15 miles of any
		other Park in operation or under
		construction upon the expiration or
a Modification of the Agreement	27 17.K.	termination of the MUDA. Any modification must be in writing
s. Modification of the Agreement	27 17.K.	and signed by both parties
t. Integration/merger clause	26 <u>17.K.</u>	Oral statements are not binding. The
		MUDA is the entire agreement. Only
		terms of the Multi-Unit Development
		Agreement MUDA are binding
		(subject to state law). Any
		representations or promises not
		contained in the Multi-Unit
		Development Agreement, Franchise Agreement MUDA or the Disclosure
		Document may not be enforceable.
		Notwithstanding the foregoing,
		nothing in any Multi-Unit
		Development Agreement MUDA is
		intended to disclaim the express
		representations made in this Disclosure
		Document.
u. Dispute resolution by arbitration or	19 <u>12</u>	All disputes will be resolved by
mediation		mediationinformal resolution and
		arbitration except for actions for
		actions for declaratory or equitable
		relief-and; actions in ejectment or for
		possession of any interest in real or
		personal property; actions which
		cannot be arbitrated under applicable law; and our decisions in the first
		iaw, and our decisions in the first

Provision	Section in Agreement	Summary
		instance to issue notices of default or termination or undertake other conduct with respect to the franchise relationship.
v. Choice of forum	19, 21 <u>12.B., 14</u>	City closest toin which our then current headquartersprincipal place of business is located (currently, Provo, Utah) (subject to applicable state law).
w. Choice of law	<u>2013</u>	MissouriState in which our principal place of business is located (currently, Utah) (subject to applicable state law).

Item 18

ITEM 18. PUBLIC FIGURES

We do not currently use any public <u>figurefigures</u> to promote our <u>franchiseSystem</u>; <u>however</u>, <u>we may use public figures in the future</u>.

Item 19

ITEM 19. FINANCIAL PERFORMANCE REPRESENTATIONS

The FTC's Franchise Rule permits a franchisor to provide information about the actual or potential financial performance of its franchised and/or franchisor-owned outlets, if there is a reasonable basis for the information, and if the information is included in the disclosure document. Financial performance information that differs from that included in Item 19 may be given only if: (1) a franchisor provides the actual records of an existing outlet you are considering buying; or (2) a franchisor supplements the information provided in this Item 19, for example, by providing information about possible performance at a particular location or under particular circumstances.

This Item includes certain information about Gross Sales and EBITDA for franchisee and corporate-owned Sky Zone Parks in the United States during the 12-month period ended December 31, 20232024 (the "20232024 Fiscal Year" or the "Reporting Period"). As used in this Item, "Gross Sales" has the same meaning as in the Franchise Agreement; that is, the total amount of all sales of products, services, programs, merchandise, food, and beverages, sold from, through or in connection with the Park, whether for cash, on credit, barter or otherwise and exclusive of applicable sales, use or service taxes. Gross Sales additionally exclude refunds that are provided to customers (not including chargebacks).

1. MODEL PARKS Model Parks

Part 1 of this Item sets forth certain Gross Sales and EBITDA information for Model Parks, a subset of all franchisee-owned Sky Zone Parks. As used in this Item, "**Model Parks**" refers to franchisee-owned Parks that have: (i) have at least 25,000 square feet of space and, (ii) have 4 or more private party rooms; and (iii) were open and operating throughout the Reporting Period.

Table 1A: Gross Sales

Model Parks			
# of Parks	311		
Average Gross Sales	\$ 2,894,787 2,929,843		
#/% Met or Exceeded Average	15 / 48% <u>14 / 45.2%</u>		
Maximum Gross Sales	\$ 6,256,721 7,090,723		
Minimum Gross Sales	\$ 1,350,657 695,009		
Median Gross Sales	\$ 2,878,874 2,726,529		
Average # of Party Rooms	4.2		
Average # of Square Feet	31,226 31,410		

Notes to Table 1A

Note 1. Table 1A includes historical Gross Sales data for the 31 Model Parks that were open and operating as of December 31, 20232024 and that reported Gross Sales and EBITDA information for the Reporting Period. As of December 31, 20232024, there were 126120 franchisee-owned Sky Zone Parks in the United States. 33 of those Parks were open and operating throughout the Reporting Period and had at least 25,000 square feet of space and 4 or more private party rooms, meeting the criteria for Model Parks. Of those 33 Model Parks, 2 were excluded because they were did not open for 12 months as of December 31, 2023 report expense & EBITDA information for the full Reporting Period.

Table 1B: EBITDA

Model Parks							
		Average		Median		;	# / % Met or
		3				Exc	ceeded Average
EBITDA ¹						1	4 / 45.2% <u>13 /</u>
		\$ 836,312 887,583		\$ 761,	384 <u>857,826</u>		41.9%
EBITDA							15 / 48.4%
% ²		30.4%	28.9	% ³⁴ =	28.0% ⁴		

Notes to Table 1B

Note 1. "EBITDA" means the earnings before interest, taxes, depreciation, and amortization. As is customary, it excludes expenses related to debt services costs, whether principal or interest. The following categories of expense items were subtracted from Gross Sales as part of the Parks' EBITDA calculations:

Expense Category	Notes
Cost of Goods Sold ("COGS")	Includes the total cost of all food, beverages, merchandise, and other costs related to products and services sold by the Parks, including distribution and delivery costs.
Occupancy Expenses	Includes rent (including both minimum rents and percentage rents), utilities (e.g., electricity, gas, water, cable, internet, telephone), real

Expense Category	Notes
	estate taxes and assessments levied against the property upon which the Park is located, and other pass-through expenses from the landlord, and any sales or other taxes imposed thereon and any pass-through expenses from the landlord.
Advertising Expenses	Includes the Parks' contributions to the National Advertising Fund as well as other digital and non-digital advertising, promotional and local marketing expenses for the Parks.
Payroll Expenses	Includes personnel wages, management salaries, benefits and payroll taxes. Excludes owner's compensation.
Insurance	Includes cost of the Sky Zone Master Insurance Program (providing general liability coverage to the Parks), property liability and workers compensation.
Other Costs	Includes royalty fees, repairs & maintenance, processing fees, and other minor expenses.

Note 2. "EBITDA %" means EBITDA as a percentage of Gross Sales.

Note 3. Average EBITDA % is the amount equal to the Average EBITDA for the group of 31 Model Parks measured divided by the Average Gross Sales for the group.

Note 4. Median EBITDA % is the amount equal to the midpoint value of the 31 separate EBITDA % amounts determined for the individual Model Parks measured.

2. Franchisee-Owned Parks (System-Wide)

Part 2 of this Item sets forth certain Gross Sales and EBITDA information for all franchisee-owned Parks in the United States (inclusive of Model Parks), broken into quartiles by Park size (square footage).

As of December 31, 20232024, there were 126120 franchisee-owned Sky Zone Parks open and operating in the United States. Part 2 of this Item sets forth Gross Sales and EBITDA information for 10197 franchisee-owned Sky Zone Parks in the United States, excluding: (i) 169 Parks that did not report expense & EBITDA information for the full Reporting Period, (ii) 45 Parks that were not open for 12 months as of December 31, 2023 opened during the Reporting Period, (iii) 47 Parks that completed rebrands which rebranded from the Defy® or Rockin' Jump® brands to the Sky Zone brand during the Reporting Period, and (iv) 1 additional Park was excluded because itwhich was not open continuously during the Reported Period (temporary closure of more than four consecutive weeks), and (v) 1 non-traditional Park.

Table 2A: Gross Sales

Cigo Dongo		Gross Sales				
Size Range (Sq Ft)	# Parks	Average	Median	Min	Max	# / % Exceeding Avg.
16,000-22,500	<u>22</u>	\$1,785,515	\$1,995,068	\$314,960	\$3,503,794	14 / 63.6%
22,501-25,000	<u>30</u>	\$2,056,800	\$2,091,207	\$782,504	\$3,984,928	<u>16 / 53.3%</u>

Sizo Dongo		Gross Sales						
Size Range (Sq Ft)	# Parks	Average Median		Min	Max	#/% Exceeding Avg.		
25,001-30,000	<u>23</u>	\$2,533,896	\$2,433,305	\$850,587	\$7,090,723	9/39.1%		
30,001-45,000	24 <u>22</u>	\$2,701,971	\$2,719,676	\$ 1,459,425	\$4,012,476	13 /		
		2,632,018	2,778,790	695,009	4,544,251	54.2 <u>59.1</u> %		
25,001-30,000	24	\$2,478,372	\$2,256,412	\$917,856	\$6,256,721	9 / 37.5%		
23,001-25,000	24	\$2,135,266	\$2,011,195	\$785,584	\$3,990,724	9 / 37.5%		
16,000-23,000	29	\$1,845,492	\$1,774,794	\$705,724	\$5,036,458	14 / 48.3%		
Aggregate	101 97	\$2,268,256	\$2,070,147	\$705,724	\$6,256,721	43/ 42.6% <u>47</u>		
riggiogate	10121	2,238,859	2,184,637	314,960	7,090,723	<u>/ 48.5%</u>		

Table 2B: EBITDA

Size Range	# Dowles	EBITDA ¹				
(Sq Ft)	# Parks	Average	Median	#/% Exceeding Avg.		
16,000-22,500	<u>22</u>	<u>\$478,919</u>	<u>\$468,767</u>	12 / 54.5%		
22,501-25,000	<u>30</u>	<u>\$522,631</u>	\$528,660	18 / 60.0%		
25,001-30,000	<u>23</u>	<u>\$776,886</u>	<u>\$684,489</u>	11 / 47.8%		
30,001-45,000	24 22	\$ 715,547 750,97				
	_	4	\$ 772,153 <u>792,655</u>	10 / 41.7 <u>45.5</u> %		
25,001-30,000	24	\$682,909	\$576,434	9 / 37.5%		
23,001-25,000	24	\$525,344	\$501,467	11 / 45.6%		
16,000-23,000	29	\$437,106	\$364,580	10 / 34.5%		
Aggregate	<u>10197</u>	\$ 582,646 <u>624,79</u> <u>3</u>	\$ 563,569 <u>595,928</u>	4 7/ 46.5% 48 / 49.5%		

Notes to Table 2B

Note 1. Please refer to Note 1 to Table 1B for the definition of EBITDA.

3. Corporate Parks

Part 3 of this Item sets forth certain information about Gross Sales for all corporate-owned Sky Zone Parks in the United States. Corporate-owned Sky Zone Parks operate substantially similar to franchisee-owned Sky Zone Parks are typically in larger metro areas.

As of December 31, 20232024, there were 74114 corporate-owned Sky Zone Parks in the United States. The financial performance representation in Part 3 of this Item relates to 3771 corporate-owned Sky Zone Parks, excluding: (a) 3 Parks that opened during the Reporting Period, (b) 617 Parks that were acquired from Sky Zone franchisees during the Reporting Period, and (c) 2825 Parks that completed rebrands which rebranded from the Defy® or Rockin' Jump® brands to the Sky Zone brand during the Reporting Period, and (d) 2 Parks that closed during the Reporting Period.

Corpora	ate <u>-Owned</u> Parks
# of Parks ¹	37 71
Average Gross Sales ¹	\$ 3,802,036 3,816,235
#/% Met or Exceeded Average	14 (37.8%) 28
	(39.4%)
Maximum Gross Sales	\$ 6,838,456 9,759,745
Minimum GossGross Sales	\$ 1,625,297 <u>1,268,575</u>
Median Gross Sales	\$ 3,440,483 <u>3,521,670</u>
Average # of Square Feet	30,421 <u>30,041</u>
Median # of Square Feet	30,250
Average # of Party Rooms	5.3 <u>4.2</u>
Median # of Party Rooms	5 <u>4</u>

Some Parks have soldearned these amounts. Your individual results may differ. There is no assurance you'll sell will earn as much.

Written substantiation of the data used in preparing this financial performance representation will be made available to you upon reasonable request. Information for franchisee-owned Sky Zone Parks is based on sales and expense information reported to us by franchisees. The We did not audit or otherwise verify the accuracy of the information submitted.

Factors that franchisees provided was not audited might adversely impact earnings for a given Park include the general public's perception of indoor trampoline parks and similar recreational facilities, increased competition in the indoor trampoline park industry, actions by franchisees that are out of our control that could adversely impact the System, and the status of the general economic environment. Factors that might adversely impact Gross Sales include the actual cost of wages paid to your employees, which could vary periodically and by market due to the status of the general economic environment. The negative impact of such factors would also adversely impact a franchisee's net income, profits and earnings.

Other than the preceding financial performance representations, we do Sky Zone Franchise Group, LLC does not make any financial performance representations. We also do not authorize our employees or representatives to make any such representations either orally or in writing. If you are purchasing an existing Sky Zone Park, however, we may provide you with the actual records of that Sky Zone Park. If you receive any other financial performance information or projections of your future income, you should report it to the franchisor's management by contacting Mike Revak, PresidentChief Operating Officer, Sky Zone Franchise Group, LLC, at 860 N. University Avenue, Provo, Utah 84604, the Federal Trade Commission, and the appropriate state regulatory agencies.

<u>Item</u>

ITEM 20. OUTLETS AND FRANCHISEE INFORMATION

Table No. 1 Systemwide Outlet Summary For years 2021 through February 29, Years 2022 to 2024(1)

Outlet Type	Year	Outlets at the Start of the Year	Outlets at the End of the Year	Net Change
	2021	120	117	-3
	2022	117	118	+1
Franchised	2023	118	126	+8
	FEB-2024	126	121 <u>120</u>	<u>-5_6</u>
	2021 <u>2022</u>	21 2 6	22<u>44</u>	<u>+1+18</u>
	2022	22	33	+11
Company- Owned*(2)	2023	33 4 4	67 7 <u>1</u>	+ <u>34</u> <u>+27</u>
	FEB 2024	€ 7 7	<u>114</u>	+4 <u>+43</u>
	2021 <u>2022</u>	141 1 43	139 <u>162</u>	-2 <u>+19</u>
Total Outlets (3)	2022	139 <u>1</u> 62	151 <u>197</u>	<u>+12+35</u>
	2023	151	193	+42
	FEB-2024	193 1 97	192 <u>234</u>	<u>-1+37</u>

Notes:

- 1. Numbers are as of December 31st each year.
- All Company-Owned outlets are owned and operated by our affiliates.
- 3. Tables 1 through 5 herein focus on Sky Zone® parks in the United States.

Table No. 2 Transfer of Outlets from Franchisees to New Owners (other Other than the Franchisor or its Affiliates)

For years 2021 Years 2022 through February 29, 2024(1)

State	Year	Number of Transfers
	2021	θ
	2022	1
Alabama	2023	0
	FEB-2024	-
	2021	0
	2022	0
Arizona	2023	0
	FEB-2024	-
	2021	1
	2022	0
California	2023	1
	FEB-2024	05
	2021	0
	2022	0
Connecticut	2023	1
	FEB-2024	01
	2021	0
	2022	0
Florida	2023	0
	FEB-2024	1
	2021	0
	2022	1
Georgia	2023	0
	FEB-2024	02
	2021 202	0
Tilin ai a	2	
Illinois	<u>2</u> 023	2
	2024	3
	2022	0
<u>Indiana</u>	2023	<u>20</u>
	<u>2024</u>	2
	<u>2022</u>	0
<u>Iowa</u>	<u>2023</u>	0
	<u>2024</u>	0
	<u>2022</u>	0
<u>Kansas</u>	<u>2023</u>	0
	FEB-2024	1
	2021	θ
Louisiana	2022	1
Louisiana	2023	0
	FEB-2024	0
	2021	1
Maryland	2022	0
iviai yiailu	2023	0
	FEB-2024	<u>01</u>
	2021	θ
Michigan		0
Michigan	2021	

State	Year	Number of Transfers
	2021 <u>202</u>	<u>01</u>
Missouri	2	
Wiissouri	<u>2023</u>	0
	<u>2024</u>	<u>1</u>
	2022	<u>10</u>
<u>Nebraska</u>	2023	0
	<u>2024</u>	0
	<u>2022</u>	0
<u>Nevada</u>	<u>2023</u>	0
	FEB-2024	<u>10</u>
	2021	0
New Jersey	2022	0
Thew Jersey	2023	2
	FEB-2024	_
	2021	0
North Carolina	2022	1
North Caronna	2023	0
	FEB-2024	0
	2021	0
North Dakota	2022	1
North Dakota	2023	0
	FEB-2024	0
	2021 <u>202</u>	2 0
Ohio	2	_
Omo	2023	0
	<u>2024</u>	0
	2022	0
<u>Oklahoma</u>	2023	0
		<u>01</u>
	<u>2022</u>	0
<u>Pennsylvania</u>	<u>2023</u>	0
	<u>2024</u>	2
	2021	0
Tennessee		0
Temiessee	2023	0
	FEB-2024	1
	2021	0
Wisconsin	2022	0
,,1500115111	2023	1
	FEB-2024	<u>23</u>
	2021	4
TOTAL	2022	6
	2023	7
	FEB-2024	<u>1030</u>

Notes:

1. Numbers are as of December 31st each year.

Table No. 3

Status of Franchised Outlets For years 2021 through February 29, Years 2022 to 2024(1)

State	Year	Outlets at Start of Year	Outlets Opened(2)	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
AL	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>AL</u>	2023	1	1	0	0	0	0	2
	FEB 2024	2	0	0	0	0	0	2
AZ	2021	3	0	0	0	0	0	3
	2022	3	1	0	0	0	0	4
AZ	2023	4	1	0	0	0	0	5
	FEB 2024	5	0	0	0	9 <u>3</u>	0	<u>52</u>
CA	2021	11	0	0	0	0	0	11
	2022	11	0	0	0	0	0	11
<u>CA</u>	2023	11	2	0	0	0	0	13
	FEB 2024	13	0 4	0	0	<u>01</u>	0	13 <u>16</u>
co	2021	3	0	0	0	θ	0	3
	2022	3	0	0	0	0	0	3
CO	2023	3	0	2	0	0	0	1
<u>==</u>	FEB 2024	1	0	0	0	0	0	1
CT	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>CT</u>	2023	1	1	0	0	0	0	2
<u>==</u>	FEB 2024	2	0	0	0	0	0	2
DE	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>DE</u>	2023	1	0	0	0	0	0	1
	FEB 2024	1	0	0	0	0	0	1
FL	2021	10	0	θ	0	0	0	10
	2022	10	0	0	0	2	0	8
FL	2023	8	0	0	0	0	0	8
	FEB 2024	8	0	0	0	0	0	8
GA	2021	4	0	0	0	0	0	4
<u>GA</u>	2022	4	0	0	0	0	0	4
<u> </u>	2023	4	1	0	0	0	0	5

State	Year	Outlets at Start of Year	Outlets Opened	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
	FEB 2024	5	0 <u>1</u>	0	<u>01</u>	1	0	4
IA	2021	2	0	0	θ	θ	0	2
_	2022	2	0	0	0	0	0	2
<u>IA</u>	2023	2	0	0	0	0	0	2
_	FEB 2024	2	0	0	0	0	0	2
IL	2021	5	θ	θ	θ	θ	θ	5
	2022	5	0	0	0	0	0	5
<u>IL</u>	2023	5	0	0	0	0	0	5
=	FEB 2024	5	€2	0	0	0	0	<u>57</u>
IN	2021	6	0	0	0	2	0	4
	2022	4	0	0	0	0	0	4
<u>IN</u>	2023	4	0	0	0	0	0	4
<u>=</u>	FEB 2024	4	0	0	0	0	0	4
KS	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
<u>KS</u>	2023	2	0	0	0	0	0	2
<u>KD</u>	FEB 2024	2	0	0	0	0	0	2
KY	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>KY</u>	2023	1	1	0	0	0	0	2
KI	FEB 2024	2	0	0	0	0	0	2
LA	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>LA</u>	2023	1	0	0	0	0	0	1
	FEB 2024	1	0	0	0	0	0	1
MA	2021	3	0	θ	θ	θ	0	3
	2022	3	0	0	0	0	0	3
MA	2023	3	0	0	0	0	0	3
	FEB 2024	3	0	0	0	3	0	0
MD	2021	5	0	θ	θ	θ	θ	5
	2022	5	0	0	0	0	0	5
<u>MD</u>	2023	5	0	0	0	0	0	5
	FEB	5	0	0	0	0	0	5

State	Year	Outlets at Start of Year	Outlets Opened ⁽²⁾	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
State	2024							
MI	2021	3	0	θ	0	0	0	3
	2022	3	0	0	0	0	0	3
MI	2023	3	0	0	0	0	0	3
<u>W11</u>	FEB 2024	3	0	0	0	0	0	3
MO	2021	3	0	0	0	θ	0	3
	2022	3	0	0	0	0	0	3
MO	2023	3	0	0	0	0	0	3
<u> </u>	FEB 2024	3	92	0	0	0	0	<u>35</u>
MS	2021	1	0	0	0	θ	θ	1
	2022	1	0	0	0	0	0	1
<u>MS</u>	2023	1	1	0	0	0	0	2
<u> </u>	FEB 2024	2	0	0	0	0	0	2
NE	2021	1	0	θ	θ	θ	θ	1
	2022	1	0	0	0	0	0	1
<u>NE</u>	2023	1	0	0	0	0	0	1
	FEB 2024	1	0	0	0	0	0	1
NH	2021	0	1	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>NH</u>	2023	1	0	0	0	0	0	1
1111	FEB 2024	1	0	0	0	0	0	1
NJ	2021	9	0	0	0	θ	0	9
	2022	9	0	1	0	0	0	8
<u>NJ</u>	2023	8	0	0	0	0	0	8
	FEB 2024	8	0	0	0	9 <u>5</u>	0	8 <u>3</u>
NM	2021	θ	0	θ	0	θ	θ	0
	2022	0	1	0	0	0	0	1
<u>NM</u>	2023	1	0	0	0	0	0	1
1111	FEB 2024	1	0	0	0	0	0	1
NY	2021	5	0	θ	0	θ	0	5
	2022	5	0	0	0	2	0	3
NY	2023	3	0	0	0	0	0	3
	FEB	3	0	0	0	0	0	3

State	Year	Outlets at Start of Year	Outlets Opened(2)	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
N. 17. 7	2024	1	0	0	0	0		1
NV	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>NV</u>	2023	1	0	0	0	0	0	1
	FEB 2024	1	0	0	0	0	0	1
NC	2021	5	0	0	0	θ	0	5
	2022	5	0	0	0	0	0	5
<u>NC</u>	2023	5	0	0	0	0	0	5
<u>= </u>	FEB 2024	5	0	0	0	0	0	5
ND	2021	2	0	0	0	θ	0	2
	2022	2	0	0	0	0	0	2
ND	2023	2	0	0	0	0	0	2
<u>ND</u>	FEB 2024	2	0	0	0	0	0	2
OH	2021	7	0	2	0	0	0	5
	2022	5	1	0	0	0	0	6
<u>OH</u>	2023	6	1	0	0	0	0	7
<u>on</u>	FEB 2024	7	0	0	0	<u>01</u>	0	76
OK	2021	2	0	θ	0	θ	θ	2
	2022	2	0	0	0	0	0	2
OV	2023	2	0	0	0	0	0	2
<u>OK</u>	FEB 2024	2	0	0	0	0	0	2
OR	2021	θ	0	0	0	0	0	0
	2022	0	0	0	0	0	0	0
OP	2023	0	1	0	0	0	0	1
<u>OR</u>	FEB 2024	1	0	0	0	0	0	1
PA	2021	9	0	0	0	0	0	9
	2022	9	0	0	0	0	0	9
DA	2023	9	0	0	0	0	0	9
<u>PA</u>	FEB 2024	9	<u> </u>	0	0	0 <u>2</u>	0	<u>98</u>
RI	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
DI	2023	1	0	0	0	0	0	1
<u>RI</u> -	FEB 2024	1	0	0	0	1	0	0
SC	2024	2	0	0	0	0	0	2

State	Year	Outlets at Start of Year	Outlets Opened(2)	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
	2022	2	1	0	0	0	0	3
<u>SC</u>	2023	3	0	0	0	0	0	3
<u>sc</u>	FEB 2024	3	0	0	0	0	0	3
SD	2021	1	0	0	0	0	0	1
	2022	1	0	0	0	0	0	1
<u>SD</u>	2023	1	0	0	0	0	0	1
<u> </u>	FEB 2024	1	0	0	0	0	0	1
TN	2021	2	0	0	0	0	0	2
	2022	2	0	0	0	0	0	2
<u>TN</u>	2023	2	0	0	0	0	0	2
<u> </u>	FEB 2024	2	0	0	0	0	0	2
TX.	2021	1	0	0	0	0	0	1
	2022	1	2	0	0	0	0	3
TX	2023	3	0	0	0	0	0	3
<u>111</u>	FEB 2024	3	0	0	0	0	0	3
IT	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>Q</u>	<u>0</u>
<u>UT</u>	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
VA	$\frac{2021}{24}$	<u>20</u>	1	0	0	0	<u> 40</u>	<u>21</u>
	2022	2	0	0	0	0	0	2
37.4	2023	2	0	0	0	0	0	2
<u>VA</u>	FEB 2024	2	0	0	0	0	0	2
₩I	2021	3	0	0	0	0	θ	3
	2022	3	0	0	0	0	0	3
<u>WI</u>	2023	3	0	0	0	0	0	3
<u>vv 1</u>	FEB 2024	3	<u> </u>	0	0	0	0	3 4
₩V	2021	1	0	0	0	0	θ	1
	2022	1	0	0	0	0	0	1
<u>wv</u>	2023	1	0	0	0	0	0	1
	FEB 2024	1	0	0	0	0	0	1
	2021 <u>2</u> 022	120 <u>117</u>	2 <u>6</u>	<u>21</u>	0	2 4	<u>10</u>	117 <u>118</u>
Totals	2022	117	6	1	0	4	0	118
	2023	118	10	2	0	0	0	126
	FEB	126	<u>012</u>	0	<u>01</u>	<u>517</u>	0	121 120

State	Year	Outlets at Start of Year	Outlets	Terminations	Non- Renewals	Reacquired by Franchisor	Ceased Operations - Other Reasons	Outlets at End of the Year
	2024							

Notes:

- 1. Numbers are as of December 31st each year.
- 2. "Outlets Opened" totals may include Rockin' Jump® or DEFY® parks which have been rebranded as Sky Zone® parks during a specified year.

Table No. 4
Status of Company-Owned Outlets
For years 2021 Years 2022 to February 29, 2024(1)

State	Year	Outlets at Start of the Year	Outlets Opened(2)	Outlets Reacquired From Franchisee Franchise es	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of the Year
AI	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>AL</u>	<u>2023</u>	0	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	0
CA	2 0 2 1 2 0 2 4	<u>©</u>	1	0 0	0	0	1
	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>AZ</u>	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2024</u>	<u>0</u>	<u>0</u>	<u>3</u>	<u>0</u>	<u>0</u>	<u>3</u>
	<u>2022</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
<u>AK</u>	<u>2023</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>0</u>
	<u>2024</u>	<u>0</u>	<u>2</u>	<u>0</u>	<u>0</u>	<u>0</u>	<u>2</u>
	2022	1	4 <u>7</u>	0	0	0	<u>58</u>
	2023	<u>58</u>	<u>78</u>	1	0	0	13 <u>17</u>
<u>CA</u>	F E B 2 0 2 4			91 0	0	0	13 <u>2</u> 0
CO	2021	0	0	0	0	0	0
<u>CO</u>	2 0 2 2	0	<u>0</u>	0 0	0	0	0

Part Part	4 4 2 5
FL 2021 2 0 0 0 0 0 0 0 0 0	2 5
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	5
EL 2023 5 2 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
$\begin{array}{ c c c c c c c c c }\hline E & F & 7 & 3 & 0 & 0 & 0 & 0 \\\hline E & B & 2 & 0 & 1 & 0 & 0 & 0 \\\hline 2 & 4 & & & & & & & & & & & & & & & & &$	
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	7
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	7 <u>10</u>
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	1
$ \begin{array}{ c c c c c c c c c } \hline & 2024 & 1 & 2 & 1 & 0 & 0 \\ \hline & 2022 & 0 & 0 & 0 & 0 & 0 & 0 \\ \hline & 2023 & 0 & 0 & 0 & 0 & 0 & 0 \\ \hline & 2024 & 0 & 1 & 0 & 0 & 0 & 0 \\ \hline & IL & 2021 & 2 & 0 & 0 & 0 & 0 \\ \hline & 2 & 2 & 0 & 0 & 0 & 0 & 0 \\ \hline & 2 & 2 & 0 & 0 & 0 & 0 & 0 \\ \hline & 2 & 2 & 0 & 0 & 0 & 0 & 0 \\ \hline \end{array} $	<u>1</u>
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	<u>4</u>
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	<u>0</u>
$\begin{array}{ c c c c c c c c c c c c c c c c c c c$	<u>0</u>
$\begin{bmatrix} 2 & 2 & \underline{0} \\ 0 & 2 \\ 2 & 2 \end{bmatrix} \qquad \begin{bmatrix} \underline{0} & 0 & \underline{0} \\ 0 & 0 & \underline{0} \end{bmatrix} \qquad 0$	<u>1</u>
	2
2023 2 1 0 0 0	2
	3
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$	3
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	θ <u>3</u>
<u>2023</u> <u>3</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>	1
<u>2024</u> <u>3</u> <u>0</u> <u>0</u> <u>0</u> <u>0</u>	<u>3</u>

State	Year	Outlets Start of Year	the		lets ned <u>(2)</u>	Reaco Fr Fran <u>Fran</u>	tlets quired om chisee echise		utlets losed		tlets Sold to anchisee	Outlets at End of the Year
	2 0 2 2	0	:	<u>0</u>		0	0		0		0	0
<u>KY</u>	2 0 2 3	0	:	<u>0</u>	-1	<u>+0</u>	θ		0		0	<u>10</u>
***	F E B 2 0 2 4	<u>0</u>	-1	12		0	θ		0		0	<u>12</u>
LA	2021	1			0		0		0		0	1
	2 0 2 2	1	:	<u>0</u>		0	θ		0		0	1
Į [2023	1		2			0		0		0	3
<u>LA</u>	F E B 2 0 2 4	3	:	<u>0</u>		0	θ		0		0	3
MA	2021	0	,		0		0		0		0	0
	2 0 2 2	0	:	<u>0</u>		0	θ		0		0	0
<u>MA</u>	2 0 2 3	0		<u>0</u>	3	<u>30</u>	θ		0		0	<u>30</u>
		9	<u>0</u>		3	θ		3		0	0	<u>63</u>
MD	2021	2			0		0		0		0	2
<u>MD</u>	2 0 2 2	2		<u>0</u>		0	θ		0		0	2

State	Year	Outlets a Start of t Year	he Out	tlets ned <u>(2)</u>	Reace Fr Fran <u>Fran</u>	tlets quired om chisee achise	utlets losed	Outlets Sold to Franchisee	Outlets at End of the Year
	2 0 2	2	<u>0</u>		0	0	0	0	2
	3 F E B 2 0 2 4	2	<u>0</u>	(0	0	0	0	2
MI	2021	0		0		θ	0	0	θ
	2 0 2 2	0	<u>0</u>		0	0	0	0	0
	2023	0		1		0	0	0	1
<u>MI</u>	E E B 2 0 2 4	1	<u>3</u>		0	θ	0	0	1 <u>4</u>
MN	2021	4		0		0	0	0	4
	2 0 2 2	4	<u>0</u>		0	0	0	0	4
	2023	4	1			0	0	0	<u>56</u>
<u>MN</u>	# E B 2 0 2 4	<u>56</u>	<u>0</u>		0	0	0	0	<u>56</u>
МО	2 0 2 1 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	<u>0</u>	1		0	0	0	0	1
1	<u>2023</u>	<u>1</u>)		0	0	<u>0</u>	<u>1</u>
	<u>2024</u>	<u>1</u>	<u>(</u>			<u>0</u>	<u>0</u>	<u>0</u>	<u>1</u>
<u>MS</u>	<u>2022</u>	<u>0</u>	<u>(</u>	<u>)</u>		0	<u>0</u>	<u>0</u>	<u>0</u>

State	Year	Outlet Start of Yea	f the	Outlets Opened(2)	Outlet Reacqui Fron Franchi <u>Franch</u> <u>es</u>	ired n isee	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of the Year
	<u>2023</u>	<u>0</u>		<u>0</u>	<u>0</u>		<u>0</u>	<u>0</u>	<u>0</u>
	<u>2024</u>	0		<u>0</u>	<u>0</u>		<u>0</u>	<u>0</u>	<u>0</u>
	2022	0		1	0		0	0	1
ļ.	2023	1		<u> 10</u>	0		0	0	<u>21</u>
<u>NJ</u>	F E B 2 0 2 4	<u>21</u>	<u>0</u>	•	95	θ	<u>θ1</u>	0	<u>25</u>
NV	2 0 2 1 2 0 2 2 2 2 2 2 2 2 2 2 2 2 2 2	0	1		0	θ	0	0	<u>θ1</u>
	2023	1		0	0		0	0	1
	2024	1		<u>0</u>	<u>0</u>		<u>0</u>	<u>0</u>	1
	2 0 2 2	θ <u>1</u>	1		0	θ	0	0	θ <u>2</u>
<u>NC</u>	2 0 2 3	9 <u>2</u>	<u>0</u>		0	0	0	0	€2
	F E B 2 0 2 4	θ <u>2</u>	2		0	0	0	0	0 4
NY	2021	1		0	θ		0	0	1
	2022	1		0	2		0	0	3
<u>NY</u>	2023 F E B 2 0 2 4	4	1	1	0	0	0 0	0	4 45
ОН	<u>:</u>	1	θ	2	<u>0</u>		0	0 0	32

State	Year	Outle Start o Yea	f the	Outlets Opened ⁽²⁾	Reace Fr Fran <u>Frar</u>	tlets quired com chisee achise	Out Clo	lets	Outlets Sold to Franchisee	Outlets at End of the Year
	<u>2023</u>	2		0		0	(<u>1</u>	<u>1</u>
	<u>2024</u>	1		0		<u>1</u>	(<u>0</u>	<u>2</u>
<u>OR</u>	2 0 2 2	<u>30</u>	<u>0</u>		0	θ		0	0	<u>30</u>
	2 0 2 3	<u>30</u>	<u>0</u>		0	0		0	0	<u>30</u>
<u>==</u>	F E B 2 0 2	<u>0</u>	3 <u>1</u>		0	0		0	0	31
PA	2021		1	0		0		9	0	1
171	2 0 2 2	1	<u>0</u>		0	θ		0	0	1
<u>PA</u>	2 0 2 3	1	<u>0</u>		0	0		0	0	1
	F E B 2 0 2 4	1	<u>0</u>		9 <u>2</u>	0		0	0	<u>43</u>
RI	2021		9	0		0	(9	0	0
<u>RI</u>	2 0 2 2 2 0 2 3	0	<u>0</u>		0	0		0	0	0
	0 2 3				v			J	Ŭ	

$ \frac{\mathbb{W}A}{\mathbb{E}} = \begin{bmatrix} 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0 & 0$	State	Year	Outlets at Start of the Year	Outlets Opened(2)	Outlets Reacquired From Franchisee Franchise es	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of the Year
SC 2022 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <td></td> <td></td> <td>0</td> <td>0</td> <td></td> <td>0</td> <td>0</td> <td>1</td>			0	0		0	0	1
SC 2023 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 4 0 0 0 0 0 0 0 4 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 <td></td> <td></td> <td>_</td> <td></td> <td></td> <td>_</td> <td>_</td> <td>_</td>			_			_	_	_
Totals 2024								
TN 2021	<u>SC</u>							
Totals	TN							
Totals C 2 2 2 2 2 2 2 2 2	114							<u> </u>
TN		0 2 2						
No. No.	_							· -
VA 2021 4 0 0 0 0 4 2022 4 1 0 0 0 0 5 2023 5 32 0 0 0 0 87 VA E 7 82 0 0 0 0 8 VA 2021 1 0 0 0 0 0 8 WA 2021 1 0 0 0 0 0 12 2 12 0 0 0 0 0 0 12 WA 3 2 0 10 0 0 0 32 WA 3 3 0 0 0 0 32 WA 3 3 0 0 0 32 WA 3 3 0 0 0 33 4 0 0	<u>TN</u>	E B 2 0 2	4	<u>0</u>	0 0	0	0	4
VA 2022 4 1 0 0 0 5 F 7 82 0 0 0 0 87 WA 2024 1 0 0 0 0 0 8 WA 2024 1 0 0 0 0 0 0 12 0 12 0 0 0 0 0 0 0 12 2 12 0 10 0 0 0 0 32 2 12 0 10 0 0 0 32 WA 32 4 0 0 0 0 32 E 32 4 0 0 0 0 32 B 2 0 0 0 0 33 2 0 0 0 0 0 33 2 0 <t< td=""><td>VA</td><td></td><td>4</td><td>Ι θ</td><td>Δ</td><td>Δ</td><td>Δ</td><td>4</td></t<>	VA		4	Ι θ	Δ	Δ	Δ	4
VA F 7 82 0 0 0 87 E B 20 82 0 0 0 0 8 WA 2021 1 0 0 0 0 0 1 20 12 0 0 0 0 0 0 12 20 12 0 10 0 0 0 0 32 2 12 0 10 0 0 0 32 E B 2 0 0 0 0 32 E B 2 0 0 0 0 33 E B 2 0 0 0 0 33 Totals 2022 22 7 4 0 0 0 33 44 2023 3344 3025 43 0 0 0 0 677	111							
VA EBB 200 24 B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B B <th< td=""><td> </td><td></td><td>5</td><td><u>32</u></td><td>0</td><td>0</td><td>0</td><td>87</td></th<>			5	<u>32</u>	0	0	0	8 7
WA 2021 1 0 0 0 0 1 2 42 0 0 0 0 0 12 2 12 0 10 1 0 0 32 2 3 0 0 0 0 32 B 3 0 0 0 33 E 0 0 0 0 36 B 2 0 0 0 33 2 0 0 0 33 2 4 0 0 0 33 2 14 0 0 0 33 26 14 0 0 0 33 44 0 0 0 0 677	<u>VA</u>	Е В 20	7	82	0 01	0	0	8
WA 2 0 12 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	₩A		1	0	0	0	0	4
WA 0 2 3		2 0 2 2	12			0	0	<u> 42</u>
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	<u>WA</u>	0 2 3			_			3 <u>2</u>
Totals $ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		E B 2 0 2	32	4	0 0	0	0	3 <u>6</u>
2025 $33\underline{44}$ $30\underline{25}$ $4\underline{5}$ 0 $0\underline{1}$ 0	Totals	2022	22 26	7 <u>14</u>	4	0	0	33 44
$oxed{\mathbb{F}}$ 71 6728 917 4 92 0 74114	1 otals				· · · · · · · · · · · · · · · · · · ·			67 71
		F	<u>71</u>	<u>6728</u>	<u>17</u> 4	92	0	71 <u>114</u>

State	Year	Outlets at Start of the Year	Outlets Opened(2)	Outlets Reacquired From Franchisee Franchise	Outlets Closed	Outlets Sold to Franchisee	Outlets at End of the Year
	B 2 0 2 4						

Notes:

- 1. Numbers are as of December 31st each year.
- 2. "Outlets Opened" totals may include Rockin' Jump® or DEFY® parks which have been rebranded as Sky Zone® parks during a specified year.

Table No. 5
Projected Openings as of February 29December 31, 2024

State	Not Opened		Projected New Company- Owned Outlet Outlets In the Next Fiscal Year		
Alabama	<u>0</u>	<u>0</u>	0		
<u>Arizona</u>	<u>0</u>	<u>0</u>	<u>0</u>		
California	7 9	4	0 5		
Colorado	1	0	0		
Connecticut	<u>10</u>	0	0		
Florida	<u> 10</u>	1	<u> 10</u>		
Georgia	1	0	0		
Illinois	2 4	2 3	<u> 43</u>		
Indiana	$\underline{40}$	0	0		
Kentucky	$\underline{0}$	$\underline{0}$	$\underline{0}$		
Maryland	2	1	0		
Michigan	<u>0</u>	<u>0</u>	<u>0</u>		
Minnesota	$\overline{0}$	$\overline{\underline{0}}$	0		
Mississippi	$\underline{0}$	$\underline{0}$	$\underline{0}$		
<u>Missouri</u>	<u>0</u>	<u>0</u>	<u>0</u>		
Nevada	1	<u>0</u>	<u>0</u>		
New York	<u>2</u>	<u>1</u>	<u>2</u>		
New Jersey	<u>20</u>	0	0		
New Hampshire	<u>0</u>	<u>0</u>	<u>0</u>		
North Carolina	<u>10</u>	0	0		
<u>Oklahoma</u>	<u>0</u>	<u>0</u>	<u>0</u>		
Oregon	0	0	1		
Pennsylvania Pennsylvania	<u>0</u>	<u>0</u>	<u>0</u>		
<u>Tennessee</u>	<u>0</u>	<u>0</u>	<u>0</u>		
Texas	<u>1</u> <u>0</u>	00	<u> 43</u>		
Utah	<u>10</u>	<u> 10</u>	0		
Virginia	<u>32</u>	1	1		

State	Franchise Agreements Signed But Outlet Not Opened	Projected New Franchised Outlet Outlets In The Next Fiscal Year	Projected New Company- Owned Outlet <u>Outlets</u> In the Next Fiscal Year		
Washington	<u>0</u>	$\underline{0}$	<u>1</u>		
Washington D.C.	<u>0</u>	0	0		
Total	23	10 11	5 <u>16</u>		

<u>Exhibit E</u> lists the names, addresses and telephone numbers of all current franchisees. In addition, as of December 31, 2024.

Exhibit E additionally lists the name, city and state, and the current business telephone number (or, if unknown, the last known home number) every franchisee telephone o f who had transferred, terminated, canceled, not renewed o r otherwise voluntarily or involuntarily ceased to do business under the franchise agreement during our most recently completed fiscal year or who has not communicated with us within 10 weeks of the date of this franchise disclosure documentFranchise Disclosure Document. If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

Lastly, Exhibit E lists the name, city and state, and the telephone numbers of all franchisees who have signed franchise agreements, but who have not yet opened a Park as of December 31, 2024 and the name, state, and the telephone numbers of all franchisees who have signed Multi-Unit Development Agreements, but who have not yet signed franchise agreements or opened Parks as of December 31, 2024.

During the last 3three fiscal years, we<u>current and former franchisees</u> have signed confidentiality clauses—with. In some instances, current or former franchisees which would restrict them from speakinghave signed provisions restricting their ability to speak openly with you about their experiences with the franchise system. You may wish to speak with current and former franchisees, but be aware that not all or certain aspects of their experience of them will be able to communicate with usyou.

We do not have any trademark specific franchisee organization.

Item 21

ITEM 21. FINANCIAL STATEMENTS

The Attached as Exhibit F to this Disclosure Document are our consolidated balance sheets as of for the fiscal years ended December 31, 2024, December 31, 2023, 2022, and 2021 December 31, 2022, and the related consolidated statements of income, of member's equity and of cash flows for the years then ended, including the related notes, are attached as Exhibit F for those fiscal years. Our fiscal year end isends on December 31.

Item 22

ITEM 22. CONTRACTS

The

<u>Attached to this Disclosure Document are the</u> following agreements are Exhibits to this Disclosure Document:

Exhibit A Franchise Agreement and Appendices

Appendix A: Franchisee Specific Terms

Appendix B: Owners' Personal Guaranty of Franchisee's Obligations Appendix C: Owner Personal Covenants Regarding Confidentiality and Non-Competition

Appendix D: Addendum To Lease

Appendix E: Protected Territory and Premises

Appendix F: Acknowledgement Addendum to the Sky Zone Franchise

Agreement

Exhibit B Multi-Unit Development Agreement and Appendices

Exhibit C State Law Addenda

Exhibit H Form of Purchase Order

Item 23

RECEIPTS

Exhibit I contains detachable documents acknowledging your receipt

Exhibit I Agreement and Conditional Consent to Transfer (including General Release)

Exhibit J Form of Renewal Addendum (including General Release)

Exhibit K Representations and Acknowledgment Statement

ITEM 23. RECEIPTS

<u>Attached as the last two pages</u> of this Disclosure Document. Two copies of the <u>are duplicate</u> receipt are attached. You must sign both copies, keeppages to be signed by you. Keep one copy for your records, and return the other one copy to us at: Mike Revak, President, 86 N. University Avenue, Suite 350, Provo, Utah 84601, with a copy to mike.revak@skyzone.com.

[REST OF PAGE LEFT BLANK INTENTIONALLY]

EXHIBIT A TO THE DISCLOSURE DOCUMENT

FRANCHISE AGREEMENT

$\begin{array}{c} \textbf{SKY ZONE}^{\circledast}_{=} \\ \textbf{FRANCHISE AGREEMENT} \end{array}$

[NAME OF FRANCHISEE]

[NAME OF AREA]

TABLE OF CONTENTS

			Page			
1.	PREA	AMBLES AND DEFINITIONS.	1			
	1.1	Preambles	1			
	1.2	Certain Definitions	1			
2.	YOU	R ORGANIZATION AND MANAGEMENT	4			
	2.1	Organizational Documents	4			
	2.2	Ownership Interests	4			
	2.3	Responsible Person/Management of Business	5			
	2.4	Control Group	5			
3.	GRA	NT OF RIGHTS.	<u>-5</u> <u>6</u>			
	3.1	Grant Of Franchise	<u>5</u> 6			
(a)	Gran	ıt	6			
(b)	Prote	ected Territory	6			
	3.2	Our Reservation of Rights.	6			
4.	LOCATION SELECTION, LEASE OR PURCHASE OF LOCATION AND LOCATION DEVELOPMENT.					
	4.1	Location Selection	7			
	4.2	Purchase Or Lease Of The Location	7			
	4.3	Location Development	<u>78</u>			
(a)	Deve	lopment	8			
(b)	Plans	5	8			
(c)	Cons	truction-8; Site Development Fee	9			
(d) —	— Forc	e Majeure				
(d)	Natu	ral Disasters	9			
	4.4	Your Obligations				
	4.5	Operating Assets	9			
	4.6	Start-Up Inventory	<u>910</u>			
	4.7	Business Commencement				
	4.8	Commencement Deadline	10			
	4.9	Grand Opening Marketing	10			
	4.10	Opening Assistance	10			

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

i

	4.11 Relocation	<u> 10<u>11</u></u>
5.	FEES.	11
	5.1 Initial Franchise Fee	11
	5.2 Royalty Fee	11
	5.3 Ad Fee	
	5.4 Other Fees Technology Fee	11
	5.5 <u>Call Center Program Fee.</u>	12
	5.6 Designated Account	
	5.65.7 Interest On Late Payments	<u> 11<u>12</u></u>
	5.75.8 Application Of Payments	<u> 41<u>12</u></u>
	5.8 Authorized 5.9 Overdue Supplier Payment	12
	5.95.10 Non-Compliance Fee	<u>12<u>13</u></u>
	5.105.11 Inspection And Compliance Reimbursement	<u> 12<u>13</u></u>
6.	TRAINING, ASSISTANCE, AND METHODS OF OPERATION	V. <u>12</u> <u>13</u>
	6.1 Training	<u> 12</u> <u>13</u>
	6.2 Refresher Training	<u> 13<u>14</u></u>
	6.3 General Guidance	<u>13<u>14</u></u>
	6.4 On-Site Consultation and Additional Guidance	<u>13<u>14</u></u>
	6.5 Operations Manual	<u> 14<u>15</u></u>
	6.6 Annual Convention	<u> 14<u>15</u></u>
7.	MARKS.	<u>14<u>16</u></u>
	7.1 Ownership and Goodwill of Marks	<u>14<u>16</u></u>
	7.2 Limitations On Your Use Of Marks	<u>15</u> <u>16</u>
	7.3 Notification Of Infringements And Claims	<u>15</u> <u>16</u>
	7.4 Discontinuance Of Use Of Marks	<u> 15<u>17</u></u>
	7.5 Indemnification Of Franchisee	<u> 15<u>17</u></u>
8.	CONFIDENTIAL INFORMATION.	<u>16</u> <u>17</u>
	8.1 Confidential Information	<u> 16</u> <u>17</u>
	8.2 Nondisclosure of Confidential Information	
	8.3 CustomerPersonal Information	<u>17</u> 19
9.	SKY ZONE METHODS OF OPERATION.	<u> 18<u>19</u></u>
	9.1 Compliance With Methods of Operation	18 19

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement

	9.2 Provisions Of This Agreement	<u>18</u> 20
	9.3 Modification Of Methods Of Operation	<u> 1820</u>
	9.4 Condition Of Your Business	<u> 19<u>20</u></u>
	9.5 Products and Services You May Offer	<u>1921</u>
	9.6 Products, Supplies, Operating Assets, and Services	20 <u>22</u>
	9.7 Compliance With Laws	22 <u>24</u>
	9.8 Personnel	22 <u>24</u>
	9.9 Insurance	23 <u>24</u>
	9.10 Quality Control	26 <u>28</u>
	9.11 Pricing Policies	26 <u>28</u>
	9.12 Technology System	<u>2628</u>
	9.13 Reciprocal Membership/Coupons	27 <u>29</u>
	9.14 Non-Compliance	27 <u>30</u>
	9.15 Innovations	<u>27<u>30</u></u>
	9.16 General Conduct	<u>2730</u>
	9.17 Safety and Security Procedures	<u>28<u>30</u></u>
0.	MARKETING.	<u>28<u>31</u></u>
	10.1 National Advertising	<u>28<u>31</u></u>
	10.2	
a)	National Advertising Fund	31
<u>b)</u>	Use of Funds	<u>31</u>
		21
<u>c)</u>	Materials	<u>31</u>
	Materials Accounting	2021
(c) (d) 10.3(Accounting	29 <u>31</u>
(<u>d)</u> 10.3 <u>(</u>	Accounting Proportionality	29 <u>31</u> 29 <u>32</u>
<u>(d)</u> 10.3 <u>(</u>	Accounting e) Proportionality f) Deferrals Or Reductions	29 <u>31</u> 29 <u>32</u> 29 <u>32</u>
<u>d)</u> 10.3 <u>(</u>	Accounting e) Proportionality f) Deferrals Or Reductions 10.5 10.2 Local Advertising	29 <u>31</u> 29 <u>32</u> 29 <u>32</u> 29 <u>32</u> 29 <u>32</u>
(<u>d)</u> 10.3 <u>(</u> 10.4 <u>(</u>	Accounting e) Proportionality f) Deferrals Or Reductions 10.5 10.2 Local Advertising Obligation to Advertise	2931 2932 2932 2932 2932 2932
d) 10.3(10.4(a) b)	Accounting Proportionality Deferrals Or Reductions 10.5 10.2 Local Advertising Obligation to Advertise Approvals	2931 2932 2932 2932 2932 2932 3032
(d) 10.3(10.4(10.4(a) (a) (b)	Accounting e) Proportionality f) Deferrals Or Reductions 10.510.2 Local Advertising Obligation to Advertise Approvals Local Advertising Funding Requirement	2931 2932 2932 2932 2932 3032 3033
(d) 10.3(10.4(10.4(10.4(10.10))) (a) (b) (c) (d)	Accounting e) Proportionality f) Deferrals Or Reductions 10.510.2 Local Advertising Obligation to Advertise Approvals Local Advertising Funding Requirement Valid Expenditures	2931 2932 2932 2932 2932 3032 3033 3033
(<u>d)</u> 10.3 <u>(</u>	Accounting e) Proportionality f) Deferrals Or Reductions 10.510.2 Local Advertising Obligation to Advertise Approvals Local Advertising Funding Requirement	2931 2932 2932 2932 2932 3033 3033 3033 3033

[City, State]

	10.8 10	<u>0.5</u> Digital Marketing	32 <u>35</u>
(a)	Restrictions on Use		
(b)	Our Website		
11.	RECORDS, REPORTS AND FINANCIAL STATEMENTS.		
	11.1	Records	32 <u>35</u>
	11.2	Periodic Reports	
	11.3	Verification	<u>3336</u>
12.	INSPI	<u>33<u>36</u></u>	
	12.1	Our Right To Inspect The Business	<u>3336</u>
	12.2	Cooperation	<u>34<u>37</u></u>
	12.3	Our Right to Audit	<u>34<u>37</u></u>
13.	TRAN	NSFER.	<u>34<u>37</u></u>
	13.1	By Us	<u>34<u>37</u></u>
	13.2	By You	<u>34<u>37</u></u>
	13.3	Conditions For Approval Of Transfer	<u>3538</u>
(a)	Contr	rol Transfers	<u>3538</u>
(b)	Other	· Transfers	<u>3640</u>
	13.4	Transfer To A Wholly Owned Entity	<u>3640</u>
	13.5	Transfer Upon Your Death Or Disability	<u>3740</u>
	13.6	Operation Upon Your Death Or Disability	<u>3740</u>
	13.7	Bona Fide Offers	<u>3741</u>
	13.8	Our Right Of First Refusal	<u>3841</u>
	13.9	Non-Exercise	<u>3842</u>
	13.10	Securities Offerings	<u>3842</u>
14.	SUCC	CESSOR FRANCHISES FRANCHISE AND EXPIRATION	
	14.1	Acquisition Of A Successor Franchise	<u>3942</u>
	14.2	Grant Of A Successor Franchise	
	14.3	Agreements/Releases	
15.	TERM	MINATION OF AGREEMENT.	
	15.1	Termination By You	
	15.2	Termination By Us	
	15.3	Our Right To Operate The Business And Management Fee	
	15.4	Alternatives To Termination	42 <u>46</u>

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement

[City, State]

	15.5	Exercise of Other Remedies	<u>4347</u>
16.	RIGHTS AND OBLIGATIONS UPON TERMINATION OR EXPIRATION OF THIS AGREEMENT.		
	16.1	Payment Of Amounts Owed To Us-: Liquidated Damages	<u>43</u> 47
	16.2	Discontinue Use of the System and the Marks	44 <u>48</u>
	16.3	Return of Confidential Information	4 <u>549</u>
	16.4	Our Right To Purchase Business	<u>4549</u>
(a)	Exerc	cise of Option	<u>4549</u>
(b)	Lease	Leasehold Rights	
(c)	Purch	nase Price	45 <u>49</u>
d)		sions	
e)		aisal	
f)		ng	
g)		uments	
h)) W	
)		ses	
j)		ation of Franchise because of your Inability-46 to Operate Business	50
	16.5	Continuing Obligations	4 6 50
7.	COVENANTS NOT TO COMPETE.		
	17.1	In-Term Covenant Not To Compete	<u>46</u> 51
	17.2	Post-Term Covenant Not To Compete	
	17.3	Reasonable Scope of Covenants	
	17.4	Reduction Of Scope of Covenants	
	17.5	Covenant Not To Compete Upon Exercise Of Right Of First Refusal	48 <u>52</u>
8.	RELA	ATIONSHIP OF THE PARTIES AND INDEMNIFICATION	
	18.1	Independent Contractors	4 8 53
	18.2	Notice of Independent Contractor	
	18.3	Taxes	40.70
	18.4	Indemnification	
	18.5	Mitigation Not Required	
19.	ENFORCEMENT AND MISCELLANEOUS MATTERS.		
	19.1 Severability And Substitution of Valid Provisions		
	19.2	Greater Notice	

	19.3 Waive ı	r Of Obligations	50 <u>55</u>
		⁷ aiver	
	19.5 Force I	Majeure	<u>5155</u>
	19.6 Out-O f	f-Stock and Discontinued	<u>51<u>56</u></u>
	 19.7 Costs And Attorneys' Fees 19.8 You May Not Withhold Payments Due To Us 19.9 Rights Of Parties Are Cumulative 		<u>51<u>56</u></u>
	19.10 Good I	Faith and Fair Dealing 52Business J	udgment 56
	19.11 Dispute Resolution		
(a)	Mediation	Resolution 56	
(b)	Arbitration		<u>5357</u>
(c)	Exceptions to	<u>54</u> 59	
(d)	Prevailing Party's Fees		
(e)	Survival		
(f)	Tolling of Statute of Limitations		<u>55</u> 60
(g)	Performance t	to Continue	<u>55</u> 60
	19.12 Govern	ning Law	<u>55</u> 60
		nt To Jurisdiction	
		r Of Punitive Damages, Jury Trial, And Class Actions	
	19.15 Limitation Of Claims		<u>55</u> <u>61</u>
	19.16 Binding Effect		<u>56</u> 61
	19.17— Withh	56	
	19.18 Construction		<u>56</u> 61
	19.19 <u>19.18</u>	Joint And Several Owners' Liability	<u>56</u> 61
	<u>19.20</u> 19.19	Right To Information	<u>56</u> 61
	<u>19.21</u> <u>19.20</u>	Multiple Copies	<u>56</u> 62
	<u>19.22</u> 19.21	Entire Agreement Between the Parties	<u>5662</u>
	<u>19.23</u> 19.22	Amendments	<u>5662</u>
	19.24 <u>19.23</u>	No Third-Party Beneficiaries	<u>5762</u>
	19.25 <u>19.24</u>	Notices	<u>57</u> <u>62</u>
20. —	—ACKNOWLEI	DGEMENTS AND REPRESENTATIONS.	57
	20.1 Acknowledgements		
	20.2 Repres	sentation_57	

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement

[City, State]

20.3 — Due	Authority 57	
20.4		
<u>19.25</u> Terr	orist Acts	<u>5863</u>
20.5 — Tim	ely Receipt and Review of Agreement and Disclosure Document	58
20.6 Fina	ncial Performance Representations	58
APPENDIX A	FRANCHISEE-SPECIFIC TERMS	
APPENDIX B	OWNERS' PERSONAL GUARANTY OF FRANCHISEE'S	
	OBLIGATIONS	
APPENDIX C	OWNER PERSONAL COVENANTS REGARDING	
	CONFIDENTIALITY AND NON-COMPETITION	
APPENDIX D	SILENT INVESTORS	
APPENDIX E	ADDENDUM TO LEASE	
APPENDIX F E	PROTECTED TERRITORY AND LOCATION	
=	ACKNOWLEDGEMENT ADDENDUM	
APPENDIX G	ACKNOWLEDGEMENT ADDENDUM	

SKY ZONE FRANCHISE AGREEMENT

THIS FRANCHISE AGREEMENT (the "Agreement") is made and entered into as of the <u>last signature</u> date <u>listed on Appendix Abelow</u> (the "Effective Date") by and between <u>Sky Zone Franchise Group SKY ZONE FRANCHISE GROUP</u>, LLC, a <u>Missouri limited liability company formed under Missouri law</u>, with its principal business address at 86 N. University Avenue, Suite 350, Provo, Utah 84601 (referred to in this Agreement as "<u>SZFG</u>," "<u>Franchisor</u>," "we," "us," or "our"), and the person or <u>entity Entity</u> identified on Appendix A as the franchisee with its principal place of business set forth on Appendix A (referred to in this Agreement as "<u>Franchisee</u>," "you," or "your").

1. PREAMBLES AND DEFINITIONS.

Preambles. This Agreement governs your ownership and operation of one (1) Sky Zone business offering a method of operating recreational centers that offer all-trampoline and other recreational Attractions (as defined below) which use our proprietary System and our Marks (a "Business"). Businesses are developed and operated in accordance with a distinctive business format and set of business methods, designs, layouts, standards, specifications, and Methods of Operation that we and our Affiliates have developed and will continue to develop (the "System"). We identify the Businesses operating under the System by means of the SKY ZONESky Zone mark and certain other trademarks, service marks, trade names, trade dress, signs, associated designs, artwork, and logos that we may specify from time to time (collectively, the "Marks"). We may change, improve, add to, and further develop the elements of the System and the Marks from time to time. We grant franchises to persons who meet our qualifications and are willing to undertake the investment and effort required to own and operate a Business offering the products and services we authorize and approve and utilizing the System and the Marks. You have indicated to us that you desire a franchise to own and operate a Business, and we are willing to grant you a license franchise to own and operate a Business on the terms and conditions of this Agreement.

1.2 Certain Definitions

. The terms listed below have the meanings which follow them and include the plural as well as the singular. Other terms are defined elsewhere in this Agreement in the context in which they arise.

"Accounting Period" - Each monthly period during the term of the Agreement.

"Affiliate" - Any person or Entity that directly or indirectly owns or controls the referenced party, that is directly or indirectly owned or controlled by the referenced party, or that is under common control with the referenced party. The term "control" means the possession, directly or indirectly, of the power to direct or cause the direction, of the management and policies of an Entity, whether through ownership of voting securities, by contract or otherwise.

"**Applicable Laws**" - All federal, state, and local laws, rules, regulations, and ordinances that are applicable in your <u>Protected</u> Territory.

1

"Attractions" means the <u>trampoline</u> playing <u>field fields</u> (including the <u>trampolinetrampolines</u>, the padding, <u>and nets</u>), and other <u>related recreational</u> equipment, including but not limited to slides, ziplines, ball courts, and foam pits.

"Competitive Business" — Any <u>business</u> (other than a Franchise operated under a separate franchise agreement entered into with us or our Affiliates) operating as, or granting franchises or license to others to operate any business operating as, any family recreational center including, but not limited to; (i) a trampoline park; (ii) centers offering inflatable indoor bouncing, rock climbing, go carts, or miniature golf; (iii) active entertainment facility, facilities; or (iv) similar family entertainment establishments. This <u>definition</u> is not intended to include any centers specializing in gymnastics, which may contain trampolines.

"Confidential Information" - Defined in Section 8.1 (Confidential Information).

"Control Group" - Defined in Section 2.4 (Control Group).

"Controlling Interest" – Thirty-three and one-third percent (33.33%) or more of your voting shares or other voting rights if you are an Entity owned by three (3) or more persons; otherwise, fifty percent (50%) or more of your voting shares or other voting rights.

"Dispute" - Any dispute, controversy or claim between <u>us</u> and <u>among the partiesyou</u> and any of our or your Affiliates, officers, directors, shareholders, members, <u>guarantors Guarantors</u>, employees or owners arising under, out of, in connection with or in relation to: (i) this Agreement, (ii) any lease or sublease for your Business, (iii) any loan or other finance arrangement between us or our Affiliates and you, (iv) the parties' relationship, (v) your Business, or (vi) any Method of Operation.

"Effective Date" - The Effective Date listed on Appendix Athe System.

- "**EFT**" The electronic transfer of funds to us from a credit card, debit card or bank account, as well as any other current or future form of pre-authorized payment.
- **"Entity"** A corporation, general or limited partnership, limited liability company, corporation, association, cooperative or other legal entity.
- "**FDD**" The Franchise Disclosure Document that was furnished to you by us, which describes this Agreement and the FranchiseSky Zone® franchise offering.

"Gross Sales" – the total amount of all sales of products, services, programs and merchandise, food, and beverages, sold from, through, or in connection with the Business whether for cash, on credit, barter or otherwise, exclusive of applicable sales, use or service taxes. Gross Sales additionally exclude refunds that are provided to customers (not including chargebacks).

"Guarantor" – Each Owner who has a greater than ten percent (10%) or greater interest in your Entity, or such other persons as we may designate (including Owners' spouses), who

must sign an agreement in the form we prescribe, undertaking to be bound jointly and severally by the terms of this Agreement, the current form of which is attached hereto as **Appendix B** (Owners' Personal Guaranty of Franchisee's Obligations). Your Responsible Person, as defined below, must additionally become a Guarantor.

"Immediate Family" Spouse, parents (including stepparents), siblings (including half siblings), and children (including stepchildren), whether natural or adopted.

"Initial Term" - Defined in Section 3.1 (Grant of Franchise).

"Internet" - All communications between computers and between computers and television, telephone, facsimile, and similar communications devices, including the World Wide Web, proprietary online services, e-mail, news groups and electronic bulletin boards.

"Location" - The franchise location identified in Section 3.1(Grant of Franchise).

"Methods of Operation" - The information we provide to you containing mandatory and suggested specifications, standards, operating procedures (including safety standards), and rules that we prescribe from time to time for the operation of a Business, including the Operations Manual and any other information we provide to you during the term of the Agreement relating to your operation of the franchise business Business or to any other of your obligations under this Agreement and related agreements.

"NAF" - Our National Advertising Fund, as <u>further</u> defined in Section 10.1 (National Advertising).

"Operations Manual" - Our confidential policy manual, as amended from time to time, which may consist of one or more manuals, including any of our operating system manuals, compliance manuals, and management training manuals, containing our Methods of Operation relating to the development and operation of Businesses and other information relating to your obligations under this Agreement. The term "Operations Manual" also includes alternative or supplemental means of communicating such information by other media, including lists, templates, bulletins, e-mails, digital newsletters and other postings, CDs, DVDs, applications, software, digital audio or video files, and any information we share with you through our eompany-intranet-or, project management system, or other software platforms for the System.

"**Operating Assets**" - All fixtures, furnishings, signs, and equipment used in the Business, including the Technology System.

"Owner" - Each person that has any direct or indirect legal or beneficial ownership interest in you, (if you are an Entity, including any person who has a direct or indirect interest in you), this Agreement, the Franchise, or the Business and any person who has any other legal or equitable interest, or the power to vest in himself any legal or equitable interest, in the revenue, profits, rights or assets thereof. As used in this Agreement, any reference to Owner includes all Owners, including Silent Investors investors who will not participate in the day-to-day operations of the Business. If the franchisee Franchisee is one or more individuals, everyone is each

<u>individual will be deemed</u> an Owner. Your Owner(s) is/are identified on **Appendix A** to this Agreement.

"Person" - A "person" refers to both an individual natural person and an Entity.

"Responsible Person" - The individual you so designate in Appendix A and any replacement thereof approved by us. The Responsible Person must be someone who has the authority to, and does in fact, actively direct your business affairs related to the Business.

"Royalty Billing Day" – Two (2) times Once per month, currently on the sixteenth (16th) and the first (1stthird (3rd) day after the preceding month ends. If the payment date falls on a holiday, the fees will be due on the next business day. On these days each calendar month that we designate Royalty Billing Days, we or our authorized designee are authorized by you to withdraw via electronic funds transfer from your designated bank account all Royalty Fees and other amounts then due to us under the terms of the this Agreement via electronic funds transfer from your designated bank account.

"Technology System" - The hardware, software, security cameras, point-of-sale system, other equipment, payment processing systems, and network connections that we specify periodically in the Operations Manuals necessary to operate our point-of-sale system, the security camera system, reservation system, waiver processing system, and other existing or future technology components or systems that we designate from time to time.

"Transfer" - The voluntary, involuntary, direct or indirect sale, assignment, transfer, license, sublicense, sublease, collateral assignment, grant of a security, collateral or conditional interest, Inter-vivos transfer, testamentary disposition or other disposition of this Agreement, any interest in or right under this Agreement, or any form of ownership interest in you or the assets, revenues or income of your Business. A Transfer includes including: (1) any transfer, redemption or issuance of a legal or beneficial ownership interest in the capital stock of, or a partnership interest in, you or of any interest convertible to or exchangeable for capital stock of, or a partnership interest in, you; (2) any merger or consolidation between you and another Entity, whether or not you are the surviving corporation; (3) any transfer in, or as a result of, a divorce, insolvency, corporate or partnership dissolution proceeding or otherwise by operation of law; (4) any transfer upon your death or the death of any of your Owners by will, declaration of or transfer in trust or under the lawsApplicable Laws of intestate succession; or (5) any foreclosure upon your Business or the transfer, surrender or loss by you of possession, control or management of your Business.

<u>"Travel and Living Expenses"</u> – The travel and living expenses (including transportation, food, and lodging) and other out-of-pocket costs incurred by the employees, managers, or other personnel of a party to this Agreement in attending meetings, events, programs, or in visiting locations.

2. YOUR ORGANIZATION AND MANAGEMENT.

2.1 Organizational Documents

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

DMS_US.363658954.1

. If you are <u>an Entity</u>, or at any time <u>become form</u> an <u>Entity and intend to assign this</u> <u>Agreement to such Entity</u>, you and each of your Owners agree and represent that:

- (a) you are duly organized and validly existing under the <u>lawsApplicable</u> <u>Laws</u> of the state of your organization, and, if a foreign Entity, you are duly qualified to transact business in the state in which your Business is located;
- (b) you have the authority to execute and deliver this Agreement and to perform your obligations hereunder; and
- (c) your activities are restricted to those necessary solely for the development, ownership, and operation of a Business in accordance with this Agreement and in accordance with any other agreements entered into with us or any of our Affiliates;
- (d) the articles or certificate of incorporation, partnership agreement or other organizational documents recite that the issuance, transfer, or pledge of any direct or indirect legal or beneficial ownership interest is restricted by the terms of this Agreement; and
- (e) all certificates representing direct or indirect legal or beneficial ownership interests in you now or hereafter issued must bear a legend in conformity with Applicable Law reciting or referring to such restrictions.

2.2 Ownership Interests

.

(a) You and each of your Owners represent, warrant, and agree that the attached **Appendix A** is a current, complete, and accurate list of all Owners. Every time there is a You must notify us of any proposed change into the persons or Entities who are your Owners, and you must receive our written approval prior to effecting such change. You must, within seven (7) calendar days from the date of such an approved change, notify us of the change and cooperate with us in updating **Appendix A** so that it is always current, complete, and accurate at all times.

Each person who is or becomes an Owner who has a greater than ten percent (10%) or greater interest in the franchisee Franchisee Entity, or such other persons as we may designate (including Owners' spouses), must execute an agreement in the form we prescribe, undertaking to be bound jointly and severally by the terms of this Agreement, the current form of which is attached hereto as **Appendix B** (Owners' Personal Guaranty of Franchisee's Obligations). However, if we are entering into this Agreement totally or partially based on the financial qualifications, experience, skills or managerial qualifications of any person or Entity who directly or indirectly owns ten percent (10%) or less interest in the franchisee Entity, we have the right to designate that person as an Owner who must sign Appendix B to this Agreement.

(b) Each person who is or becomes an Owner that (i) has a greater than ten percent (10%) or greater interest in the franchisee Franchisee Entity or (ii) is given access to our

Confidential Information must execute an agreement in the form we prescribe, undertaking to be bound by the confidentiality and non-competition covenants contained in the Agreement, the current form of which is attached hereto as **Appendix C**.

(c) If you are a partnership Entity, then each person or Entity who, now or hereafter, is or becomes a general partner is deemed an Owner who must sign **Appendix B** and **Appendix C**, regardless of the percentagetheir ownership interest percentage.

2.3 Responsible Person/Management of Business

. If you are an Entity, or at any time become, form an Entity and intend to assign this Agreement to such Entity, you must designate in Appendix A as the "Responsible Person" an individual approved by us who must have the authority to bind you regarding all operational decisions with respect to your Business; and have completed our training program to our satisfaction. You (or your Responsible Person) shall exert your best efforts totoward the development and operation of your Business and all other Businesses you own; and, absent our prior approval, you or your Responsible Person may not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitments or which may otherwise may conflict with your obligations hereunder. WeNotwithstanding the foregoing, we shall have no responsibility, liability, or obligation to any party to any such arrangement, agreement or contract, or any amendments thereto, on account of our approval thereof or otherwise you, your Responsible Person (as applicable), or your Owners regarding your assignment or allocation of management responsibilities, and you agree to indemnify and hold us harmless with respect thereto. You must notify us of any proposed change of the Responsible Person and receive our written approval prior to such change. If such change results from the voluntary or involuntary termination of the Responsible Person, you must submit a new proposed Responsible Person within fifteen (15) days after such termination. Neither you nor your ownersOwners will, directly or indirectly, take any actions to avoidnegate, circumvent, or otherwise restrict the authority requirement for the your Responsible Person's authority to bind you with respect to the Business. Your Business must always be managed by you (or your Responsible Person) or by an on-site general or assistant manager or a shift supervisoran hourly team lead who has completed theall appropriate training programs.

2.4 Control Group

. If you or one of your Affiliates have entered into a Multi-Unit Development Agreement with us and are entering into this Agreement pursuant thereto, and you are an Entity, you acknowledge and agree that the Owner or group of Owners described in **Appendix A** hereof that has, directly or indirectly, fifty-one percent (51%) or more ownership interest in you and voting control over its ownership interests in you ("**Control Group**"), has the same ownership interest in and voting control of the Entity that executed the Multi-Unit Development Agreement. Furthermore, you acknowledge and agree that we have the right to approve in advance your ownership structure.

3. GRANT OF RIGHTS.

3.1 Grant Of Franchise

- (a) <u>Grant</u>. Subject to the terms of and upon the conditions contained in this Agreement, we hereby grant you a non-exclusive license (the "Franchise") to operate one <u>(1)</u> Business using the System and the Marks at the location identified on **Appendix E** (the "Location"), for a term commencing on the Effective Date and expiring on the tenth (10th) anniversary of that date ("Initial-Term"), unless sooner terminated in accordance with Section 15 (Termination of Agreement). You hereby accept the Franchise and agree to operate the Business according to the provisions of this Agreement for the entire <u>Initial-Term</u>. You have no right to: (i) sublicense the Marks or the System to any other person or entity, (ii) use the Marks or the System at any location other than the Location, or (iii) to use the Marks or the System in any wholesale, e-commerce, or other channel of distribution besides the operation of the Business at the Location.
- (b) **Protected Territory**. This Agreement grants to you a Franchise for a single Business with ato be located within the protected territory identified on Appendix E (the "Protected Territory"). We reserve the right, in our sole discretion, to modify the Protected Territory and revise Appendix E upon: (i) termination of the Initial Term or termination of this Agreement; (ii) any transfer, redemption or issuance of a legal or beneficial ownership interest in the capital stock of, or a partnership interest in, you or of any interest convertible to or exchangeable for capital stock of, or a partnership interest in, you; (iii) an merger or consolidation between you and another Entity, whether or not you are the surviving Entity; (iv) any transfer in, or as a result of, a divorce, insolvency, corporate or partnership dissolution proceeding or otherwise by operation of law; (v) any transfer upon your death or the death of any of your Owners by will, declaration or transfer in trust or under the laws of intestate succession; or (vi) any foreclosure upon your Business or the transfer, surrender or loss by you of possession, control or management of your Business any Transfer. Except as provided in Section 3.2 (Our Reservation of Rights) and provided that you are in full compliance with this Agreement and all other agreements entered into with us (or any of our Affiliates), we will not openown and operate, or license a third party to openown and operate, a Business within your Protected Territory during the Term of this Agreement.

3.2 Our Reservation of Rights.

Except as otherwise expressly provided in Section 3.1(b) (Protected Territory), we and all our Affiliates (and our and their respective successors and assigns, by purchase, merger, consolidation or otherwise) retain all—our rights with respect to the Marks, the System and Businesses anywhere in the world, and <u>retain</u> the right to engage in any business whatsoever <u>and</u> any other right not expressly granted to you in this Agreement, including the <u>right</u>rights to:

(a) <u>own and operate</u>, and/or grant to others the right to <u>own and operate</u>, Competitive Businesses <u>or other businesses</u> offering similar or identical products, <u>and services</u>, and classes and using the System or elements of the System: (i) under the Marks anywhere

outside of the Protected Territory, or (ii) under names, symbols, or marks other than the Marks anywhere, including inside and outside of the Protected Territory;

- (b) offer to sell, or sell and distribute, any products or services under any trade names, trademarks, service marks or trade dress, including the Marks, anywhere through any distribution channels or methods, which may include(including the Internet, retail stores, wholesale, and the Internet (gift cards, distribution or fulfillment centers, or any other existing or future form of electronic commerce); and regardless of the nature or location of the customers with whom such other businesses and distribution channels do business, even if such businesses sell products or services that are identical or similar to, and/or competitive with those that Businesses customarily sell;
- (c) acquire, be acquired by the assets or ownership interests of, merge, affiliate with or engage in any transaction with other businesses (whether competitive or notincluding Competitive Businesses), with units located anywhere or business conducted anywhere and, even if such businesses are located in the Protected Territory, and: (i) rebrand the other businesses to operate under the Sky Zone® name, Marks, and System; (ii) franchise, license or create similar arrangements with respect to the other businesses and/or permit the other businesses to continue to operate under another name, and/or (iiii) permit the businesses to operate under another name and convert existing Businesses (including yours) to such other name;
- by a business providing products or services similar to those provided at Businesses, or by any other business, even if such business operates, franchises, and/or licenses Competitive Businesses; and
- <u>(e)</u> <u>Operate, or grant any third party the right to operate, any Businesses that we or our designees acquire as a result of the exercise of a right of first refusal or purchase right that we have under any agreements.</u>

4. LOCATION SELECTION, LEASE OR PURCHASE OF LOCATION AND LOCATION DEVELOPMENT.

4.1 Location Selection

E You acknowledge that, following your signingthe execution of this Agreement, you (with or without our assistance) will find and submit to us for our acceptance a Location for your Business. You acknowledge and agree that our recommendation or acceptance of the Location, and any information regarding the Location communicated toor financial projections we may provide you regarding the Location or our standard site selection criteria for Businesses, do not constitute a representation or warranty of any kind, express or implied, as to the suitability of the Location for a Business or for any other purpose. Our recommendation or acceptance of the Location indicates only that we believe that the Location falls within the acceptable criteria for locations that we have established as of the time of our recommendation or acceptance of the Location. You acknowledge and agree that your selection of the Location is based on your own

independent investigation of the suitability of the Location. We have the right to grant or withhold acceptance of any proposed Location in our business judgment.

4.2 Purchase Or Lease Of The Location

- You must lease, sublease, or purchase the Location within nine (9) months after signing this Agreement. Your failure to sign a lease within this timeframe constitutes grounds for immediate termination of this Agreement under Section 15.2 (Termination By Us) and the loss of your non-refundable Initial Franchise Fee. You may request a three (3) month extension-of this time frame to lease, sublease or purchase the Location by sending us a written request at least thirty (30) days before the deadline ("First Extension Period"). We may grant the First Extension Period, without your payment of any extension fee, if we believe you have engaged in a good faith effort to find a Location. If no less than thirty (30) days prior to the expiration of the First Extension Period, you request an additional extension of time to lease, sublease or purchase the Location ("Additional Extension Period"), we shall have the right, in our sole discretion, to grant such Additional Extension Period provided you pay us an extension fee in an amount equal to theour then-current initial franchise fee. We have the right, but not the obligation, to review the business terms of any lease, sublease, or purchase contract for the Location, and you agree to deliver a copy to us for our review before you sign it. You agree that any lease or sublease for the Location must, in form and substance satisfactory to us, include all the provisions set forth on **Appendix D** attached hereto. You may not execute a lease, sublease purchase contract or any modification thereof without our approval. Our approval of the lease, sublease or purchase contract does not constitute a warranty or representation of any kind, express or implied, as to its fairness or suitability or as to your ability to comply with its terms. We do not, by virtue of approving the lease, sublease, or purchase contract, assume any liability or responsibility to you or to any third parties. Such approval indicates only that we believe that the Location and certain terms of the lease, sublease or purchase contract fall within the acceptable criteria we have established as of the time of our approval. You further acknowledge that we have advised you to seek legal counsel to review and evaluate the lease, sublease, or purchase contract. You must deliver a copy of the fully signed lease, sublease, or purchase contract to us within five (5) days after its execution.

4.3 <u>Location Development</u>

(a) <u>Development</u>. You are solely responsible for developing the Business and for—all expenses associated with <u>itsuch development</u>. We will furnish you with mandatory specifications and layouts for a Business, including requirements for dimensions, design, image, interior layout, decor, Operating Assets, <u>and</u>-color scheme, and other suggestions. The mandatory specifications and layouts we provide will not contain the requirements of any <u>federal</u>, <u>state</u>, or <u>local laws</u>, <u>codes</u>, or <u>regulations</u>. You are <u>obligated to have prepared Applicable Laws</u>. We will <u>furnish guidance to you in developing the Location as we deem appropriate. You will consult with our development team to obtain the services of an architect approved by us to produce <u>construction plans</u>, based on the <u>preliminary layout provided</u> by us, to suit the shape and dimensions of the Location. You are responsible for the cost of the architect's services.</u>

Sky Zone Franchise Group, LLC April 2025 FDD

Franchise Agreement

[City, State]

Notwithstanding the foregoing, you must prepare, at your expense, all required construction plans and specifications to suit the shape and dimensions of the Location and to ensure that such plans and specifications comply with all Applicable Laws, building codes and permit requirements and with lease requirements and restrictions, including those concerning the Americans with Disabilities Act ("ADA") or similar rules governing public accommodations for persons with disabilities. We will furnish such guidance to you in developing the Location as we deem appropriate. You will consult with our real estate team to obtain the services of an architect approved by us to produce construction plans, based on the preliminary layout provided by us, to suit the shape and dimensions of the location.

- Plans. After the preliminary layout is completed, you will hire a local (b) licensed architect that meets our standards, or may use an architect approved by us, to detail the layout into construction plans. You are responsible for the cost for both the approved architect's services and the local architect's services. We shall provide the first three (3) space plans ("Space Plan") to you at no additional cost. If additional Space Plans are necessary, the cost shall be Three Hundred Fifty Dollars (\$350) for each additional Space Plan. You must submit all such modified plans and specifications, including design specifications, to us for our approval before starting to develop the Location. You acknowledge that design quality is important to us. All final plans are subject to our approval. At our request, you must submit all revised or "as built" plans and specifications. All development and any signage must be in accordance with the plans and specifications we have approved and must comply with all Applicable Laws, ordinances and local rules and regulations. We may periodically inspect the Location during its development. We do not, by approving your plans or specifications or inspecting the Location, assume any liability or responsibility to you or to any third parties. Such approvals and inspections shall be solely for the purpose of assuring determining compliance with our standards and shall not be construed as any express or implied representation or warranty that your Business complies with any Applicable Laws (including the ADA or any other laws regulating standards for the access to, use of, or modifications of buildings for any by persons whose disabilities are protected by law) or that the construction thereof is sound or free from defects. All prototype and modified plans and specifications for your Business remain our sole and exclusive property, and you may claim no interest therein.
- contractor acceptable to us. You must procure all applicable construction insurance in amounts and coverages acceptable to us. You must provide us with weekly progress reports during construction in a format acceptable to us. We have the right to visit and inspect, the site during the construction phase. Such visits shall be at our expense, except for visits made upon your request, which shall be at your expense. The requirement to complete construction of your Business includes obtaining You must further obtain all required construction and occupancy licenses and permits, developing develop the Location (including all outdoor features and landscaping of the Location, if applicable), installing install all required Operating Assets, and doing complete all other things tasks as may be required pursuant to this Agreement or by practical necessity to have open your Location ready to Business for use by customers. In consideration for the foregoing site development activities and guidance set forth in this Section 4.3, no later than ten (10) days before the date that you open the Business for business use by

customers, you must pay to us a non-refundable site development fee equal to Five Thousand Dollars (\$5,000) (the "Site Development Fee") in the form of a lump sum payment via EFT from your designated bank account (as further described in Section 5.7 below).

(d) <u>Force MajeureNatural Disasters</u>. Notwithstanding anything to the contrary <u>containedset forth</u> in this Section 4.3, you shall not be deemed to be in breach of this Section 4.3 if your failure to start construction, finish construction or open your Business as <u>described</u> above <u>provided</u> results solely from windstorms, rains, floods, earthquakes, typhoons, mudslides, fires, or other natural disasters. Any delay resulting from any of such causes shall extend performance accordingly, in whole or in part, as may be reasonable, except that no such cause, alone or in combination with other causes, shall extend performance more than <u>ninety (90)</u> days without our prior written consent, which consent may be withheld.

4.4 **Your Obligations**

. You agree, at your own expense, to $\frac{docomplete}{developing}$ the following with respect to developing the Business at the Location:

- (a) secure all financing required to develop and operate the Business;
- (b) obtain all permits and licenses required to construct and operate the Business;
- (c) construct all required improvements to the Location and decorate the Business in compliance with plans and specifications we have approved, and which comply with all governmental requirements;
- (d) purchase, or lease, and install all required Operating Assets required for the Business; and
- (e) purchase an initial inventory of authorized and approved products, materials and supplies in accordance with our Methods of Operation; and
- (f) comply with all applicable federal, state, and local laws, ordinances, regulations, Applicable Laws and best practices of the industry.

4.5 Operating Assets

½ You agree to use in developing and operating the Business only those Operating Assets that we have approved for Businesses as meeting our specifications and standards for quality, design, appearance, function, and performance. You agree to place or display at the Location (including both its interior and exterior) only such signs, emblems, lettering, logos, and display materials that we approve from time to time. See Section 9.6 (Products, Supplies, Operating Assets, and Services) for details relating to the acquisition of the Operating Assets.

4.6 **Start-Up Inventory**

<u>After your execution of this Agreement and prior to your commencement of operations hereunder, we We will giveprovide you lists of the start-up inventory we require that you to must obtain prior to commencing operations of your Business hereunder. See Section 9.5 (Products and Services You May Offer) and 9.6 (Products, Supplies, Operating Assets, and Services) for details relating to your inventory.</u>

4.7 Business Commencement

- . You agree that you shall not to-commence operation of the Business until:
- (a) we approve the Business as developed in accordance with our specifications and standards;
- (b) pre-opening training has been completed by you, your Responsible Person, and/or your employees to our as provided described in Section 6.1 (Training);
- (c) you have given us a copy of your <u>approved and executed</u> lease, sublease or purchase contract for the Location;
- (d) the Initial Franchise Fee and all other amounts then due to us have been paid;
- (e) <u>weyou</u> have <u>been</u> furnished <u>with</u> copies of all insurance policies required by this Agreement <u>to us</u> and <u>been</u> named <u>us</u> as an additional insured, or <u>provided</u> such other evidence of insurance coverage and payment of premiums as we request or accept; and
- (f) you have obtained all required permits, licenses and certifications for operating the Business, and the Location is in compliance with all Applicable Laws.

4.8 Commencement Deadline

. You <u>agree to must</u> commence Business operations within eighteen (18) months after the execution of this Agreement and within five (5) days after we notify you that the conditions set forth in this Section have been satisfied.

4.9 Grand Opening Marketing

You agree to conduct grand opening marketing for the Business. The grand opening marketing period is the period beginning a minimum of one hundred twenty (120) days immediately preceding the date that you commence regular operations at the Business to the date you commence regular operations at the Business. You must spend a minimum of Twenty Thirty-Five Thousand Dollars (\$25,000) for your 35,000) toward pre-opening marketing obligations and promotional activities (the "Grand Opening Marketing Expense"). You must spend one half (1/2) of such Grand Opening Marketing Expense prior to the opening of the Business and the remainder in the month in which the Business opens. Such You must provide to us documentation evidencing such expenditures in the form, and including the details and copies of the advertising and materials and receipts, as we request at any time. All of your

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

[City, State]

pre-opening marketing will activities must utilize marketing and public relations programs and media and advertising materials we have approved.

4.10 **Opening Assistance**

. We will send at least one (1) representative to provide on-site opening assistance for at least five (5) days during the opening of the Business. If additional training is required on-site, you will be responsible for all <u>expenses, Travel and Living Expenses</u> incurred by the trainers and we will have the right to charge a reasonable training and assistance fee.

4.11 Relocation

. You may not relocate the Business without our prior written consent. Such approval will not be unreasonably withheld, provided that: (i) the new location for the Business premises is satisfactory to us and you comply with our then-current real estate requirements, (ii) your lease, if any, for the new location complies with our then-current requirements, (iii) you comply with our then-current requirements for constructing and furnishing the new location, (iv) the new location will not, as determined in our sole discretion, materially and adversely affect the revenue of any other Business, (v) you have fully performed and complied with each provision of this Agreement within the last three years prior to, and as of, the date we consent to such relocation (the "Relocation Request Date"), and (vi) you are not in default under this Agreement as of the Relocation Request Date. If your Location lease expires or is otherwise terminated, you must secure our approval of another site and enter a lease, sublease, or purchase agreement for the new approved site within ninety (90) days of such expiration or termination. If we consent to the Business's relocation, we have the right to charge you for the expenses we incur in connection with the relocation. We reserve the right to terminate this Agreement if you fail to secure a new approved site within ninety (90) days after you losefollowing the original expiration or termination of the Location's lease or sublease.

5. FEES.

5.1 <u>Initial Franchise Fee</u>

. You agree to pay us a nonrecurring and nonrefundable initial franchise fee in the amount of Seventy-Five Thousand Dollars (\$75,000) (the "**Initial Franchise Fee**") when you execute this Agreement. However, if you are approved to establish a Business by rebranding an existing trampoline park that you have operated in one of our affiliated franchise systems, you will not be required to pay the Initial Franchise Fee.

5.2 Royalty Fee

. You agree to pay us a nonrefundable royalty (the "**Royalty Fee**") per Accounting Period via EFT. The Royalty Fee your Business will pay shall beis equal to six percent (6%) of the Gross Sales of the Business. We will collect the Royalty Fee on the Royalty Billing Day, pursuant to our Methods of Operation, via the EFT initiated by us or by a third party authorized by us from the designated account identified in Section 5.55.6 (Designated Account), or by such other means as we may authorize and approve.

April 2025 FDD

Franchise Agreement

[City, State]

5.3 Ad Fee

. You must pay to us a continuing advertising fee in the amount of two percent (2%) of the Gross Sales for the Business (the "Ad Fee") per Accounting Period, payable in the same manner as the Royalty Fee. The Ad Fee will be contributed toward the NAF and may be increased or decreased by us withupon at least thirty (30) days' notice to you but the Ad Fee shall not exceed two and one-half percent (2.5%) during the Term of this Agreement.

5.4 Other Fees Technology Fee

• You agree to pay us or our Affiliates such other fees that are set forth in this Agreement or that we otherwise impose. Such fees shall be due at the same time and payable in the same manner as the Royalty Fee, unless different payment terms are expressly stated for such fees at the time they are imposed or thereafter.

. We require you to pay a fee to us, or a service provider we designate (which may be one of our Affiliates), for technology-related services, including, but not limited to, website or email hosting, help desk support, software or website development, enterprise solutions, point-of-sale systems, and other services associated with your Technology System (the "Technology Fee"). The Technology Fee is currently One Thousand Four Hundred and Eighty Dollars (\$1,480) per month and payable in the same manner as the Royalty Fee, beginning two (2) months prior to your Business' opening. The Technology Fee may be increased or decreased by us based upon changes in amounts billed to us from service providers upon thirty (30) days' prior written notice to you; provided, however, the Technology Fee will not be more than our costs plus twenty percent (20%) during the Term of this Agreement. Notwithstanding our collection of the Technology Fee, you may be required to purchase hardware, software, or other components of the Technology System from required vendors.

5.5 Call Center Program Fee. You may be required to participate in a call center program, as may be modified by us from time to time (the "Call Center Program"). Participation in the Call Center Program may include, without limitation, using and publishing a telephone number designated by us; engaging a designated service provider (which may be us, our Affiliate, or approved third-party service providers) to answer calls and handle customer service matters; and acquiring, installing, and using additional hardware, software, or other technology. You agree to pay all fees imposed by us or our approved third-party service providers for these services. We do not currently collect Call Center Program fees, which are instead payable to approved third-party service providers. If imposed, we may increase the Call Center Program fees up to twenty percent (20%) once per calendar year during the Term of this Agreement.

5.6 5.5 Designated Account

. Prior to the opening of the Business, and as a condition thereof, you shall establish a designated bank account from which we or our authorized designeed shall be authorized to withdraw in any manner which we prescribe, which may include EFT, ACH, or wire transfer, any amounts due to us, our Affiliates, or any supplier from you under this Agreement, including

but not limited to, Royalty Fees, Ad Fees and AdTechnology Fees. We have the right to review your sales numbers daily. On the days designated as your Royalty Billing Days, we or our authorized designee shall calculate the Royalty Fee due for that Accounting Period and withdraw such amount and any other amounts due under this Agreement, including any Ad Fees, Technology Fees, or other fees, directly from the designated account. All costs and expenses of establishing and maintaining such designated account, including transaction fees and wire transfer fees, shall be paid by you. You agree to always maintainensure that sufficient funds are available in such your designated bank accounts for such account at all times to cover our withdrawals.

5.7 <u>5.6-Interest On Late Payments</u>

. All amounts which you owe us and do not pay us when due will bear interest after their due date <u>until payment is received in full</u> at the lesser of: (a) the highest rate of interest permitted by <u>lawApplicable Law</u>; or (b) 18% per annum. You acknowledge that this Section does not constitute our agreement to accept any payments after they are due or our commitment to extend credit to, or otherwise finance your operation of, the Business.

5.8 5.7 Application Of Payments

. Notwithstanding any designation you might make, we have the right to apply any of your payments to any of your past due indebtedness to us. You acknowledge and agree that we have the right to set off any amounts you or your <u>ownersOwners</u> owe us against any amounts we might owe you or your <u>ownersOwners</u>.

5.9 5.8 Authorized Overdue Supplier Payment

§. You acknowledge and agree that in order to ensure quality and consistency at all Businesses, we may require that you obtain goods or services from certain designated designate, approve or develop standards and specifications for manufacturers, distributors, and suppliers. Pursuant to our Methods of Operation, we may identify certain suppliers, that of products and services to your Business, which may include us, or any Affiliate, as a mandatory or approved suppler or our Affiliates. You hereby acknowledge and agree that in the event we receive notice from any supplier that you are over 60 days past due on any payment to such supplier, and you have not provided any notice to the supplier disputing such overdue amount prior to our receipt of notice from the supplier concerning any such past due amount, you hereby authorize us to make payment on your behalf of any such overdue amount to such supplier. You acknowledge and agree we may pay any such overdue amount by withdrawing from your designated bank account an amount equal to the overdue amount owed to the supplier. Notwithstanding the foregoing, you acknowledge and agree that nothing in this Section requires us to pay your overdue amounts in lieu of or as a condition precedent to issuing any written notice of default pursuant to Section 15.2(k) herein.

5.10 5.9 Non-Compliance Fee

In the event you are found in violation of receive a written notice of default for failing to comply with any of the terms of this Agreement beyond any applicable and fail to cure such default within the specified cure period (if any), we may impose a fee of Two Hundred Fifty Dollars (\$250) per day for each day you remain out of compliance. If there is no cure period, the Two Hundred Fifty Dollars (\$250) per day finenon-compliance fee shall begin the day after we give you written notice. For any jumper who Notwithstanding the foregoing, if you do not obtain a waiver prior to them using the trampolines or other attractions for any jumper before they use any Attractions at your Business, we have the right to charge you a non-compliance fee of Two Thousand Five Hundred Dollars (\$2,500) per violation. Nothing herein is intended to limit our rights of termination based on your violation of this Agreement.

5.11 5.10 Inspection And Compliance Reimbursement

. You agree to reimburse us for our actual costs if, after an inspection of your Business, we determine (in our business judgment) that additional follow up inspections or assessments are required. Our actual costs may include (but are not necessarily limited to) travel, meals, the Travel and hourly wage expenses Living Expenses of our employees or designees who perform such follow up inspections or assessments.

6. TRAINING, ASSISTANCE, AND METHODS OF OPERATION.

6.1 Training

. Before the Business begins operating, we will furnish initial training on the operation of a Business to you (or, if you are an Entity, your Responsible Person), and your management team that consists of a minimum of three (3) managers and one (1) additional leader (collectively, "Management Team"). The initial training will be furnished at such time and place as we may designate, which may include another Business. You (or your Responsible Person), and your Management Team are required to complete the initial training to our satisfaction. If you are an existing franchisee and you have previously completed our initial training program, you willmay not be required to attend the initial training program, however depending on various factors, including but not limited to how long it has been since you completed the initial training program, your compliance with our Methods of Operation in operating your existing Business(es), and the extent of changes to the initial training program since you last completed such program. Notwithstanding the foregoing, we may require that certain of your employees and that any new general manager complete the initial training program. You also are required to participate in all other activities required to operate the Business. Although we will furnish initial training to you (or your Responsible Person) and your Management Team, at no additional fee or other charge, you will be responsible for all travelTravel and living expensesLiving Expenses and compensation which you (or your Responsible Person) and your Owners or employees incur in connection with such initial training. If we determine that you (or your Responsible Person) are unable to complete initial training to our satisfaction, we have the right to terminate this Agreement pursuant to Section 15.1 (Termination By Us). As of the Effective Date, we do not have plans to change the initial training program, but we reserve the right to do so. If the initial training program changes before you enrollprior to your enrollment in the training such program, you, your Owners and your employees will be required to complete the initial training program that is Feather of franchisees at such time.

April 2025 FDD
Franchise Agreement
[City, State]

6.2 Refresher Training

. We may require you (or your Responsible Person) and/or previously trained and experiencedany of your employees to attend periodic refresher training courses at such times and locations that we designate, and ("Refresher Courses"). Refresher Courses may additionally include electronic training courses, trade shows, ongoing education or certification programs, franchisor-sponsored performance groups, webinars, and courses and programs provided by third parties we designate. You may charge reasonable required to pay fees for such Refresher Courses to third-parties or pay our then-current fees for such courses to us (as applicable). We also may require you to pay us fees for training additional employees or your-new employees you have hired after your Business commences operations.

6.3 General Guidance

- . We may advise you from time to time regarding operating operational issues concerning the Business disclosed by reports you submit to us or on-site inspections we make from time to time. Such guidance may be furnished in our Operations Manual, bulletins, e-mails, through our company intranet, or other written materials or by any electronic transmission and/or during telephone consultations and/or consultations at our office or the Business. In addition, we may furnish guidance to you with respect to:
- (a) standards, specifications and operating procedures and methods utilized by the Business, including the Methods of Operation;
 - (b) <u>equipment and facility maintenance;</u>
- $\underline{\underline{(c)}}$ (b) purchasing required Operating Assets, products, materials, and supplies and inventory management;
 - (d) (e)-advertising and marketing program approvals;
 - (e) (d) employee training your employees;
 - (f) (e)-administrative, bookkeeping and accounting procedures;
 - (g) (f)-use of the authorized and approved Technology Systems System; and
- (h) (g)-sample or template forms or documents, including but not limited to any Membership Agreements (as defined below) or customer waivers. ANY SUCH SAMPLE OR TEMPLATE FORMS ARE PROVIDED FOR INFORMATIONAL PURPOSES ONLY AND ARE NOT INTENDED AS LEGAL ADVICE. WE DO NOT WARRANT THE COMPLETENESS, LEGALITY OR ENFORCEABILITY OF ANY DOCUMENT OR INFORMATION PROVIDED BY US. YOU SHOULD RETAIN YOUR OWN LEGAL COUNSEL TO REVIEW AND REVISE ANY SUCH DOCUMENT OR INFORMATION TO CONFORM TO ALL APPLICABLE-FEDERAL, STATE AND LOCAL LAWS.

6.4 On-Site Consultation and Additional Guidance

- . During the $\underline{\text{term}}\underline{\text{Term}}$ of this Agreement, additional guidance may be provided in any of the following ways:
- (a) Internet and telephone consultation during such times as are outlined in the Operations Manual;
- (b) wholesaling services whereby we may ourselves act as an approved or designated source for products, merchandise, accessories, Operating Assets, etc.;
 - (c) website and social media marketing assistance;
- (d) ongoing marketing programs (notexcluding collateral) to fulfill our obligations in Section 10 (Marketing);
- (e) meetings, seminars or conferences whereby we may get together meet with you and other franchisees for business or social purposes;
 - (f) research and development regarding Methods of Operation; and/or
- (g) at your request, we may furnish additional guidance and assistance and, in such a case if approved, we may charge a reasonable fee for such assistance. If you request, or if we require, additional or special training for your employees, all of the expenses that we incur in connection with such training, including per diem charges Travel and travel and living expenses Living Expenses for our personnel, will be your responsibility alongside a reasonable fee for such training.

6.5 Operations Manual

. During the term of this Agreement, we will allow you to access or borrow our then-current Operations Manual. In our discretion, we may provide you the Operations Manual in paper or electronic form. The information provided Operations Manual contains mandatory and suggested specifications, standards, operating procedures, and rules that we prescribe from time to time for the operation of a Business and information relating to your other obligations under this Agreement and related agreements. We may modify the Operations Manual from time to time to reflect changes in Applicable Laws, the law, marketplace, or our Methods of Operation. The Operations Manual and our Methods of Operations constitute confidential trade secrets and will remain our property. You agree to keep your copy of any bound operations manual and any information we share with you that is a part of our Methods at all times. If there is a Dispute over the Operations Manual's contents, our most recently updated master copy of Operation and/orthe Operations Manual shall control. You agree that the Operations Manual's contents are confidential pursuant to your confidentiality obligations set forth in this Agreement and that you will not disclose the Operations Manual to any person other than any employee of yours who needs to know its contents. You may not at any time copy, duplicate, record, or otherwise reproduce any part of the Operations Manual. You further agree to keep all such information the Operations Manual current and in a secure whether hard copies located location at theyour Business or by keeping and to keep all password -protected electronic

DMS_US.363658954.1

information secure. You agree that these requirements are reasonable and necessary to preserve the identity, reputation, value, and goodwill of the systemSystem.

6.6 Annual Convention

Tyou, or at least one (1) representative identified by you, will be required to attend our annual convention at a location of this Agreement. You will pay all Convention fees established by us for each person attending the Convention, and you will also be responsible for all expenses, including travel meals Travel and living costs, wages Living Expenses and other expenses incurred by the persons attending the Convention on your behalf. Annual Convention fees will not be more than our costs plus twenty percent (20%). You must pay all Convention fees established by us, whether you attend the Convention or not. If you have not paid the Convention fees owed to us, at least fifteen (15) days before the Convention begins, the Convention fees will be automatically debited from your designated bank account. No Convention fees shall be refundable.

7. MARKS.

7.1 Ownership and Goodwill of Marks

. Your right to use the Marks is derived solely from this Agreement and limited to your operation of the Business pursuant to and in compliance with this Agreement and the Methods of Operation. Your unauthorized use of the Marks will be a breach of this Agreement and an infringement of our rights in and to the Marks. You acknowledge and agree that your usage of the Marks and any goodwill established by such use will be exclusively for our and our Affiliates' benefit and that this Agreement does not confer any goodwill or other interests in the Marks upon you (other than the right to operate the Business in compliance with this Agreement). All provisions of this Agreement applicable to the Marks apply to any additional proprietary trademarks and service marks and commercial symbols we authorize you to use. You will not represent in any manner that you have any ownership in the Marks or the right to use the Marks except as provided in thethis Agreement and in the Operations Manual. You will not register or attempt to register any of the Marks or any marks or names confusingly similar to the Marks. You agree not to challenge any of our Marks or any applications or registrations relating thereto at any time.

7.2 Limitations On Your Use Of Marks

. You agree to use the Marks as the sole identification of the Business, except that you agree to identify yourself as the independent owner thereof in the manner we prescribe. You may not use any Marks, or any name that is confusingly similar with any Marks: (i) as part of any corporate or legal business name—or; (ii) as part of an Internet domain name or Internet e-mail address or (other than an e-mail address we issue to you); (iii) with any prefix, suffix or other modifying words, terms, designs or symbols (other than logos licensed to you hereunder), or; (iv) in any modified form, nor may you use any Marks; (v) in connection with the performance or sale of any unauthorized services or products—or in any other manner we have not expressly

authorized in writing. No Marks may be used; (vi) in any advertising concerning the transfer, sale or other disposition of the Business or an ownership interest in you; or (vii) in any other manner we have not expressly authorized in writing. You agree to display the Marks in the manner we prescribe at the Business, on supplies or materials we designate and in connection with forms—and, advertising, promotional, and marketing materials we designate. You agree to giveutilize such notices of trademark and service marks registrations; (i.e., "®", or "TM", as we specify and to obtain any fictitious or assumed name registrations required under Applicable LawLaws. YouFollowing any termination or expiration of this Agreement, you agree to immediately withdraw any fictitious or assumed name registrations immediately upon termination or expiration of this Franchise Agreementwhich use any Marks.

7.3 <u>Notification Of Infringements And Claims</u>

You agree to notify us immediately of any known, actual, suspected, threatened, or apparent infringement or challenge to your use of any Marks, or of any claim by any person of any rights in any Marks and agree not to communicate with any person other than us, our attorneys, and your attorneys in connection with any such potential infringement, challenge or claim. We have the right to take such action as we deem appropriate and the right to control exclusively any litigation, United States Patent and Trademark Office ("USPTO") proceeding or any other administrative proceeding arising out of any such infringement, challenge or claim or otherwise relating to any Marks. You agree to sign all instruments and documents, render such assistance, and do such acts and things as, in the opinion of our attorneys, may be necessary or advisable to protect and maintain our interests in any litigation or USPTO proceeding or other proceeding or otherwise to protect and maintain our interests in the Marks. We will reimburse you for your reasonable costs of taking any action that we have asked you to take pursuant to this Section 7.3.

7.4 Discontinuance Of Use Of Marks

. If it becomes advisableWe may at any time for us and/or require you to modify or discontinue the use of using any MarksMark and/or use one (1) or more additional or substitute trademarks or service marks, you Marks. You agree to replace the Marks at your Business with the modified, additional, or substitute Marks we specify and comply with our all other directions we give regarding the Marks at your Business within a reasonable time after receiving notice thereoffrom us. We willare not be obligated required to reimburse you for any costs or expenses or associated with making such changes, for any loss of revenue attributable due to any modified or discontinued MarksMark, or for your expenses of promoting any expenditures you make to promote a modified or substitute trademark or service markMark. Our rights in this Section 7.4 apply to any and all of the Marks (and any portion of any Mark) that we authorize you to use pursuant to this Agreement and our Methods of Operation. You acknowledge both our right to take these actions and your obligations to comply with our directions.

7.5 Indemnification Of Franchisee

. We agree to indemnify you against, and to reimburse you for, all damages for which and expenses that you are held liable incur in responding to any trademark infringement proceeding

arising out of disputing your authorized use of any Mark pursuant to and in compliance with Marks under this Agreement and, except as provided herein, for all costs you reasonably incur in defending any such claim brought against you, provided if you have timely notified us of such claim and provided further that you and your Owners and Affiliates are in compliance the proceeding, complied with our directions in responding to the proceeding, and have used the Marks in accordance with the terms of this Agreement and all other agreements entered into with us or any of our Affiliates. We are entitled to prosecute our Methods of Operation; provided, however, we will not pay any of your attorneys' costs or fees if you hire your own attorney. At our option, we may defend and/or settle control the defense of any proceeding arising out of from your use of any Mark pursuant to under this Agreement, and, if we undertake to prosecute, defend and/or settle any such matter, we have no obligation to indemnify or reimburse you for any fees or disbursements of any legal counsel retained by you our Methods of Operation.

8. CONFIDENTIAL INFORMATION.

8.1 Confidential Information

. We possess (and will continue to develop and acquire), and may disclose to you either orally or in writing, certain <u>proprietary and</u> confidential information relating to the development and operation of Businesses (the "Confidential Information"), which may include:

- (a) site selection criteria and methodologies;
- (b) plans and specifications for the development of Businesses;
- (c) the Methods of Operation methods and knowledge and experience used in developing and operating Businesses, including and the Operations Manuals Manual;
 - (d) sales, marketing and advertising programs and techniques for Businesses;
- (e) marketing research and promotional, marketing, advertising, public relations, customer relationship management, and other brand-related materials and programs for Businesses;
- (f) knowledge of specifications for and suppliers of certain Operating Assets, products, materials and supplies (including Proprietary Products, as such term is defined below);
- (g) knowledge of the operating results and financial performance of Businesses other than theyour Business;
 - (h) methods of training and management relating to Businesses;
- (i) customer communication and retention programs, along with data used or generated in connection with those programs;
 - (j) computer system and software programs used or useful in Businesses;

- (k) <u>information generated by, or used or developed in, your Business'</u> operation, including customer waivers and information relating to customers, such as customer names, addresses, telephone numbers, email addresses, membership information, buying habits, preferences, demographic information and related information, and any other information contained from time to time in the Technology System (collectively, "Customer Information"); and
- (l) (k) all other information related to the Business or Businesses generally that we reasonably designate from time to time as confidential or proprietary. This includes all customer and membership lists and information for the Business and Businesses generally.

"Confidential Information" does not include: (i) information that is part of the public domain or becomes part of the public domain through no fault of you, (ii) information disclosed to you by a third party having legitimate and unrestricted possession of such information, or (iii) information that you can demonstrate by clear and convincing evidence was within your legitimate and unrestricted possession when the parties began discussing the sale of the franchise prior to disclosure.

8.2 Nondisclosure of Confidential Information

. We will disclose the Confidential Information to you solely for your use in the operation of your Business pursuant to this Agreement. The Confidential Information is proprietary to us and our Affiliates and includes our trade secrets. During the termTerm of this Agreement and thereafter: (a), you and your Owners: (a) may not use the Confidential Information in any other business or capacity; (b) you and your Owners-must exert your best efforts to maintain the confidentiality of the Confidential Information; (c) you and your Owners may not make unauthorized copies of any portion of the Confidential Information disclosed in written, electronic or other form; and (d) you and your Owners must implement all reasonable procedures we prescribe from time to time to prevent unauthorized use or disclosure of the Confidential Information, including the use of nondisclosure agreements (in a form that we prescribe or approve that identifies us as a third party beneficiary of such agreements with the independent right to enforce the agreements with your Owners, Responsible Person, officers, directors, managers, assistant managers, shift supervisors, and any other individuals who have access to the Confidential Information, which you must deliver to us. At the end of the term; and (e) will not sell, trade or otherwise profit in any way from the Confidential Information, except using methods approved by us. Upon any expiration or termination of this Agreement, you and your Owners must deliver to us all such Confidential Information in your possession. You acknowledge and agree that you will not acquire any interest in Confidential Information, other than the right to utilize Confidential Information disclosed to you in operating the Business during the term of this Agreement, and that the use or duplication of any Confidential Information in any other business or capacity will constitute an unfair method of competition and a violation of this Agreement.

8.3 Customer Personal Information

- (a) <u>Protection of CustomerPersonal Information</u>. You must comply with our Methods of Operation, other directions from us, and all Applicable Laws <u>(including privacy and data protection laws)</u> regarding—the organizational, physical, administrative and technical measures and security procedures to safeguard the confidentiality and security of <u>CustomerPersonal</u> Information on your Technology System or otherwise in your possession or control and, in any event, employ reasonable means to safeguard the confidentiality and security of <u>CustomerPersonal</u> Information, and comply with all applicable privacy and data protection laws and regulations. You shall always comply with the Payment Card Industry Data Security Standards. "<u>CustomerPersonal</u> Information" means refers to any information that can be used to identify an individual, including names, contact information, financial information, and other personal information of or relating to the <u>Business's</u> customers and prospective customers <u>of the Business</u>.
- <u>(b)</u> <u>Breach or Unauthorized Access</u>. If there is a suspected or actual breach of security or unauthorized access involving <u>your CustomerPersonal</u> Information, you must notify us immediately after becoming aware of such actual or suspected occurrence and specify the extent to which <u>CustomerPersonal</u> Information was compromised or disclosed. You are responsible for any financial losses you incur or remedial actions that you must take because of a breach of security or unauthorized access to <u>CustomerPersonal</u> Information in your control or possession.
- (c) (b)—Ownership of CustomerPersonal Information. You agree that all CustomerPersonal Information that you collect in connection with your Business is a subset of Customer Information and shall be deemed to be owned by us and must be furnished to us at any time that we request it. In addition, we and our affiliates may, through the Technology System or otherwise, have independent access to CustomerPersonal Information, in compliance with applicable privacy and data protection laws and regulationsApplicable Laws.
- CustomerPersonal Information whileduring the Term of this Agreement or any successor franchise agreement is in effect, but only to market SKY ZONEapproved products and services for your Business to customers in accordance with the policies that we establish periodically and Applicable Laws. You may not sell, transfer, or use CustomerPersonal Information for any purpose other than marketing SKY ZONEapproved products and services for your Business and in accordance with this Agreement and our Methods of Operation. We and our Affiliates may use CustomerPersonal Information in any manner or for any legal purpose. You must secure from your actual and prospective customers and others all consents and authorizations from, and provide them all required disclosures to, that actual and prospective customers and other individuals as required by Applicable Law requires Laws to transmit Customer and Information to us and our Affiliates, and for us and our Affiliates to use that Customer such Personal Information, in the manner that this Agreement contemplates.

9. SKY ZONE METHODS OF OPERATION.

9.1 Compliance With Methods of Operation

. You acknowledge that each aspect of the interior and exterior appearance, layout, decor, services, and operation of your Business is important to protect our reputation and goodwill and to maintain uniform operating standards under the Marks. Any required standards exist to protect our interest in the System and the Marks and are not for the purpose of establishing any control, or the duty to take control, over those matters that are clearly reserved to you. You agree to operate and maintain the Business in full compliance with any mandatory specifications, standards, and operating procedures set forth in our Methods of Operation and Operations Manual, as we periodically modify and supplement them during the term of this Agreement, relating to the appearance, function, cleanliness, or operation of a Business. Your Business may not be used for any purpose, other than the operation of a Business in compliance with this Agreement. You agree that your Business will offer courteous and efficient service and a pleasant ambiance. You must keep the Business open for business to the public at least during the hours we prescribe from time to time in the Operations Manual or otherwise approve, unless prohibited by Applicable Laws or by the lease for the Location. You expressly acknowledge and agree that our minimum required hours of operation as set forth in the Operations Manual may require you to keep the Business open for business on Sundays. You acknowledge and agree that such mandatory standards also include that you shall, under no circumstances, exceed the maximum capacity of the Business as determined by us, including maximum capacity for individual pieces of equipment, and that you shall always ensure that itsyour security camera system is operational and that we have independent access to such security camera system. You further acknowledge and agree that the maximum capacity set by us as described herein may be more stringent than maximum capacity requirements of local laws. Applicable Laws, but we have established the maximum capacity for safety reasons based on the size, layout and design of your Business.

9.2 **Provisions Of This Agreement**

. You agree that the Methods of Operation prescribed from time to time in the Operations Manual, or otherwise communicated to you in writing or other tangible form, constitute provisions of this Agreement as if fully set forth herein. All references to this Agreement include the Operations Manual and all of the Methods of Operation as may be periodically modified.

9.3 Modification Of Methods Of Operation

. We may periodically modify the Methods of Operation, which modifications may include regional or local variations as we determine, provided, however, that such modifications will not alter your fundamental status and rights under this Agreement. You acknowledge that any modifications to the Methods of Operation may obligate you to invest additional capital in the Business ("Capital Modifications") and/or incur higher operating costs. We will not obligate you to make any Capital Modifications when such investment cannot, in our reasonable judgment, be amortized during the remaining term Term of this Agreement, unless we agree to extend the term Term of this Agreement so that such additional investment, in our reasonable judgment, may be amortized, or unless such investment is necessary to comply with Applicable Laws or safety standards we prescribe.

9.4 Condition Of Your Business

- Business so that it is attractive, clean, and efficiently operated in accordance with the Operations Manual. You agree to make such modifications and additions to the layout, decor, operations, and general theme of the Business as we require from time to time, including replacement of worn-out or obsolete Operating Assets, repair of the interior and exterior and appurtenant parking areas, and periodic cleaning and redecorating. If at any time the general state of repair, appearance or cleanliness of your Business, or its Operating Assets, does not meet our standards, we may notify you and specify the action you must take to correct such deficiency. If, within ten (10) days after receiving such notice, you fail or refuse to initiate and thereafter continue in good faith and with due diligence a bona fide programefforts to complete such required maintenance, we have the right (in addition to our rights under Section 15 (Termination of this Agreement), but not the obligation, to enter the Location and do such maintenance on your behalf and at your expense. You must promptly reimburse us for such expenses.
- (b) Remodeling and Upgrades. You must periodically re-equip, upgrade and/or remodel your Business pursuant to our plans and specifications, provided, however, that, except for signage, we will not require substantial remodeling more often than every four (4) years during the term Term of this Agreement. In addition, on or about the tenth (10th) anniversary of the Effective Date, we have the right, in our business judgment, to require that you fully remodel and/or expand the Business, add or replace improvements, equipment and signs and otherwise modify the Business as we require to bring it into compliance with specifications and standards then applicable for Businesses, or if If you are unable to maintain possession of the Location, or if in our judgment the Business should be relocated, you must secure substitute premises we approve, develop such premises in compliance with specifications and standards then applicable for Businesses and continue to operate the Business at the Location until operations are transferred to the substitute premises.
- (c) Repair and Reconstruction. If your Business is damaged or destroyed by fire or other casualty, you must initiate within thirty (30) days (and continue until completion) all repairs or reconstruction to restore your Business to its original condition. If, in our reasonable judgment, the damage or destruction is of such a nature that it is feasible, without incurring substantial additional costs, to repair or reconstruct your Business in accordance with our then-standard layout and decor specifications, we may require you to repair or reconstruct your Business in accordance with those specifications. You may not make any alterations to your Business, nor any replacements, relocations, or alterations of Operating Assets, without our written approval. We have the right, at your expense, to rectify any replacements, relocations or alterations not previously approved by us.

9.5 Products and Services You May Offer

(a) <u>Offerings</u>. You agree that your Business will offer for sale such services, products, and merchandise related to the <u>conceptSystem</u> that we determine from time to time to

be appropriate for your Business. You further agree that your Business will not, without our written approval, offer any services, products (including promotional items), or attractions not then authorized by us. In addition, you must offer the specific products, services, and attractions that we require in the Operations Manual or otherwise in writing. You may sell products and services only in the varieties, forms, and packages that we have approved in accordance with our Methods of Operation. We may change these specifications periodically, and we may designate specific products, services, or attractions as optional or mandatory. You must maintain a sufficient supply of required products to meet the inventory standards we prescribe in the Operations Manual (or to meet reasonably anticipated customer demand if we have not prescribed specific standards). YouExcept as otherwise set forth in the Operations Manual, you may not install or operate any video, arcade, lottery games or games of chance, which are strictly prohibited from your Business.

(b) <u>Test Marketing</u>. We may conduct market research to determine consumer trends and salability of new services and products. You agree to cooperate by participating in our market research programs; by test marketing new services and merchandise in your Business and providing us timely reports and other relevant information regarding such market research. You must purchase a reasonable quantity of such test products and make a reasonable effort to sell them.

9.6 Products, Supplies, Operating Assets, and Services

.

- Qperating Assets, and services that you purchase for resale or purchase or lease for use in your Business: (i) meet standards and specifications that we establish from time to time; (ii) be a specific brand, kind, or model; (iii) be purchased or leased only from suppliers or service providers that we have expressly approved; and/or (iv) be purchased or leased only from a single source that we designate (which may include us or our Affiliates or a buying cooperative organized by us or our Affiliates). To the extent that we establish specifications, require approval of suppliers or service providers, or designate specific suppliers or service providers for items or services, we will publish our requirements in the Operations Manual or otherwise in writing. You will establish independent commercial relationships with our approved suppliers for specific items and with other suppliers for the goods and services for which we only provide specifications.
- (b) Required Purchases from Us or Our Affiliates. So long as During the Term of this Agreement is in effect, you agree to purchase the products which may be listed, if we or any of our Affiliates is identified as the sole designated supplier for certain products, services, or supplies, as set forth in the Operations Manual or otherwise in writing, you must purchase such products, services, or supplies from us or our Affiliates. Currently these products are We or our Affiliates are currently the sole designated supplier for the Attractions (including their installation), balls, stickers, wristbands—and, SkySocks. You may purchase certain marketing materials, and certain merchandise from us (the "Proprietary Products"). We or our Affiliates.

Sky Zone Franchise Group, LLC
April 2025 FDD

April 2025 FDD

Franchise Agreement

These may add or remove items may change in from the future and other items may be added ist of Proprietary Products at any time. You acknowledge that your agreement to comply with the foregoing sourcing requirements from us or our Affiliates is a material condition upon which this franchise Franchise is granted. Further, to maintain the high standards of quality and uniformity associated with the System, you acknowledge that the purchase and use of any proprietary products which you must purchase from us or our Affiliates, are such Proprietary Products is an integral part of operating the Business. You acknowledge that we and our Affiliates may be approved suppliers and may be the only approved suppliers for certain products. If we or our Affiliates are unable to offer these items any Proprietary Products, we may designate an engagine or approved supplier or suppliers for these items.

- (c) <u>Availability</u>. Contingent upon the availability of <u>productsProprietary Products</u>, we and our Affiliates will use commercially reasonable efforts to supply such <u>productsProprietary Products</u> to you within a reasonable time after the receipt of said orders, provided, however, that neither we nor our Affiliates <u>warrantswarrant</u> that: (i) we or our Affiliates will be able to <u>obtainfulfill</u> all <u>such productsorders</u>, or (ii) that <u>the productsany Proprietary Products</u> will be <u>obtained</u> available to you by <u>the datesyour</u> requested <u>dates</u>, if any.
- Affiliates, to preserve our legitimate interest in protecting protect the quality of the Sky Zone Indoor Trampoline BusinessesSystem and the SKY ZONE brand, its reputation and goodwill, we and our Affiliates reserve the right to, to the fullest extent allowed by applicable law, establish and have exclusive control over the prices, discounts, specifications, and all other terms and conditions governing the sale of products Proprietary Products, including Attractions, to you. Pricing of product or Proprietary Products is subject to change at any time or from time to time by us or our Affiliates effective upon thirty (30) days prior written notice to you. Generally, prices will change when the manufacturer's prices to us or our Affiliates change or in connection with manufacturers price changes, tariffs, or based upon increases in the cost of oil—or, steel increases. You must submit for our approval any attempts to discount or promote the jump time pricing because it may impact brand equity or other raw materials. Section 9.11 (Pricing Policies) sets forth information about our abilities to periodically set maximum or minimum advertised pricing for the products or services offered at your Business.
- (e) <u>Terms of Purchase</u>. All purchases of products, supplies and equipment from us or our Affiliates, must be personally guaranteed by the <u>individuals who are required to guarantee this Agreement pursuant to the Guarantee attached Guarantors</u> hereto. The terms of payment will be established by us and our Affiliates from time to time.
- Affiliates may derive revenue based on your purchases and leases, including from charging you for products and services we or our Affiliates provide to youany Proprietary Product sales and from promotional allowances, volume discounts, and other payments made to us by third-party manufacturers, suppliers and/or distributors that we designate or approve for some or all our franchisees. We and our Affiliates may use all amounts received from suppliers and/or distributors, whether based on your or other franchisees' actual or prospective dealings with them, without restriction for any purposes we or our Affiliates deem appropriate. If you intend to Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

[City, State]

derive any revenue based on payments or promotional allowances received from suppliers and/or distributors, you must report to us the details of the proposed arrangement and accompanying details to us prior to formalizing such arrangement and if approved, such revenue shall be included as part of your Gross Sales.

- Approval Process. If you would like to offer products, services, or classes (g) or use any supplies, Operating Assets, or services that we have not approved or to purchase or lease from a supplier, suppliers or service provider that we have not approved, you must submit a written request for approval and provide us with any information that we request. We have the right to inspect the proposed supplier's facilities and test samples of the proposed products and to evaluate the proposed service provider and the proposed service offerings. We may require the proposed supplier or service provider to visit our headquarters to evaluate the proposed supplier or service provider in person. You agree to pay us a charge not to exceed the reasonable cost of the inspection and our actual eostcosts of testing the proposed product or evaluating the proposed service or service provider, including Travel and Living Expenses for our personnel-and travel costs, whether or not the itemproduct, service, supplier, or service provider is approved. We have the right to grant, deny, or revoke approval of products, services, suppliers, or service providers based solely on our judgment. We will notify you in writing of our decision as soon as practicable following our evaluation. If you do not receive our approval within ninety (90) days after submitting all the information that we request, our failure to respond will be deemed a disapproval of the request. You acknowledge that the products and services that we approve for you to offer or use in your Business may differ from those that we permit or require to be offered or used in other Businesses.
- Revocation of Approval. We reserve the right to reinspect the facilities (h) and products of any approved supplier and to reevaluate the services provided by at any service provider at and totime. We may, with or without cause, revoke our approval of the item, service, any supplier, or service provider if any fail to meet any of our then current criteria. If you receive a notice of revocation of approval, you agree to cease purchasing or leasing the formerly approved item or service or any items or otherwise revise our supplier approval process at any time. If we revoke our approval for a supplier or service provider, you must immediately stop purchasing products, supplies, or services from the formerly approved such supplier or service provider and you must dispose of your remaining inventory of the formerly approved items and services as we direct. If we revoke approval of a previously approved product that you have been selling to customers or a service that you have been offering to customers, you must immediately discontinue offering the service and may continue to sell the product only from your existing inventory for up to thirty (30) days following our disapproval. We have the right to shorten this period if, in our opinion, the continued sale of the product would prove detrimental to our reputation or to the goodwill and reputation of the System. After the thirty (30-) day period, or such shorter period that we may designate, you must dispose of your remaining formerly approved inventory as we direct.

9.7 Compliance With Laws

. You must <u>secure and</u> maintain in force in your name all required <u>license_licenses</u>, permits, and certificates relating to the operation of your Business. You must operate your <u>Sky Zone Franchise Group, LLC</u>

April 2025 FDD

Franchise Agreement

Business in full compliance with all Applicable Laws. You agree to comply and assist us in our compliance efforts with any Applicable Laws. You must notify us in writing immediately upon the commencement of any legal or administrative action, or the issuance of an order of any court, agency, or other governmental instrumentality, which may adversely affect the development, occupancy or operation of your Business or your financial condition; or the delivery of any notice of violation or alleged violation of any Applicable Law, including those relating to health or sanitation at your Business. All your advertising and promotion must be completely factual and must conform to the highest standards of ethical advertising. In all dealings with us, as well as your customers, suppliers, lessors, and the public, you must adhere to the highest standards of honesty, integrity, fair dealing, and ethical conduct. You agree to refrain from any business or advertising practice which may be injurious to our business, to the business of other Businesses or to the goodwill associated with the Marks and System. You agree to keep up to date on the developing ASTM standards for the safe operations of trampoline parks and to comply with such standards.

9.8 Personnel

.

- Your Responsibility. You are solely responsible for all employment decisions with respect to your personnel, including hiring, firing, compensation, benefits, discipline, regardless whether training, supervision, and of you receive advicerecommendations or other guidance from us on any of these subjects. We do not require youTo the extent that we elect to implement provide any sample employment-related policies or procedures or security-related policies or procedures that we (at our option) may make available to you inwithin the Operations Manuals Manual or otherwise for your, such policies and procedures are optional use. You shall determine to what extent, if any, these such sample policies, and procedures may be applicable to your operations at the Business. Any training that we provide to your employees will be limited to teaching them-brand standards and about the System related information to promote the quality and uniformity associated with the System. You must perform background checks (meeting the standards we set forth in the Operations Manual, which may change from time to time based on insurance carriers' requirements) for all employees you hire and periodically update such background checks as specified in the Operations Manual. These requirements may affect who you may hire.
- (b) <u>Management</u>. Your Business must be staffed by at least one <u>(1)</u> trained general manager and appropriate numbers of assistant managers, shift supervisorsteam leads, and personnel so that all shifts are staffed by at least one <u>(1)</u> assistant manager or shift supervisor team lead, unless otherwise approved by us.

9.9 Insurance

.

(a) <u>Master Insurance Program</u>. In conjunction with our broker and underwriting partners, we and Sky Zone, LLCour Affiliate Loscann Insurance Company have

developed an insurance and risk management program, which we may amend from time to time in our sole discretion (the "Master Insurance Program"). You agree to participate in the Master Insurance Program and any subsequent or additional mandatory insurance program we or our affiliates Affiliates may create. You also agree to pay via EFT, on a monthly basis, your allocated share of the overall systemwide Master Insurance Program costs (the "Total Cost of Risk"), which we or our affiliates Affiliates determine in our sole discretion. We or our affiliates Affiliates may recalculate the Total Cost of Risk every six (6) months (each six (6)-month period, a "Coverage Period"), at which time the Total Cost of Risk may increase or decrease. We will notify you of your Total Cost of Risk prior to the commencement of each Coverage Period. We and our affiliates Affiliates may use or distribute any Total Cost of Risk payments that we collect from you that are related to the Master Insurance Program for any purpose, without limitation. The continuation of the Master Insurance Program is dependent on many factors and nothing herein shall constitute a guarantee or promise that we will continue to offer the Master Insurance Program, or any particular coverage or service included therein, that the Total Cost of Risk or other charges related to the Master Insurance Program will remain the same, or that the Master Insurance Program will continue being offered during the term of this Agreement or for any other period of time. We cannot guarantee that the Master Insurance Program coverage will remain the same or that the Total Cost of Risk, Captive SIR Program (defined below) amounts, or other fees related to the Master Insurance Program will not change in the future. We may discontinue or modify the Master Insurance Program at any time and may require you to participate in additional or replacement mandatory insurance programs. If we at any time discontinue, or reduce the scope of, the Master Insurance Program, you will be required to buy and maintain at your own expense, insurance with at least the same level of coverage as provided under the then current Master Insurance Program, from an approved supplier of general liability and excess insurance. If your state requires greater coverage amounts than that offered in the Master Insurance Program, you must obtain and maintain any additional coverage as required by your state.

Captive Self-Insured Retention Program. Under the Master Insurance Program, each claim has a \$250,000 self-insured retention. We have established the Captive Self-Insured Retention Program (the "Captive SIR Program") as part of the Master Insurance Program in which you must participate. The Captive SIR Program Terms and Conditions included in the Operations Manual shall govern your participation in the Captive SIR Program. The Captive SIR Program will be used to fund claims that are above \$10,000 and up to the \$250,000 self-insured retention per claim. The funds will be held in a separate account and managed by us or one of our affiliates Affiliates and are nonrefundable. All costs of administering the Captive SIR Program will come from amounts contributed to the Master Insurance Program by individual Businesses through the Total Cost of Risk payments. A regular actuarial analysis of the amount of estimated losses covered by the \$250,000 self-insured retention and the estimated losses covered by each Business's \$10,000 deductible responsibility for the per-claim self-insured retention will be completed by our-insurance broker and their actuarial team. Each Business must make non-refundable monthly payments into the Captive SIR Program (as part of the Total Cost of Risk payments) and pay up to the \$10,000 per claim costs as part of the self-insured retention. Contributions to the Captive SIR Program and the up to \$10,000 per claim may be collected via EFT from your designated bank accounts, in our sole

discretion. All administrative costs and expenses associated with the Captive SIR Program will be paid out of the Total Cost of Risk payments.

- (c) The continuation of the Master Insurance Program and/or Captive SIR Program is dependent on many factors and nothing herein shall constitute a guarantee or promise that we will continue to offer the Master Insurance Program and/or Captive SIR Program, or any particular coverage or service included therein, that any fees related to the Master Insurance Program and/or Captive SIR Program will remain the same, or that the Master Insurance Program and/or Captive SIR Program will continue being offered during the Term of this Agreement or for any other time period. We reserve the right to discontinue or modify the Master Insurance Program and/or Captive SIR Program at any time. In addition, without limiting any rights we have reserved in Section 9.9(a) above, we reserve the right to implement new insurance programs in which you must participate at your sole cost and expense, including programs made available by a captive self-insured retention company.
- Notwithstanding the Master Insurance Program and Captive SIR Program, during the Term, you must maintain in force at your sole expense insurance coverage for the Business in the amounts, and covering the risks, we periodically specify in the Operations Manual. We may require some or all your insurance policies to provide for waiver of subrogation in favor of us and certain of our affiliates. Your insurance carriers must be licensed to do business in the state in which the Business is located and be rated "A" or higher by A.M. Best and Company, Inc. (or such similar criteria we periodically specify). Insurance policies must be in effect before you begin constructing the Business. We may periodically increase the amounts of coverage required under those insurance policies and/or require different or additional insurance coverage at any time we deem necessary or appropriate to reflect, among other factors, inflation, identification of new risks, changes in LawApplicable Laws or standards of liability, higher damage awards, or relevant changes in circumstances. Insurance policies must name us and any affiliates Affiliates we periodically designate as additional insureds and provide for no less than thirty (30) days' prior written notice to us of any policy's material modification, cancellation, or non-renewal or any non-payment. You must periodically, including before the Business opens, send us a valid certificate of insurance or duplicate insurance policy evidencing the coverage specified here or in the Operations Manual and the payment of premiums. We will require you to use our designated insurance broker or vendors, which may include us or our Affiliates, to facilitate your compliance with the these insurance requirements. We have the right to obtain insurance coverage for the Business at your expense if you fail to do so, in which case you must reimburse our costs in a timely fashion. We also have the right to defend claims in our sole discretion.
- (e) You must, at your sole cost and expense, procure, prior to the commencement of construction or any operations under this Agreement, and shall maintain insurance of the types and in such amounts as may be required below or as set forth in the then current Operations Manual, and by the terms of any lease, sublease, mortgage or deed of trust for the Business and in no event less than the following coverage in the following minimum amounts:

- (i) Workers compensation with minimum amounts as follows: \$500,000 per accident, \$500,000 disease policy limit, and \$500,000 disease limit per employeerequired by Applicable Law (with All States Broad Form);
- (ii) Participation in the Master Insurance Program and Captive SIR Program, or if such programs are discontinued, comprehensive general liability coverage in an amount not less than \$1,000,000.002,000,000.00 per occurrence and \$3,000,000.004,000,000.00 aggregate with an "A" AM Best rated carrier. The total liability limit will be \$6,000,000. Such coverage will include: (a) Bodily injury to or death of one or more persons; (b) Property insurance including damage or destruction and builder's risk; (c) Product liability and completed operation liability; (d) Broad form contractual liability, specifically including the indemnification of us and our affiliates; (e) Fire legal liability; (f) Sexual Misconduct Liability; and (fg) non-owned automobile liability; and
- (iii) Excess or umbrella liability insurance with limits of not less than \$1,000,000 excess over \$5,000,000 with an "A" AM Best rated carrier.
- (f) WeNotwithstanding anything to the contrary in this Section 9.9, we may, from time to time, in our sole discretion, make such changes in minimum types of policies, policy limits, coverage, endorsements and other provisions as it may determine.
- (g) Regardless of the Master Insurance Program and Captive SIR Program or the amounts set forth in this Agreement or the Operations Manual, it is your responsibility to maintain adequate insurance coverage at all times during the term of and after the expiration of this Agreement, so that coverage, including but not limited to any policies that are on a "claims made" basis, which through the purchase of an extended reporting endorsement (i.e., "tail" insurance") will be in effect for acts or omissions that occurred prior to the termination of the policy and are reported within a <u>twenty-four (24)</u> month period following the end of the policy period.
 - i. All general liability insurance policies will provide that we must receive <u>no less than thirty (30)</u> days prior written notice of any termination, expiration or cancellation of the insurance policy. Each insurance policy maintained by you for the Business must: (a) name you as the first named insured; (b) must name us, and our Affiliates, successors, and assigns as additional insureds; (c) include a waiver of the insurer's right of subrogation against any additional insureds; and (d) provide coverage for your indemnification obligations under this Agreement.
 - ii. If you fail to maintain such insurance, we may procure such insurance on behalf of you, and will be entitled to reimbursement from you, in addition to any other rights and remedies under this Agreement. However, we are not obligated to obtain such insurance on behalf of you.
 - iii. You should determine, through consultation with your advisors, if additional insurance is necessary, and you recognize that any recommended levels are merely minimum requirements.

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

DMS_US.363658954.1

- iv. Your failure to maintain coverage will not relieve you of any contractual responsibility or obligation or liability under this Agreement.
- v. You must use an approved supplier for general liability and excess insurance, but all other insurance may be obtained from any licensed insurance agent or broker.
- (h) <u>Business Interruptions</u>. For any interruption in the operation of the Business, you shall continue to pay us, during such period of interruption, continuing Royalty Fees based on the average monthly Royalty Fees paid by you during the <u>twelve (12)</u> months immediately preceding the period of interruption if you have business interruption insurance.

9.10 Quality Control

. We have the right to establish "quality control" programs, such as a "including but not limited to mystery shopper" or "secret shopper" program programs, a customer satisfaction measurement programprograms, client surveys, and/or a "customer intercept" programprograms, to ensure the highest quality of service and products in all Businesses. You shall participate in anyall such quality control programs, we designate and bear your pro rata share, as determined by us, of the costs of any such programwe reserve the right to seek reimbursement of all associated costs and expenses.

9.11 **Pricing Policies**

If we determine Subject to Applicable Laws, we may periodically set a maximum or minimum price that we may lawfully require you to charge certain pricesyou may advertise for goods or products and services, certain minimum prices for goods or services, or certain maximum prices for goods or services, you must adhere to our pricing policies offered by your Business (as set forth in the Operations Manual or otherwise in writing from time to time). If we impose a maximum advertised price for any product or service, you may not advertise a higher price for the product or service than the maximum advertised price we impose. If we impose a minimum advertised price for any product or service, you may not advertise a lower price for such product or service than the minimum advertised price we impose. Further, you must comply with any advertising policy we adopt which may prohibit you from advertising any price for a product or service that is different than our suggested retail price. You must provide us with your current price list upon our request. If you wish to discount your Business' jump time pricing, you must submit such requests for our approval in accordance with Section 10.2(b) herein to ensure compliance with our marketing and advertising policies.

9.12 Technology System

.

(a) <u>Use of Technology System</u>. You must obtain, maintain, and use the Technology System that we specify in your Business. You must use the Technology System to: (i) enter and track purchase orders and receipts, attendance, and <u>eustomer informationCustomer</u>

- <u>Information</u>, (ii) update inventory, (iii) enter and manage <u>your</u> customers' <u>contact</u> information Personal Information, (iv) create management contracts: (v) generate sales reports and analysis relating to the Business, (vi) manage waivers, and (vi) provide other services relating to the operation of the Business. If we require you to use any proprietary software or to purchase any software from a designated vendor, you must execute and pay any fees associated with any software license agreements or any related software maintenance agreements that we or the licensor of the software require.
- (b) <u>Changes to Technology System</u>. We may add, remove, or modify components of the Technology System periodically and may designate approved suppliers or specifications for such items. You must replace, upgrade, or update at your expense the Technology System at your expense as we may require periodically without limitation. You may be required to invest in and implement new technology initiatives at your own expense, which may include changes related to the acceptance of implementing new payment methods, monitors, music, Internet TV broadcast broadcasts, software management applications, surveillance systems, e-learning modules, and software applications designed to better manage business functions and control costs. We will establish reasonable deadlines for implementation of any changes to our Technology System requirements.
- System will be dedicated for business uses relating to the operation of the Business; (ii) to use the Technology System in accordance with our policies and operational procedures; (iii) to transmit financial and operating data to us as required by the Operations Manual and this Agreement; (iv) to do all things necessary to give us unrestricted access to the Technology System at all times (including users IDs and passwords, if necessary) so that we may independently download and transfer data via a modem or other connection that we specify; (v) to maintain the Technology System in good working order at your own expense; (vi) to ensure that your employees are adequately trained in the use of the Technology System and our related policies and procedures; and (vii) not to load or permit any unauthorized programs or games on any hardware included in the Technology System. If you use or download any unauthorized software, you shall be liable for all damages and problems caused by the unauthorized software in addition to the other remedies provided under this Agreement.
- (d) <u>Security Cameras</u>. As part of the Technology System, you must install and maintain security cameras to our specifications that will be used to ensure that quality and safety standards are maintained at the <u>Business</u>. You must additionally record and retain all video recordings from your security camera system for at least one (1) year. You must ensure that your security camera system is always operational and that we have independent access to your security system and video recordings at all times.
- (e) <u>Digital Communication</u>. You must, at your expense, communicate electronically with us, using methods that we designate, including e-mail and use of our Intranet. Your right to participate in the <u>SKY ZONESky Zone</u> web site or any intranet system we may develop or otherwise use the Marks or System on the <u>internetInternet</u> or other on-line communicators terminates when this Agreement expires or terminates.

(f) Media. We reserve the right to require you to present music, movies, television and other media at the Business. You may be required to pay your pro-rata share of music, movie, and other media licensing fees to show or otherwise present media at your Business. If we have not identified any approved vendors for media content and the display of such media content is not otherwise prohibited by our Methods of Operation, you are solely responsible for ensuring that you have paid all necessary licensing fees to show or otherwise present such media content at your Business.

9.13 Reciprocal Membership/Coupons

- (a) You must offer and sell rights of access to your Business, referred to individually as a "Membership" or collectively as "Memberships." All Memberships must be evidenced by a written agreement (a "Membership Agreement") and may not be for a term that extends beyond the expiration of this Agreement. When selling Memberships, you will use the form of Membership Agreement that we will provide to you, and you must make modifications to the form to ensure it complies with all Applicable Laws for your Park, subject to our final written approval. We may modify the types and terms of Memberships to be offered, terminate your right to offer certain types of Memberships, and/or approve or require other types of Memberships for sale at any time, including before your Business opens for use by customers.
- Methods of Operation, which may regulate, among others, the following topics: (1) the types and terms of Memberships you may offer; (2) the form(s) of Membership Agreement; (3) the terms and conditions upon which a customer may transfer their Membership from your Business to another Business and vice versa; (4) admission of clients of your Business to other Businesses, or vice versa, including payment allocations due to Businesses which provide services (if any); (5) procedures to follow when customers transfer to or from your Business; (6) use and acceptance of coupons, passes, and certificates related to Memberships; (7) group accounts and group Memberships (and discounts applicable thereto); and (8) payment terms for Memberships.
- you, honor those Memberships on the terms and conditions of the existing Membership Agreements, and to accept as remuneration only such payments as accrue pursuant to the applicable Membership Agreement from the time of assignment.
- (d) ___You must participate fully in any reciprocal access program or company discount coupons that we may design or implement. You agree and acknowledge that when a person redeems any discounted coupons at your Business or uses your Business pursuant to a reciprocal access program, you are may not be entitled to reimbursement for membership Membership fees, or the cost of goods or services provided to the member as a reciprocal access benefit.

9.14 Non-Compliance

. While the day-to-day operation of your Business and the safety and well-being of-your customers is your responsibility, we reserve the right to turn off your access to the point-of-sale system and other components of the Technology System, or limit the number of jumpers who can simultaneously be entered through the point-of-sale system, if we believe the safety or well-being of any-of-your customers are in danger, or if there are other significant quality breaches in the operation of your Business. Any such business interruption shall be without liability to us, and we will not be responsible for any lost profits or other damages you sustain.

9.15 Innovations

. All ideas, concepts, procedures, techniques, <u>materials</u>, or processes relating to a Business or the System ("**Innovations**"), whether or not protectable intellectual property and whether created by or for you or your Owners or employees, must be promptly disclosed to us and will be deemed to be our sole and exclusive property, part of the System, and works made-for-hire for us. To the extent any item does not qualify as a "work made-for-hire" for us, you <u>willmust</u> assign ownership of that item, and all related rights to that item, to us and agree to sign (and to cause your Owners, employees, and contractors to sign) whatever assignment or other documents we request to evidence our ownership or to help us obtain intellectual property rights in the Innovation. We and our Affiliates have no obligation to make any payments to you or any other person with respect to any Innovations. You may not use any Innovation in operating the Business or otherwise without our prior approval.

9.16 General Conduct

. You will not, and will not allow your Owners or employees to, engage in conduct that, in our sole determination, may result in or tends to: (a) degrade, offend, shock, or insult the community, (b) ridicule public morals or decency, or (c) prejudice or harm us, our Affiliates, the Marks, or the System generally. You will (and will ensure that your Owners and employees) conduct yourself with due regard to public conventions and morals.

9.17 Safety and Security Procedures

You are solely responsible to take appropriate security and safety measures to protect itsyour employees, guestscustomers, those engaging in business with you, those coming on the premises of the Business, and the public at large. We do not in any way share any of that responsibility. You are responsible for obtaining its own legal advice with respect to the preparation and enforceability of any guest waiver formsensuring that your customer waiver forms comply with all Applicable Laws and you may modify the customer waiver forms to the extent necessary to comply with such Applicable Laws, provided that you provide us with immediate written notice of all such modifications.

10. MARKETING.

10.1 National Advertising

.

- (a) <u>National Advertising Fund</u>. Recognizing the value of advertising and marketing to the goodwill and public image of Businesses and the <u>SKY ZONESky Zone®</u> brand, we have established and administer a National Advertising Fund ("NAF") for the creation and development of marketing, advertising and related programs and materials, including electronic, print and Internet media as well as the planning and purchasing of <u>national and/or regional</u> network advertising. You agree to contribute the Ad Fee to the NAF <u>as further set forth in Section 5.3 (Ad Fee)</u>. We will direct all programs financed by the NAF and <u>we retain the right to determine the creative concepts materials and endorsements used therein and the geographic market and media placement and allocation thereof.</u>
- Use of Funds. You agree that the NAF may be used to pay the costs of (b) researching, developing, administering and preparing national, regional, point -of -sale and-local direct sales advertising and marketing strategy materials for use by the SKY ZONESky Zone® brand and Businesses, including preparing and producing video, audio and written advertising materials; website design, development and updating; electronic advertising efforts, including search engine optimization and social media networks and related platforms; administering regional and multi-regional advertising programs, including purchasing direct mail and other media advertising; administrative and other costs associated with all NAF efforts; and employing advertising, promotion and marketing agencies to assist therewith and supporting public relations, market research and other advertising promotion and marketing activities and amounts expended pursuant to this Section. All payments, plus income earned therefrom, shall be used exclusively for the above-stated purposes, and shall not be used to defray any of our general operating expenses, except for reasonable salaries, administrative costs, travel expenses, overhead, and similar expenses we may incur in activities related to the administration of the NAF and all costs of development and preparing national, regional, point of sale, and local advertising materials for use within the System.
- (c) <u>Materials</u>. We will furnish you with samples of advertising, marketing formats, promotional formats and other materials produced by the NAF at no additional cost to you when we deem appropriate. Multiple copies of such materials will be furnished to you at your direct cost of producing them plus any related shipping handling and storage charges.
- (d) Advisory Council. We currently have set up an advisory council that works with us to improve the System, including the products and services offered, advertising campaign, and other matters of interest to us and our franchisees to improve the System ("FAC"). FAC will not have any decision making authority and will be constituted, modified and dissolved by us, in our sole discretion. We have the right to form, change, merge or dissolve any advisory council at any time.
- delight 10.2 Accounting. The NAF will be accounted for separately from our other funds. The NAF will have the right to negotiate and retain any commissions or marketing payments received from suppliers of any marketing or other materials or products. We may spend on behalf of the NAF, in any fiscal year, an amount that is greater or less than the aggregate contribution of all Businesses to the NAF in that year and the NAF may borrow from us or others to cover deficits or invest any surplus for future use. All interest earned on monies contributed to the NAF will be used to pay advertising costs before other assets of the NAF are Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

expended. We <u>are not required to audit the NAF. We</u> will prepare an annual <u>unaudited</u> statement of monies collected and costs incurred by the NAF and furnish the statement to you upon <u>reasonable</u> written request. We have the right to cause the NAF to be incorporated or operated through a separate Entity at such time as we deem appropriate and such successor Entity will have all of the rights and duties specified herein. <u>We do not act as trustee or in any other fiduciary capacity with respect to the NAF.</u>

- (e) 10.3—Proportionality. You acknowledge that the NAF is intended to maximize recognition of the Marks and patronage of Businesses. Although we will endeavor to utilize the NAF to develop advertising and marketing materials and programs and to place advertising that will benefit all Businesses, we are not obligated to ensure that expenditures by the NAF in or effecting any geographic area are proportionate or equivalent to the contributions to the NAF by Businesses operating in that geographic area. Nor, nor are we under any obligation to ensure that any Business will benefit directly or in proportion to its NAF contributions paid to the NAF from the development of advertising and marketing materials or the placement of advertising. Except as expressly provided in this Section, we assume no direct or indirect liability or obligation to you with respect to collecting amounts due to, or maintaining, directing, or administering the NAF. We do not act as trustee or in any other fiduciary capacity with respect to the NAF.
- (f) 10.4 Deferrals Or Reductions. We may forgive, waive, settle, and compromise all claims by or against the NAF. We reserve the right to defer or reduce contributions of any Business franchisee and, upon thirty (30) days' prior written notice to you, to reduce or suspend your payment of contributions to the NAF and suspend operations of the NAF for one or more periods of any length and to terminate (and if terminated to reinstate) the NAF. If the NAF is terminated, all unspent monies on the date of termination will be distributed to our franchisees in proportion to their respective contributions to the NAF during the preceding three (3) month period, and amounts required to be paid by you pursuant to Section Sections 5.3 (Ad Fee) and 10.1 (National Advertising) shall be added to amounts required to be expended by you pursuant to Section 10.5 10.2 (Local Advertising).

10.2 10.5 Local Advertising

.

(a) Obligation to Advertise. You must participate in such advertising, promotional, and community outreach programs that we may specify from time to time, at your own expense. You must use your best efforts to promote the use of the Marks in your Protected Territory. You must ensure that all your advertising, marketing, promotional, customer relationship management, public relations and other brand related programs and materials that you or your agents or representatives develop or implement relating to the Business isare completely clear, factual, and not misleading, complies comply with all Applicable Laws, and conforms conform to the highest ethical standards and the advertising and marketing policies that we periodically specify. Any media advertising or direct mail marketing that you conduct must be predominantly focused within your Protected Territory unless we agree otherwise. There are no territorial restrictions from accepting business from retail customers that reside or work or are

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement

Franchise Agreement [City, State]

otherwise based outside of your <u>Protected</u> Territory if these customers contact you, but we reserve the right to implement <u>rules and restrictions</u> <u>Methods of Operation</u> regarding soliciting such customers in the future in <u>our Manualsthe Operations Manual</u> or otherwise in writing.

- Approvals. You must submit to us at least 30 days prior to their intended (b) use, for our prior approval, a marketing plan and samples of all advertising and promotional materials not prepared or previously approved by us in the prior 12 months. You must obtain our advance written approval prior to using or producing any advertising or marketing materials using any of the Marks, in whole or in part. If you elect to work with an advertising agency, you must obtain our written approval of such advertising agency before you sign any contracts or share any Confidential Information with the such advertising agency. You agree to conduct all advertising in a dignified manner and to conform to conformity with the standards and requirements we specify in the Operations Manuals Manual. We will have the final decision on all creative development of advertising and promotional messages and materials. If you wish to use any advertising and promotional materials which were not prepared or previously approved by us during the preceding twelve (12) months, you must submit samples of your proposed materials to us for our approval at least thirty (30) days prior to their intended use. If we do not provide our written approval is not received within fifteen (15) days from the date we received the material, the material is receive your samples, our failure to respond will be deemed approveda disapproval of the request. We will own the copyright to anything so submitted, whether approved by us or not. We reserve the right to require you to discontinue the use of any advertising or marketing materials at any time. We reserve the right to require you to include certain language in your local advertising materials about Sky Zone franchise opportunities.
- (c) <u>Local Advertising Funding Requirement</u>. In addition to the contributions, you pay to the NAFAd Fees and the <u>Pre-Grand Opening Marketing Expense</u>, in the second, third and fourth months in which the Business is open, you must spend at least Twelve Thousand Dollars (\$12,000) per month (depending on the size of the Business and its location) on local advertising on local advertising and promotional activities in your <u>Protected Territory (during the "Local Advertising Funding Requirement") first month following the opening of your Business.</u> Thereafter, you must spend a minimum of four percent (4%) of the Business' Gross Sales per month on local advertising on a quarterly basis (the "Local Advertising Funding Requirement").
- (d) <u>Valid Expenditures</u>. The following <u>expensesexpenditures</u> may count towards <u>theyour</u> Local Advertising Funding Requirement: (i) amounts contributed to advertising <u>eooperativesCooperatives (as defined below)</u>; (ii) amounts spent by you for advertising media, such as television, radio, Internet, newspaper, billboards, posters, direct mail, collateral and promotional items, advertising on public vehicles (transit and aerial), and (iii) if not provided by us, <u>the</u> cost of producing approved materials necessary to participate in these media. The following expenses do not count towards <u>theyour</u> Local Advertising <u>Funding</u> Requirement: (a) permanent on-premises signs, <u>whether featured on the interior or exterior of the Business</u>, (b) <u>lighting</u>, personnel salaries, or administrative costs, (c) transportation vehicles (even though such vehicles may display the Marks), (d) <u>Yellow Pages advertising</u>, (ed) discounts, (fe) free offers, and (gf) any employee incentive programs. We you develop and offer. Notwithstanding the

<u>foregoing</u>, <u>we</u> have the right to modify or designate in the Operations Manual the types of expenditures that will or will not count toward the Local Advertising Funding Requirement.

(e) Payment to Us. At our request, you must submit appropriate documentation to verify your compliance with the Local Advertising Funding Requirement. If you fail to spend (or prove that you spent) the Local Advertising Funding Requirement in any month, then we may, in addition to and without limiting our other rights and remedies, require you to pay us the shortfall as an additional Ad Fee or to pay us the shortfall for us to spend on local marketing for your Business. In addition, we may require you to pay the Local Advertising Funding Requirement to us for us to use to administer your local advertising (or to reimburse you, up to an amount not to exceed the monies so collected, for the costs incurred by you in implementing local marketing plans developed by you and approved by us) if, in our business judgment, we determine that: (i) you are your Business is under-performing, (ii) our participation is appropriate or necessary, or (iii) you have failed to comply with the Local Advertising Funding Requirement. If we require you to pay the Local Advertising Funding Requirement to us, you must pay it to us in the same manner and at the same time as the Royalty Fees.

10.3 10.6 Advertising Cooperatives

. We have the right to establish or approve local and/or regional advertising cooperatives for Businesses—in your local or regional areas, covering such geographical areas as we may designate from time to time (each a "Cooperative"). You must participate in any such eooperative Cooperative and its programs and abide by its by-laws. If your Business is within the territorygeographic area of an existing Cooperative at the time your Business opens for business, you agree to immediately become a member of the Cooperative. If a Cooperative applicable to your Business is established during the term of this Agreement, you agree to become a member no later than thirty (30) days after the date approved by us for the Cooperative to commence operation. The following provisions shall apply to each Cooperative:

- (a) Each Cooperative shall utilize a voting system of one (1) vote per one (1) eligible Business.
- (b) Each Cooperative shall be organized and governed in a form and manner, and shall commence operations on a date, approved in advance by us in writing. No changes in the by-laws or other governing documents of a Cooperative shall be made without our prior written consent. We shall have the right to unilaterally change the by-laws or other governing documents of a Cooperative.
- (c) Each Cooperative shall be organized for the exclusive purpose of administering advertising programs and—developing, subject to our approval, developing promotional materials for use by the members in the Cooperative.
- (d) No advertising or promotional plans or materials may be used by a Cooperative or furnished to its members without prior approval by us in accordance with the procedure and standards set forth in Section 10.510.2(b) (Approvals).

You and each other member of the Cooperative shall contribute to the Cooperative, using a collection structure selected and established by us, the amount determined in accordance with the Cooperative's by-laws. Any Businesses owned by us or any of our Affiliates located in such designated local or regional geographic area(s) will contribute to the Cooperative on the same basis as Businesses which are owned and operated by franchisees. At our request, you shall furnish us with copies of such information and documentation evidencing your Cooperative contributions as we may require to evidence your compliance with this Section 10.6. Contributions to such local and/or regional advertising cooperatives are Cooperatives will be credited towards theyour Local Advertising Funding Requirement; however, if we provide you and your Cooperative ninety (90) days' notice of a special promotion, including any regional promotions, you must participate in such promotion and pay to us any special promotion advertising fees assessed in connection therewith, beginning on the effective date of such notice and continuing until such special promotion is concluded. Any such special promotion advertising fees shall be in addition to, and not credited towards, the other advertising expenditures and commitments required of you bypursuant to this Section 10, including your Local Advertising Funding Requirement.

10.4 10.7 Community Marketing Programs

. You must participate in, at your expense, <u>ourany</u> guerrilla marketing programs and campaigns that we develop and administer from time to time <u>and specify</u> in our Operations Manual or other communications with you. Community marketing may <u>additionally</u> include participation in community events, street fairs, seasonal events and festivals, farmers markets and sponsorships of adult and children's sports teams. You must also sponsor local fundraising events by making your <u>parkBusiness</u> available as a venue <u>and revenue sharing programsfor such</u> events.

10.5 10.8-Digital Marketing

.

Restrictions on Use. We or our Affiliates may, in our sole discretion, (a) establish and operate websites, social media accounts (such as Facebook®, X/Twitter®, Bluesky®, LinkedIn[®], Yelp[®], Instagram[®], Pinterest[®], <u>YouTube[®], TikTok[®]</u>, etc.), applications, keyword or ad word purchasing programs, accounts with websites featuring gift certificates or discounted coupons (such as Groupon®, Living Social®, etc.), mobile applications, or other means of digital advertising on the Internet or any electronic communications network (collectively, "Digital Marketing") that are intended to promote the Marks, your Business, and the entire network of Businesses System. We will have the sole right to control all aspects of any Digital Marketing, including those related to your Business. Unless we consent otherwise in writing, you and your employees may not, directly, or indirectly, conduct or be involved in any Digital Marketing that use the Marks or that relate to the Business or the network System. You may not separately register any domain name, create any username, or operate any web site containing any of the Marks without our written approval. If we do permit you or your employees to conduct any Digital Marketing, you or your employees must comply with any policies, standards, guidelines, or content requirements, or vendor requirements that we establish periodically and must

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

immediately modify or delete any Digital Marketing that we determine, in our sole discretion, is not compliant with such policies, standards, guidelines, or requirements. We may withdraw our approval for any Digital Marketing at any time.

(b) <u>Our Website</u>. As part of our Digital Marketing, we or one of our designees will operate and maintain a <u>SKY ZONESky Zone</u> website, which will include basic information related to the Business. You must promptly provide us with any information that we request regarding your Business for inclusion on the website. We retain all rights relating to the <u>SKY ZONESky Zone</u> web site and may alter or terminate the <u>web sitewebsite</u>.

11. RECORDS, REPORTS AND FINANCIAL STATEMENTS.

11.1 Records

. You agree to establish and maintain at your own expense a bookkeeping, accounting and record-keeping system conforming to the requirements and formats we prescribe from time to time. You agree to prepare and to maintain for three (3) years complete and accurate books, records (including invoices and records relating to your advertising expenditures) and accounts (using our then-current standard chart of accounts) for your Business, copies of your sales tax returns and such portions of your state and federal income tax returns as relate to your Business. All such books and records shall be kept at your principal address indicated on the first page of this Agreement Appendix A, unless we otherwise approve. You must record all sales using the Technology System, which we have the independent right to access at any time. All data pertaining to your Business, and all data you create or collect in connection with the System, or in connection with your operation of the Business (including data pertaining to or otherwise concerning your membersCustomer Information) or otherwise provided by you (including data uploaded to, or downloaded from your computer systemTechnology System) is and will be owned exclusively by us, and we will have the right to use such data in any manner that we deem appropriate without compensation to you. Such data will be part of the Confidential Information. We hereby license use of such data back to you for the term of this Agreement, at no additional cost, solely for your use in connection with the Business conducted under this Agreement.

11.2 Periodic Reports

- . You must furnish us, in the manner and format(s) that we prescribe from time to time:
- (a) within <u>thirty (30)</u> days after the end of each fiscal quarter, a quarterly balance sheet and income statement and statement of cash flow of your Business for such quarter, reflecting any adjustments and accruals;
- (b) within <u>ninety (90)</u> days after the end of each fiscal year, a year-end balance sheet and income statement and statement of cash flow of your Business for such year, reflecting all year-end adjustments and accruals; and
- (c) within <u>thirty (30)</u> days of our request, such other information as we may require from time to time, including sales data and labor cost reports and sales and income tax statements. All such reports shall use our then-current standard chart of accounts.

April 2025 FDD

Franchise Agreement

11.3 Verification

. You agree to verify and sign each report and financial statement in the manner we prescribe. We reserve the right to require that your annual financial statements be audited, at your expense, by an independent certified public accountant approved by us. We reserve the right to publish or disclose information that we obtain under this Section in any data compilations, collections, or aggregations that we deem appropriate so long as we do not disclose information relating to performance of your individual Business, unless such disclosure is required by Applicable LawLaws or order of a court.

12. INSPECTIONS AND AUDITS.

12.1 Our Right To Inspect The Business

. To determine whether you and the Business are complying with this Agreement and Methods of Operation, we and our designated agents have the right at any time during your regular business hours, and without prior notice to you, to:

- (a) inspect the Business;
- (b) observe, photograph and videotape the operations of the Business for such consecutive or intermittent periods as we deem necessary;
- (c) remove samples of any products, materials or supplies for testing and analysis;
 - (d) interview personnel and customers of the Business;
- (e) inspect and copy any books, records (whether electronic or hard copy) and documents relating to your operation of the Business, including member and membership informationCustomer Information;
- (f) retrieve such data and information from your security camera system or other components of the Technology System, including obtaining such information from third parties or vendors.

12.2 Cooperation

. You agree to cooperate with us fully in connection with any such inspections, observations, photographingphotography, videotaping, product removal and interviews. You agree to present to your customers such evaluation forms that we periodically prescribe and to participate and/or request-your customers to participate in any surveys performed by us or on our behalf.

12.3 Our Right to Audit

DMS_US.363658954.1

. We have the right at any time during your business hours, and without prior notice to you, to inspect and audit, or cause to be inspected and audited, your (if you are an Entity) and the Business's business, bookkeeping and accounting records, sales and income tax records and returns and other records. You agree to cooperate fully with our representatives and independent accountants we hire to conduct any such inspection or audit. If any inspection or audit discloses an understatement of your Business' Gross Sales, you agree to pay us the Royalty Fees, Ad Fees, and any other fees understated, plus interest on the understated amounts from the date originally due until the date of payment, within fifteen (15) days after receiving the inspection or audit report. In the event such inspection or audit is made necessary by your failure to furnish reports, supporting records or other information as herein required, or to furnish such items on a timely basis, you agree to reimburse us for the reasonable cost of such inspection or audit, including the charges of attorneys and independent accountants and the travel expenses, room and board and compensation of our employees. In the event anor if our inspection or audit reveals that any payments have been understated in any report to us, then you shall immediately pay to us the amount understated upon demand, in addition to interest from the date such amount was due until paid, at the highest contract rate of interest permitted by Applicable Law. If an inspection or audit discloses an understatement in any report of two percent (2%) or more, you shall, in addition to repayment the payment of monies owed with interest, reimburse us for all costs and expenses connected with the inspection or audit, including the charges of professional fees for attorneys and independent accountants and the travel expenses, room and board and compensationTravel and Living Expenses of our employees. The foregoing remedies are in addition to our other remedies and rights under this Agreement and Applicable Law.

13. TRANSFER.

13.1 By Us

. We have the right to sell or assign, in whole or in part, our interests in this Agreement without your approval, and any such sale or assignment will inure to the benefit of any assignee or other legal successor to our interests herein.

13.2 By You

. You understand and acknowledge that the rights and duties created by this Agreement are personal to you (or, if you are an Entity, to your Owners) and that we have granted the Franchise to you in reliance upon our perceptions of your (or your Owners') individual or collective character, skill, aptitude, attitude, business ability, acumen, and financial capacity.

Accordingly, you and your Owners may not conduct a Transfer without our prior written approval. Any Transfer without such approval constitutes a breach of this Agreement and is void and of no effect. For any proposed Transfers, you must promptly give us written notice and submit a copy of all proposed contracts and other information concerning the Transfer and transferee that we may request. We have sole and absolute discretion to withhold our consent, except as otherwise provided in this Section 13. We have the right to communicate with both you, your counsel, and the proposed transferee on any aspect of a proposed Transfer. You agree to provide any information and documentation relating to the proposed Transfer that we

reasonably require. No Transfer that requires our consent may be completed until at least <u>sixty</u> (60) days after we receive written notice of the proposed Transfer. Our consent to a Transfer <u>is not a representation of the fairness of the terms of any contracts between you and the transferee, a guarantee of the Business' or the transferee's prospects of success. Further, our consent to a <u>Transfer</u> does not constitute a waiver of any claims that we have against the transferor, nor is it a waiver of our right to demand exact compliance with the terms of this Agreement. If your Business is not open and operating, we will not consent to your Transfer of this Agreement, and we are under no obligation to do so.</u>

13.3 <u>Conditions For Approval Of Transfer</u>

. If you (and your Owners) are in full compliance with this Agreement and the conditions of this Section 13.3 are met, we will not unreasonably withhold our consent to Transfer, but we may give our consent subject to reasonable conditions. A Transfer of ownership, possession or control of the Business may be made only in conjunction with a Transfer of this Agreement. We may further described below, we may, if we deem it necessary in our sole discretion, require the transferee in any Transfer to renovate, modify, or re-design the Business to the then-current standards and specifications of a Sky Zone Parkpark.

- (a) <u>Control Transfers</u>. If the Transfer is of this Agreement, the <u>Park</u>, or a controlling interest in you, or is one of a series of Transfers which in the aggregate constitute the Transfer of this Agreement, the <u>Park</u>, or a controlling interest in you, all of the following conditions must be met prior to or concurrently with the effective date of the Transfer:
- (i) the transferee and its direct and indirect owners have the moral character, aptitude, attitude, experience, references, acumen and financial capacity to operate the Business and meet our then-current standards for franchisees, and the proposed transferee may not be an Entity, or be affiliated with an Entity, that is required to comply with reporting and information requirements of the Securities Exchange Act of 1934, as amended. Neither the proposed transferee nor its owners (if an Entity) or Affiliates may have a direct or indirect ownership interest in, or perform services for, a Competitive Business. At the time you sign a conditional consent to transfer, you must pay us, by wire transfer, a deposit of Fifteen Thousand Dollars (\$15,000) ("Fee Deposit"). We will refund you the Fee Deposit, less any amounts which may be due under this Agreement or any other agreement between you (or your Affiliates) and us (or our Affiliates), within thirty (30) days following the effective date of the Transfer or the date on which you and the transferee have complied with all terms set forth in any applicable consent to transfer that we and you sign in connection with the Transfer, whichever is later;
- (ii) you have paid all Royalties; Ad Fees; Technology Fees,; fund contributions, actuarial-based adjustment payments, Captive SIR Program amounts, and other required payments due pursuant to the Master Insurance Program and Captive SIR Program for all incidents that arise or are related to injuries or other claims, demands, or judgments through the effective date of the Transfer; amounts owed for purchases from us; and all other amounts owed to us or to third party creditors and have submitted all required reports and statements;

DMS_US.363658954.1

- (iii) the transferee (or its Responsible Person) and its managers, shift supervisors and personnel must have completed our initial training program or must be currently certified by us to operate and/or manage a Business to our satisfaction prior to closing;
- (iv) the transferee has agreed to be bound by all of the terms and conditions of this Agreement for the remainder of its term or, at our option, must execute our then-current standard form of franchise agreement and related documents, including our then-current standard form of personal guarantee, used in the state in which your Business is located (which may provide for different royalties, advertising contributions and expenditures, duration and other rights and obligations than those provided in this Agreement);
- (v) the transferee (and, if the transferee is an individual, such owners of a legal or beneficial interest in the transferee as we may request) must (i) enter into a written assignment, in a form satisfactory to us, assuming and agreeing to discharge and guarantee all of your obligations under this Agreement and (ii) must execute our then current form of personal guarantee you have provided us with executed versions of any documents executed by you (or your transferring Owners) and the transferee to effectuate the Transfer;
- (vi) you pay us a transfer fee equal to fifty percent (50%) of the <u>our</u> then-current initial franchise fee <u>for new franchisees</u>. If the proposed Transfer is among your <u>existing</u> Owners, this Section 13.3(a)(vi) will not apply, although you are required to reimburse <u>usinstead pay us Two Thousand Five Hundred Dollars (\$2,500)</u> for <u>any reasonable the</u> legal (including in-house counsel) and administrative costs we incur in connection with the Transfer;
- (vii) you (and your transferring Owners) have executed a general release, in form satisfactory to us, of any and all claims against us, our Affiliates, and our and their shareholders, officers, directors, employees and agents;
- (viii) we have approved the material terms and conditions of such Transfer and determined that the price and terms of payment will not adversely affect the transferee's operation of the Business;
- (ix) if you or your Owners finance any part of the sale price of the transferred interest, you and/or your Owners have agreed that all of the transferee's obligations pursuant to any promissory notes, agreements or security interests that you or your Owners have reserved in the Business are subordinate to the transferee's obligation to pay Royalty Fees, Ad Fees, and other amounts due to us or our Affiliates and otherwise to comply with this Agreement the transferee's franchise agreement with us;
- (x) you must provide us with written notice from your landlord indicating that your landlord has agreed to transfer the lease for the Location to your transferee; if the proposed Transfer (including any assignment of the lease or subleasing of the Location)

requires notice to or approval from your landlord, or any other action under the terms of the lease, you have taken such appropriate action and delivered us evidence of the same;

(xi) your proposed transferee must covenant that it will continue to operate you have corrected any existing deficiencies of the Business under the Marks of which we have notified you, and using/or the System and must arrange transferee agrees to modernize, renovate, or upgrade the Business, at its expense, to conform to in accordance with our then-current standards requirements and specifications for new Businesses; and within the time period we specify following the effective date of the Transfer (we will advise the transferee before the effective date of the Transfer of the specific actions that it must take and the time period within which such actions must be taken);

(xii

(xii) you provide us the evidence we reasonably request to show that appropriate measures have been taken to effectuate the Transfer as it relates to the operation of your Business, including, by transferring all necessary and appropriate business licenses and material agreements, or obtaining new business licenses and material agreements; and

(xiii) you and/or any transferring Owner(s) have executed an agreement in favor of us agreeing to remain bound by the restrictions contained in Sections 16.2 (Discontinue Use of the System and the Marks), 16.3 (Return of Confidential Information), and 17.2 (Post-Term Covenant Not to Compete) as if this Agreement had terminated. You agree that the restrictions referenced in the immediately preceding sentence will continue to apply regardless of whether you and/or any transferring Owner(s) execute an agreement confirming the survival of these restrictions. You and each of your Owners further agree that the provisions of Section 19.12 (Governing Law) and 19.13 (Consent to Jurisdiction) survive the partial or full Transfer of an Owner's interest in you.

(b) <u>Other Transfers</u>. For all other Transfers other than a Transfer of the Agreement, the Park, or a controlling interest in you, we may require you to comply with all or a selection of conditions in Section 13.3(a), which we will identify on a case-by-case basis, provided that Section 13.3(e) will not apply, although you are required to reimburse usthe transfer fee will be Two Thousand Five Hundred Dollars (\$2,500) for any reasonablethe legal (including in-house counsel) and administrative costs we incur in connection with the Transfer.

13.4 Transfer To A Wholly Owned Entity

. Notwithstanding Section 13.3 (Conditions for Approval of Transfer), if you are in full compliance with this Agreement, you may Transfer this Agreement to an Entity which conducts no business other than the Business and, if applicable, other Businesses, in which you maintain management control and of which you own and control 100% of the equity and voting power of all issued and outstanding capital stock or membership interests, and further provided that all assets of the Business are owned, and the entire business of the Business is conducted, by a single Entity. Transfers of shares in such Entity will be subject to the provisions of Section 13.3. Notwithstanding anything to the contrary herein, you agree to remain personally liable under this Agreement as if the Transfer to such Entity had not occurred.

April 2025 FDD
Franchise Agreement
[City, State]

13.5 Transfer Upon Your Death Or Disability

. Upon your death or permanent disability or, if you are an Entity, the death or permanent disability of the Owner of a controlling interest in you, your or such Owner's executor, administrator, conservator, guardian, or other personal representative must Transfer your interest in this Agreement or such Owner's interest in you to a third party. Such disposition of this Agreement or the interest in you (including Transfer by bequest or inheritance) must be completed within a reasonable time, not to exceed six (6) months from the date of death or permanent disability and will be subject to all the terms and conditions applicable to Transfers contained in this Section. A failure to Transfer your interest in this Agreement or the ownership interest in you within this period constitutes a breach of this Agreement. For purposes hereof, the term "permanent disability" means a mental or physical disability, impairment or condition that is reasonably expected to prevent or does prevent you or an Owner of a controlling interest in you from managing and operating the Business for a period of three months from the onset of such disability, impairment, or condition.

13.6 Operation Upon Your Death Or Disability

. If, upon your death or permanent disability or the death or permanent disability of the Owner of a controlling interest in you, the Business is not being managed by a trained manager, your or such Owner's executor, administrator, conservator, guardian, or other personal representative must within a reasonable time, not to exceed fifteen (15) days from the date of death or permanent disability, appoint a manager to operate the Business. Such manager will be required to successfully complete training at your expense within sixty (60) days of being appointed to operate the Business. Pending the appointment of a manager as provided above or if, in our judgment, the Business is not being managed properly any time after your death or permanent disability or after the death or permanent disability of the Owner of a controlling interest in you, we have the right, but not the obligation, to appoint a manager for the Business. All funds from the operation of the Business during the management by our appointed manager will be kept in a separate account, and all expenses of the Business, including compensation, other costs and travelthe Travel and living expenses Living Expenses of our manager, will be charged to this account. We also have the right to charge a reasonable management fee (in addition to the Royalty Fees-and, Ad Fees, and other amounts payable under this Agreement) during the period that our appointed manager manages the Business. Operation of the Business during any such period will be on your behalf, provided that we only have a duty to utilize reasonable efforts in doing so and will not be liable to you or your Owners for any debts, losses or obligations incurred by the Business or to any of your creditors for any products, materials, supplies or services the Business purchases during any period it is managed by our appointed manager.

13.7 Bona Fide Offers

. If you (or any of your Owners) at any time determine to sell, assign or Transfer for consideration an interest in this Agreement and the Business or an ownership interest in you, you (or such Owner) agree to obtain a bona fide, executed written offer and earnest money deposit (in the amount of $\underline{\text{five percent }(5\%)}$ or more of the offering price) and a complete franchise

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

application from a fully disclosed offeror including lists of the owners of record and beneficially of any corporate or limited liability company offeror and all general and limited partners of any partnership and immediately submit to us a true and complete copy of such offer, which includes details of the payment terms of the proposed sale. To be a valid, bona fide offer, the proposed purchase price must be denominated in a dollar amount. The offer must apply only to an interest in you or in this Agreement and the Business and may not include an offer to purchase any of your (or your Owners') other property or rights. However, if the offeror proposes to buy any other property or rights from you (or your Owners) under a separate, contemporaneous offer, such separate, contemporaneous offer must be disclosed to us, and the price and terms of purchase offered to you (or your Owners) for the interest in you or in this Agreement and the Business must reflect the bona fide price offered therefor and not reflect any value for any other property or rights. Any Transfer in violation of our right of first refusal is null and void.

13.8 Our Right Of First Refusal

• We have the right, exercisable by written notice delivered to you or your selling Owners within thirty (30) days from the date of the delivery to us of both an exact copy of sucha bona fide offer (as described in Section 13.7) and all other information we request, to purchase such interest for the price and on the terms and conditions contained in such bona fide offer, provided that:

- (a) we may substitute cash for any form of payment proposed in such offer;
- (b) our credit will be deemed equal to the credit of any proposed purchaser;
- (c) we will have not <u>lessfewer</u> than <u>sixty (60)</u> days after giving notice of our election to purchase to prepare for closing; and
- (d) we are entitled to receive, and you and your Owners agree to make, all customary representations and warranties given by the seller of the assets of a business or the capital stock of an incorporated business, as applicable, including representations and warranties as to: (i) ownership and condition of and title to stock or other forms of ownership interest and/or assets; (ii) liens and encumbrances relating to the stock or other ownership interest and/or assets; and (iii) validity of contracts and the liabilities, contingent or otherwise, of the corporation whose stock is being purchased.

13.9 Non-Exercise

. If we do not exercise our right of first refusal, you or your Owners may complete the sale to such purchaser pursuant to and on the exact terms of such bona fide offer, subject to our approval of the Transfer as provided in Sections 13.2 (By You)., 13.3 (Conditions for Approval of Transfer), and 13.4 (Transfer to a Wholly Owned Entity). If the sale to such purchaser is not completed within one hundred and twenty (120) days after delivery of such bona fide offer to us, or if there is a material change in the terms of the sale (which you agree promptly to communicate to us), the sale will be treated as a new sale subject to our right of first refusal as provided in Section 13.8 (Our Right of First Refusal).

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

DMS_US.363658954.1

13.10 Securities Offerings

. Neither you nor any of your Owners may issue or sell, or offer to issue or sell, any of your securities or any securities of any of your Affiliates, regardless of whether such sale or offer would be required to be registered pursuant to the provisions of the Securities Act of 1933, as amended, or the securities laws of any other jurisdiction, without obtaining our prior consent and complying with all of our requirements and restrictions concerning use of information about us and our Affiliates. Neither you nor any of your Owners may issue or sell your securities or the securities of any of your Affiliates if: (1) such securities would be required to be registered pursuant to the Securities Act of 1933, as amended, or such securities would be owned by more than 35 persons; or (2) after such issuance or sale, you or such Affiliate would be required to comply with the reporting and information requirements of the Securities Exchange Act of 1934, as amended. Any proposed private placement of your or of your Affiliate's securities must be approved by us.

14. SUCCESSOR FRANCHISES FRANCHISE AND EXPIRATION.

14.1 Acquisition Of A Successor Franchise

- . Upon the expiration of the <u>Initial</u> Term, <u>if (a) you will have the option to acquire a successor franchise to continue to operate the Business for one (1) additional term of ten (10) years if you meet the following conditions:</u>
- Agreement during the Term; <u>you (and each of your Owners) have substantially complied with this</u>
- (b) you (and each of your Owners) are in full compliance with this Agreement during, both on the Initial Term, (b)date you give us written notice of your desireelection to acquire a successor franchise not less than six months nor more than 12 months prior to the expiration(as provided in Section 14.2 below) and on the date on which the term of this Agreement (the "Successor Request"), and (c) (i) successor franchise would commence, in full compliance with this Agreement and all Methods of Operation;
- <u>(c)</u> you maintain possession of <u>the Location</u> and agree to remodel and/or expand the Business, add or replace improvements, <u>Operating Assets</u>, equipment and signs and otherwise modify the Business as we require to bring it into compliance with <u>our</u> specifications and standards then applicable for Businesses or (ii) and the System;
- (i) if you are unable to maintain possession of the Location, or if in our judgment the Business should be relocated, you <u>agree to</u> secure substitute premises we approve, develop such premises in compliance with Section 4.11 (Relocation) and continue to operate the Business at the Location until operations are transferred to the substitute premises, then, subject to the;
- <u>you execute the then-current standard form of franchise agreement and related documents (modified as necessary to reflect that such agreement will be for a successor franchise), which may contain materially different terms and fees than this Agreement; Sky Zone Franchise Group, LLC</u>

April 2025 FDD

Franchise Agreement

- (e) you and your Owners agree to execute, in a form satisfactory to us, guarantees and general releases of all claims against us, our Affiliates, and our and their shareholders, officers, directors, employees, agents, successors and assigns; and
- (f) you pay a non-refundable successor franchise fee equal to twenty-five percent (25%) of our then-current initial franchise fee for new franchises, in the form of a lump sum payment by wire transfer or via EFT from your designated bank account; and
- Section $14_{\frac{1}{2}}$ <u>you comply with all additional</u> terms and conditions set forth in this

If you will have the right(and your Owners) fail to meet the conditions set forth in this Section, you acknowledge that we need not grant you a successor franchise, whether or not we had, or chose to exercise, the right to terminate this Agreement during its Term pursuant to Section 15.2 herein. Further, any failure by you or your Owners to sign the agreements and releases described in this Section 14 and deliver them to us for acceptance and execution within thirty (30) days after their delivery to you will be deemed an election not to acquire a successor franchise to continue to operate the Business.

14.2 Grant Of A Successor Franchise

- (a) to grant you one ten-year successor franchise;
- (b) to grant you one ten-year successor franchise on the condition that deficiencies of the Business, or in your operation of the Business, are corrected within a reasonable cure period that we specify; or
- (c) not to grant you a successor franchise based on our determination (which will be described in the notice) that you and your Owners have not substantially complied with this Agreement during the Initial Term

If you fail to provide us with your Successor Request within the prescribed time period, we need not grant you a successor franchise. If we fail to provide Our Notice to you within sixty (60) days after our receipt of your Successor Request, it is your responsibility to provide us with written notice of our failure to respond, which we must cure within fifteen (15) days after our receipt of such notice.

April 2025 FDD Franchise Agreement

Your right to acquire a successor franchise is subject to your continued compliance with all of the terms and conditions of this Agreement through the date of the expiration of the Initial Term, in addition to your compliance with any obligations described in Our Notice. If you fail to Our Notice states that you must remodel your Business and/or must cure anycertain deficiencies specified in Our Notice in the specified of your Business or its operation as a condition to our granting you a successor franchise, and you fail to complete the remodeling and/or to cure period those deficiencies, you will not have the right to a successor franchise, unless we, in our sole discretion, decide to extend the Initial Term to allow you additional time to complete the remodeling and/or to cure the deficiencies. If we fail to provide Our Notice to you within 60 days after our receipt of your Successor Request, it is your responsibility to provide us with written notice of our failure to respond, which we must cure within 15 days after our receipt of such notice.

14.3 Agreements/Releases

If we grant you a successor franchise, you and your Owners must execute the form of franchise agreement and any ancillary agreements we are then customarily using in connection with the grant of successor franchises for Businesses (which may include materially different terms and fees than this Agreement), provided, however, that the successor initial franchise fee will be equal to 25% of the initial franchise fee we are then currently customarily charging for new franchises. You and your Owners further agree to execute general releases, in a form satisfactory to us, of all claims against us, our Affiliates, and our and their shareholders, officers, directors, employees, agents, successors and assigns. Failure by you or your Owners to sign such agreements and releases and deliver them to us for acceptance and execution within 30 days after their delivery to you will be deemed an election not to acquire a successor franchise. Additionally, we may require you to, at your expense, upgrade the Business to then System standards.

15. TERMINATION OF AGREEMENT.

15.1 Termination By You

. If you and your Owners are in compliance with this Agreement and we materially fail to comply with this Agreement and do not correct such failure within <u>sixty (60)</u> days after written notice of such material failure is delivered to us, you may terminate this Agreement, effective <u>thirty (30)</u> days after delivery to us of written notice of termination. Your termination of this Agreement for any other reason or without such notice will be deemed null and void.

15.2 Termination By Us

. In addition to our right to terminate pursuant to other provisions of this Agreement and under Applicable <u>LawLaws</u>, we have the right to terminate this Agreement, effective upon delivery of notice of termination to you, if:

(a) you become insolvent, file for bankruptcy, or make an assignment for the benefit of your creditors, execution is levied against your business assets, or a suit to foreclose any lien or mortgage is instituted against you and not dismissed within thirty (30) days;

April 2025 FDD Franchise Agreement

- (b) you: (i) fail to lease, sublease or purchase the Location in accordance with the timeframe set forth in Section 4.2 (Purchase or Lease of the Location), (ii) fail to meet any construction deadline set forth in Section 4, or (iii) fail to open your Business in the time period set forth in Section 4.8 (Commencement Deadline);
- (c) you abandon or fail to actively operate your Business for three (3) consecutive business days, except where such failure to actively operate results solely from causes beyond your reasonable control and you provide written notice to us regarding such causes as soon as practicable;
- (d) you or any of your Owners surrender or transfer control of the operation of your Business or make any unauthorized Transfer without our written consent;
- (e) you or any of your Owners have made any material misrepresentation or omission in connection with your purchase of the Franchise;
- (f) you suffer cancellation or termination of the lease or sublease for your BusinessLocation;
- (g) you, any Owner, or any of your officers or directors are convicted of or plead nolo contendere to a felony, a crime involving moral turpitude or consumer fraud, or any other crime or offense that we believe is likely to have an adverse effect on our franchise system, the Marks and any associated goodwill, or the SKY ZONESky Zone concept (an "Adverse Effect") or you, any Owner, or any of your officers or directors has engaged in or engages in activities that, in our reasonable opinion, have an Adverse Effect;
- (h) you or any of your Owners make any unauthorized use or disclosure of any Confidential Information or use, duplicate or disclose any portion of the Operations Manual in violation of this Agreement;
- (i) you fail or refuse to comply with any mandatory specification, standard, or operating procedure prescribed by us relating to the <u>safety</u> cleanliness, or sanitation of your Business or violate any health, safety or sanitation law, ordinance or regulation, that we reasonably believe may pose harm to the public or to your or our reputation, and do not correct such failure, refusal or violation within <u>twenty-four (24)</u> hours after written notice thereof is delivered to you;
- (j) you fail, refuse, or neglect to pay any monies owing to us or our affiliates or fail to make sufficient funds available to us as provided in Section 5.5 (**Designated Account**) within ten (10) days after receiving written notice of your default or thirty (30) days after due date of the payment, whichever is the shorter period, or (ii) you have previously been given at least two (2) notices of nonpayment for any reason within the last twenty-four (24) months and you subsequently fail to timely pay when due any monies; or (iii) you fail to do all things necessary to give us access to the information contained in your Technology System within ten (10) days after receiving notice;

- you fail to make a timely payment of any amount due to a supplier unaffiliated with us (other than payments which are subject to bona fide dispute), and do not correct such failure within thirty (30) days after we deliver to you notice of such failure to comply;
- you underreport Gross Sales by more than two percent (2%) two times or (1)more in any two-year period or by five percent (5%) or more for any Accounting Period;
- you fail to comply with any other provision of this Agreement or any other agreement between you (or any of your Owners or Affiliates) and us or our Affiliates, and do not correct such failure within thirty (30) days after notice of such failure to comply is delivered to you;
- you fail on three (3) or more separate occasions within any period of twelve (12) consecutive months to submit when due reports or other data, information or supporting records or to pay when due Royalty Fees, Ad Fees, or other payments due us, any of our Affiliates or any unaffiliated suppliers or otherwise failcomply with this Agreement, or on two (2) or more separate occasions within any six (6) consecutive month period to comply with the same obligation under this Agreement, whether or not such failure is corrected afterwe notify you of the failures and, if we do notify you of the failures, whether or not you correct the failures following our delivery of notice is delivered to you;
- you fail to pay when due any federal or state income, service, sales, employment related or other taxes due on the operations of the Business, unless you are, in good faith, legally contesting your liability for such taxes;
- you fail to appoint a manager within fifteen (15) days after your death or permanent disability or the death or permanent disability of the Owner of a controlling interest in you or such manager fails to complete our training within sixty (60) days after being appointed;
- you or any of your Owners or employees violate Section 6.89.16 (General (q) Conduct):
- you or any Owner violates the noncompete covenants in Section 17 (Covenants Not to Compete); or
- you are in default three or more times within any 18 month period, whether or not the defaults are similar and whether or not they are cured.fail to perform background checks for employees you hire pursuant to Section 9.8 and as specified in the Operations Manual.

Additionally, we will have the right to terminate this Agreement if we terminate any other franchise agreement, multi-unit development agreement, or any other agreement between you (or any of your Owners) or your Affiliates and us or any of our Affiliates.

We have no obligation whatsoever to refund any portion of the Initial Franchise Fee or any other monies upon any termination of this Agreement. Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

15.3 Our Right To Operate The Business And Management Fee

. If we issue you a notice of default and you fail to cure such default within any applicable period, we have the right, in our sole discretion, to assume the operation of the Business for such length of time as we determine in our business judgment. You authorize us to operate the Business for so long as we deem necessary and practical, and without waiver of any other rights or remedies we may have under this Agreement. All monies from the operation of the Business during such period of operation by us shall be accounted for separately and the expenses of the business, including travel, food, lodging, the Travel and salariesLiving Expenses of our representatives who operate the Business, shall be charged to such account. We shall be entitled to retain fifty percent (50%) of the Gross Sales of your Business as our management fee after operating expenses are paid. You shall indemnify us and our representatives from all claims arising from the acts and omissions of us and our representatives pursuant to this Section 15.3.

15.4 Alternatives To Termination

. In addition to our rights under Section 15.3 (Our Right to Operate the Business and Management Fee), if we issue you a notice of default and you fail to cure such default within any applicable time period, we have the right, in our sole discretion, upon delivery of notice to you, and without waiving our right to terminate this Agreement as a result of such failure, to take any or all of the following actions without terminating this Agreement:

- (a) provide you with written notice that we have temporarily elected not to terminate this Agreement and allow you additional time to cure any default(s), if you pay to us the non-compliance fee set forth in Section 5.9 (Non-Compliance Fee) until such time you cure the default. Our collection of the non-compliance fee and temporary extension of the applicable cure period does not waive our right to require your full compliance with this Agreement;
- (b) temporarily or permanently limit, curtail, or remove certain services or benefits provided or required to be provided to you hereunder, including: (i) restricting your or any of your staff's attendance at any training, meetings, workshops, or conventions (including Refresher Courses); (ii) refusing to sell or furnish to you any advertising or promotional materials; (iii) refusing to provide you with ongoing advice about the operation of the Business; (iv) refusing any of your requests to approve a new supplier or the use of any advertising or promotional materials; and (v) refusing to permit you to enter into a new franchise agreement for a Business at any other location;
- (c) temporarily or permanently reduce the size of the Protected Territory, in which event the restrictions on us and our affiliates under Section 3.1(b) (Protected Territory) will not apply in the geographic area that was removed from the Protected Territory;
- (d) temporarily remove information concerning the Business from our website and/or stop your or the Business's participation in any other programs or benefits offered on or through our website;
- (e) suspend or terminate any temporary or permanent fee reductions to which we might have agreed (whether as a policy, in an amendment to this Agreement, or otherwise);

April 2025 FDD Franchise Agreement [City, State]

- (f) suspend our performance of, or compliance with, any of our obligations to you under this Agreement or other agreements; and/or
- (g) undertake or perform on your behalf any obligation or duty that you are required to, but fail to, perform under this Agreement. You will reimburse us upon demand for all costs and expenses that we reasonably incur in performing any such obligation or duty.

15.5 Exercise of Other Remedies

Our exercise of our rights under Section 15.3 (Our Right to Operate the Business and Management Fee) and 15.4 (Alternatives to Termination) will not: (i) be a defense for you to our enforcement of any other provision of this Agreement or waive or release you from any of your other obligations under this Agreement or any other agreement, (ii) constitute an actual or constructive termination of this Agreement, or (iii) be our sole or exclusive remedy for your default. You shall hold us harmless with respect to any action we take pursuant to Section 15.3 and 15.4; and you agree that we shall not be liable for any loss, expense, or damage you incur because of any action we take pursuant to Section 15.3 and 15.4. You must continue to pay all fees and otherwise comply with all your obligations under this Agreement following our exercise of any of these rights. If we exercise any of our rights under Section 15.3 or 15.4, we may thereafter terminate this Agreement without providing you any additional corrective or cure period, unless the default giving rise to our right to terminate this Agreement has been cured to our reasonable satisfaction. We may, in our business judgment, reinstate any services or benefits removed, curtailed, or limited pursuant to this Section 15.4, and you agree to accept immediately any such reinstatement of services or benefits so removed, curtailed, or limited.

16. RIGHTS AND OBLIGATIONS UPON TERMINATION OR EXPIRATION OF THIS AGREEMENT.

16.1 Payment Of Amounts Owed To Us-; Liquidated Damages

. You agree to pay us within 15 days after the effective date of termination, for any reason, or expiration of this Agreement, or on such later date that the amounts due to us are determined, suchany Royalty Fees, Ad Fees, amounts owed for purchases from us or our Affiliates, interest due on any of the foregoing, and all other amounts owed to us or our Affiliates which are then unpaid. If within fifteen (15) days following any termination or expiration of this Agreement, or following any later date that we calculate such unpaid amounts. Subject to Applicable Laws, if we terminate this Agreement for cause or if you terminate this Agreement without cause before its expiration, you also must immediately pay us (as liquidated damages for premature termination and not as a penalty or as damages for any breach of the Agreement or as a substitute for other payments due us) an amount and we agree that it would be difficult if not impossible to determine the amount of damages that we would suffer due to the loss or interruption of the revenue stream we otherwise would have derived from your continued payment of Royalty Fees, Ad Fees, and Cooperative contributions, less any cost savings, through the remainder of the term of this Agreement (the "Liquidated Damages"). Liquidated Damages will be equal to the greater of: (x) One Hundred Thousand Dollars (\$100,000), or (y) equal to the combined monthly average monthly of Royalty Fees, Ad Fees, and any other fees payable to us

or our Affiliates under this Agreement (without regard to any fee waivers or other reductions) paid by you during the 24twelve (12) months of your Business' actual operations (or such shorter period as your Business has been open if less than 24 months) preceding the date of termination, multiplied by: (A) the number of remaining months in the term of this Agreement after termination, or (B) twelve (12) if the Business had been open for less than twelve (12) months upon the date of termination. The present value of the total calculated at a discount rate of eight percent (8%), assuming payment at the end of each month, will be our Liquidated Damages. You and we agree that the calculation described in this Section is a calculation only of Liquidated Damages and that nothing herein shall preclude or limit us from proving and recovering any other damages caused by your breach of this Agreement.

16.2 Discontinue Use of the System and the Marks

- . Upon the termination, for any reason, or expiration of this Agreement:
- (a) you must immediately cease using, by advertising or in any other manner; (i) the Marks, (ii) the System and all other elements associated with the System, and (iii) any colorable imitation of any of the Marks or any trademark, service mark, trade dress, or commercial symbol that is confusingly similar to any of the Marks or that indicates or suggests a connection or association with us;
- (b) you may not directly or indirectly at any time or in any manner (except with respect to other Businesses you own and operate) identify yourself or any business as a current or former Business, or as one of our licensees or franchisees;
- (c) you agree to take such action as may be required to cancel all fictitious or assumed names or equivalent registrations relating to your use of any Marks and to change your corporate or legal business name, if necessary, so that it does not contain any of the Marks;
- (d) if we do not exercise our option to purchase the Business pursuant to Section 16.4 (Our Right to Purchase Business);
- (i) you agree to comply with our then-current de-branding checklist, which requires you to—(i): deliver to us within thirty (30) days after the Notification Date (as defined below) the Operations Manual, all signs, sign-faces, sign-cabinets, marketing materials, uniforms, forms, stationary, packaging, promotional materials, and other materials containing any Marks or otherwise identifying or relating to a Business and allow us, without liability to you or third parties, to remove all such items from the Business;(e)—if weyou fail to do not exercise our option to purchase the Business pursuant to Section 16.4 (Our Right to Purchase Business), so;
- your own expense make such alterations as we may specify to distinguish the Business clearly from its former appearance and from other Businesses so as to prevent confusion therewith by the public;

DMS_US.363658954.1

- (iii) (f) if we do not exercise our option to purchase the Business pursuant to Section 16.4 (Our Right to Purchase Business), you agree that, after the Notification Date to cancel or transfer to us or our designee all authorized and unauthorized domain names, social media accounts, telephone numbers, post office boxes, and classified and other directory listings relating to, or used in connection with, the Business or the Marks (collectively, "Identifiers"). You acknowledge that as between you and us, we have the sole rights to and interest in all Identifiers. If you fail to comply with this Section—16.2(f), you hereby authorize us and irrevocably appoint us or our designee as your attorney-in-fact to direct the telephone company, postal service, registrar, Internet Service Provider and all listing agencies to transfer such Identifiers to us. The telephone company, the postal service, registrars, Internet Service Providers and each listing agency may accept such direction by us pursuant to this Agreement as conclusive evidence of our exclusive rights in such Identifiers and our authority to direct their transfer; and
- (e) (g)-you agree to furnish us, within thirty (30) days after the Notification Date, with evidence satisfactory to us of your compliance with the foregoing obligations.

16.3 Return of Confidential Information

. You agree that, upon termination of this Agreement (including the full or partial transfer of rights by you or any Owner), for any reason, or expiration of this Agreement, you will immediately and forever cease to use any of our Confidential Information in any business or otherwise and return to us (or at our direction, destroy or delete) all copies of the Operations Manual and any other confidential materials, including computer software and any mechanisms (such as electronic keykeys) used to access the software, that we have allowed you to use.

16.4 Our Right To Purchase Business

.

(a) Exercise of Option. Upon termination or expiration of this Agreement in accordance with its terms and conditions, we have the option, exercisable by giving written notice thereof to you within sixty (60) days from the date of such termination or expiration, to purchase the Business from you, including the leasehold rights to the Location, free and clear of all liens, restrictions, or encumbrances. (The date on which we notify you whether or not we are exercising our option is referred to in this Agreement as the "Notification Date.") We have the unrestricted right to assign this option to purchase the Business. We will be entitled to all customary warranties and representations in connection with our asset purchase, including representations and warranties as to ownership and condition of and title to assets; liens and encumbrances on assets; validity of contracts and agreements; and liabilities effecting the assets, contingent or otherwise.

(b)

(b) <u>Leasehold Rights</u>. You agree, at our election, to assign your leasehold interest in the Location to us or, to enter a sublease for the remainder of the lease term on the same terms (including renewal options) as the prime lease.

April 2025 FDD Franchise Agreement [City, State]

- (c) <u>Purchase Price</u>. The purchase price for the Business will be its fair market value at the time of the Notification Date, determined in a manner consistent with reasonable depreciation of the Business's equipment, signs, inventory, materials, and supplies, provided that the Business will be valued as an independent business and its value will not include any value for the Franchise or any rights granted by this Agreement, <u>any goodwill</u>, <u>Customer Information</u>, the Marks, or participation in the <u>network of BusinessesSystem</u>. The Business's fair market value will include the reasonable goodwill you developed since your commencement of operations that exists independent of the goodwill of the Marks and the System. The length of the remaining term of the lease for the Location will also be considered in determining the Business's fair market value.
- (d) <u>Exclusions</u>. We may exclude from the assets purchased hereunder cash or its equivalent and any Operating Assets, inventory, materials, and supplies that are not reasonably necessary (in function or quality) to the Business's operation or that we have not approved as meeting standards for Businesses, and the purchase price will reflect such exclusions.
- (e) <u>Appraisal</u>. If we and you are unable to agree on the Business's fair market value, its fair market value will be determined by one independent qualified appraiser that we appoint in our business judgment. We will appoint the appraiser within <u>fifteen (15)</u> days after the date we determine we are unable to agree on the Business's fair market value. You and we will share equally the fees and expenses of the appraiser. You and we further agree to take reasonable actions to cause the appraiser to complete the appraisal within <u>thirty (30)</u> days after the appraiser's appointment.
- (f) <u>Closing</u>. The purchase price will be paid at the closing of the purchase, which will take place not later than <u>ninety (90)</u> days after determination of the purchase price. We have the right to set off against the purchase price, and thereby reduce the purchase price by, all amounts you or your Owners owe to us <u>or our Affiliates</u>.
- (g) <u>Instruments</u>. At the closing, you agree to deliver instruments transferring: (i) good and merchantable title to the assets purchased, free and clear of all liens and encumbrances (other than liens and security interests acceptable to us, if any), with all sales and other transfer taxes paid by you; and all licenses and permits of the Business which may be assigned or transferred; and (ii) the leasehold interest in the Location and improvements thereon.
- (h) <u>Escrow</u>. If you cannot deliver clear title to all the purchased assets, or if there are other unresolved issues, the closing of the sale will, at our election, be accomplished through an escrow arrangement with an independent escrow agent selected by us.
- (i) <u>Releases</u>. You and your owners agree to execute general releases, in form satisfactory to us, of any and all claims against us and our shareholders, officers, directors, employees, agents, successors and assigns.
- (j) <u>Operation of Franchise because of your Inability to Operate Business</u>. To prevent any interruption of the Business which would cause harm to the Business, if you are

unable to operate the Business for any reason whatsoever, you authorize us and our agents and Affiliates to operate the Business for so long as we deem necessary and practical. All income from the operation of the Business will be kept in a separate account, and the expenses of the business, including reasonable compensation and expenses of us and our agents, will be charged to said account. Nothing contained herein will be construed to require us to operate the Business in the case of your inability to operate same, and the rights set forth herein may be exercised in our sole and absolute discretion.

16.5 <u>Continuing Obligations</u>

. All of our and your (and your Owners' and Affiliates') obligations which expressly or by their nature survive the expiration or termination of this Agreement will continue in full force and effect subsequent to and notwithstanding its expiration or termination and until they are satisfied in full or by their nature expire, including Sections 7 (Marks), 8 (Confidential Information), 17 (Covenants Not to Compete), 18.4 (Indemnification), 19.11 (Dispute Resolution), and 16 (Rights and Obligations Upon Termination or Expiration of this Agreement).

17. COVENANTS NOT TO COMPETE.

17.1 In-Term Covenant Not To Compete

You specifically acknowledge that, pursuant to this Agreement, you will receive valuable, specialized training, Confidential Information, and other proprietary and specialized information and knowledge that provide a valuable, competitive advantage in operating a trampoline park. You further acknowledge that we would be unable to protect the Confidential Information against unauthorized use or disclosure or to encourage the free exchange of ideas and information among our franchisees if you were permitted to hold interests in or perform services for a Competitive Business, and we have granted you the rights hereunder in consideration of, and in reliance upon, your agreement to deal exclusively with us. You therefore covenant that during the term of this Agreement (except as otherwise approved in writing by us), you, your Owners, and youyour and their Immediate Families immediate family members shall not, either directly, indirectly or through, on behalf of, or in conjunction with any person or legal Entity:

- (a) <u>Have any direct or indirect interest as a disclosed or beneficial owner in a Competitive Business, wherever located or operating;</u>
- (b) <u>Perform services as a director, officer, manager, employee, consultant, lessor, representative, agent or otherwise for a Competitive Business, wherever located or operating;</u>
- (c) (a)—Divert or attempt to divert any present or prospective business or customer of related to the Business or any other franchisee's Business—to any Competitive Business, by direct or indirect—inducement or otherwise, or dodivert or perform, directly attempt to divert the employment of any employee of ours, any of our Affiliates, or indirectly, any other act injurious or prejudicial to another franchisee to any Competitive Business;

April 2025 FDD Franchise Agreement [City, State]

- (d) Engage in any other activity which might adversely affect the goodwill associated withof the Marks and the System; or
- (b) Own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in (as owner or otherwise) any Competitive Business
- (e) Solicit, interfere, or attempt to interfere with our or our Affiliates' relationships with any customers, vendors, consultants, or other franchisees.

17.2 Post-Term Covenant Not To Compete

. You covenant that, except as otherwise approved in writing by us, you and your Owners and your and their immediate family members shall not, for a continuous, uninterrupted period of two (2) years commencing upon the date of: (a) a Transfer permitted under Section 13 (Transfer), with respect only to Transfers that result in you no longer owning and operating the Business (or, in the case of an Owner, results in that Owner no longer having any direct or indirect ownership in you); (b) expiration of this Agreement; (c) termination or non-renewal of this Agreement (regardless of the cause for termination or non-renewal); or (d) a final order of a duly authorized arbitrator, panel of arbitrators, or a court of competent jurisdiction (after all appeals have been taken) with respect to any of the foregoing or with respect to enforcement of this Section 17.2, either directly or indirectly, for yourself or your Immediate Family, your Owners or your or their immediate family members or through, on behalf of, or in conjunction with any person or legal Entity, own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in (as an owner or otherwise) any Competitive Business that is, or is intended to be, located: (a) at the Location, (b) within fifteen (15) miles of the Location, or (c) within fifteen (15) miles of any Business in operation or under construction as of the date that you are required to comply with this Section 17.2. Additionally, neither you nor your Owners or any of your or their immediate family members shall: (x) solicit, interfere, or attempt to interfere with our or our Affiliates' relationships with any customers, vendors, consultants, or other franchisees; or (y) engage in any other activity that might adversely affect the goodwill of the Marks and the System. You agree and acknowledge that the two-year period of this restriction these restrictions shall be tolled during any time period in which you are in violation of this restriction.

The restrictions in Sections 17.1(c) and 17.2 do not apply to: $(a\underline{1})$ interests in or operation of a Business underpursuant to a written Franchise Agreement with us; or $(b\underline{2})$ the ownership of shares of a class of securities that are listed on a public stock exchange or traded on the over-the-counter market and that represent less than five percent (5%) of that class of securities.

17.3 Reasonable Scope of Covenants

. You acknowledge that the scope of the restrictions in Sections 17.1 (In-Term Covenant Not to Compete) and 17.2 (Post-Term Covenant Not to Compete) are reasonable and necessary to protect us, the Confidential Information, and the System, and that such restrictions are designed solely to prevent you from taking information, materials, training, and know-how that

we provided to you and using them to compete with us. In addition, your operation of a Competitive Business in violation of Section 17.1 and 17.2 would necessarily involve your use of Confidential Information that would result in an unfair competitive advantage vis-à-vis other SKY ZONESky Zone franchisees. You further acknowledge that you and your Owners possess skills and abilities of a general nature and have other opportunities for exploiting these skills. Consequently, our enforcement of the covenant in Section 17.1 and 17.2 will not deprive you or your Owners of personal goodwill or the ability to engage in a lawful trade or business and earn a living.

17.4 Reduction Of Scope of Covenants

. You understand and acknowledge that we shall have the right, in our business judgment, to reduce the scope of any covenant set forth in Sections 17.1 (In-Term Covenant Not to Compete) and 17.2 (Post-Term Covenant Not to Compete), or any portion thereof, without your consent, effective immediately upon receipt by you of written notice thereof; and you agree that you shall comply forthwith with any covenant as so modified, which shall be fully enforceable. If any covenant herein which restricts competitive activity is deemed unenforceable by virtue of its scope in terms of area, <u>prohibited</u> business activity prohibited and/or length of time, but would be enforceable by reducing any part or all thereof, you and we agree that such covenant will be enforced to the fullest extent permissible under the laws and public policies applied in the jurisdiction whose law is applicable to the validity of such covenant.

17.5 <u>Covenant Not To Compete Upon Exercise Of Right Of First Refusal</u>

. If we exercise our right of first refusal pursuant to Section 13.8 (Our Right of First Refusal), you and your selling Owner(s) agree that, for a period of two (2) years commencing on the date of the closing, you and they will be bound by the noncompetition covenant contained in Section 17.2 (Post-Term Covenant Not to Compete).

18. RELATIONSHIP OF THE PARTIES AND INDEMNIFICATION.

18.1 Independent Contractors

. Neither this Agreement nor the dealings of the parties pursuant to this Agreement shall create any fiduciary relationship or any other relationship of trust or confidence between the parties hereto. You acknowledge and agree that we and you are and shall be independent contractors. Nothing contained in this Agreement, or arising from the conduct of the parties hereunder, is intended to make either party a general or special agent, joint venturer, partner, or employee of the other for any purpose whatsoever. We have no relationship with your employees, and you have no relationship with our employees. You may not make any express or implied agreements, warranties, guarantees or representations or incur any debt in our name or on our behalf or using the Marks, and you may not represent that the relationship of the parties hereto is anything other than that of independent contractors. We shall not be construed to have any liability, including joint liability, for any of your acts or omissions under any circumstances. We will not be obligated for any damages to any person or property arising directly or indirectly out of the operation of your Business.

18.2 Notice of Independent Contractor

<u>.</u> You agree to hold yourself out to the public as an independent contractor operating your Business <u>under licenseas a Franchise</u> from us, and you must display in a conspicuous location in or upon the Business, or in a manner that we specify, a sign containing the following notice or an alternative notice that we specify: "This business is owned and operated independently by [name of franchisee] who is an authorized licensed user of the trademark SKY ZONE, which is a trademark licensed by Sky Zone Franchise Group, LLC." You must include this notice or other similar language that we specify on all forms, advertising, promotional materials, business cards, receipts, letterhead, contracts, stationary, and other written materials we designate.

18.3 Taxes

. We will have no liability for any sales, use, service, occupation, employment related, excise, gross receipts, income, property, or other taxes, whether levied upon you or the Business, in connection with the business you conduct (except any taxes we are required by law to collect from you with respect to purchases from us). Payment of all such taxes is your sole responsibility. Further, you will pay all state and local taxes, including sales, use, service, occupation, employment related, excise, gross receipts, income, property or other taxes, that may be imposed on us as a result of our receipt or accrual of the Initial Franchise Fee, Royalty Fees, Ad Fees, extension fees, and all other fees that are referenced in this Agreement, whether assessed against you through withholding or other means or whether paid by us directly, unless the tax is credited against income tax otherwise payable by us. In such event, you will pay to us (or to the appropriate governmental authority) such additional amounts as are necessary to provide us, after taking such taxes into account (including any additional taxes imposed on such additional amounts), with the same amounts that we would have received or accrued had such withholding or other payment, whether by you or by us, not been required.

18.4 Indemnification

. You, and each of the Guarantors, agree that you shall, at all times, indemnify, exculpate, defend and hold harmless, to the fullest extent permitted by law, us, our successor, assigns, and Affiliates, and the respective officers, directors, shareholders, agents, representatives, independent contractors, servants, and employees of us and each of them (collectively, the "Indemnified Parties") from all Losses (defined below) directly or indirectly incurred in connection with any action, suit, proceeding, claim, demand, investigation, or inquiry (formal or informal), or any settlement thereof arising out of or relating to: (i) the infringement, alleged infringement or any other violation by you, your Guarantors or principals of any patent, mark, copyright, or other proprietary right owned or controlled by third parties due to your unauthorized use of all or any portion of the Marks and/or System; (ii) the violation, breach, or asserted violation or breach by you, your Guarantors or principals of any Applicable LawLaws, rulingrulings, or industry standards; (iii) libel, slander, or any other form of defamation by you or your Guarantors or principals; (iv) the violation or breach by you or by your Guarantors or principals of any warranty, representation, agreement, or obligation of this Agreement or in any other agreement between you and us or our Affiliates; (v) acts, errors, omissions of you, any of your affiliates Affiliates, any of your principals, officers, directors,

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

shareholders, agents, representatives, independent contractors, and employees of you and your affiliates Affiliates in connection with the establishment and operation of the Business, including any acts, errors, or omissions of any of the foregoing in the operation of any motor vehicle or in the establishment or implementation of security for the Business; (vi) any of the foregoing that are alleged to be caused by an Indemnified Party's negligence, unless (and then only to the extent that) the claims, obligations, and damages are determined to be caused solely by the Indemnified Party's gross negligence or willful misconduct according to a final, unappealable ruling issued by a court or arbitrator with competent jurisdiction; and (vii) any allegation that we or another Indemnified Party is a joint employer or otherwise responsible for your acts or omissions relating to your employees. For purposes of this indemnification, "Losses" means and all losses, expenses, obligations, liabilities, damages (actual, consequential, or otherwise), and reasonable defense costs that an Indemnified Party incurs, including accountants', arbitrators', mediators', attorneys', and expert witness fees, costs of investigation and proof of facts, court costs, travel Travel and living expenses Living Expenses, and other expenses of litigation, arbitration, or alternative dispute resolution, regardless of whether litigation, arbitration, or alternative dispute resolution is commenced. We have the right to defend any such claim against us at your expense. This indemnity will continue in full force and effect after and notwithstanding the expiration or termination of this Agreement.

18.5 Mitigation Not Required

. Under no circumstances will we or any other Indemnified Party be required to seek recovery from any insurer or other third party, or otherwise to mitigate our, their or your losses and expenses, to maintain and recover fully a claim against you. You agree that a failure to pursue such recovery or mitigate a loss will in no way reduce or alter the amounts we or another Indemnified Party may recover from you.

19. ENFORCEMENT AND MISCELLANEOUS MATTERS.

19.1 Severability And Substitution of Valid Provisions

. Each provision of this Agreement is severable from the others. If any provision (or portion of a provision) of this Agreement or any of the documents executed in conjunction with this Agreement is for any reason determined by a court to be invalid, illegal, or unenforceable, the invalidity, illegality, or unenforceability will not affect any other remaining provisions of this Agreement or any other document. The remaining provisions will continue to be given full force and effect and bind us and you.

19.2 **Greater Notice**

. If any Applicable Law requires a greater prior notice than is required hereunder of the termination of this Agreement or of our refusal to enter into a successor franchise agreement, or the taking of some other action not required hereunder, or if, under any Applicable Law, any provision of this Agreement or any part of Methods of Operation is invalid or unenforceable, the prior notice and/or other action required by such law or rule will be substituted for the comparable provisions hereof, and we will have the right to modify such invalid or

unenforceable provision or unenforceable part of this Agreement or the Operations Manual or any part of Methods of Operation to the extent required to be valid and enforceable. You agree to be bound by any promise or covenant imposing the maximum duty permitted by law which is subsumed within the terms of any provision hereof as though it were separately articulated in and made a part of this Agreement, that may result from striking from any of the provisions hereof, or any part of Methods of Operation, any portion or portions which a court or arbitrator may hold to be unenforceable in a final decision to which we are a party, or from reducing the scope of any promise or covenant to the extent required to comply with such a court order or arbitration award. Such modifications to this Agreement will be effective only in such jurisdiction, unless we elect to give them greater applicability, and will be enforced as originally made and entered into in all other jurisdictions.

19.3 Waiver Of Obligations

. We and you may by written instrument unilaterally waive or reduce any obligation of or restriction upon the other under this Agreement, effective upon delivery of written notice thereof to the other or such other effective date stated in the notice of waiver. Any waiver we grant will be without prejudice to any other rights we may have, will be subject to our continuing review and may be revoked at any time and for any reason, effective upon delivery to you of <u>ten (10)</u> days' prior written notice.

19.4 Non-Waiver

. We and you will not be deemed to have waived or impaired any right, power or option reserved by this Agreement (including the right to demand exact compliance with every term, condition and covenant herein or to declare any breach thereof to be a default and to terminate this Agreement prior to the expiration of its term) by virtue of: (i) any custom or practice at variance with the terms hereof; (ii) our or your failure refusal or neglect to exercise any right under this Agreement or to insist upon exact compliance by the other with our and your obligations hereunder including Methods of Operation; (iii) our waiver, forbearance, delay, failure, or omission to exercise any right, power or option whether of the same, similar or different nature with respect to other Businesses; (iv) the existence of other franchise agreements for Businesses which contain different provisions from those contained herein; or (v) our acceptance of any payments due from you after any breach of this Agreement. No special or restrictive legend or endorsement on any check or similar item given to us will constitute a waiver, compromise, settlement or accord and satisfaction. We are authorized to remove or obliterate any legend or endorsement, and such legend or endorsement will have no effect.

19.5 Force Majeure

. Neither we nor you will be liable for loss or damage or deemed to be in breach of this Agreement if our or your failure to perform our or your obligations is not our or your fault and results from: (i) transportation shortages, inadequate supply of equipment, products, merchandise, supplies, labor, material or energy or the voluntary foregoing of the right to acquire or use any of the foregoing in order to accommodate or comply with the orders, requests, regulations, recommendations or instructions of any federal, state or municipal government or

any department or agency thereof; (ii) acts of nature; (iiii) fires, strikes, embargoes, war or riot; (iviii) failure to obtain land use or environmental approvals from the applicable government body or agency, so long as you diligently pursue any such required approvals; or (viv) any other similar event or cause. Any delay resulting from any of said force majeure causes will extend performance accordingly or excuse performance, in whole or in part, as may be reasonable, except that said causes will not excuse payments of amounts owed at the time of such occurrence or payment of Royalty Fees and Ad Fees due on any sales thereafter.

19.6 Out-Of-Stock and Discontinued

. We <u>and our Affiliates</u> are not liable to you for any loss or damage, or deemed to be in breach of this Agreement, if we cannot deliver, or cause to be delivered, or if our Affiliates or designated sources or approved suppliers cannot deliver, all your orders for products <u>(including Proprietary Products)</u>, merchandise, Operating Assets, supplies, <u>etc.</u>, <u>where and other goods if such thingsproducts</u>, supplies, and other goods are out-of-stock or discontinued.

19.7 Costs And Attorneys' Fees

. If we incur expenses in connection with your failure to pay when due amounts owed to us or to submit when due any reports, information or supporting records or otherwise to comply with this Agreement, you agree to reimburse us for any of the costs and expenses which we incur, including reasonable accounting, attorneys', arbitrators', and related fees.

19.8 You May Not Withhold Payments Due To Us

. You agree that you will not withhold payment of any amounts owed to us on the grounds of our alleged nonperformance of any of our obligations hereunder. You agree that all such claims will, if not otherwise resolved by us, be submitted to <a href="mailto:mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation_mediation

19.9 Rights Of Parties Are Cumulative

. Our and your rights hereunder are cumulative, and no exercise or enforcement by us or you of any right or remedy hereunder will preclude our or your exercise or enforcement of any other right or remedy hereunder which we or you are entitled by law to enforce.

19.10 Good Faith and Fair DealingBusiness Judgment

If Applicable Law shall imply a covenant of good faith and fair dealing in this Agreement, we and you agree that (i) such covenant shall not imply any rights or obligations that are inconsistent with a fair construction of the terms of this Agreement; (ii) this Agreement (and the relationship of the parties which arises from this Agreement) grants us the right to make decisions, take actions and/or refrain from taking actions not inconsistent with your explicit rights and obligations hereunder that may affect favorably or adversely your interests; (iii) we will use our judgment in exercising such rights based on our assessment of our own interests and balancing those interests against the interests of the owners of Businesses generally (including

ourselves, and our Affiliates and other franchisees), and specifically without considering your individual interests or the individual interests of any other particular franchisee; (iv) we will have no liability to you for the exercise of our rights in this manner so long as such rights are not exercised in bad faith toward you; and (v) in the absence of such bad faith, no trier of fact in any legal action or arbitration proceeding shall substitute its judgment for our judgment so exercised

Whenever we reserve discretion in a particular area or where we agree or are required to exercise our rights reasonably or in good faith, we will satisfy our obligations whenever we exercise "reasonable business judgment" in making our decision or exercising our rights. A decision or action by us will be deemed to be the result of "reasonable business judgment," even if other reasonable or even arguably preferable alternatives are available, if our decision or action is intended to promote or benefit the System generally even if the decision or action also promotes a financial or other of our individual interests. Examples of items that will promote or benefit the System include enhancing the value of the Marks, improving customer service and satisfaction, improving product quality, improving uniformity, enhancing or encouraging modernization, and improving the competitive position of the System. Neither you nor any third party (including a trier of fact) will substitute their judgment for our reasonable business judgment.

19.11 <u>Dispute Resolution</u>

- (a) <u>MediationInformal Resolution</u>. Except as provided in Section 19.11(c) (Exceptions to Arbitration), prior to filing any demand for arbitration, the parties agree to <u>mediateattempt to informally resolve</u> any Dispute in accordance with the following procedures:
- by sending the other party, in accordance with Section 19.26 (Notices), a written notice of its request for mediation headed <u>Dispute</u> with the heading "Notification of Dispute." The Notification of Dispute will specify, to the fullest extent possible, the party's version of the facts surrounding the dispute; the amount of damages; and the nature of any injunctive or other relief such party claims. The party (or parties as the case may be) receiving a Notification of Dispute will respond within 20thirty (30) days after receipt thereof, stating its version of the facts and, if applicable, its position as to damages sought by the party initiating the dispute procedure; provided, however, that if the <u>disputeDispute</u> has been the subject of a default notice given under Section 15 (Termination of Agreement), the other party will respond within <u>ten (10)</u> business days.
- (ii) Upon receipt of a Notification of Dispute and response under Section 19.11(a)(i), the parties will endeavor, in good faith, for sixty (60) days (or an extended timeframe as mutually agreed upon between the parties) to resolve the dispute outlined in the Notification of Dispute and the other party's response. If In particular, the parties have been unable to resolve a dispute outlined in a Notification of Dispute or a response thereto within 20 days after receipt of the response, either party may initiate a mediation procedure with the American Arbitration Association ("AAA"), pursuant to its Commercial Mediation Procedures,

in the city of our then current corporate headquarters, unless both parties agree to a different location. The parties must select a mediator jointly.

- (iii) All mediation sessions must be attended by an Owner (who has authority to settle the dispute on your behalf) and our representative(s) who is/are authorized to settle the dispute. The parties may be represented by counsel at the mediation. The parties agree to participate in the mediation proceedings in good faith and with the intention of resolving the dispute, if at all possible, within 30 days of the notice from the party seeking to initiate the mediation procedures. If the dispute is not resolved within 30 days, any party may initiate an arbitration pursuant to Section 19.11(b) (Arbitration). In addition, if the party receiving notice of mediation has not responded within 5 days of delivery of the notice or a party fails to participate in the mediation, this Section 19.11(a) will no longer be applicable, and the other party can pursue arbitration. The parties agree that the costs of the mediator will be split equally between the parties. Each party must pay its own fees and expenses incurred in connection with the mediation. The mediation proceeding and any negotiations and results thereof will be treated as a compromise settlement negotiation and the entire process is confidential. At least five days prior to the initial mediation session, each party must deliver a written statement of positions Responsible Person will engage in all discussions with a designated SZFG executive to attempt to resolve the Dispute during such timeframe.
- Arbitration. Except as provided in Section 19.11(c) (Exceptions to Arbitration), any Dispute, including any disputeDispute regarding the scope or validity of the arbitration obligation under this Section, not resolved by mediation the informal process set forth in Section 19.11(a) must be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association then in effect to be heard by one (1) arbitrator who has a minimum of five (5) years' experience in franchising or distribution law.
- **Counterclaims**. In connection with any arbitration proceeding, (i) each party will submit or file any claim which would constitute a compulsory counterclaim (as defined by the then-current Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding as the claim to which it relates. Any such claim which is not submitted or filed in such proceeding will be barred.
- Waiver of Class Actions. Any You (and your Owners) expressly acknowledge and agree that any arbitration must be on an individual basis only as to a single franchise franchisee (and not as or through an association) and the parties and the arbitrator will have no authority or power to proceed with any claim on a class-wide basis or otherwise to join or consolidate any claim with any claim or any other proceeding involving third parties. If a court or arbitrator determines that this limitation on joinder of or class-wide claims is unenforceable, then the agreement to arbitrate the dispute Dispute will be null and void and the parties must submit all claims to the jurisdiction of the courts, in accordance with Section 19.13 (Consent to Jurisdiction).
- (iii) **Venue**. The arbitration must take place in the city where our headquarters principal place of business is located at the time of the dispute Dispute (currently, Provo, Utah), unless the parties mutually agree upon an alternate location.

- (iv) Arbitrator. The arbitrator must follow the law and not disregard the terms of this Agreement. The arbitrator will have subpoena powers limited only by the laws of the state in which our corporate headquartersprincipal place of business is then located. The arbitrator may not consider any settlement discussions or offers that might have been made by either you or us. The arbitrator may not under any circumstance: (a) stay the effectiveness of any pending termination of this Agreement, (b) assess punitive or exemplary damages, (c) certify a class or a consolidated action, or (d) make any award which extends, modifies, or suspends any lawful term of this Agreement or any reasonable standard of business performance that we set. The arbitrator will have the right to decide as to any procedural matters as would a court of competent jurisdiction be permitted to make in the state in which our corporate headquarters is then located. The arbitrator will also decide any factual, procedural, or legal questions relating in any way to the dispute Dispute between the parties, including: any decision as to whether Section 19.11(b) (Arbitration) is applicable and enforceable as against the parties, subject matter, timeliness, scope, remedies, unconscionability, and any alleged fraud in the inducement.
- (v) <u>Orders</u>. The arbitrator can issue summary orders disposing of all or part of a claim and provide for temporary restraining orders, preliminary injunctions, injunctions, attachments, claim and delivery proceedings, temporary protective orders, receiverships, and other equitable and/or interim/final relief. The judgment of the arbitrator on any preliminary or final arbitration award will be final and binding and may be entered in any court having jurisdiction. Each party consents to the enforcement of such orders, injunctions, etc., by any court having jurisdiction.
- (vi) <u>Procedure</u>. The parties to the <u>dispute Dispute</u> will have the same discovery rights as are available in civil actions under the laws of the state in which our <u>corporate headquartersprincipal place of business</u> is <u>then_located_at the time of the Dispute (currently, Utah)</u>. All other procedural matters will be determined by applying the statutory, common laws, and rules of procedure that control a court of competent jurisdiction in which our corporate headquartersprincipal place of business is then_located at the time of the Dispute.
- (vii) <u>Confidential</u>. Other than as may be required by law <u>(including but not limited to state and federal franchise laws)</u>, the entire arbitration proceedings (including any rulings, decisions, or orders of the arbitrator), will remain confidential and will not be disclosed to anyone other than the parties to this Agreement.
- (viii) <u>Advances</u>. We reserve the right, but have no obligation, to advance your share of the costs of any arbitration proceeding for such arbitration proceeding to take place and by doing so will not be deemed to have waived or relinquished our right to seek recovery of those costs in accordance with Section 19.7 (Cost and Attorneys' Fees) or 19.11(d) (Prevailing Party's Fees).
- three (3) depositions, unless the parties mutually agree to additional depositions. Each side's depositions are to consume no more than a total of fifteen (15) hours, and each deposition shall be limited to five (5) hours, unless the parties mutually agree to additional time.

- (x) Electronic Discovery. With respect to any discovery of electronically stored information, you and we agree that such requests must balance the need for production of electronically stored information relevant and material to the outcome of a disputed issue against the cost of locating and producing such information. You and we agree that:
- <u>1. Production of electronically stored information need only</u> be from sources used in the ordinary course of business. No party shall be required to search for or produce information from back-up servers, tapes, or other media;
- 2. The production of electronically stored information shall normally be made on the basis of generally available technology in a searchable format which is usable by the party receiving the information and convenient and economical for the producing party. Absent a showing of compelling need, the parties need not produce metadata, with the exception of header fields for email correspondence;
- 3. The description of custodians from whom electronically stored information may be collected shall be narrowly tailored to include only those individuals whose electronically stored information may reasonably be expected to contain evidence that is relevant and material to the outcome of a Disputed issue;
- 4. The parties shall attempt to agree in advance upon, and the arbitrator may determine, reasonable search parameters;
- 5. Where the costs and burdens of electronic discovery are disproportionate to the nature of the Dispute or to the amount in controversy, or to the relevance of the materials requested, the arbitrator shall either deny such requests or order disclosure on condition that the requesting party advance the reasonable cost of production to the other side, which cost advance will not be awarded to the prevailing party in any final award.
- (c) <u>Exceptions to Arbitration</u>. Notwithstanding Sections 19.11(a) (<u>Mediation Informal Resolution</u>) and 19.11(b) (Arbitration), the parties agree that the following Disputes will not be subject to arbitration or mediation:
- (i) any action for equitable relief, including seeking preliminary or permanent injunctive relief, specific performance, declaratory relief, other relief in the nature of equity to enjoin any harm or threat of harm to such party's tangible or intangible property, brought at any time, including prior to or during the pendency of any arbitration proceedings initiated hereunder;
- (ii) any action in ejectment or for possession of any interest in real or personal property;
 - (iii) any action which by Applicable LawLaws cannot be arbitrated; or

- (iv) our decision in the first instance to issue a notice of default and/or notice of termination or undertake any other conduct with respect to the franchise relationship that might later result in a disputeDispute or controversy between us.
- (d) <u>Prevailing Party's Fees</u>. The prevailing party in any action or proceeding arising under, out of, in connection with, or in relation to this Agreement will be entitled to recover its reasonable costs and expenses (including attorneys' fees, arbitrator's fees and expert witness fees, costs of investigation and proof of facts, court costs, and other arbitration or litigation expenses) incurred in connection with the claims on which it prevailed.
- (e) <u>Survival</u>. The provisions of this Section 19.11 are intended to benefit and bind certain third-party non-signatories and will continue in full force and effect after and notwithstanding the expiration or termination of this Agreement.
- (f) <u>Tolling of Statute of Limitations</u>. All applicable statutes of limitation and defenses based on the passage of time are tolled while the <u>disputeDispute</u> resolution procedures in this Section 19.11 are pending. The parties will take such action, if any, required to effectuate such tolling.
- (g) <u>Performance to Continue</u>. Each party must continue to perform its obligations under this Agreement pending final resolution of any Dispute pursuant to this Section 19.11, unless to do so would be impossible or impracticable under the circumstances.

19.12 Governing Law

. This Agreement will be governed by, construed, and enforced in accordance with the laws of the State of Missouri, although any Missouri franchise law only applies in the event you are a Missouri resident or the Location is located in Missouri. In the event of any conflict-of-law question, the laws of Missouri shall prevail, without regard to the application of Missouri conflict of law rules ALL MATTERS RELATING TO ARBITRATION WILL BE GOVERNED BY THE UNITED STATES FEDERAL ARBITRATION ACT (9 U.S.C. §§ 1 ET SEQ.). EXCEPT TO THE EXTENT GOVERNED BY THE FEDERAL ARBITRATION ACT, THE UNITED STATES TRADEMARK ACT OF 1946 (LANHAM ACT, 15 U.S.C. SECTIONS 1051 ET SEQ.) (THE "LANHAM ACT"), OR OTHER UNITED STATES FEDERAL LAW, THIS AGREEMENT, THE DEVELOPMENT RIGHTS, AND ALL CLAIMS ARISING FROM THE RELATIONSHIP BETWEEN US AND YOU WILL BE GOVERNED BY THE LAWS OF THE STATE IN WHICH OUR PRINCIPAL PLACE OF BUSINESS IS LOCATED (CURRENTLY, UTAH) WITHOUT REGARD TO ITS CONFLICT OF LAWS RULES, EXCEPT THAT ANY STATE LAW REGULATING THE OFFER OR SALE OF FRANCHISES OR GOVERNING THE RELATIONSHIP OF A FRANCHISOR AND ITS FRANCHISEE WILL NOT APPLY UNLESS ITS JURISDICTIONAL REQUIREMENTS ARE MET INDEPENDENTLY WITHOUT REFERENCE TO THIS PARAGRAPH.

19.13 Consent To Jurisdiction

Subject to Section 19.11 (Dispute Resolution), you and the Owners must file any suit against us, and we may file any suit against you, in federal or state courts located in the state where our principal place of business is then located. You (and each Owner) irrevocably submit to the jurisdiction of such courts and waive any objection you (or he or she) may have to either the jurisdiction of or venue in such courtsSUBJECT TO SECTION 19.11 (DISPUTE RESOLUTION), YOU (AND YOUR OWNERS) AGREE THAT ALL ACTIONS ARISING OUT OF OR RELATING TO THIS AGREEMENT OR OTHERWISE AS A RESULT OF THE RELATIONSHIP BETWEEN YOU AND US MUST BE COMMENCED EXCLUSIVELY IN A STATE OR FEDERAL COURT IN THE LOCATION IN WHICH OUR PRINCIPAL PLACE OF BUSINESS IS LOCATED (CURRENTLY, PROVO, UTAH), AND WE AND YOU (AND EACH OWNER) IRREVOCABLY CONSENT TO THE JURISDICTION OF THOSE COURTS AND WAIVE ANY OBJECTION TO EITHER THE JURISDICTION OF OR

19.14 Waiver Of Punitive Damages, Jury Trial, And Class Actions

. Except with respect to your obligation to indemnify us pursuant to Section 18.4 (Indemnification) and 18.5 (Mitigation Not Required) and claims we bring against you for your unauthorized use of the Marks or unauthorized use or disclosure of any Confidential Information, we and you and your respective Owners waive to the fullest extent permitted by law any right to or claim for any punitive or exemplary damages against the other and agree that, in the event of a dispute between us, the party making a claim will be limited to equitable relief and to recovery of any actual damages it sustains. We and you irrevocably waive, to the fullest extent permitted by law, trial by jury in any action, proceeding or counterclaim, whether at law or in equity, brought by either of us. Any claims must be brought on an individual basis only as to a single franchise (and not as or through an association or class). This waiver is effective even if a court of competent jurisdiction decides that the arbitration provision in Section 19.11(b) (Arbitration) is unenforceable. Each party acknowledges that it has had a full opportunity to consult with counsel concerning this waiver, and that this waiver is informed, voluntary, intentional, and not the result of unequal bargaining powerEXCEPT WITH RESPECT TO YOUR OBLIGATION TO INDEMNIFY US PURSUANT TO SECTION 18.4 (INDEMNIFICATION) AND 18.5 (MITIGATION NOT REQUIRED) AND CLAIMS WE BRING AGAINST YOU FOR YOUR UNAUTHORIZED USE OF THE MARKS OR UNAUTHORIZED USE OR DISCLOSURE OF ANY CONFIDENTIAL INFORMATION, WE AND YOU AND YOUR RESPECTIVE OWNERS WAIVE TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT TO OR CLAIM FOR ANY PUNITIVE OR EXEMPLARY DAMAGES AGAINST THE OTHER AND AGREE THAT, IN THE EVENT OF A DISPUTE BETWEEN US, THE PARTY MAKING A CLAIM WILL BE LIMITED TO EQUITABLE RELIEF AND TO RECOVERY OF ANY ACTUAL DAMAGES IT SUSTAINS. WE AND YOU IRREVOCABLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY LAW, TRIAL BY JURY IN ANY ACTION, PROCEEDING OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, BROUGHT BY EITHER OF US. WE AND YOU AGREE THAT ANY PROCEEDING WILL BE

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

City, State

CONDUCTED ON AN INDIVIDUAL BASIS AND THAT ANY PROCEEDING BETWEEN US AND ANY OF OUR AFFILIATES, OR OUR AND THEIR RESPECTIVE SHAREHOLDERS, OFFICERS, DIRECTORS, AGENTS, AND EMPLOYEES, ON THE ONE HAND, AND YOU (OR YOUR OWNERS, GUARANTORS, AFFILIATES, AND EMPLOYEES), ON THE OTHER HAND, MAY NOT BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II) COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER PROCEEDING, (III) JOINED WITH ANY CLAIM OF AN UNAFFILIATED THIRD PARTY, OR (IV) BROUGHT ON YOUR BEHALF BY ANY ASSOCIATION OR AGENT. THIS WAIVER IS EFFECTIVE EVEN IF A COURT OF COMPETENT JURISDICTION DECIDES THAT THE ARBITRATION PROVISION IN SECTION 19.11(B) (ARBITRATION) IS UNENFORCEABLE. EACH PARTY ACKNOWLEDGES THAT IT HAS HAD A FULL OPPORTUNITY TO CONSULT WITH COUNSEL CONCERNING THIS WAIVER, AND THAT THIS WAIVER IS INFORMED, VOLUNTARY, INTENTIONAL, AND NOT THE RESULT OF UNEQUAL BARGAINING POWER.

19.15 Limitation Of Claims

. Except for claims arising from your nonpayment or underpayment of amounts you owe us pursuant to this Agreement, or claims related to your unauthorized use of the Marks, any and all claims arising out of or relating to this Agreement or our relationship with you will be barred unless a Notification of Dispute is sent or a judicial proceeding is commenced within one (1) year from the date on which the party asserting such claim knew or should have known of the facts giving rise to such claims.

19.16 Binding Effect

. This Agreement is binding upon us and you and our respective executors, administrators, heirs, beneficiaries, assigns and successors in interest.

19.17 Withhold Approval

<u>.</u> Except where this Agreement expressly obligates us to reasonably approve or to not unreasonably withhold our approval of any of your actions or requests, we have the absolute right to refuse any request you make or to withhold our approval of any of your proposed or effected actions that require our approval.

19.17 19.18 Construction

. The headings of the several Sections hereof are for convenience only and do not define, limit, or construe the contents of such Sections. Unless otherwise specified, all references to a number of days shall mean calendar days and not business days.

The words "**include**," "**including**," and words of similar import shall be interpreted to mean "including, but not limited to" and the terms following such words shall be interpreted as examples of, and not an exhaustive list of, the appropriate subject matter.

19.18 19.19 Joint And Several Owners' Liability Sky Zone Franchise Group, LLC

. If two_(2) or more persons or Entities are at any time Owners, whether as partners or joint venturers, their obligations, and liabilities to us will be joint and several.

19.19 19.20 Right To Information

. You consent to us obtaining, using, and disclosing to third parties (including financial institutions, legal and financial advisors, and prospective franchisees), for any purpose we specify or as may be required by law, all financial and other information (including CustomerPersonal Information) contained in or resulting from information, data, materials, statements and reports related, directly or indirectly, to the Business.

19.20 19.21 Multiple Copies

<u>; Electronic Signatures</u>. This Agreement may be executed in counterparts, each of which will be deemed an original, and all of which when taken together shall constitute one and the same document. This Agreement shall not be binding on either party until it is executed by both parties. <u>Electronic signatures or signatures which are transmitted via facsimile or scanned and emailed shall have the same force and effect as originals.</u>

19.21 19.22 Entire Agreement Between the Parties

. This Agreement, including any addenda and appendices and our Operations Manual, and the documents referred to therein constitute the sole agreement between you and us with respect to the entire subject matter of this Agreement and embody all prior agreements and negotiations with respect to your Business authorized hereunder. We expressly disclaim any understandings, agreements, inducements, course(s) of dealing, representations (financial or otherwise), promises, options, rights of first refusal, guarantees, warranties (express or implied) or otherwise (whether oral or written) which are not fully expressed in this Agreement. Nothing in this or any related agreement, however, is intended to disclaim the representations made by us in the FDD.

19.22 **19.23 Amendments**

. No change, modification, amendment, or waiver of any of the provisions hereof, including by custom, usage of trade, or course of dealing or performance, shall be effective and binding upon either party unless it is in writing, specifically identified as an amendment hereto and signed by a duly authorized representative of both parties. The Operations Manual and any Methods of Operation that we adopt, and implement may be changed by us unilaterally from time to time.

19.23 19.24 No Third-Party Beneficiaries

. Except as expressly otherwise provided herein, no third party shall have the right to claim any of the benefits conferred under this Agreement.

19.24 **19.25** Notices

. All notices and other communications required or permitted under this Agreement will be in writing and will be given by one of the following methods of delivery: (i) personally; (ii) by certified or registered mail, postage prepaid; (iii) by overnight delivery service; or (iv) by electronic mail to the email address on file for you with us (with a <u>coycopy</u> sent by U.S. Mail or overnight delivery services). Notices to you will be sent to the address set forth on **Appendix A**. Notices to us must be sent to:

Sky Zone Franchise Group, LLC 86 N. University Avenue, Suite 350 Provo, Utah 84601 Attn: Franchise Operations (franchiseops@skyzone.com) with copy to Sky ZoneSZFG's Legal Department (legal@skyzone.com)

Either party may change its mailing address or electronic mail address by giving notice to the other party. Notices will be deemed received the same day when delivered personally, upon attempted delivery when sent by registered or certified mail or overnight delivery service, or when sent by electronic mail.

20. ACKNOWLEDGEMENTS AND REPRESENTATIONS.

20.1 Acknowledgements

¿You acknowledge that you have read this Agreement and our FDD and understand and accept the terms, conditions and covenants contained in this Agreement as being reasonably necessary to maintain our high standards of quality and service and the uniformity of those standards at each Business and thereby to protect and preserve the goodwill of the Marks. You acknowledge that you have conducted an independent investigation of the business venture contemplated by this Agreement and recognize that, like any other business, the nature of the business conducted by a Business may evolve and change over time, that an investment in a Business involves business risks and that your business abilities and efforts are vital to the success of the venture. You acknowledge that, in all of their dealings with you, our officers, directors, employees and agents act only in a representative, and not in an individual, capacity. All business dealings between you and such persons as a result of this Agreement are solely between you and us. You further acknowledge that we have advised you to seek franchise counsel to review and evaluate this Agreement.

20.2 Representation

• You represent to us, as an inducement to our entry into this Agreement, that all statements you have made and all materials you have submitted to us in connection with your purchase of the franchise are accurate and complete and that you have made no misrepresentations or material omissions in obtaining the Franchise. We have approved of your purchasing a Franchise in reliance upon all of your representations.

20.3 Due Authority

• You represent that this Agreement has been duly authorized and executed by you or on your behalf and constitutes your valid and binding obligation, enforceable in accordance with its terms, subject to applicable bankruptcy, moratorium, insolvency, receivership, and other similar laws affecting the rights of creditors generally.

19.25 20.4 Terrorist Acts

. You acknowledge that under applicable U.S. law, including Executive Order 13224, signed on September 23, 2001 (the "Order"), we are prohibited from engaging in any transaction with any person engaged in, or with a person aiding any person engaged in, acts of terrorism, as defined in the Order. Accordingly, you represent and warrant to us that, as of the date of this Agreement, neither you nor any person holding any ownership interest in you, controlled by you, or under common control with you is designated under the Order as a person with whom business may not be transacted by us, and that you: (i) do not, and hereafter will not, engage in any terrorist activity; (ii) are not affiliated with and do not support any individual or Entity engaged in, contemplating, or supporting terrorist activity; and (iii) are not acquiring the rights granted under this Agreement with the intent to generate funds to channel to any individual or Entity engaged in, contemplating, or supporting terrorist activity, or to otherwise support or further any terrorist activity.

20.5 Timely Receipt and Review of Agreement and Disclosure Document

¿You have received an execution ready copy of this Agreement at least seven calendar days before you executed this Agreement or any related agreements or paid any consideration to us. You have also received a FDD required by applicable state and/or federal laws, including a form of this Agreement, at least 14 calendar days (or such longer period as required by applicable state law) before you executed this Agreement or any related agreements or paid any consideration to us. You have reviewed this Agreement and the FDD and have been given ample opportunity to consult with, and ask questions of, our representatives regarding the documents. You have no knowledge of any representations made about the SKY ZONE franchise opportunity by us, our Affiliates, or any of our or their officers, directors, owners, or agents that are contrary to the statements made in our FDD or to the terms and conditions of this Agreement.

20.6 Financial Performance Representations

Except as may be stated in the FDD, you and your Owners acknowledge that neither we, nor any of our Affiliates, nor any of our or our Affiliates' officers, agents, employees, or representatives have made any representation to you, express or implied, as to the historical revenues, earnings, or profitability of any Business or the anticipated revenues, earnings, or profitability of the Business subject to the Franchise or any other business operated by us, our licensees, our franchisees, or our Affiliates. In entering into this Agreement, you are not relying upon any information furnished by us or our representatives other than the information contained in this Agreement and the FDD. Any information you have acquired from other SKY ZONE franchisees regarding their sales, profits or cash flows is not information obtained from us, and we make no representation about that information's accuracy. YOU AND EACH OF YOUR OWNERS WARRANT AND REPRESENT THAT HE/SHE HAS NOT RELIED UPON

ANY GUARANTEES CONCERNING REVENUE PROFIT OR THE SUCCESS OF THIS FRANCHISE IN DECIDING TO SIGN THIS AGREEMENT.

20.7 No Waiver Or Disclaimer Of Reliance In Certain States

<u>.</u> The following provision applies only to franchisees and franchises that are subject to state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

No statement, questionnaire, or acknowledgment signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the franchise.

[Signature page follows]

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the Effective Date.

FRANCHISOR	<u>FRANCHISEE</u>
SKY ZONE FRANCHISE GROUP, LLC	(IF ENTITY):
By:	By:
Name:	Name:
Title:	Title:
Date*:	Date:
*This is the Effective Date	
	By:
	Name:Title:
	Date:
	(IF INDIVIDUALS):
	[Signature]
	[Print Name]
	Date:
	[Signature]
	[Print Name]
	Date:
	[Signature]
	[Print Name]
	Date:
	[Signature]

Date:		

APPENDIX A

FRANCHISEE-SPECIFIC TERMS

1	Effective Date. The Effective Date is:	
2.	Franchisee's Name:	
<u>32</u> .	Franchisee's Address and Electronic Mail	Address For Notices:
<u>43.</u>	Form of Entity of Franchisee.	
+⊇.	•	Company Franchises was organized on
		<u>Company</u> . Franchisee was organized on s of the State of . Its
	Federal Employer Identification Number	is It has not
	conducted business under any name other tha	n its corporate or company name.
	Name of Each Director/Officer/Managing Member	Position(s) Held
		neral or limited partnership formed on of the State of . Its
	, under the laws Federal Employer Identification Number is	
	business under any name other than its partne	ership name.
	Name of Each General Partner	

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

DMS_US.363658954.1

<u>5</u> .	Owners.		
	4A. Franchisee and each of its Owners represents complete and accurate list of all Owners of any including the full name and mailing address of each of and extent of each Owner's interest in Franchisee. From ownership interest, represents and warrants that each legal and beneficial owner of his ownership interest liens, restrictions, agreements and encumbrances of required or permitted by this Agreement.	interest whatsoever in Franchisee, Owner, and fully describes the nature franchisee and each Owner as to his ch Owner is the sole and exclusive in Franchisee, free and clear of all	
	Owner's Name and Address	Percentage and Nature of Ownership Interest	
	2 <u>B</u> . Control Group. You represent and warrant that the following Owner or group of Owners has, directly or indirectly, 51% or more ownership interest in you and voting control over its ownership interests in you and constitutes your Control Group as described in Section 2.4 (Control Group) of the Franchise Agreement.		
	Owner's Name and Address	Percentage and Nature of Ownership Interest	

[Appendix A Signature Page Follows]

This **Appendix A** is deemed accepted and made a part of the Franchise Agreement as of the Franchise Agreement's Effective Date.

FRANCHISOR	<u>FRANCHISEE</u>
SKY ZONE FRANCHISE GROUP, LLC	(IF ENTITY):
By:	By:
Name:	Name:
Title:	Title:
Date:	Date:
	By:
	Name:
	Title:
	Date:

(IF INDIVIDUALS): [Signature] [Print Name] Date:_____ [Signature] [Print Name] Date:_____ [Signature] [Print Name] Date:_____ [Signature] [Print Name]

Date:_____

APPENDIX B

OWNERS' PERSONAL GUARANTY OF FRANCHISEE'S OBLIGATIONS ("GUARANTY")

Each of the undersigned waives: (a) acceptance and notice of acceptance by Franchisor of the foregoing undertakings; (b) notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed; (c) protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed; (d) any right hethe undersigned may have to require that an action be brought against Franchisee or any other person as a condition of liability; (e) notice of any amendment to the agreement; and (f) any and all other notices and legal or equitable defenses to which hethe undersigned may be entitled.

Each of the undersigned consents and agrees that: (i) their direct and immediate liability under this guaranty Guaranty shall be joint and several; (ii) they shall render any payment or performance required under the Agreement upon demand if Franchisee fails or refuses to do so punctually; (iii) such liability shall not be contingent or conditioned upon pursuit by Franchisor of any remedies against Franchisee or any other person; and (iv(iv) Franchisor may proceed against the undersigned and Franchisee jointly and severally, or Franchisor may, at its option, proceed against the undersigned without having commenced any action or having obtained any judgment against Franchisee; (v) such liability shall not be diminished, relieved or otherwise effected by any extension of time, credit or other indulgence which the Franchisor may from time to time grant to Franchisee or to any other person including, without limitation, the acceptance of any partial payment or performance or the compromise or release of any claims, none of which shall in any way modify or amend this guaranty Guaranty, which shall be continuing and irrevocable until satisfied in full; and (vi) the undersigned agrees to pay all reasonable attorneys' fees and all costs and other expenses incurred in any collection or attempt to collect amounts due pursuant to this undertaking or any negotiations relative to the obligations hereby guaranteed or in enforcing this undertaking against the undersigned.

Sky Zone Franchise Group, LLC

April 2025 FDD

Franchise Agreement

[City, State]

It is further understood and agreed by the undersigned that the provisions, covenants, and conditions of the <u>guarantyGuaranty</u> will inure to the benefit of our successors and assigns.

This guaranty Guaranty shall be governed by the governing law provisions set forth in Section 19.12 (Governing Law) of the Agreement and all disputes related to it shall be resolved in accordance with the dispute resolution provisions set forth in Sections 19.11 (Dispute Resolution), 19.13 (Consent to Jurisdiction), 19.14 (Waiver of Punitive Damages, Jury Trial, and Class Actions), and 19.15 (Limitation of Claims) of the Agreement.

Each of the undersigned represents and warrants that, if no signature appears below for such undersigned's spouse, such undersigned is either not married or, if married, is a resident of a state which does not require the consent of both spouses to encumber the assets of a marital estate.

[Signature Page Follows]

DMS_US.363658954.1

GUARANTOR(S):	
Signature:	Signature:
Print Name:	Print Name:
Signature:	Signature:
Signature.	<u>Signature.</u>
Print Name:	Print Name:
guaranty given herein by his/her spouse. Such to Guarantor's performance of this Guaranty. PERCENTAGE OF OWNERSHIP	ntor indicated below, acknowledges and conserconsent also serves to bind the assets of the mari
	consent also serves to bind the assets of the mari
guaranty given herein by his/her spouse. Such to Guarantor's performance of this Guaranty. PERCENTAGE OF OWNERSHIP	Consent also serves to bind the assets of the mari
guaranty given herein by his/her spouse. Such to Guarantor's performance of this Guaranty. PERCENTAGE OF OWNERSHIP	Consent also serves to bind the assets of the mari
guaranty given herein by his/her spouse. Such to Guarantor's performance of this Guaranty. PERCENTAGE OF OWNERSHIP	GUARANTOR (S) (Signature)
guaranty given herein by his/her spouse. Such to Guarantor's performance of this Guaranty. PERCENTAGE OF OWNERSHIP	GUARANTOR (S) (Signature) (Print Name)

		(Print Name)
	Name of Guarantor	Name of Guarantor
	Name of Guarantor's Spouse	Name of Guarantor's Spouse
	Signature of Guarantor's Spouse	Signature of Guarantor's Spouse
	Name of Guarantor	Name of Guarantor
	Name of Guarantor's Spouse	Name of Guarantor's Spouse
l	Signature of Guarantor's Spouse	Signature of Guarantor's Spouse

APPENDIX C

OWNER PERSONAL COVENANTS REGARDING CONFIDENTIALITY AND NON-COMPETITION ("AGREEMENT")

In conjunction with your investment in	("Franchisee") a
, you (" Owner " or " y	you "), acknowledge and agree as follows:

- 1. Franchisee owns and operates, or is developing, a Business located or to be located in the Protected Territory pursuant to a franchise agreement ("Franchise Agreement") with Sky Zone Franchise Group, LLC, which Franchise Agreement requires persons with legal or beneficial ownership interests in Franchisee under certain circumstances to be personally bound by the confidentiality and noncompetition covenants contained in the Franchise Agreement. All capitalized terms contained herein shall have the same meaning set forth in the Franchise Agreement.
- 2. You own or intend to own a legal or beneficial ownership interest in Franchisee and acknowledge and agree that your execution of this agreement ("Agreement") is a condition to such ownership interest and that you have received good and valuable consideration for executing this Agreement. We may enforce this Agreement directly against you and Your Owners (as defined below).
- 3. If you are an Entity, all persons who have a legal or beneficial interest in you ("Your Owners") must also execute this Agreement.
- 4. You and Your Owners, if any, may gain access to parts of our Confidential Information (as defined in Section 8.1 (Confidential Information) of the Franchise Agreement) as a result of investing in Franchisee. The Confidential Information is proprietary and includes our trade secrets. You and Your Owners hereby agree that while you and they have a legal or beneficial ownership interest in Franchisee and thereafter you and they: (a) will not use the Confidential Information in any other business or capacity (such use being an unfair method of competition); (b) will exert best efforts to maintain the confidentiality of the Confidential Information; and (c) will not make unauthorized copies of any portion of the Confidential Information disclosed in written, electronic or other form. If you or Your Owners cease to have an interest in Franchisee, you and Your Owners, if any, must deliver to us any such Confidential Information in your or their possession.
- 5. You specifically acknowledge that you will receive valuable, specialized training, Confidential Information, and other proprietary and specialized information and knowledge that provide a valuable, competitive advantage in operating a trampoline park. You further acknowledge that we would be unable to protect the Confidential Information against unauthorized use or disclosure or to encourage the free exchange of ideas and information among our franchisees if you were permitted to hold interests in or perform services for a Competitive Business (as defined in Section 1.2 (Definitions) of the Franchise Agreement), and we have granted you and the Franchisee certain rights

under the Franchise Agreement in consideration of, and in reliance upon, your agreement to deal exclusively with us. You therefore covenant that during the term of the Franchise Agreement (except as otherwise approved in writing by us), you, Your Owners, and youyour and their Immediate Families (as defined in Section 1.2 of the Franchise Agreement)immediate family members shall not, either directly, indirectly or through, on behalf of, or in conjunction with any person or legal entity:

- (a) Have any direct or indirect interest as a disclosed or beneficial owner in a Competitive Business, wherever located or operating;
- (b) Perform services as a director, officer, manager, employee, consultant, lessor, representative, agent or otherwise for a Competitive Business, wherever located or operating:
- 1.—Divert or attempt to divert any present or prospective business or customer of related to the Business or any other franchisee's Business to any Competitive Business, by direct or indirect inducement or otherwise, or dodivert or perform, directly or indirectly, any other act injurious or prejudicial to attempt to divert the employment of any employee of ours, any of our affiliates, or another franchisee or developer, to any Competitive Business;
- <u>Engage in any other activity which might adversely affect</u> the goodwill associated withof the Marks and the System; or
- 2. Recruit, employ or seek to employ any person who is at that time, or has been within the past six months, employed by us or one of our affiliates, or otherwise directly or indirectly induce such person to leave his or her employment; or
- 3. Own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in (as owner or otherwise) any Competitive Business
 - (e) Solicit, interfere, or attempt to interfere with our or our Affiliates' relationships with any customers, vendors, consultants, or other franchisees.
- 6. You covenant that, except as otherwise approved in writing by us, you and Your Owners and your and their immediate family members shall not, for a continuous, uninterrupted period of two years commencing upon the date of (a) a Transfer (as defined in Section 1.2 of the Franchise Agreement) permitted under Section 13 (Transfer) of the Franchise Agreement, (b) expiration of the Franchise Agreement, (c) termination or non-renewal of the Franchise Agreement (regardless of the cause for termination or non-renewal), or (d) a final order of a duly authorized arbitrator, panel of arbitrators, or a court of competent jurisdiction (after all appeals have been taken) with respect to any of the foregoing or with respect to enforcement of this Paragraph, either directly or indirectly, for yourself,

Your Owners, or your Immediate Family, or their immediate family members or through, on behalf of, or in conjunction with any person or legal entity, own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in (as owner or otherwise) any Competitive Business that is, or is intended to be, located (a) at the Location of the Business, (b) within fifteen (15) miles of the Location, or (c) fifteen (15) miles of any Business in operation or under construction as of the date that you are required to comply with this Paragraph 6Section 6. Additionally, neither you nor Your Owners or any of your or their immediate family members shall: (x) solicit, interfere, or attempt to interfere with our or our Affiliates' relationships with any customers, vendors, consultants, or other franchisees; or (y) engage in any other activity that might adversely affect the goodwill of the Marks and the System. You agree and acknowledge that the two-year period of this restrictionthese restrictions shall be tolled during any time period in which you are in violation of this restriction.

7. You and each of Your Owners expressly acknowledge the possession of skills and abilities of a general nature and the opportunity to exploit such skills in other ways, so that enforcement of the covenants contained in Sections 5 and 6 will not deprive any of you of your personal goodwill or ability to earn a living. If any covenant herein which restricts competitive activity is deemed unenforceable by virtue of its scope or in terms of geographical area, type of business activity prohibited and/or length of time but could be rendered enforceable by reducing any part or all of it, you and we agree that it will be enforced to the fullest extent permissible under Applicable LawLaws and public policy. We may obtain in any court of competent jurisdiction any injunctive relief, including temporary restraining orders and preliminary injunctions, against conduct or threatened conduct for which no adequate remedy at law may be available or which may cause it irreparable harm. You and each of Your Owners acknowledges that any violation of Sections 4, 5 or 6 hereof would result in irreparable injury for which no adequate remedy at law may be available. If we file a claim to enforce this Agreement and prevail in such proceeding, you agree to reimburse us for all its costs and expenses, including reasonable attorneys' fees.

IN WITNESS WHEREOF, each of the undersigned has hereunto affixed his signature, under seal, as of the Effective Date of the Franchise Agreement.

OWNER If an individual:	If a corporation, partnership, limited liability company or other legal entity:
(Signature)	
(Print Name)	

TO BE NEGOTIATED WITH LANDLORD

APPENDIX D

ADDENDUM TO LEASE

This	addendum ("Addendum") is executed as of this	day	of
	, by and between ("Franchisee or "Tenant") and ("Landlord")	as	an
addendum to	the lease (as amended, renewed, and/or extended from time to time,	"the Leas	e")
for the premi	ises located at		
state of	, (the " Premises "), dated as of	•	

WHEREAS, Franchisee has entered into a franchise agreement ("**Franchise Agreement**") with Sky Zone Franchise Group, LLC ("**Franchisor**") for the operation of a Sky Zone Park ("**Business**") at the Premises, and as a requirement thereof, the Lease for the Premises must be subject and subordinate to the provisions contained in this Addendum; and WHEREAS, Landlord and Franchisee agree that in the event of any conflict, the terms contained herein shall supersede any terms to the contrary set forth in the Lease.

- 1. Landlord shall deliver to Franchisor a copy of any notice of default or termination of the Lease at the same time such notice is delivered to Franchisee.
- 2. Franchisee hereby assigns to Franchisor, with Landlord's irrevocable and unconditional consent, all of Franchisee's rights, title and interests to and under the Lease upon any termination or non-renewal of the Franchise Agreement, but no such assignment shall be effective unless: (a) the Franchise Agreement is terminated or expires without renewal; (b) Franchisor has exercised its option to purchase, if any, under the Franchise Agreement; and (eb) Franchisor notifies the Franchisee and Landlord in writing that Franchisor assumes Franchisee's obligations under the Lease as of the date of such assumption. Landlord acknowledges that Franchisor may elect to assume the Lease in the name of an affiliate of Franchisor. In no event shall any assumption of the Lease by Franchisor or Franchisor's affiliate release Tenant under the Lease or any guarantor of Tenant under any guaranty agreement executed in connection with the Lease.
- 3. Franchisor shall have the right, but not the obligation, upon giving written notice of its election to Franchisee and Landlord, to cure any breach of the Lease within five (5) days following expiration of any notice and cure period afforded to Tenant under the Lease (or, if no notice and cure period, within five (5) days), and, if so stated in the notice, to also succeed to Franchisee's rights, title and interests thereunder.
- 4. The Lease may not be modified, amended, supplemented, renewed, extended or assigned by Franchisee without Franchisor's prior written consent.
- 5. Franchisee shall have the right to assign the Lease or sublet the Premises to <u>Franchisor or an affiliate of</u> Franchisor without Landlord's approval, and any assignment or subletting provisions set forth in the Lease shall not apply to any such assignment or subletting.

Sky Zone Franchise Group, LLC
April 2025 FDD
Franchise Agreement
City, State

DMS_US.363658954.1

- Franchisee and Landlord acknowledge and agree that Franchisor shall have no liability or obligation whatsoever under the Lease unless and until Franchisor assumes the Lease in writing pursuant to Section 2 or 3 above.
- If Franchisor (or Franchisor's affiliate) assumes the Lease as provided above, Franchisor may, without Landlord's consent, further assign the Lease to a franchisee of Franchisor to operate the Premises, provided that the proposed franchisee has met all of Franchisor's applicable program criteria and requirements and has executed Franchisor's standard franchise agreement. Landlord agrees to execute such further documentation to confirm its consent to the assignment permitted under this Exhibit as Franchisor may reasonably request. Upon such assignment to a franchisee of Franchisor, Franchisor and, if applicable, Franchisor's affiliate, shall be released from any further liability under the terms and conditions of the Lease.
- 8. Landlord and Franchisee hereby acknowledge that Franchisee has agreed under the Franchise Agreement that Franchisor and its employees or agents shall have the right to enter the Premises for certain purposes. Landlord hereby agrees not to interfere with or prevent such entry by Franchisor, its employees or agents. Landlord and Franchisee hereby further acknowledge that in the event the Franchise Agreement expires (without renewal) or is terminated, Franchisee is obligated to take certain steps under the Franchise Agreement to de-identify the Premises. Landlord agrees to permit Franchisor, its employees or agent, to enter the Premises and remove signs (both interior and exterior), decor and materials displaying any marks, designs or logos owned by Franchisor, provided Franchisor shall, at its expense, repair any damage to the Premises as a result thereof. In no event shall Landlord charge Franchisor rent for Franchisor's exercise of rights pursuant to this paragraph.
- Landlord agrees to allow Franchisee to remodel, equip, paint and decorate the interior and exterior of the Premises pursuant to the terms of the Franchise Agreement and any successor Franchise Agreement under which Franchisee may operate the Business at the Premises.
- 10. Landlord agrees that during and after the term of the Lease, it will not disclose or use Franchisor's Confidential Information (as defined below) for any purpose other than for the purpose of fulfilling Landlord's obligations under the Lease. "Confidential Information" as used herein shall mean all non-public information and tangible things, whether written, oral, electronic or in other form, provided or disclosed by or on behalf of Franchisee to Landlord, or otherwise obtained by Landlord, regarding the design and operations of the business located at the Premises, including, without limitation, all information identifying or describing the floor plan, equipment, furniture, fixtures, wall coverings, flooring materials, shelving, decorations, trade secrets, trade dress, "look and feel," layout, design, menus, recipes, formulas, manner of operation, suppliers, suppliers, and all other products, goods, and services used, useful or provided by or for Franchisee on the Premises. Landlord acknowledges that all Confidential Information belongs exclusively to Franchisor. Landlord agrees that should it breach or threaten to breach this provision of this Exhibit, Franchisor will suffer irreparable damages and Franchisor's remedy at law will be inadequate. Therefore, if Landlord threatens or breaches this provision, Franchisor shall be entitled to all remedies available to Franchisor at law or in equity, including, without limitation, injunctive relief.

Sky Zone Franchise Group, LLC April 2025 FDD Franchise Agreement

[City, State]

- 11. Landlord agrees that: (a) Franchisor has solely granted to Franchisee the right to use Franchisor's proprietary trade name, trademarks, service marks logos, insignias, slogans, emblems, symbols, designs and indicia of origin (collectively the "Marks") at the Premises under the terms of the Franchise Agreement; and (b) Franchisor has not granted any rights or privileges to Landlord to use the Marks at the Premises or anywhere else; and (c) Landlord's unauthorized use of the Marks during or after the term of the Lease shall cause irreparable harm to Franchisor and Franchisor's remedy at law will be inadequate. Therefore, if Landlord threatens or actually breaches this provision, Franchisor shall be entitled to all remedies available to Franchisor at law or in equity, including, without limitation, injunctive relief.
- 12. Copies of all notices required or permitted hereby or by the Lease shall also be sent to Franchisor at the following addresses, or such other addresses as Franchisor shall specify by written notice to Landlord.

Sky Zone Franchise Group, LLC 86 N. University Avenue, Suite 350 Provo, Utah 84601 Attn: Franchise Operations

With copies at all times to:

Sky Zone Legal Department: legal@skyzone.com

13. Under the Franchise Agreement, any lease for the location of a Business is subject to Franchisor's approval. Accordingly, the Lease is contingent upon such approval.

14

And with further copies to:

Arnall Golden Gregory LLP
171 17th Street, Suite 2100
Atlanta, Georgia 30363
Attention: Jonathan L. Neville, Esq.

- 13. Any exclusive use rights, option terms or other provisions of the Lease which may be stated as personal to Tenant and not to survive any assignment, sublet, or transfer shall nonetheless survive any assignment or sublease to Franchisor, Franchisor's affiliate and/or any franchisee of Franchisor..
- 14. For so long as Franchisor and/or Franchisor's affiliate is the named Tenant under the Lease (and only if Franchisor elects for Lease assumption), notwithstanding anything in the Lease to the contrary or in conflict: (a) no radius provisions within the Lease shall apply, (b) in no event shall any direct or indirect change of control of Franchisor, whether by stock exchange, equity offering or otherwise, constitute a transfer, assignment or subletting under the Lease

giving rise to Landlord approval, and (c) Franchisor or Franchisor's affiliate shall have the unfettered right (without Landlord approval) to transfer/assign the Lease to any affiliate of Franchisor or pursuant to merger, consolidation, reorganization, or sale of all or materially all of Franchisor's (or Franchisor's affiliate's) assets.

- <u>15</u>. Franchisor, along with its successors and assigns, is an intended third-party beneficiary of the provisions in this Addendum.
- 1516. References to the Lease and the Franchise Agreement include all amendments, addenda, extensions and renewals to such documents.
- 1617. References to the Landlord, Franchisee and Franchisor include the successors and assigns of each of the parties.

[Signature blocks on next page]

DMS_US.363658954.1

VITNESS the execution hereof under seal.	
	LANDLORD:
	Date:
	FRANCHISEE:
	Date:

APPENDIX E

PROTECTED TERRITORY AND PREMISES LOCATION

The Protected Territory	includes the following zip codes:	and
surrounding areas as ou	tlined in Appendix E-1. See attached map attached as A	Appendix E-1.
The	<u>Premises</u> <u>Location</u>	is:
Appendix E will be up by us. The Premises Low months of the Effective	a is <u>designated or accepted</u> at the time this Agreemed dated when a <u>Premises Location</u> has been designated by <u>cation</u> must be designated and your lease <u>must be signated</u> . Date of this Agreement and must be within the Protect identified, we may revise the Protected Territory.	y you and accepted ned within nine (9)
FRANCHISEE:	FRANCHISOR:	
	SKY ZONE FRANCHISE G	ROUP, LLC:
Ву:	By:	
Its:	Its:	
Date:	Date:	

APPENDIX E-1

MAP OF PROTECTED TERRITORY AND PREMISES LOCATION

DMS_US.363658954.1

APPENDIX F

ACKNOWLEDGMENT ADDENDUM TO THE SKY ZONE FRANCHISE AGREEMENT

THIS DOCUMENT SHALL NOT BE SIGNED BY YOU, AND WILL NOT APPLY, IF THE OFFER OR SALE OF THE FRANCHISE IS SUBJECT TO THE STATE FRANCHISE REGISTRATION/DISCLOSURE LAWS IN THE STATES OF CALIFORNIA, HAWAII, ILLINOIS, INDIANA, MARYLAND, MICHIGAN, MINNESOTA, NEW YORK, NORTH DAKOTA, RHODE ISLAND, SOUTH DAKOTA, VIRGINIA, WASHINGTON, OR WISCONSIN.

DO NOT SIGN THE ACKNOWLEDGEMENT IF THE FRANCHISE IS TO BE OPERATED IN, OR YOU ARE A RESIDENT OF, MARYLAND.

As you know, you and we are entering into a Franchise Agreement for the operation of a SKY ZONE franchise. The purpose of this Acknowledgment Addendum is to determine whether any statements or promises were made to you that we have not authorized or that may be untrue or inaccurate. Please review each of the following questions carefully and provide honest responses to each question.

Acknowledgments and Representations*.

- 1. Did you receive a copy of our Franchise Disclosure Document (and all exhibits and attachments) ("FDD") at least (a) 14 calendar days prior to signing the Franchise Agreement, or (b) if you are a resident of Iowa, New York or Oklahoma, at the earlier or the first personal meeting or 10 business days before the execution of the franchise or other agreement or payment of any consideration, or if you are a resident of Michigan, Oregon or Wisconsin, at the earlier of 10 business days before the execution of any binding agreement or payment of any consideration, or (d) if you are a resident of Washington, 14 calendar days prior to payment or execution of the Franchise Agreement? Check one:

 Yes. Po.
- 2. Have you studied and reviewed carefully our FDD and Franchise Agreement? Check one:

 Yes. □ No.
- 3. Is the name, address and phone number of any broker and each of our employees or representatives who was involved in offering you this franchise listed on the FDD receipt you signed (or on any updated receipt we provided to you)? Check one: ☐ Yes. ☐ No.
- 4. Do you understand that that, unless there exists a currently effective Multi Unit Development Agreement between you and us, the franchise granted is for the right to operate a single SKY ZONE facility at the authorized location only and includes no exclusive area or protected territory, and that we and our affiliates have the right to issue franchises or operate competing businesses for or at locations, as we determine, near your authorized location? Check one: □ Yes. □ No.

5.	post-term) listed in Section 17 of the appropriate remedy to protect the intercovenant(s)? Further, do you understand covenants in defined broadly in Section in violation of the covenants by those he	I by the non compete covenants (both in term and Franchise Agreement and that an injunction is an rests of the SKY ZONE system if you violate the that the term "you" for purposes of the non compete 17 of the Franchise Agreement, such that any actions olding any interest in the franchisee entity may result of the Franchise Agreement? Check one: Yes.
6.	negative impact on the trampoline park	onomic crisis and financial situation could have a industry, the SKY ZONE franchise system and your the economic situation may worsen? Check one:
	If you answered "No" to questions necessary):	1-8, please explain (attached additional sheets if
7. -	Was any oral, written or visual claim or disclosures in the FDD? Check one: ☐ Y	r representation made to you which contradicted the
TH TR AN	IAT YOU HAVE CONSIDERED EACH UTHFULLY TO THE ABOVE QUESTI ISWER, CONTINUE ON A SEPARATE S OTE: IF THE RECIPIENT IS A (CORPORATION, PARTNERSHIP, LIMITED ENTITY, EACH OF ITS OWNERS MUST
Sig	ened:	Signed:
Pri	nt Name:	Print Name:
Dat	te:	Date:
Sig	gned:	Signed:
_	nt Name:	Print Name:
	te:	Date:
AP	PROVED ON BEHALF OF SKY ZON	E FRANCHISE GROUP. LLC:
	ened:	,
U	nt Name:	•
	te:	
Apri Fran	Zone Franchise Group, LLC 12 2025 FDD 1 chise Agreement 2 State	

Appendix E-2

*Such representations are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Illinois Franchise Disclosure Act or under the Maryland Franchise Registration and Disclosure Law, to the extent applicable. In addition, except to the extent we have negotiated changes to the Franchise Agreement and/or Multi-Unit Development Agreement that differ from the FDD, nothing in this Acknowledgement Addendum or in any related agreement is intended to disclaim representations made in Sky Zone Franchise Group, LLC's 2024 FDD that was provided to you.

80103924v1

DMS_US.363658954.1

EXHIBIT B TO THE DISCLOSURE DOCUMENT

MULTI-UNIT DEVELOPMENT AGREEMENT

${\bf SKY\ ZONE}^{\underline{e}}_{\underline{\ }}$ MULTI-UNIT DEVELOPMENT AGREEMENT

[NAME OF DEVELOPER]
[NAME OF DEVELOPMENT AREA]

1.	DEVELOPMENT AREA 1 Preambles and Grant of Development F	≀ights
2.	≟ GRANT OF DEVELOPMENT RIGHTS	1
2. 3.	TERM	
3.		
4. —	INITIAL SERVICES AND ONGOING OBLIGATIONS.	2
5. —	OUR RESERVATION OF RIGHTS	2
6. —	DEVELOPMENT FEE	2
7. —	EXECUTION OF FRANCHISE AGREEMENTS	2
8.	OWNERSHIP; CONTROL GROUP; RESPONSIBLE PERSON PERSONAL GUARANTY	3
9.	DEFAULT AND TERMINATION	
10. —	FRANCHISE AGREEMENTS MAY NOT BE AFFECTED	
11. —	FUTURE DEVELOPMENT	4
12. —	COMPLIANCE WITH APPLICABLE LAWS	
13.	YOUR NON COMPETITION OBLIGATIONS	
10.	13.1	
	A. Preambles	
		•
	C. Protected Territories	4
	D. Liquidity and Working Capital Availability	
	E. Reports	
	F. No Ongoing Obligations	
	G. Compliance with Applicable Laws	2
2.	Exercise of Development Rights	;
	A. Execution of Franchise Agreements	
	B. Timing for Executing Franchise Agreements	3
	C. Proposed Sites for Businesses	
	E. Failure to Comply with Development Schedule	
	D. Development Schedule Compliance E. Failure to Comply with Development Schedule F. Future Development	
	- T didio Dottolopillotti	
3.	Our Reservation of Rights	4
4.	Term	į
<u>5</u> .	Restrictive Covenants During Term Terms	
≅.	13.2 After Term 5	
	13.3 Other	
	i	
	ne Franchise Group, LLC	
	25 FDD nit Development Agreement	

[City, State]

<u>i</u>

TABLE OF CONTENTS (continued)

		Page
	A. Competitive Business	5
	B. Covenants-5 Against Competition	6
	13.4 Exception 5	
	13.5—Injunctive	
	C. Covenants from Others	6
	D. Exceptions	6
6.	Transfers	6
	A. Transfers By Us	6
	B. Transfers By You	6
7.	Confidential Information; Innovations	8
	A. Confidential Information	8
	B. Return or Destruction Upon Termination	8
	C. Innovations	9
	<u>D. Equitable</u> Relief	<u>69</u>
14. —	NO RIGHTS TO DEVELOPMENT AREA UPON TERMINATION	
15. —	ASSIGNMENT BY US	6
16. —	ASSIGNMENT BY YOU	6
17.	SEVERABILITY	6
18. –	WAIVERS	
19. _	DISPUTE RESOLUTION	
	19.1 Mediation 6	
	19.2	
8.	Development Fee	9
9.	Ownership; Control Group; Responsible Person; Personal Guaranty	9
<u> </u>		9
	A. Ownership and Control Group B. Personal Guaranty C. Responsible Person	10
	C. Responsible Person	10
	D. Accuracy of Appendix A	10
<u>10.</u>	Termination of Agreement	10
	A. Default; Events of Termination	10
	B. Cross-Termination of Agreement C. Effects of Termination or Expiration	11
		11
	D. Franchise Agreements	12
	E. Restrictive Covenants Not to Compete F. Survival of Covenants	12
	F. Survival of Covenants	13

TABLE OF CONTENTS (continued)

Page

<u>11.</u>	Relationship of the Parties; Indemnification A. Independent Contractors	
	A. Independent Contractors B. Indemnification	
10	Diamete Benefiting	
<u>12.</u>	Dispute Resolution A. Informal Resolution	
	B. Arbitration	7
	19.3C. Exceptions to Arbitration	
	19.4D. Prevailing Party's Fees	9
	19.5 <u>E.</u> Survival	9
	19.6F. Tolling of Statute of Limitations	
	19.7G. Performance to Continue	9
19.8 1	3.	Governing L
	<u>917</u>	
19.9 1	<u> 4. </u>	ent to Jurisdicti
_	9 <u>17</u>	
20. —	—WAIVER OF PUNITIVE DAMAGES, JURY TRIAL AND CLASS AC	TIONS
21.	NOTICES	
22.	FORCE MAJEURE	
23. —	MULTIPLE COPIES	
24.	ENTIRE AGREEMENT	
25. —	MODIFICATION	
26. —	NO THIRD PARTY BENEFICIARIES	
27.	OTHER FRANCHISEES/DEVELOPERS	
28. —	BINDING EFFECT	
29.	NO GUARANTEE OF SUCCESS	
<u>15.</u>	Waiver of Punitive Damages, Jury Trial, and Class Actions	
16.	Limitations of Claims	
<u>17.</u>	Miscellaneous A. Notices	
	C. Other Developers	
	D. Terrorist Acts	
	E. Joint and Several Obligations	

TABLE OF CONTENTS (continued)

		Page
F.	Severability	19
G.	Headings and Construction	19
<u>H.</u>	Waiver	19
Ī.	Costs and Attorneys' Fees	19
J.	Rights of Parties Are Cumulative	20
K.	Entire Agreement; Modification	20
L.	Binding Effect	20
M.	No Third-Party Beneficiaries	20
N.	Counterparts; Electronic Signatures	20

SKY ZONE MULTI-UNIT DEVELOPMENT AGREEMENT

This Multi-Unit Development Agreement (the "this "Agreement") is made and entered into as of the <u>last signature</u> date <u>listed on Appendix A (the "below (the "Effective Date") Sky Zone Franchise Group by and between SKY ZONE FRANCHISE GROUP</u>, LLC, a <u>Missouri limited liability company formed under Missouri law</u>, with its principal business address at 86 N. University Avenue, Suite 350, Provo, Utah 84601 (referred to in this Agreement as "SZFG," "we," "us," or "our"), and the person(s) or entity identified on <u>Appendix A</u> as the "Developer" with its principal place of business as set forth on <u>Appendix A</u> (referred to in this Agreement as "you" or "your").

WHEREAS, you have provided us with all financial information about your shareholders, partners, officers, directors, guarantors and other persons as we have requested; and

WHEREAS, you desire to develop, own and operate,

1. <u>Preambles and Grant of Development Rights.</u>

- A. <u>Preambles.</u>
- i. We grant, to persons or entities who we determine meet our qualifications, franchises (each a "Franchise") for the development and operation of indoor trampoline parks featuring trampoline attractions to be used for sports and recreational activities, and other active entertainment attractions (each a "Business") using the Sky Zone® service mark and other trademarks we authorize from time to time (the "Marks") and the system and methods under which Businesses are developed and operated (the "System").
- <u>Each Franchise is granted solely pursuant to a written franchise agreement and related documents and agreements signed by us and a franchisee (each a "Franchise Agreement") under which we grant a geographic area in which neither we nor any of our affiliates will establish or authorize any person or entity to establish another Business using the Marks and the System (a "Protected Territory").</u>
- We also grant, to persons or entities who we determine meet certain additional qualifications and who are willing to commit, the right to acquire multiple Franchises for the development and operation of Businesses within a defined area (the "Development Area") pursuant to an agreed upon schedule (the "Development Schedule").
- iv. You have requested that we grant you such rights, either through yourself or an affiliate in which your Control Group (as defined in Section 89 below) owns and controls fifty-one percent (51%) or moregreater interest (each an "Affiliated Entity"), multiple SKY ZONE trampoline parks (each, a "Business") in the development area specified in Appendix A (the "Development Area") and we are willing to do so in reliance on all of the information, representations, warranties

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

and acknowledgements you have provided to us in support of your request, and subject to the terms and conditions set forth in this Agreement.

NOW THEREFORE, in consideration of the mutual covenants contained herein, and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto, intending to be legally bound, do hereby mutually agree as follows:

1. Development Area. With respect to the Development Area specified in Appendix A, all political boundaries shall be considered fixed as of the date of this Agreement and shall not change for the purpose hereof, notwithstanding a political reorganization or change to such boundaries or regions. All street boundaries shall be deemed to end at the street center line unless otherwise specified above. In case of inconsistency between the written description above and any map attached to Appendix A, the written description of the Development Area controls.

2.

B. <u>Grant of Development Rights</u>.

Subject to the terms and conditions of this Agreement and provided that you and your Affiliated Entities are in full compliance with this Agreement and any other agreement (including any Franchise Agreement) between you or your Affiliated Entities and us, we We grant you the right, and you undertake the obligation, to establish and operate in the Development Area for your own accounteither yourself or through an Affiliated Entity, the number of Businesses Entities, to acquire Franchises to develop and operate Businesses within the Development Area and in accordance with the Development Schedule specified in the development schedule in Appendix A hereto (the "Development Schedule Rights"). This Agreement does not grant you any right to use the Sky Zone trademark or any other marks that we designate (the "The Development Rights are limited to the rights to acquire Franchises in accordance with and as described in this Agreement. Rights to develop and operate Businesses or to use the Marks") or the system and methods used in Businesses (the "System"), which are each granted only by the execution of apursuant to individual Franchise Agreements, and you agree that the Development Rights do not include any such rights. You also acknowledge that we grant rights only pursuant to the expressed provisions of written agreements and not in any other manner, including orally or by implication, innuendo, extension or extrapolation. Nothing in this Agreement provides you with us. You may not the right to develop theany Businesses at locations outside of the Development Area and you may not sublicense, subcontract, or delegate any of the rights Development Rights granted to you herein. In case of inconsistency between any written description in this Agreement and any map attached to Appendix A, the map attached to Appendix A controls.

You must have open and in continuous operation in the Development Area, pursuant to Franchise Agreements, the cumulative number of Businesses specified in the Development Schedule by each of the development deadlines specified in the Development Schedule (the "Development Deadlines"). It is of material importance to us that you timely perform all obligations under this Agreement and the Franchise Agreement for each Business.

2 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

You represent that you conducted your own independent investigation and analysis of the prospects for the establishment of Businesses within the Development Area, approve the Development Schedule as being reasonable and viable, and recognize that failure to achieve the results described in the Development Schedule will constitute material breach of this Agreement.

3. <u>Term.</u> Unless sooner terminated in accordance with Section 9, the term of this Agreement will expire on the earlier of: (i) the opening of the last Business listed in the Development Schedule; or (ii) the last Development Deadline set forth in Appendix A. There is no right to renew this Agreement.

4. Initial Services and

- C. Protected Territories. Within each Franchise Agreement executed pursuant to this Agreement, we will assign its corresponding Business a Protected Territory; however, these Protected Territories for your Businesses will be separate and distinct from the Development Area. Your Development Area automatically excludes the Protected Territories for any existing Businesses therein as of the Effective Date of this Agreement, regardless of whether such Protected Territories are defined, described, or otherwise illustrated in any maps attached to Appendix A. If a third-party's development area is adjacent to your Development Area, you acknowledge and agree that the Protected Territories for the third-party's Businesses may extend into your Development Area depending on those Businesses' final locations. If multiple maps are attached to Appendix A as the Development Area, once you develop a specified number of Businesses within a specific map (which, if applicable, would be set forth in Appendix A) in accordance with this Agreement and sign a Franchise Agreement for each Business within such map, we shall have the right to grant Franchises (or authorize the grant of Franchises) to any other person or entity to own new Businesses to be located within such map but not within the Protected Territories set forth in the Franchise Agreements for your Businesses.
- D. Liquidity and Working Capital Availability. We have granted the Development Rights to you based, in part, on your representations to us, and our assessment of your levels of liquidity as of the Effective Date. You will ensure that, throughout the Term (as defined below), you will maintain sufficient liquidity and working capital reserves to meet your obligations under this Agreement. At our request, you will provide us with evidence of your liquidity and working capital availability.
- E. Reports. Upon our request, you agree to provide us with a report of your business activities concerning the Development Rights, including information about your efforts to find sites for Businesses in the Development Area and the status and projected opening date for each Business under development in the Development Area.
- F. No Ongoing Obligations. You acknowledge and agree that our initial service under this Agreement is solely to identify the Development Area and that we have no ongoing obligations such as training or operational assistance to you underpursuant to this Agreement. All ongoing and further obligations to you in opening your locations Businesses shall be provided pursuant to any Franchise Agreements between you (or your Affiliated Entities) and us.

5. Our Reservation of Rights. Although

3 Sky Zone 2024 MUDA<u>Franchise Group, LLC</u> DMS_US.363658145.1

G. Compliance with Applicable Laws. You must, at your expense, comply with all federal, state, city, municipal and local laws, ordinances, rules and regulations in the Development Area pertaining to the development of your Businesses.

2. **Exercise of Development Rights**

- ÷
- A. Execution of Franchise Agreements. You or an Affiliated Entity must execute our then-current form of Franchise Agreement for each Business to be established by you in the Development Area; provided, however, you will not be required to pay us any Initial Franchise Fees under such Franchise Agreements. Further, if you are in good standing at the time each Franchise Agreement is executed, the "Royalty Fee" in each such agreement shall be equal to six percent (6%) of the Gross Sales (as defined in such Franchise Agreement) of the Business. If any conflict arises between this Agreement and any individual Franchise Agreement as to any individual Business, the terms of such individual Franchise Agreement will control.
- B. Timing for Executing Franchise Agreements. Unless we otherwise specify a later time period, simultaneously with signing this Agreement, you or an Affiliated Entity must sign and deliver to us a Franchise Agreement (which will reflect that no Initial Franchise Fee shall be due) and related documents representing the first Franchise you are obligated to acquire under this Agreement. Thereafter, you or an Affiliated Entity must sign our then-current form of Franchise Agreement and related documents, the terms of which may differ substantially from the terms contained in the Franchise Agreement in effect as of the Effective Date of this Agreement, except for the Royalty Fee set forth in Section 2.A above, once we have approved a site for a Business and prior to your execution of a lease or otherwise securing possession of the site for such Business. The Franchise Agreement will govern the development and operation of the Business at the approved site identified therein.
- C. Proposed Sites for Businesses. You must give us all information and materials we request to assess each Business site you propose and to assess your (and your Affiliated Entity's) financial and operational ability to fund the development and operation of each proposed Business. We have the absolute right to disapprove any site or any Affiliated Entity: (1) that does not meet our criteria, or (2) if you or your Affiliated Entities are not then in compliance with any existing Franchise Agreements in effect with us. We will use our reasonable efforts to review and approve or disapprove any sites and any Affiliated Entity you propose within thirty (30) days after we receive all requested information and materials. Once we approve a proposed, you or your approved Affiliated Entity must sign a separate Franchise Agreement as described above. If you or your Affiliated Entity fail to do so within fifteen (15) days after we provide you with an execution copy of the Franchise Agreement, we may withdraw our approval for the site.
- D. Development Schedule Compliance. Each period described in the Development Schedule is a "Development Deadline." You or your Affiliated Entity must satisfy the obligations described on the Development Schedule (reflected on Appendix A) during and as of the end of each Development Deadline. The Development Schedule is not our representation, express or implied, that the Development Area can support, or that there are or will be sufficient sites for, the number of Businesses specified in the Development Schedule or during any
- 4 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

particular Development Deadline. Except for the final Business to be developed pursuant to the Development Schedule, we will count a Business toward the Development Schedule only if it is actually open and operating for full use by guests within the Development Area and substantially complying with the terms of its Franchise Agreement as of the end of the Development Deadline. However, a Business which is, with our approval or because of fire or other casualty, permanently closed during the last ninety (90) days of a Development Deadline, after having been open and operating for a period of at least ninety (90) days, will be counted toward the development obligations for the Development Deadline in which it closed, but not thereafter. We are relying on your representation that you conducted your own independent investigation and analysis of the prospects for the establishment of Businesses within the Development Area, approve the Development Schedule and its Development Deadlines as being reasonable and viable, and recognize that failure to comply with the Development Schedule will constitute a breach of this Agreement.

- Failure to Comply with Development Schedule. If you fail to comply with the E. Development Schedule as of the end of any Development Deadline, in addition to terminating this Agreement under Section 10 and asserting any other rights we have under this Agreement as a result of such failure, we may (but need not) elect to revoke our agreement under Section 3 not to grant similar development or franchise rights to others within the Development Area. Alternatively, and without waiving any rights afforded to us under this Agreement or any Franchise Agreement in which you or an Affiliated Entity own or hold any interest, we have the right, but no obligation, to refrain from exercising our termination rights in favor of granting you a written extension to the Development Schedule. Such an extension may, in our business judgment, be conditioned on any or all of the following: (a) a reduction in the size of the Development Area; (b) a modified Development Schedule (in terms of timing and/or number of Businesses to be opened); (c) your execution of our then-current form of general release; and (d) your execution of our then-current form of Development Agreement, which shall supersede and replace this Agreement and the terms of which may differ substantially from the terms contained herein. In addition, if we grant you a written extension, you may be required to pay to us an extension fee equal to half of the Development Fee (as defined below and set forth in Appendix A). Nothing obligates us to grant you an initial or any subsequent extension to the Development Schedule. We reserve the right to terminate this Agreement at any time if you fail to comply with its terms, including at the end of any unfulfilled extension period.
- F. Future Development. You recognize and acknowledge that this Agreement requires you to open Businesses in the future pursuant to the Development Schedule. You further acknowledge that the estimated expenses and investment requirements set forth in Items 6 and 7 of our Franchise Disclosure Document ("FDD") are subject to increase over time, and that future Businesses may require greater initial investment and operating capital requirements than those stated in the FDD provided to you prior to the execution of this Agreement. You must execute Franchise Agreements and open all required Businesses by the dates set forth on the Development Schedule, regardless of: (i) the requirement of any greater investment, (ii) the financial condition or performance of your prior Businesses, or (iii) any other circumstances, financial or otherwise. The foregoing will not be interpreted as imposing any obligation upon us to execute the Franchise Agreements under this Agreement if you have not complied with each and every condition necessary to develop the Businesses, if you do not meet our then-current requirements for Franchisees at the time you are otherwise expected to execute a Franchise Agreement, or if you (or your Affiliated Entities) are not in compliance with the terms of any Franchise Agreement in effect with us.

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

3. Our Reservation of Rights

. Except as described in Sections 1.C. and 2.E. and this Section 3 and provided that you are in full compliance with this Agreement and all Franchise Agreements and other agreements entered into with us (or any of our affiliates), we will not develop, own and operate, or license a third party to own and operate, or franchise a Business physically located within the Development Area while this Agreement is in effect. Notwithstanding the foregoing, we, our parent and our affiliates (and our and their respective successors and assigns, by purchase, merger, consolidation or otherwise) retain all rights with respect to the Marks, the System and Businesses anywhere in the world, and retain the right to engage in any business whatsoever, and any other right not expressly granted to you in this Agreement, including the right to:

- A. 5.1—Own and operate, and/or grant to others the right to own and operate, any trampoline based recreational facility ("Competitive Businesses") (as defined in Section 5 of this Agreement) or other businesses offering similar or identical products, and services, and classes and using the System or elements of the System: (i) under the Marks anywhere outside of the Development Area, or (ii) under names, symbols, or marks other than the Marks anywhere, including inside and outside of the Development Area and Protected Territories of your Businesses;
- B. 5.2 offer Offer to sell, or sell and distribute, any products or services under any trade names, trademarks, service marks or trade dress, including the Marks, anywhere through any distribution channels or methods, which may include (including the internet, retail stores, wholesale, and the Internet (gift cards, distribution or fulfillment centers, or any other existing or future form of electronic commerce); and regardless of the nature or location of the customers with whom such other businesses and distribution channels do business, even if such businesses sell products or services that are identical or similar to, and/or competitive with those that Businesses customarily sell;
- C. 5.3 acquire, be acquired by Acquire the assets or ownership interests of, merge, affiliate with or engage in any transaction with other businesses (whether competitive or not including Competitive Businesses), with units located anywhere or business conducted anywhere—and, even if such businesses are located in the Development Area or within the Protected Territories of your Businesses, and: (i) convertrebrand the other businesses to operate under the Sky Zone name, Marks, and System, (ii) franchise, license or create similar arrangements with respect to the other businesses and/or permit the other businesses to continue to operate under another name, and/or (iii) permit the businesses to operate under another name and convert existing Businesses (including yours) to such other name.;
- D. Be acquired or become controlled (regardless of the form of transaction) by a business providing products or services similar to those provided at Businesses, or by any other business, even if such business operates, franchises, and/or licenses Competitive Businesses; and
- E. Operate, or grant any third party the right to operate, any Businesses that we or our designees acquire as a result of the exercise of a right of first refusal or purchase right that we have under any agreements.

<u>4.</u> <u>Term</u>

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

. Unless sooner terminated in accordance with Section 10, the term of this Agreement will expire on the earlier of: (i) the execution of the Franchise Agreement for the final Business required to be developed pursuant to the Development Schedule; or (ii) the last day of the last Development Deadline set forth in Appendix A. There is no right to renew this Agreement (the "Term").

5. Restrictive Covenants During Terms

÷

- A. Competitive Business. The term "Competitive Business" for purposes of this Agreement means any business (other than a Franchise operated under a Franchise Agreement with us or our affiliates) operating as, or granting franchises or licenses to others to operate any business operating as, any family recreational center, including but not limited to:

 (i) a trampoline park; (ii) centers offering inflatable indoor bouncing, rock climbing, go carts, or miniature golf; (iii) active entertainment facilities; or (iv) similar family entertainment establishments. The term Competitive Business is not intended to include any centers specializing in gymnastics, which may contain trampolines.
- B. Covenants Against Competition. We have granted you the Development Rights in consideration of and in reliance upon your agreement to deal exclusively with us. Therefore, during the Term, you agree that, other than in accordance with this Agreement, neither you, your Affiliated Entities, nor any of your or their officers, directors, shareholders, members, partners or other Owners, nor any immediate family members of any of these individuals (collectively, "Bound Parties"), shall:
 - i. have any direct or indirect interest as a disclosed or beneficial owner in a Competitive Business, wherever located or operating:
 - <u>ii.</u> <u>perform services as a director, officer, manager, employee, consultant, lessor, representative, agent or otherwise for a Competitive Business, wherever located or operating;</u>
 - divert or attempt to divert any present or prospective business or customer related to the Businesses or any other franchisee's Business by direct inducement or otherwise, or divert or attempt to divert the employment of any employee of ours, any of our affiliates, or another franchisee or developer, to any Competitive Business;
 - <u>iv.</u> <u>engage in any other activity which might adversely affect the goodwill of the Marks and System; or</u>
 - v. directly or indirectly assist any immediate family member of any of the Bound Parties in engaging in any of the activities listed in the foregoing subsections.
- C. Covenants from Others. To the extent permitted by applicable law, you agree to obtain from the Bound Parties similar agreements regarding non-competition (Section 5.B. and Section 10.10.E.) and Confidential Information (Section 7). You must provide us with copies of

7 Sky Zone 2024 MUDAFranchise Group, LLC DMS_US.363658145.1

all such agreements upon our request. We may regulate the forms of agreement that you use and be a third-party beneficiary of the agreement with independent enforcement rights. You may not assume that any such form we provide you is or will be enforceable in a particular jurisdiction. You are solely responsible for obtaining your own professional advice with respect to the adequacy of the terms and provisions of such forms, but you must secure our approval, prior to use, of any form that contains variations from the form we provide.

D. Exceptions. Non-competition covenants related to Competitive Businesses do not apply to the ownership of shares of a class of securities that are listed on a public stock exchange or traded on the over-the-counter market and that represent less than five percent (5%) of such class of securities.

6. **Transfers**

÷

- A. <u>Transfers By Us. This Agreement inures to the benefit of us and our successors and assigns, and we have the right to transfer or assign all or any part of our interest, rights, privileges, duties and obligations hereunder to any person or legal entity without your approval.</u>
- B. Transfers By You. You acknowledge that we have granted you the Development Rights in reliance upon, and in consideration of, your business skills, financial capacity and other required qualifications. Accordingly, the rights and duties created under this Agreement are personal to you. Your interest in this Agreement, any of your rights under this Agreement, the Development Rights or any interest therein, and any direct or indirect ownership interest in you (if you are an Entity and regardless of its size), any approved Affiliated Entity, or any of your Owners (if such Owners are legal entities) may not be assigned, transferred, shared or divided, voluntarily or involuntarily, directly or indirectly, by operation of law or otherwise, in any manner (collectively, a "Transfer"), without our prior written consent. Any actual or intended assignment, transfer or sale made in violation of the terms of this Section 6.B. shall be null and void and shall constitute a material breach of this Agreement, constituting good cause for termination of this Agreement. Additionally, you may not pledge or encumber this Agreement, your Development Rights or an ownership interest in you or your Owners (to someone other than us) as security for any loan or other financing, unless: (1) we grant our prior written consent and (2) the lender agrees that its claims will be subordinate to all amounts you owe at any time to us or our affiliates.

The Development Rights may not be transferred. For purposes of further clarity, if you or your Affiliated Entities intend to transfer any open and operating Businesses to third-parties, any such transfers must be made in accordance with all applicable terms set forth in each Business' corresponding Franchise Agreement.

If you are an Entity, we will not unreasonably withhold our consent to a Transfer of any direct or indirect ownership interests in you which are less than or equal to forty-nine percent (49%) of your ownership interests, provided all of the following conditions are met before or concurrently with the effective date of the transfer:

& Sky Zone 2024 MUDAFranchise Group, LLC DMS_US.363658145.1

- i. <u>you are in compliance with the Development Schedule, this Agreement, and all Franchise Agreements in effect by and between you (or your Affiliated Entities) and us;</u>
- ii. the proposed transferee provides information to us sufficient for us to assess the transferee's business experience and aptitude, and if the Transfer is approved and completed, you will retain sufficient financial resources to satisfy the remaining requirements of the Development Schedule;
- iii. <u>neither you nor your Affiliated Entities have violated any provision of this Agreement or any other agreement with us or our affiliates during both the sixty (60) day period before you requested our consent to the Transfer and the period between your request and the effective date of the Transfer;</u>
- iv. <u>neither the transferee nor its owners or affiliates have an ownership interest</u> (direct or indirect) in or perform services for a Competitive Business;
- v. you pay us a transfer fee equal to two thousand five hundred dollars (\$2,500) for our administrative costs and expenses in connection with documenting and otherwise processing such Transfer, including reasonable legal fees;
- vi. your transferring Owners execute a consent to transfer, which will include a general release and a non-disparagement clause, in a form satisfactory to us, releasing us and our affiliates, and our and their respective officers, directors, members, shareholders, employees and agents, in their corporate and individual capacities, from any and all claims, including, without limitation, claims arising under federal, state and local laws, rules and ordinances;
- <u>vii.</u> <u>all individuals and entities comprising the proposed transferee who will have a ten percent (10%) or greater interest in you, or such other persons we designate, must execute or have executed a guaranty in the form we prescribe;</u>
- we have determined that the proposed Transfer will not adversely affect your ability to meet the Development Schedule or operate any of the open and operating Businesses developed pursuant to this Agreement;
- ix. the transferee's obligations under promissory notes, agreements, or security interests are subordinate to your obligations to pay any amounts due to us, our affiliates, and third-party vendors related to the operation of the Businesses and otherwise to comply with the Development Schedule and this Agreement; and
- x. your transferring Owners agree to comply with all applicable post-termination obligations, including by complying with the restrictive covenants found in Section 10.10.E. of this Agreement.

You acknowledge that the proposed transferee will be evaluated by us based on the same criteria as those currently being used to assess new developers and that the proposed transferee will be provided with the disclosures required by law. We may review all information

9 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

regarding the Development Rights that you give the transferee, and we may give the transferee copies of any reports or information that you have given us or that we have made regarding the Development Rights. Our consent to a Transfer pursuant to this Section 6.B. is not a representation of the fairness of the terms of any contract between your transferring Owners and the transferee, a guarantee of any prospects of success, or a waiver of any claims we have against your transferring Owners or of our right to demand the transferee's full compliance with this Agreement.

7. **Confidential Information; Innovations**

÷

- Confidential Information. All information furnished to you by us, whether orally or in writing, including this Agreement, plans, specifications, financial or business data or projections, all documents, data, information, materials, reports, proposals, procedures, financial information, models, drawings, programs, compilations, devices, methods, designs, techniques and specifications, inventions, know-how, processes, business plans, marketing techniques, supplier lists, supplier information, advertising strategies, operations, our trade secrets, or any other forms of business information, whether or not marked as confidential (collectively, the "Confidential Information"): (1) shall be deemed proprietary and shall be held by you in strict confidence; (2) shall not be disclosed or revealed or shared with any other person except to your employees or contractors who have a need to know such Confidential Information for purposes of this Agreement and who are under a duty of confidentiality no less restrictive than your obligations hereunder, or to individuals or entities specifically authorized by us in advance; and (3) shall not be used except to the extent necessary to exercise the Development Rights. You will protect the Confidential Information from unauthorized use, access, or disclosure in the same manner as you protect your own confidential or proprietary information of a similar nature and with no less than reasonable care.
- B. Return or Destruction Upon Termination. All Confidential Information will at all times remain our sole property. You agree to return to us or destroy, at our election, all Confidential Information in your possession or control and permanently erase all electronic copies of such Confidential Information promptly upon our request or upon the expiration or termination of this Agreement, whichever comes first. At our request, you will certify in writing signed by one of your officers that you have fully complied with the foregoing obligations.
- C. Innovations. You agree that, as between us, we or our affiliates own the Confidential Information, and that your rights to use the Confidential Information, derive solely from this Agreement. All improvements, developments, derivative works, enhancements, or modifications to any Confidential Information (collectively, "Innovations") made or created by you, your employees or your contractors, whether developed separately or in conjunction with us, shall be owned solely by us or our affiliates. You represent, warrant, and covenant that your employees and contractors are bound by written agreements assigning all rights in and to any Innovations developed or created by them to you. To the extent that you, your employees or your contractors are deemed to have any interest in such Innovations, you hereby agree to assign, and do assign, all right, title and interest in and to such Innovations to us. To that end, you shall execute, verify, and deliver such documents (including assignments) and perform such other acts (including appearances as a witness) as we may reasonably request for use in

10

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

applying for, obtaining, perfecting, evidencing, sustaining, and enforcing such ownership rights in and to the Innovations, and the assignment thereof. Your obligation to assist us with respect to such ownership rights shall continue beyond the expiration or termination of this Agreement. In the event we are unable for any reason, after reasonable effort, to secure your signature on any document needed in connection with the actions specified in this Section, you hereby irrevocably designate and appoint us and our duly authorized officers and agents as your agent and attorney in fact, which appointment is coupled with an interest and is irrevocable, to act for and on your behalf to execute, verify, and file any such documents and to do all other lawfully permitted acts to further the purposes of this Section with the same legal force and effect as if executed by you.

D. Equitable Relief. If you breach any of the provisions of this Section 7, we will be entitled to equitable relief, including in the form of injunctions and orders for specific performance, in addition to all other remedies available at law or equity. The obligations under this Section 7 shall survive any expiration or termination of the Agreement.

8. 6. Development Fee

. For the rights we grant you under the terms of this Agreement, you agree to pay us a development fee as specified in <u>Appendix A</u> (the "**Development Fee**") upon execution of this Agreement. The Development Fee is fully earned <u>by us</u> on receipt and is not refundable for any reason.

7. <u>Execution of Franchise Agreements</u>.

You or an Affiliated Entity must execute a separate Franchise Agreement in our then-current form (a "Franchise Agreement") for each Business to be established by you in the Development Area, except that you are not required to pay us any Initial Franchise Fee under such Franchise Agreement. Further, if you are in good standing at the time each subsequent Franchise Agreement is executed, we will modify each such Franchise Agreement so that the "Royalty Fee" in each such agreement shall be equal to 6% of the Total Gross Revenue (as defined in such Franchise Agreement) of the Business.

You or an Affiliated Entity must sign the Franchise Agreement for a Business the earlier of: (a) the date you or an Affiliated Entity sign a lease for the Business location; or (b) 60 days prior to the Development Deadline for such Business. If you fail to provide us with an executed then-current form of Franchise Agreement (as modified consistent with this Section 7) at least 60 days prior to Development Deadline, your failure will be deemed a material breach of this Agreement and we will have the right to terminate this Agreement as provided herein.

8.

9. Ownership; Control Group; Responsible Person; Personal Guaranty

A. Ownership and Control Group. Any person holding an ownership interest in you is an "Owner" for purposes of this Agreement. You acknowledge and agree that we have granted the rights in Section 1 above Development Rights to you; (a) based on you, your Owner 11

Sky Zone <u>2024 MUDA</u>Franchise Group, LLC DMS_US.363658145.1

or the group of Owners described in Appendix A hereof that has, directly or indirectly, fifty-one percent (51%) or more ownership interestinterests in you and voting control over its ownership interests in you (the "Control Group") and (b) based on the same ownership and voting control of such Control Group, in the same percentage as hereunder, in each Franchisee that executes a Franchise Agreement hereunderpursuant to this Agreement. If you are a business corporation, partnership, limited liability company or other legal entity (an "Entity"), the same Control Group must own a minimum of fifty-one percent (51%) in you and in any Affiliated Entity executing a Franchise Agreement as a SKY ZONESky Zone® franchisee (each a "Franchisee") pursuant to this Agreement. All such ownership interests of Owners in the Control Group shall be in the same percentages in your Entity under this Agreement as it shall be in any Franchisee Entity executing a Franchise Agreement. Furthermore, you acknowledge and agree that we have the right to approve in advance: (i) the ownership structure of each Franchisee executing a Franchise Agreement pursuant to this Agreement; and (ii) any changes to the persons or Entities who are your Owners.

- B. EachPersonal Guaranty. If you are an Entity, each Owner thatwho has a ten percent (10%) or greater than 10%-interest in you, if you are an Entityor such other persons as we designate (including Owners' spouses), must sign Appendix B to this Agreement (Owners' Personal Guaranty of Franchisee's Obligations). However, if we are entering into this Agreement totally or partially based on the financial qualifications, experience, skills or managerial qualifications of any person or Entity who directly or indirectly owns 10% or less interest in you, we have the right to designate that person as an Owner who must sign Appendix B to this Agreement. In addition, if you are a partnership Entity, then each person or entity who, now or hereafter is or becomes a general partner is deemed an Owner who must sign Appendix B, regardless of their percentage ownership interest.
- C. Responsible Person. You must designate one individual, who shall be set forth in Appendix A, who has the authority to, and does in fact, actively direct your business affairs related to your obligations under this Agreement (the "Responsible Person"). Your Responsible Person shall exert their best efforts to the development of your Businesses; and absent our prior approval may not engage in any other business or activity, directly or indirectly, that requires substantial management responsibility or time commitments or which may otherwise may—conflict with the obligations hereunder. You must notify us of any proposed change of the Responsible Person and receive our written approval prior to such change. If such change results from the termination (whether voluntary or involuntary) of the Responsible Person, you must submit a new proposed Responsible Person within fifteen (15) days after such termination. Neither you nor your Owners will, directly or indirectly, take any actions to avoid_negate, circumvent, or otherwise restrict the authority requirement for theyour Responsible Person's authority to bind you with respect to this Agreement.
- <u>D.</u> <u>Accuracy of Appendix A.</u> Your Responsible Person and other Owners are identified in <u>Appendix A</u> to this Agreement. You and each of your Owners represent, warrant and agree that the attached <u>Appendix A</u> is a current, complete and accurate list of all Owners. <u>Every time there is a Following any</u> change in the persons or Entities who are your Owners, you must, within seven <u>(7)</u> calendar days from the date of such change, <u>notify us of the change and cooperate with us in updating <u>Appendix A</u> so that it is <u>at all times</u> current, complete and accurate <u>at all times</u>.</u>

9.

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

10. <u>Termination of Agreement</u>

÷

A. <u>Default and</u>; Events of Termination-

If (i) you fail to comply with the Development Schedule for any location at any time, (ii) you fail to comply with. You will be deemed in default under this Agreement if you breach any of the other terms of this Development Agreement or if you or your Affiliated Entities breach any of the terms of any other multi-unit development agreement, or (iii) we terminate a Franchise Agreement, or any other agreement between you or anyour Affiliated EntityEntities and us, we may or any of our affiliates. Notwithstanding anything to the contrary in this Agreement, we will have the right to terminate this Agreement immediatelyat any time and without notice, upon any one or more of the occurrence of (iii) and following our giving you 30 days' notice and opportunity to cure upon the occurrence of (i) or (ii), without further recourse to you.

Without waiving any rights afforded to us under this Agreement or any Franchise Agreement in which you or an Affiliated Entity own or hold any interest, we have the right, but no obligation, to refrain from exercising our termination right in favor of granting you a written extension on the Development Schedule. Such an extension may, in our business judgment, be conditioned on any or all of the following: (a) a reduction in the size of the Development Area; (b) a modified Development Schedule (in terms of timing and/or number of units to be opened); (c) your execution of our then-current form of general release; and/or (d) your execution of our then-current form of Development Agreement, which shall replace this Agreement and which may contain materially different terms and conditions. In addition, if we grant you a written extension, you must pay to us an extension fee of \$250 per day. Nothing obligates us to grant you an initial or any subsequent extension on the Development Schedule. We reserve the right to terminate this Agreement at any time if you fail to comply with its terms, including at the end of any unfulfilled extension period.

10. Franchise Agreements May Not be Affected. Except as expressly provided in this Agreement, all individual Franchise Agreements that we and you or an Affiliated Entity sign for Businesses within the Development Area are independent of this Agreement. The continued effectiveness of any individual Franchise Agreement does not depend on the continued effectiveness of this Agreement. If any conflict arises between this Agreement and any individual Franchise Agreement as to any individual Business, the latter will control.

Upon termination of this Agreement, neither you nor any of your Affiliated Entities has a right to establish or operate any individual Business for which an individual Franchise Agreement has not been executed by us at the time of termination. We may establish, and to license others to establish, Businesses in the Development Area, except as may be otherwise provided underevents:

- i. you fail to pay the Development Fee;
- <u>ii.</u> <u>you cease or threaten to cease to carry on the obligations granted to you under this Agreement, or take or threaten to take any action to liquidate your assets, or if you do not pay any debts or other amounts incurred by you in exercising the</u>

13 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

<u>Development Rights hereunder when such debts or amounts are due and payable;</u>

- iii. you fail to comply with the Development Schedule or any Development Deadline;
- iv. you make or purport to make a general assignment for the benefit of creditors; or if you hereto institute any proceeding under any statute or otherwise relating to insolvency or bankruptcy, or should any proceeding under any such statute or otherwise be instituted against you; or if a custodian, receiver, manager or any other person with like powers is appointed to take charge of all or any part of the business granted hereunder or of the shares or documents of title owned by any of your shareholders or title holders;
- <u>v.</u> <u>you fail to furnish reports or any other documentation required pursuant to this Agreement and do not correct such failure within ten (10) days following notice;</u>
- you or any of your Owners has made any material misrepresentation or omission in your or their application and the documents and other information provided to us to support your or their application to acquire the rights granted in this Agreement;
- <u>you (or any of your Owners) make or attempt to make an unauthorized transfer</u>
 (as further described in Section 6 herein);
- you (or any of your Owners): (A) are convicted of or plead guilty or "no-contest" to a felony, (B) are convicted of or plead guilty or "no contest" to any crime or other offense likely to adversely affect the reputation of Businesses or the goodwill of the Marks, or (C) engage in any conduct which, in our opinion, adversely affects or, if you were to continue as a Developer under this Agreement, is likely to adversely affect the reputation of the business you conduct pursuant to this Agreement, the reputation and goodwill of Businesses generally or the goodwill associated with the Marks;
- we provide written notice of your (or any of your Owners') failure: (A) on three (3) or more separate occasions within any twelve (12) consecutive month period to comply with this Agreement, or (B) on two (2) or more separate occasions within any six (6) consecutive month period to comply with the same obligation under this Agreement, in any case, whether or not you correct the failures within the timeframe that we designate after our delivery of notice to you; or
- x. you fail to observe, perform or comply with any other of the terms or conditions of this Agreement not listed in Sections 10.A.i through 10.A.ix above, and such failure continues for a period of ten (10) days after written notice thereof has been given by us to you.
- B. <u>Cross-Termination of Agreement. We will have the right to terminate this Agreement if we terminate any other multi-unit development agreement, Franchise Agreement, or any other agreement which has been executed between you (or anany of your Owners) or your Affiliated EntityEntities and us.</u>

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

- 11. Future Development. You recognize and acknowledge that this Agreement requires you to open Businesses in the future pursuant to the Development Schedule. You further acknowledge that the estimated expenses and investment requirements set forth in Items 6 and 7 of our Franchise Disclosure Document are subject to increase over time, and that future Businesses likely will involve greater initial investment and operating capital requirements than those stated in the Franchise Disclosure Document provided to you prior to the execution of this Agreement. You must execute all the Franchise Agreements and open all the Businesses by the dates set forth on the Development Schedule, regardless of (i) the requirement of a greater investment, (ii) the financial condition or performance of your prior Businesses, or (iii) any other circumstances, financial or otherwise. The foregoing will not be interpreted as imposing any obligation upon us to execute the Franchise Agreements under this Development Agreement if you have not complied with each and every condition necessary to develop the Businesses, or if you do not meet our then-current requirements for franchisees at the time you are scheduled to execute a Franchise Agreement.
- **12.** <u>Compliance with Applicable Laws</u>. You must, at your expense, comply with all federal, state, city, municipal and local laws, ordinances, rules and regulations in the Development Area pertaining to the development of your Businesses.

13. Your Non-Competition Obligations.

- **13.1.** <u>During Term</u>. You, any personal guarantors, each of your Owners and your and your Owner's respective spouse, parent (including stepparents), sibling (including half siblings), or child (including step children), whether natural or adopted (such relatives, the "Immediate Family"), will not, during the term of this Agreement, directly, indirectly or through, on behalf of, or in conjunction with any person or legal entity:
 - (a) Divert or attempt to divert any present or prospective business or customer of any Business to any Competitive Business, by direct or indirect inducement or otherwise, or do or perform, directly or indirectly, any other act injurious or prejudicial to the; or
 - (b) Own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in any Competitive Business.
- 13.2 After Term. You covenant that, except as otherwise approved in writing by us, you and your Owners shall not, for a continuous, uninterrupted period of two years commencing upon the date of (a) a transfer permitted under Section 16, (b) expiration of this Agreement, (c) termination of this Agreement (regardless of the cause for termination), or (d) a final order of a duly authorized arbitrator, panel of arbitrators, or a court of competent jurisdiction (after all appeals have been taken) with respect to any of the foregoing or with respect to enforcement of this Section 13.2, either directly or indirectly, for yourself or your spouse, parents (including step parents), siblings (including half siblings), and children (including step children), whether natural or adopted (your "Immediate Family"), or through, on behalf of, or in conjunction with any person or legal entity, own, maintain, operate, engage in, be employed by, act as a consultant for, perform services for, provide assistance to, or have any interest in (as owner or otherwise) any Competitive Business that is, or is intended to be, located (a) within the Development Area, (b) within 15 miles of any Business developed pursuant to this Agreement, or (c) within 15 miles

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

of any other SKY ZONE Business in operation or under construction as of the date that you are required to comply with this Section 13.2. You agree and acknowledge that the two-year period of this restriction shall be tolled during any time period in which you are in violation of this restriction.

- 13.3 Other Covenants. Furthermore, you agree and acknowledge that any in-term covenant to compete that applies to you, your affiliates, your Owners or your Immediate Family under any Franchise Agreement with us shall continue to apply and will not be superseded or amended by the termination of this Agreement and the applicability of Section 13.3.
- **13.4** Exception. The restrictions in Sections 13.1.3 and 13.2 do not apply to: (a) interests in or operation of a Business under a written Franchise Agreement with us; or (b) the ownership of shares of a class of securities that are listed on a public stock exchange or traded on the over-the-counter market and that represent less than 5% of that class of securities.
- 13.5 <u>Injunctive Relief</u>. You agree that damages alone cannot adequately compensate us if there is a violation of these noncompetitive covenants and that injunctive relief is essential for our protection. You therefore agree that in case of any alleged breach or violation of this Section, we may seek injunctive relief without posting any bond or security, in addition to all other remedies that may be available to us at equity or law.
 - 14. No Rights to Development Area Upon or any of our affiliates.
- <u>C.</u> <u>Effects of Termination. After or Expiration. Upon the expiration or termination or expiration of this Agreement for any reason, any and the following provisions apply:</u>
 - i. all of your rights you had in and tounder this Agreement will cease, and you will no longer be entitled to exercise the Development Rights or hold yourself out to the public as being a developer of Businesses within the Development Area-shall cease, and;
 - ii. we will have the absolute and unrestricted right to develop the Development Area or to contract with other <u>franchiseesparties</u> for the future development of the Development Area.;
- **15.** Assignment by Us. We have the right to sell or assign, in whole or in part, our interests in this Agreement, and any such sale or assignment shall inure to the benefit of any assignee or other legal successor to our interest.
- 16. Assignment by You. You may only transfer your rights and interests under this Agreement if you obtain our prior written consent and transfer all of your, and your Affiliated Entities' respective rights and interests under all Franchise Agreements for the Businesses in the Development Area. Accordingly, the assignment terms and conditions in the Franchise Agreements apply to any transfer of your rights and interests under this Agreement or any ownership in you, except that you cannot assign your rights and interests in this Agreement for an amount of consideration greater than the Development Fee specified in Section 6 (Development Fee) unless we otherwise agree and you comply with all supplemental assignment conditions we specify in our business judgment, including, without limitation, your payment of supplemental transfer fee in the amount we specify.

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

- 17. <u>Severability</u>. To the extent that this Agreement is judicially determined to be unenforceable by virtue of its scope or in terms of area or length of time, but may be made enforceable by reductions of any or all thereof, the same shall be enforced to the fullest extent permissible under the laws and public policies applied in the jurisdiction where enforcement is sought.
- **18.** <u>Waivers</u>. Failure of either party to insist, in one or more instances, on performance by the other in strict accordance with the terms and conditions of this Agreement shall not be deemed a waiver or relinquishment of any right granted hereunder or of the future performance of any such term or condition or of any other term or condition of this Agreement, unless such waiver is in a writing signed by or on behalf of both parties.

19.—

- you must immediately cease using all of our Confidential Information and return to us, or destroy, any Confidential Information in your possession or control as further described in Section 7.B. above (except that you may retain and continue to use any Confidential Information that you are permitted to use under any Franchise Agreements which remain in full force and effect); and
- <u>without limiting any other rights or remedies to which we may be entitled, you must pay any amounts owing to us pursuant to this Agreement through the effective date of termination.</u>
- D. <u>Franchise Agreements. Unless concurrently terminated, any Franchise Agreements in effect by and between us and you (or your Affiliated Entities) will remain in full force and effect following the expiration or termination of this Agreement. For purposes of further clarity, any covenants pertaining to confidentiality, non-competition, and non-solicitation set forth in such Franchise Agreements additionally remain in full force and effect.</u>
 - E. Restrictive Covenants Not to Compete.
 - i. Upon termination or expiration of this Agreement, or upon a Transfer, you and your Owners agree that, for two (2) years beginning upon the effective date of termination, expiration or transfer, or the date on which all persons restricted by this Section 10.E. begin to comply with this Section 10.E., whichever is later, neither you nor your Affiliated Entities, nor any of the Bound Parties will have any direct or indirect interest as an owner (whether of record, beneficially, or otherwise), investor, partner, director, officer, employee, consultant, lessor, representative, agent, or in any other capacity in any Competitive Business located or operating:
 - 1. Within the Development Area;
 - 2. Within a fifteen (15) mile radius of any Business developed pursuant to this Agreement; and
 - 3. Within a fifteen (15) mile radius of any other Business in operation or under construction on the later of the effective date of the termination or

17
Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

expiration of this Agreement or the date upon which all persons restricted by this Section 10.E. begin to comply with this Section 10.E.

- ii. You agree and acknowledge that the two (2) year period of these restrictions shall be tolled during any time period in which you, your Owners, or any Bound Parties are in violation of these restrictions.
- F. Survival of Covenants. Notwithstanding the expiration, termination or transfer of this Agreement for any reason whatsoever, all covenants and agreements to be performed or observed by you will survive any such termination, expiration or transfer.

11. Relationship of the Parties; Indemnification

÷

- A. Independent Contractors. Each of us is an independent contractor, and neither is considered to be the agent, representative, master or servant of the other for any purpose. Neither of us has any authority to enter into any contract, to assume any obligations or to give any warranties or representations on behalf of the other. Nothing in this Agreement may be construed to create a relationship of partners, joint venturers, fiduciaries, agency or any other similar relationship between us and you.
- Indemnification. To the fullest extent permitted by law, you agree to indemnify, defend, and hold harmless us, our affiliates, and our and their respective current and former owners, managers, directors, officers, employees, agents, successors, and assignees (the "Indemnified Parties") against, and to reimburse any one or more of the Indemnified Parties for, all claims, obligations, and damages directly or indirectly arising out of the operation of the business you conduct under this Agreement, or your breach of this Agreement, including those alleged to be caused by the Indemnified Party's negligence, unless (and then only to the extent that) the claims, obligations, or damages are determined to be caused solely by the Indemnified Party's intentional misconduct in a final, unappealable ruling issued by a court with competent jurisdiction. For purposes of this indemnification, "claims" include all obligations, damages (actual, consequential, or otherwise), and costs that any Indemnified Party reasonably incurs in defending any claim against it, including reasonable accountants', arbitrators', attorneys', and expert witness fees, costs of investigation and proof of facts, court costs, travel and living expenses, and other expenses of litigation or alternative dispute resolution, regardless of whether litigation or alternative dispute resolution is commenced. Each Indemnified Party may defend any claim against it at your expense and agree to settlements or take any other remedial, corrective, or other actions. This indemnity will continue in full force and effect subsequent to and notwithstanding this Agreement's expiration or termination. An Indemnified Party need not seek recovery from any insurer or other third party, or otherwise mitigate its losses and expenses, in order to maintain and recover fully a claim against you under this Section 11.B. You agree that a failure to pursue a recovery or mitigate a loss will not reduce or alter the amounts that an Indemnified Party may recover from you under this Section 11.B.

12. <u>Dispute Resolution</u>

18

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

- A. 19.1 MediationInformal Resolution. Except as provided in Section 19.312.C. (Exceptions to Arbitration), prior to filing any demand for arbitration, the parties agree to mediateattempt to informally resolve any dispute, controversy or claim between us and among the partiesyou and any of our or your affiliates Affiliated Entities, officers, directors, shareholders, members, guarantors, employees or ewners Owners arising under, out of, in connection with or in relation to: (i1) this Agreement; (ii2) any lease or sublease for your Business, Businesses: (iii3) any loan or other finance arrangement between us or our Affiliates and you; (iv4) the parties' relationship; (v5) your Business, Businesses; or (vi6) the System (each a "Dispute") in accordance with the following procedures:
 - i. (a) The party seeking mediation to informally resolve a Dispute must commence mediation the process by sending the other party, in accordance with Section 2317.A. (Notices), a written notice of its request for mediation headed Dispute featuring the heading "Notification of Dispute." The Notification of Dispute will specify, to the fullest extent possible, the party's version of the facts surrounding the dispute Dispute; the amount of damages; and the nature of any injunctive or other relief such party claims. The party (or parties as the case may be) receiving a Notification of Dispute will respond within 20thirty (30) days after receipt thereof, stating its version of the facts and, if applicable, its position as to damages sought by the party initiating the dispute procedure; provided, however, that if the dispute Dispute has been the subject of a default notice given under Section 910 (Default and Termination of Agreement), the other party will respond within ten (10) business days.
 - ii. (b)—Upon receipt of a Notification of Dispute and response under Section 19.1(a)12.A.i., the parties will endeavor, in good faith, for sixty (60) days (or an extended timeframe as mutually agreed upon between the parties) to resolve the disputeDispute outlined in the Notification of Dispute and the other party's response. If In particular, the parties have been unableResponsible Person will engage in discussions with a designated SZFG executive to attempt to resolve a dispute outlined in a Notification of the Dispute or a response thereto within 20 days after receipt of the response, either party may initiate a mediation procedure with the American Arbitration Association ("AAA"), pursuant to its Commercial Mediation Procedures, in the city of our then-current corporate headquarters, unless both parties agree to a different location. The parties must select a mediator jointlyduring such timeframe.
 - (c) All mediation sessions must be attended by an Owner (who has authority to settle the dispute on your behalf) and our representative(s) who is/are authorized to settle the dispute. The parties may be represented by counsel at the mediation. The parties agree to participate in the mediation proceedings in good faith and with the intention of resolving the dispute if at all possible within 30 days of the notice from the party seeking to initiate the mediation procedures. If the dispute is not resolved within 30 days, any party may initiate an arbitration pursuant to Section 19.2 (Arbitration). In addition, if the party receiving notice of mediation has not responded within 5 days of delivery of the notice or a party fails to participate in the mediation, this Section 19.1 will no longer be applicable and the other party can pursue arbitration. The parties agree that the costs of the mediator will be split equally between the parties. Each party must

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

pay its own fees and expenses incurred in connection with the mediation. The mediation proceeding and any negotiations and results thereof will be treated as a compromise settlement negotiation and the entire process is confidential. At least five days prior to the initial mediation session, each party must deliver a written statement of positions.

19.2—

- B. <u>Arbitration</u>— Except as provided in Section <u>19.3-12.C.</u> (Exceptions to Arbitration), any Dispute, including any <u>disputeDispute</u> regarding the scope or validity of the arbitration obligation under this Section, not resolved by <u>mediationthe informal process set forth in Section 12.A.</u> must be submitted to binding arbitration in accordance with the Commercial Arbitration Rules of the American Arbitration Association then in effect to be heard by one <u>(1)</u> arbitrator who has a minimum of five (5) years' experience in franchising or distribution law.
 - i. (a) Counterclaims. In connection with any arbitration proceeding, each party will submit or file any claim which would constitute a compulsory counterclaim (as defined by the then-current Rule 13 of the Federal Rules of Civil Procedure) within the same proceeding as the claim to which it relates. Any such claim which is not submitted or filed in such proceeding will be barred.
 - ii. (b) Waiver of Class Actions. Any You (and your Owners) expressly acknowledge and agree that any arbitration must be on an individual basis only as to a single franchisefranchisee (and not as or through an association) and the parties and the arbitrator will have no authority or power to proceed with any claim on a class-wide basis or otherwise to join or consolidate any claim with any claim or any other proceeding involving third parties. If a court or arbitrator determines that this limitation on joinder of or class-wide claims is unenforceable, then the agreement to arbitrate the dispute will be null and void and the parties must submit all claims to the jurisdiction of the courts, in accordance with Section 2014 (Consent to Jurisdiction).
 - iii. (c) Venue. The arbitration must take place in the city where our headquartersprincipal place of business is located at the time of the dispute-currently, Provo, Utah), unless the parties mutually agree upon an alternate location.
 - Arbitrator. The arbitrator must follow the law and not disregard the terms of this Agreement. The arbitrator will have subpoena powers limited only by the laws of the state in which our corporate headquartersprincipal place of business is then located. The arbitrator may not consider any settlement discussions or offers that might have been made by either you or us. The arbitrator may not under any circumstance: (a) stay the effectiveness of any pending termination of this Agreement, (b) assess punitive or exemplary damages, (c) certify a class or a consolidated action, or (d) make any award which extends, modifies or suspends any lawful term of this Agreement or any reasonable standard of business performance that we set. The arbitrator will have the right to make a determination decide as to any procedural matters as would a court of competent jurisdiction be permitted to make in the state in which our corporate

20 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1 headquarters is then located. The arbitrator will also decide any factual, procedural, or legal questions relating in any way to the dispute between the parties, including: any decision as to whether Section 49.212.B. (Arbitration) is applicable and enforceable as against the parties, subject matter, timeliness, scope, remedies, unconscionability, and any alleged fraud in the inducement.

- v. (e) Orders. The arbitrator can issue summary orders disposing of all or part of a claim and provide for temporary restraining orders, preliminary injunctions, injunctions, attachments, claim and delivery proceedings, temporary protective orders, receiverships, and other equitable and/or interim/final relief. The judgment of the arbitrator on any preliminary or final arbitration award will be final and binding and may be entered in any court having jurisdiction. Each party consents to the enforcement of such orders, injunctions, etc., by any court having jurisdiction.
- vi. (f) Procedure. The parties to the dispute will have the same discovery rights as are available in civil actions under the laws of the state of Missouriwhere our principal place of business is located at the time of the Dispute (currently, Utah). All other procedural matters will be determined by applying the statutory, common laws, and rules of procedure that control a court of competent jurisdiction in Missourithe state where our principal place of business is located at the time of the Dispute.
- vii. (g) Confidential. Other than as may be required by law (including but not limited to state and federal franchise laws), the entire arbitration proceedings (including any rulings, decisions, or orders of the arbitrator), will remain confidential and will not be disclosed to anyone other than the parties to this Agreement.
- viii. (h) Advances. We reserve the right, but have no obligation, to advance your share of the costs of any arbitration proceeding for such arbitration proceeding to take place and by doing so will not be deemed to have waived or relinquished our right to seek recovery of those costs in accordance with Section 19.412.D. (Prevailing Party's Fees).
- ix. Depositions. In any arbitration each side may take no more than three (3) depositions, unless the parties mutually agree to additional depositions. Each side's depositions are to consume no more than a total of fifteen (15) hours, and each deposition shall be limited to five (5) hours, unless the parties mutually agree to additional time.
- x. Electronic Discovery. With respect to any discovery of electronically stored information, you and we agree that such requests must balance the need for production of electronically stored information relevant and material to the outcome of a disputed issue against the cost of locating and producing such information. You and we agree that:
 - 1. Production of electronically stored information need only be from sources used in the ordinary course of business. No party shall be required to

21 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

- search for or produce information from back-up servers, tapes, or other media;
- 2. The production of electronically stored information shall normally be made on the basis of generally available technology in a searchable format which is usable by the party receiving the information and convenient and economical for the producing party. Absent a showing of compelling need, the parties need not produce metadata, with the exception of header fields for email correspondence;
- 3. The description of custodians from whom electronically stored information may be collected shall be narrowly tailored to include only those individuals whose electronically stored information may reasonably be expected to contain evidence that is relevant and material to the outcome of a disputed issue:
- 4. The parties shall attempt to agree in advance upon, and the arbitrator may determine, reasonable search parameters;
- 5. Where the costs and burdens of electronic discovery are disproportionate to the nature of the dispute or to the amount in controversy, or to the relevance of the materials requested, the arbitrator shall either deny such requests or order disclosure on condition that the requesting party advance the reasonable cost of production to the other side, which cost advance will not be awarded to the prevailing party in any final award.
- C. <u>19.3</u> Exceptions to Arbitration. Notwithstanding Sections <u>19.112.A.</u> (<u>MediationInformal Resolution</u>) and <u>19.212.B.</u> (Arbitration), the parties agree that the following Disputes will not be subject to arbitration or mediation:
 - i. (a) any action for equitable relief, including seeking preliminary or permanent injunctive relief, specific performance, declaratory relief, other relief in the nature of equity to enjoin any harm or threat of harm to such party's tangible or intangible property, brought at any time, including prior to or during the pendency of any arbitration proceedings initiated hereunder;
 - ii. (b) any action in ejectment or for possession of any interest in real or personal property;
 - iii. (c) any action which by Applicable Law Laws cannot be arbitrated; or
 - iv. (d) our decision in the first instance to issue a notice of default and/or notice of termination or undertake any other conduct with respect to the franchise relationship that might later result in a dispute or controversy between us.
- D. 19.4 Prevailing Party's Fees. The prevailing party in any action or proceeding arising under, out of, in connection with, or in relation to this Agreement will be entitled to recover its reasonable costs and expenses (including attorneys' fees, arbitrator's fees and

22 Sky Zone 2024 MUDAFranchise Group, LLC DMS_US.363658145.1 expert witness fees, costs of investigation and proof of facts, court costs, and other arbitration or litigation expenses) incurred in connection with the claims on which it prevailed.

- E. 19.5 Survival. The provisions of this Section 1912 are intended to benefit and bind certain third-party non-signatories and will continue in full force and effect subsequent to and notwithstanding the expiration or termination of this Agreement.
- F. 19.6 Tolling of Statute of Limitations. All applicable statutes of limitation and defenses based on the passage of time are tolled while the dispute resolution procedures in this Section 1912 are pending. The parties will take such action, if any, required to effectuate such tolling.
- G. 19.7 Performance to Continue. Each party must continue to perform its obligations under this Agreement pending final resolution of any Dispute pursuant to this Section 1912, unless to do so would be impossible or impracticable under the circumstances.
- 13. 19.8 Governing Law. This Agreement will be governed by, construed, and enforced in accordance with the laws of the State of Missouri although any Missouri franchise law only applies in the event you are a Missouri resident or the Development Territory is located in Missouri. In the event of any conflict-of-law question, the laws of Missouri shall prevail, without regard to the application of Missouri conflict-of-law rules.
- ALL MATTERS RELATING TO ARBITRATION WILL BE GOVERNED BY THE UNITED STATES FEDERAL ARBITRATION ACT (9 U.S.C. §§ 1 ET SEQ.). EXCEPT TO THE EXTENT GOVERNED BY THE FEDERAL ARBITRATION ACT, THE UNITED STATES TRADEMARK ACT OF 1946 (LANHAM ACT, 15 U.S.C. SECTIONS 1051 ET SEQ.) (THE "LANHAM ACT"), OR OTHER UNITED STATES FEDERAL LAW, THIS AGREEMENT, THE DEVELOPMENT RIGHTS, AND ALL CLAIMS ARISING FROM THE RELATIONSHIP BETWEEN US AND YOU WILL BE GOVERNED BY THE LAWS OF THE STATE IN WHICH OUR PRINCIPAL PLACE OF BUSINESS IS LOCATED (CURRENTLY, UTAH) WITHOUT REGARD TO ITS CONFLICT OF LAWS RULES, EXCEPT THAT ANY STATE LAW REGULATING THE OFFER OR SALE OF FRANCHISES OR GOVERNING THE RELATIONSHIP OF A FRANCHISOR AND ITS FRANCHISEE WILL NOT APPLY UNLESS ITS JURISDICTIONAL REQUIREMENTS ARE MET INDEPENDENTLY WITHOUT REFERENCE TO THIS PARAGRAPH.
- 14. 19.9 Consent to Jurisdiction. Subject to Section 19 (Dispute Resolution), you and the Owners must file any suit against us, and we may file any suit against you, in federal or state courts located in the state where our principal place of business is then located. You (and each Owner) irrevocably submit to the jurisdiction of such courts and waive any objection you (or he or she) may have to either the jurisdiction of or venue in such courts.
- **20.** Waiver of Punitive Damages, Jury Trial and Class Actions. Except with respect to any obligation to indemnify us and claims we bring against you for your unauthorized use of the Marks or unauthorized use or disclosure of any confidential information, we and you and your respective Owners waive to the fullest extent permitted by law any right to or claim for any punitive or exemplary damages against the other and agree that, in the event of a dispute between us, the party making a claim will be limited to equitable relief and to recovery of any actual damages it sustains. We and you irrevocably waive, to the fullest extent permitted by

23

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

law, trial by jury in any action, proceeding or counterclaim, whether at law or in equity, brought by either of us. Any claims must be brought on an individual basis only as to a single franchise (and not as or through an association or class). Each party acknowledges that it has had a full opportunity to consult with counsel concerning this waiver, and that this waiver is informed, voluntary, intentional, and not the result of unequal bargaining power.

21.—

. SUBJECT TO SECTION 12, YOU (AND YOUR OWNERS) AGREE THAT ALL ACTIONS ARISING OUT OF OR RELATING TO THIS AGREEMENT OR OTHERWISE AS A RESULT OF THE RELATIONSHIP BETWEEN YOU AND US MUST BE COMMENCED EXCLUSIVELY IN A STATE OR FEDERAL COURT IN THE LOCATION IN WHICH OUR PRINCIPAL PLACE OF BUSINESS IS LOCATED (CURRENTLY, PROVO, UTAH), AND WE AND YOU (AND EACH OWNER) IRREVOCABLY CONSENT TO THE JURISDICTION OF THOSE COURTS AND WAIVE ANY OBJECTION TO EITHER THE JURISDICTION OF OR VENUE IN THOSE COURTS.

15. Waiver of Punitive Damages, Jury Trial, and Class Actions

EXCEPT WITH RESPECT TO ANY OBLIGATION TO INDEMNIFY US AND ANY CLAIMS WE BRING AGAINST YOU FOR YOUR UNAUTHORIZED USE OF THE MARKS OR UNAUTHORIZED USE OR DISCLOSURE OF ANY CONFIDENTIAL INFORMATION, WE AND YOU AND YOUR RESPECTIVE OWNERS WAIVE TO THE FULLEST EXTENT PERMITTED BY LAW ANY RIGHT TO OR CLAIM FOR ANY PUNITIVE OR EXEMPLARY DAMAGES AGAINST THE OTHER AND AGREE THAT, IN THE EVENT OF A DISPUTE BETWEEN US, THE PARTY MAKING A CLAIM WILL BE LIMITED TO EQUITABLE RELIEF AND TO RECOVERY OF ANY ACTUAL DAMAGES IT SUSTAINS. WE AND YOU IRREVOCABLY WAIVE, TO THE FULLEST EXTENT PERMITTED BY LAW, TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM, WHETHER AT LAW OR IN EQUITY, BROUGHT BY EITHER OF US. WE AND YOU AGREE THAT ANY PROCEEDING WILL BE CONDUCTED ON AN INDIVIDUAL BASIS AND THAT ANY PROCEEDING BETWEEN US AND ANY OF OUR AFFILIATES, OR OUR AND THEIR RESPECTIVE SHAREHOLDERS, OFFICERS, DIRECTORS, AGENTS, AND EMPLOYEES, ON THE ONE HAND, AND YOU (OR YOUR OWNERS, GUARANTORS, AFFILIATED ENTITIES, AND EMPLOYEES), ON THE OTHER HAND, MAY NOT BE: (I) CONDUCTED ON A CLASS-WIDE BASIS, (II) COMMENCED, CONDUCTED OR CONSOLIDATED WITH ANY OTHER PROCEEDING, (III) JOINED WITH ANY CLAIM OF AN UNAFFILIATED THIRD PARTY, OR (IV) BROUGHT ON YOUR BEHALF BY ANY ASSOCIATION OR AGENT. THIS WAIVER IS EFFECTIVE EVEN IF COURT OF COMPETENT JURISDICTION DECIDES THAT THE ARBITRATION PROVISION IN SECTION 12.B. IS UNENFORCEABLE. EACH PARTY ACKNOWLEDGES THAT IT HAS HAD A FULL OPPORTUNITY TO CONSULT WITH COUNSEL CONCERNING THIS WAIVER, AND THAT THIS WAIVER IS INFORMED, VOLUNTARY, INTENTIONAL, AND NOT THE RESULT OF UNEQUAL BARGAINING POWER.

16. <u>Limitations of Claims</u>

. Except for claims arising from your nonpayment or underpayment of amounts you owe us pursuant to this Agreement, or claims related to your unauthorized use of the Marks, any and

24

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

all claims arising out of or relating to this Agreement or our relationship with you will be barred unless a Notification of Dispute is sent or a judicial proceeding is commenced within one (1) year from the date on which the party asserting such claim knew or should have known of the facts giving rise to such claims.

17. Miscellaneous

÷

A. <u>Notices</u>. All notices and other communications required or permitted under this Agreement will be in writing and will be given by one of the following methods of delivery: (i) personally; (ii) by certified or registered mail, postage prepaid; (iii) by overnight delivery service; or (iv) by electronic mail (with a copy sent by <u>USU.S.</u> Mail or <u>Overnight overnight</u> delivery services). Notices to you will be sent to the address set forth on <u>Appendix A</u>. Notices to us must be sent to <u>both</u>:

Sky Zone Franchise Group, LLC 86 N. University Avenue, Suite 350 Provo, Utah 84601

Attn: Franchise Operations (franchiseops@skyzone.com)

with a copy to Sky ZoneSZFG's Legal Department (legal@skyzone.com)

Either party may change its mailing address or electronic mail address by giving notice to the other party. Notices will be deemed received the same day when delivered personally, upon attempted delivery when sent by registered or certified mail or overnight delivery service, or when sent by electronic mail.

- B. 22. Force Majeure. Neither we nor you will be liable for any loss or damage or deemed to be in breach of this Agreement if our or your failure to perform our or your obligations is not our or your fault and results from: (i) transportation shortages, inadequate supply of equipment, products, merchandise, supplies, labor, material or energy or the voluntary foregoing of the right to acquire or use any of the foregoing in order to accommodate or comply with the orders, requests, regulations, recommendations or instructions of any federal, state or municipal government or any department or agency thereof; (ii) acts of nature; (iiii) fires, strikes, embargoes, war or riot; (iviii) failure to obtain land use or environmental approvals from the applicable government body or agency, so long as you diligently pursue any such required approvals; or (viv) any other similar event or cause. Any delay resulting from any of said force majeure causes will extend performance accordingly or excuse performance, in whole or in part, as may be reasonable, except that said causes will not excuse payments of amounts owed at the time of such occurrence.
- 23. <u>Multiple Copies</u>. This Agreement may be executed in counterparts, each of which will be deemed an original, and all of which when taken together shall constitute one and the same document. This Agreement shall not be binding on either party until it is executed by both parties.

24.

25

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

- C. Other Developers. You acknowledge and agree that the provisions of Franchise Agreements and Multi-Unit Development Agreements by and among us and other Sky Zone® franchise owners and developers may vary substantially from those contained in this Agreement and the Franchise Agreements to be executed by you in connection herewith.
- D. Terrorist Acts. You acknowledge that under applicable U.S. law, including Executive Order 13224, signed on September 23, 2001 (the "Order"), we are prohibited from engaging in any transaction with any person engaged in, or with a person aiding any person engaged in, acts of terrorism, as defined in the Order. Accordingly, you represent and warrant to us that, as of the date of this Agreement, neither you nor any person holding any ownership interest in you, controlled by you, or under common control with you is designated under the Order as a person with whom business may not be transacted by us, and that you: (i) do not, and hereafter will not, engage in any terrorist activity; (ii) are not affiliated with and do not support any individual or Entity engaged in, contemplating, or supporting terrorist activity; and (iii) are not acquiring the rights granted under this Agreement with the intent to generate funds to channel to any individual or Entity engaged in, contemplating, or supporting terrorist activity, or to otherwise support or further any terrorist activity.
- E. <u>Joint and Several Obligations. If two (2) or more persons or Entities are at any time Owners, whether as partners or joint venturers, their obligations, and liabilities to us will be joint and several.</u>
- F. Severability. If any term or condition of this Agreement is, to any extent, declared to be invalid or unenforceable, all other terms and conditions of this Agreement, other than those as to which it is held invalid or unenforceable, will not be affected thereby and each term and condition of this Agreement will be separately valid and enforceable to the fullest extent permitted by law.
- G. Headings and Construction. Headings preceding the text, sections and subsections hereof have been inserted solely for convenience of reference and will not be construed to affect the meaning, construction or effect of this Agreement. Whenever this Agreement allows or requires us to take actions or make decisions, we may do so in our sole and unfettered discretion, even if you believe our action or decision is unreasonable, unless the Agreement expressly and specifically requires that we act reasonably or refrain from acting unreasonably in connection with the particular action or decision. The term "including" means "including, without limitation" unless otherwise noted. The term "control" means the right and power to direct or cause the direction of an Entity's management and policies.
- H. Waiver. The waiver by either you or us of a breach of any term or condition contained in this Agreement will not be deemed to be a waiver of such term or condition or any subsequent breach of the same or any other term or condition herein contained unless such waiver is expressly set forth in writing. A party's failure to exercise any right to demand exact compliance and any custom or practice at variance with the terms and conditions of this Agreement will not constitute a waiver of the right to demand exact compliance with the terms and conditions hereof.
- I. Costs and Attorneys' Fees. If we incur expenses in connection with your failure to pay when due amounts owed to us or to submit when due any reports, information or

26

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

supporting records or otherwise to comply with this Agreement, you agree to reimburse us for any of the costs and expenses which we incur, including reasonable accounting, attorneys', arbitrators', and related fees.

- J. Rights of Parties Are Cumulative. Our and your rights hereunder are cumulative, and no exercise or enforcement by us or you of any right or remedy hereunder will preclude our or your exercise or enforcement of any other right or remedy hereunder which we or you are entitled by law to enforce.
- K. Entire Agreement; Modification. This Agreement, together with any addenda and appendices, constitutes the sole agreement between you and us with respect to the entire subject matter of this Development-Agreement and embodies all prior agreements and negotiations with respect to your Businessesthe Development Rights authorized hereunder. We expressly disclaim any understandings, agreements, inducements, course(s) of dealing, representations (financial or otherwise), promises, options, rights of first refusal, guarantees, warranties (express or implied) or otherwise (whether oral or written) which are not fully expressed in this Agreement. Nothing in this or any related agreement, however, is intended to disclaim the representations made by us in the Franchise Disclosure DocumentFDD that was furnished to you by us-
- **25.** <u>Modification</u>. This Agreement shall not be modified or amended except by an instrument in writing signed by or on behalf of the parties hereto.
- L. <u>Binding Effect. This Agreement is binding upon us and you and our respective executors, administrators, heirs, beneficiaries, assigns and successors in interest.</u>
- M. **26.** No Third-Party Beneficiaries. Except as expressly otherwise provided herein, no third party shall have the right to claim any of the benefits conferred under this Agreement.
- **27.** Other Franchisees/Developers. You acknowledge that other SKY ZONE franchisees/developers have or will be granted franchises or multi-unit development rights at different times and/or in different situations, and further acknowledge that the provisions of such agreements may vary substantially from those contained in this Agreement.
- **28.** <u>Binding Effect</u>. This Agreement is binding upon us and you and our respective executors, administrators, heirs, beneficiaries, assigns and successors in interest.
- 29. No Guarantee of Success. YOU AND EACH OF THE UNDERSIGNED OWNERS WARRANT AND REPRESENT THAT HE/SHE HAS NOT RELIED UPON ANY GUARANTEES CONCERNING REVENUE, PROFIT OR THE SUCCESS OF THIS FRANCHISE IN SO SIGNING.

No Waiver or Disclaimer Of Reliance In Certain States

<u>.</u> The following provision applies only to franchisees and franchises that are subject to state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, or Wisconsin:

27 Sky Zone 2024 MUDA Pranchise Group, LLC DMS_US.363658145.1

No statement, questionnaire, or acknowledgment signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the franchise.

N. Counterparts; Electronic Signatures. This Agreement may be executed in counterparts, each of which will be deemed an original, and all of which when taken together shall constitute one and the same document. This Agreement shall not be binding on either party until it is executed by both parties. Electronic signatures or signatures which are transmitted via facsimile or scanned and emailed shall have the same force and effect as originals.

[Signature page follows]

IN WITNESS WHEREOF, the parties hereto have executed and delivered this Agreement as of the Effective Date.

FRANCHISOR	<u>DEVELOPER</u>	
SKY ZONE FRANCHISE GROUP, LLC	(IF ENTITY):	
Ву:		
Name: Title:	[Name]	
Date <u>*</u> :	Pv <i>r</i>	
*This is the Effective Date	By: Name:	
	Title:	
	Date:	
	Date:(IF INDIVIDUALS):	
	[Signature]	
	[Print Name]	
	Date:	
	[Signature]	
	[Print Name]	
	Date:	
	[Signature]	
	[Print Name]	
	Date:	
	[Signature]	

29 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

[Print Name]	
Date:	

30 Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

APPENDIX A TO THE DEVELOPMENT AGREEMENT DEVELOPER-SPECIFIC TERMS

Developer's Name			
Developer's Name:			
Developer's Address And Electronic Mail Address For Notices:			
Development Area	(Section 1): [attach ma	o if necessary]	
			_
	For numbers of the De-	ralammant Calessius as t	auth hale f
[AS APPLICABLE:	For purposes of the Dev	DIODMANT SCHARIIIA CAT TR	
Rusinesses must be			
	e opened in Map #1 attach		
opened in Map #2 at	e opened in Map #1 attach ttached hereto.]		
opened in Map #2 at	e opened in Map #1 attach ttached hereto.]		
opened in Map #2 at [City, State] (Park 1)	e opened in Map #1 attach ttached hereto.] [see attached map]		
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2)	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map]		
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2)	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map]		
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Development Sche of Busine	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map]	ed hereto and [] E	Businesses must n and operate a t this Agreement.
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Park 2) Development Sche of Busine Businesses r	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map] Section 68): \$ dule (SectionSections 1 8 esses within the Developments be open and operating	ed hereto and [] E 2): You agree to establishent Area during the term of the property of the secondance with the following the secondance with the following the secondance with the following the secondance with the second	Businesses must n and operate a t this Agreement.
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Park 2) Development Sche of Busine Businesses r	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map] Section 68): \$ dule (Section Sections 1 8 esses within the Developme	ed hereto and [] E	Businesses must n and operate a this Agreement.
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Park 2) Development Sche of Busine Businesses r	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map] Section 68): \$ dule (SectionSections 1 8 esses within the Development of BUSINESSES The Minimum Number of	ed hereto and [] E 2): You agree to establishent Area during the term of the in accordance with the fole DEVELOPMENT DEADLINE Deadline for Having the	Businesses must n and operate a this Agreement.
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Park 2) Development Sche of Busine Businesses r	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map] Section 68): \$ dule (SectionSections 1 8 esses within the Development of BUSINESSES The Minimum Number of Businesses Open and	ed hereto and [] E 2): You agree to established the following the term of the following the term of the following the following the headline for Having the Minimum Number of	Businesses must n and operate a this Agreement.
opened in Map #2 at [City, State] (Park 1) [City, State] (Park 2) Development Fee (Park 2) Development Sche of Busine Businesses r	e opened in Map #1 attach ttached hereto.] [see attached map] [see attached map] Section 68): \$ dule (SectionSections 1 8 esses within the Development of BUSINESSES The Minimum Number of	ed hereto and [] E 2): You agree to establishent Area during the term of the in accordance with the fole DEVELOPMENT DEADLINE Deadline for Having the	Businesses mus a and operate a this Agreement.

A-1

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> DMS_US.363658145.1

, 20
, 20
, 20
, 20
, 20
(the Expiration Date of
the Agreement)

DEVELOPMENT DEADLINE	NUMBER OF NEW BUSINESSES TO OPEN DURING DEVELOPMENT DEADLINE	CUMULATIVE NUMBER OF BUSINESSES TO BE OPERATING BY END OF DEVELOPMENT DEADLINE
Months from the Effective Date of this Agreement	[1 Business]	[1 Business]
Months from the Effective Date of this Agreement	[1 Business]	[2 Businesses]

Form of Entity of Developer.

Corporation Or Limited Liability Compan	y . Developer was organized on
under the laws of the State of	Its Federal Employer
dentification Number is It ha	s not conducted business under any name
other than its corporate or company name.	
Name of Each Director/Officer/Managing Member	Position(s) Held
Partnership. Developer is a general or limit	
	Its Federal Employer as not conducted business under any name
Name of Each General Partner]

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

	nsible Person (Sect Person	t ion 8.3<u>9.C.</u>). The name is	e and hom as		Respons follo
Owner	s and Control Group) (Section 8.1 <u>9.A.</u>).			
	including the full nar nature and extent of a to his ownership into exclusive legal and a clear of all liens, res	scurate list of all Owners me and mailing address each Owner's interest in erest, represents and wo beneficial owner of his ostrictions, agreements a lired or permitted by this	of each C Developer. arrants tha ownership ind encumb	Owner, and fully de Developer and each Owner is the Interest in Developerances of any kindrances of an	escribes ch Owne he sole er, free
	Owner ²	's Name and Address		Percentage and Ownership Ir	
	has, directly or indirectits ownership interest	represent and warrant the ctly, 51% or more owner sts in you and constitudership and Control Grou	ship interes tes your C	et in you and voting Control Group as o	control (
	has, directly or indirectits ownership interest Section 8.19.A. (Own	ctly, 51% or more owner sts in you and constitu	ship interes tes your C	et in you and voting Control Group as o	control described

A-3

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

ŀ	

[Appendix A Signature Page Follows]

A-4

Sky Zone 2024 MUDA Franchise Group, LLC DMS_US.363658145.1

This <u>Appendix A</u> is deemed accepted and made a part of the Multi-Unit Development Agreement as of the <u>Multi-Unit Development Agreement's its</u> Effective Date.

FRANCHISOR	DEVELOPER
SKY ZONE FRANCHISE GROUP, LLC	(IF ENTITY):
Ву:	
Name: Title:	[Name]
Date:	Ву:
	Name: Title:
	Date:(IF INDIVIDUALS):
	[Signature]
	[Print Name]
	Date:
	[Signature]
	[Print Name]
	Date:
	[Signature]
	[Print Name]
	Date:
	[Signature]
Sky Zone 2024 MUDA Franchise Group, LLC	A-5

DMS_US.363658145.1

April 2025 FDD Multi-Unit Development Agreement [City, State]

[Print Name]	
Date:	

A-6

Sky Zone <u>2024 MUDA</u>Franchise Group, <u>LLC</u> <u>DMS_US.363658145.1</u>

April 2025 FDD Multi-Unit Development Agreement [City, State]

DEVELOPMENT AREA (SECTION 1)

(Park 1)

(Park 2)

Sky Zone 2024 MUDA-BFranchise Group, LLC
April 2025 FDD
Multi-Unit Development Agreement
[City, State]

APPENDIX B OWNERS' PERSONAL GUARANTY OF DEVELOPER'S OBLIGATIONS (attached)

OWNERS' PERSONAL GUARANTY OF DEVELOPER'S OBLIGATIONS ("GuarantyGUARANTY")

In consideration of, and as an inducement to, the execution of the Sky Zone Franchise
Group, LLC® Multi-Unit Development Agreement dated as of
20 (the "Agreement") by and between the Sky Zone Franchise Group, LLC
("Franchisor"), and ("Developer"), each of the
undersigned Owners of a greater than 10% interest in Developer hereby personally and
unconditionally: (1) guarantees to Franchisor and its successors and assigns, for the term of
the Agreement and thereafter as provided in the Agreement, that Developer shall punctually
pay and perform each and every undertaking, agreement and covenant set forth in the
Agreement (and any amendments) and that each and every representation of Developer made
in connection with the Agreement (and any amendments) are true, correct and complete in all
respects at and as of the time given; and (2) agrees personally to be bound by, and personally
liable for the breach of, each and every provision in the Agreement (and any amendments),
including, without limitation, the confidentiality obligations and non-competition
provisions covenants set forth therein.

Each of the undersigned waives: (a) acceptance and notice of acceptance by Franchisor of the foregoing undertakings; (b) notice of demand for payment of any indebtedness or nonperformance of any obligations hereby guaranteed; (c) protest and notice of default to any party with respect to the indebtedness or nonperformance of any obligations hereby guaranteed; (d) any right hethe undersigned may have to require that an action be brought against Developer or any other person as a condition of liability; (e) notice of any amendment to the agreement; and (f) any and all other notices and legal or equitable defenses to which hethe undersigned may be entitled.

Each of the undersigned consents and agrees that: (i) histheir direct and immediate liability under this quarantyGuaranty shall be joint and several; (ii) hethey shall render any payment or performance required under the Agreement upon demand if Developer fails or refuses to do so punctually; (iii) such liability shall not be contingent or conditioned upon pursuit by Franchisor of any remedies against Developer or any other person; (iv) Franchisor may proceed against the undersigned and Developer jointly and severally, or Franchisor may, at its option, proceed against the undersigned without having commenced any action or having obtained any judgment against Developer; and (ivv) such liability shall not be diminished, relieved or otherwise effected by any extension of time, credit or other indulgence which the Franchisor may from time to time grant to Developer or to any other person including, without limitation, the acceptance of any partial payment or performance or the compromise or release of any claims, none of which shall in any way modify or amend this guaranty Guaranty, which shall be continuing and irrevocable until satisfied in full; and (vi) the undersigned agrees to pay all reasonable attorneys' fees and all costs and other expenses incurred in any collection or attempt to collect amounts due pursuant to this undertaking or any negotiations relative to the obligations hereby guaranteed or in enforcing this undertaking against the undersigned.

It is further understood and agreed by the undersigned that the provisions, covenants and conditions of the Guaranty will inure to the benefit of our successors and assigns.

Sky Zone Franchise Group, LLC
April 2025 FDD
Multi-Unit Development Agreement
[City, State]

This Guaranty shall be governed by the governing law provisions set forth in Section 2013 (Governing Law) of the Agreement and all disputes related to it shall be resolved in accordance with the dispute resolution provisions set forth in Sections 4912 (Dispute Resolution), 2014 (Consent to Jurisdiction), and 2115 (Waiver of Punitive Damages, Jury Trial and Class Actions), 16 (Limitation of Claims) of the Agreement.

Each of the undersigned represents and warrants that, if no signature appears below for such undersigned's spouse, such undersigned is either not married or, if married, is a resident of a state which does not require the consent of both spouses to encumber the assets of a marital estate.

[Signature Page Follows]

IN WITNESS WHEREOF, each of the undersigned has hereunte affixed his or her signature, under seal on the same day and year as the Agreement was executed.

GUARANTOR(S):

Signature:

Print Name:

Signature:

Print Name:

Signature:

Print Name:

The undersigned, as the spouse of the Effective Date of the Agreement Guarantor indicated below, acknowledges and consents to the guaranty given herein by his/her spouse. Such consent also serves to bind the assets of the marital estate to Guarantor's performance of this Guaranty.

PER	CEN	TAGE	OF	OWI	NERS	HIP
INTI	ERES	TIN	DEV	ELO	PER	

GUARANTOR (S)

		(Signature)
		(Print Name)
		(Signature)
		(Print Name)
		(Signature)
		(Print Name)
l	Name of Guarantor	Name of Guarantor
l	Name of Guarantor's Spouse	Name of Guarantor's Spouse
	Signature of Guarantor's Spouse	Signature of Guarantor's Spouse

Sky Zone Franchise Group, LLC
April 2025 FDD
Multi-Unit Development Agreement
City, State

Name of Guarantor	Name of Guarantor
Name of Guarantor's Spouse	Name of Guarantor's Spouse
Signature of Guarantor's Spouse	Signature of Guarantor's Spouse

80103914v1

EXHIBIT C TO THE DISCLOSURE DOCUMENT

STATE LAW ADDENDA-TO-FDD

STATE LAW ADDENDUM TO THE DISCLOSURE DOCUMENT

The following provision is applicable to franchisees and franchises subject to state franchise registration/disclosure laws in California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin:

No statement, questionnaire, or acknowledgment signed or agreed to by you in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by us, any franchise seller, or any other person acting on behalf of us. This provision supersedes any other term of any document executed in connection with the franchise.

STATE LAW ADDENDUM - CALIFORNIA

"THE CALIFORNIA FRANCHISE INVESTMENT LAW REQUIRES THAT A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE BE DELIVERED TOGETHER WITH THE FRANCHISE DISCLOSURE DOCUMENT."

Notwithstanding anything contained in the foregoing Franchise Agreement, Multi-Unit Development Agreement and Franchise Disclosure Document ("FDD") to the contrary, the following provisions of the California Investment Law shall apply to any franchise or franchisee located in the State of California, which shall control to the extent of any inconsistency:

The following language is added to the "Special Risks to Consider About *This* Franchise" page:

Spousal Liability. While your spouse need not sign a personal guarantee unless he or she is an owner of the legal entity that is the franchisee, the fact that California is a community-property state means that both your and your spouse's marital and personal assets, including your house, could be lost if your franchise fails.

Item 1 of the FDD

This following language is added to the end of Item 1 of the Franchise Disclosure Document:

Covid-19 governmental orders resulted in temporary closings of some Parks and have led to requirements that all Parks comply with Center for Disease Control guidelines.

Item 3 of the FDD

Neither we, nor any person or franchise broker in Item 2 of the Franchise Disclosure Document is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities Exchange Act of 1934, 15 U.S.C.A. 78a et seq., suspending or expelling such persons from membership in such association or exchange.

Item 6 of the FDD

The following language is added to the end of the "Remarks" column for Line Item entitled "Interest on Late Payments": In California, the highest interest rate permitted by law is 10%.

Item 17 of the FDD

California Business and Professions Code Sections 20000 through 20043 provide rights to the franchisee concerning termination, transfer, or non-renewal of a franchise. If the Franchise Agreement contains a provision that is inconsistent with the law, the law will control.

The Franchise Agreement provides for termination upon bankruptcy. This provision may not be enforceable under federal bankruptcy law (11 U.S.C.A. Sec. 101 et seq.).

The Franchise Agreement contains a covenant not to compete that extends beyond the termination of the franchise. These provisions may not be enforceable under California law.

Section 31125 of the California Corporations Code requires us to give you a disclosure document, in a form containing the information that the commissioner may by rule or order require, before a solicitation of a proposed material modification of an existing franchise.

The Franchise Agreement requires application of the law of <u>Missourithe state in which</u> the franchisor's principal place of business is located. This provision may not be enforceable under California law.

You must sign a general release if you renew or transfer your franchise. California Corporations Code Section 31512 voids a waiver of your rights under the Franchise Investment Law (California Corporations Code Sections 31000 through 31516). Business and Professions Code Section 20010 voids a waiver of your rights under the Franchise Relations Act (Business and Professions Code Sections 20000 through 20043).

OUR WEBSITE HAS NOT BEEN REVIEWED OR APPROVED BY THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION, ANY COMPLAINTS CONCERNING THE CONTENT OF THIS WEBSITE MAY BE DIRECTED TO THE CALIFORNIA DEPARTMENT OF FINANCIAL PROTECTION & INNOVATION at www.dfpi.ca.gov.

Prospective franchisees are encouraged to consult private legal counsel to determine the applicability of California and federal laws (such as Business and Professions Code Section 20040.5, Code of Civil Procedure Section 1281, and the Federal Arbitration Act) to any provisions of a franchise agreement restricting venue to a forum outside the State of California.

The Franchise Agreement and Multi-Unit Development Agreement may limit franchisee's rights and may not be enforceable including, but not limited to: a class action waiver, jury trial waiver, limitation of time to bring claims and punitive damages waiver.

The Franchise Agreement and Multi-Unit Development Agreement are amended to add the following language:

"No disclaimer, questionnaire, clause, or statement signed by a franchisee in connection with the commencement of the franchise relationship shall be construed or interpreted as waiving any claim of fraud in the inducement, whether common law or statutory, or as disclaiming reliance on or the right to rely upon any statement made or information provided by any franchisor, broker or other person acting on behalf of the franchisor that was a material inducement to a franchisee's investment. This provision supersedes any other or inconsistent term of any document executed in connection with the franchise."

Other Provisions Unaffected: All other terms and provisions contained in the Franchise Agreement, Multi-Unit Development Agreement, and Franchise Disclosure Document shall remain in full force and effect, except to the extent specifically modified herein.

Addendum dated this day of	· · · · · · · · · · · · · · · · · · ·
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	By:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE LAW ADDENDUM - HAWAII

THESE FRANCHISES WILL BE/HAVE BEEN FILED UNDER THE FRANCHISEE INVESTMENT LAW OF THE STATE OF HAWAII. FILING DOES NOT CONSTITUTE APPROVAL, RECOMMENDATION OR ENDORSEMENT BY THE DIRECTOR OF REGULATORY AGENCIES OR A FINDING BY THE DIRECTOR OF REGULATORY AGENCIES THAT THE INFORMATION PROVIDED HEREIN IS TRUE, COMPLETE AND NOT MISLEADING.

THE FRANCHISE INVESTMENT LAW MAKES IT UNLAWFUL TO OFFER OR SELLA NYSELL ANY FRANCHISE IN THIS STATE WITHOUT FIRST PROVIDING TO THE PROSPECTIVE FRANCHISEE, OR SUBFRANCHISOR, AT LEAST SEVEN DAYS PRIOR TO THE EXECUTION BY THE PROSPECTIVE FRANCHISEE OF ANY BINDING FRANCHISE OR OTHER AGREEMENT, OR AT LEAST SEVEN DAYS PRIOR TO THE PAYMENT OF ANY CONSIDERATION BY THE FRANCHISEE, OR SUBFRANCHISOR, WHICHEVER OCCURS FIRST, A COPY OF THE OFFERING CIRCULAR, TOGETHER WITH A COPY OF ALL PROPOSED AGREEMENTS RELATING TO THE SALE OF THE FRANCHISE.

THIS OFFERING CIRCULAR CONTAINS A SUMMARY ONLY OF CERTAIN MATERIAL PROVISIONS OF THE FRANCHISE AGREEMENT AND MULTI-UNIT DEVELOPMENT AGREEMENT. THE CONTRACT OR AGREEMENT SHOULD BE REFERRED TO FOR A STATEMENT OF ALL RIGHTS, CONDITIONS, RESTRICTIONS AND OBLIGATIONS OF BOTH THE FRANCHISOR AND THE FRANCHISEE.

IN WITNESS WHEREOF,	the parties	hereto	have c	luly executed	d, sealed a	and delivered	d this
Addendum dated this	_day of		,	20			
				SKY ZONE	FRANCHI	SE GROUP, L	<u>LC</u>
				<u>By:</u>			
				=			
				Name:			
				=			
				Title:			
				=			
				FRANCHISI	EE:		

STATE LAW ADDENDUM - ILLINOIS

Notwithstanding anything contained in the foregoing Franchise Agreement, Multi-Unit Development Agreement and Franchise Disclosure Document to the contrary, the following provisions of the Illinois Franchise Disclosure Act ("Act") shall apply to any franchise located in the State of Illinois, which shall control to the extent of any inconsistency:

1. The following statements are added to the end of Item 17:

Except for the Federal Arbitration Act that applies to arbitration, Illinois law governs the Franchise Agreement and Multi-Unit Development Agreement.

In conformance with Section 4 of the Illinois Franchise Disclosure Act, any provision in a franchise agreement or multi-unit development agreement that designates jurisdiction and venue in a forum outside of the State of Illinois is void. However, a franchise agreement or multi-unit development agreement may provide for arbitration to take place outside of Illinois.

Franchisees' rights upon termination and non-renewal are set forth in Sections 19 and 20 of the Illinois Franchise Disclosure Act.

In conformance with Section 41 of the Illinois Franchise Disclosure Act, any condition, stipulation or provision purporting to bind any person acquiring any franchise to waive compliance with the Illinois Franchise Disclosure Act or any other law of Illinois is void.

2. <u>Other Provisions Unaffected:</u> All other terms and provisions contained in the Franchise Agreement or Multi-Unit Development Agreement shall remain in full force and effect, except to the extent specifically modified herein.

IN WITNESS WHEREOF, the parties here Addendum dated this day of	eto have duly executed, sealed and delivered this, 20
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	Ву:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE ADDENDUM FOR THE STATE OF INDIANA

- 1. To be added to <u>Item 3</u> of the Disclosure Document, is the following statement: There are presently no arbitration proceedings to which the Franchisor is a party.
- 2. <u>Item 17</u> of the Disclosure Document is amended to reflect the requirement under Indiana Code 23-2-2.7-1 (9), which states that any post term non-compete covenant must not extend beyond the franchisee's exclusive territory.
- 3. Item 17 is amended to state that this is subject to Indiana Code 23-2-2.7-1 (10).
- 4. Under Indiana Code 23-2-2.7-1 (10), jurisdiction and venue must be in Indiana if the franchisee so requests. This amends Article XVII of the Franchise Agreement and Article X12 of the Multi-Unit Development Agreement.
- 5. Under Indiana Code 23-2-2.7-1 (10), franchisee may not agree to waive any claims or rights.

Addendum dated this day of	to have duly executed, sealed and delivered this, 20
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	By:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE LAW ADDENDUM - MARYLAND

REQUIRED BY THE STATE OF MARYLAND

This will serve as the State Addendum for Sky Zone Franchise Group, LLC for the State of Maryland for Sky Zone Franchise Group, LLC's Franchise Disclosure Document and for its Franchise and Multi-Unit Development Agreements. The amendments to the Franchise and Multi-Unit Development Agreements included in this addendum have been agreed to by the parties.

- 1. The provision contained in the termination sections of the Franchise and Multi-Unit Development Agreements may not be enforceable under federal bankruptcy law (11 U.S.C. Section 101 et seq.).
- 2. <u>Items 5 and 7</u> of the Franchise Disclosure Document, Section 5.1 of the Franchise Agreement, and Section 6 of the Multi-Unit Development Agreement are modified to include the following statement: "Based upon the franchisor's financial condition, the Maryland Securities Commissioner has required a financial assurance. Therefore, all initial fees and payments owed by franchisees shall be deferred until the franchisor completes its pre-opening obligations under the franchise agreement. In addition, all development fees and initial payments by area developers shall be deferred until the first franchise under the development agreement opens."
- 3. <u>Item 11</u> of the Franchise Disclosure Document shall be amended to state that a franchisee may obtain an accounting of the advertising fund as required by the Maryland Franchise Registration and Disclosure Law, by requesting same in a written request to Franchisor.
- 4. <u>Item 17</u> of the Franchise Disclosure Document shall be amended at the sections dealing with the issuance of general releases to the effect that the general release required as a condition of renewal and/or assignment/transfer are not intended to nor shall they act as a release, estoppel or waiver of any liability under the Maryland Franchise Registration and Disclosure Law. The appropriate sections of the Franchise Agreement and Multi-Unit Development Agreement are hereby deemed to be amended accordingly.
- 5. The Maryland Franchise Registration and Disclosure Law prohibits a franchisor from requiring a prospective franchisee to assent to any release, estoppel or waiver of liability as a condition of purchasing a franchise. Any disclaimer regarding the occurrence and/or acknowledgment of the nonoccurrence of acts that would constitute a violation of the Franchise Law in order to purchase the franchise are not intended to nor shall they act as a release, estoppel or waiver of any liability incurred under the Maryland Franchise Registration and Disclosure Law. The appropriate sections of the Franchise Agreement and Multi-Unit Development Agreement and Franchisee Disclosure Acknowledgement Statement are amended to comply with this provision.
- 6. <u>Item 17</u> of the Franchise Disclosure Document and the appropriate sections of the Franchise Agreement and Multi-Unit Development Agreement are amended to state that any claims arising under the Maryland Franchise Registration and Disclosure Law must be brought within 3 years after the grant of the franchise.

<u>Item 17</u> of the Franchise Disclosure Document is hereby amended to state that the Franchise Agreement and Multi-Unit Development Agreement require binding arbitration, the site of which is in the state of Missourithe franchisor's principal place of business (currently, Utah), the costs of which are borne by the parties equally and any issues not decided by arbitration may be brought in a court of competent jurisdiction. The law of the Statestate of Missourithe franchisor's principal place of business (currently, Utah) governs the arbitration. However, pursuant to the Maryland Franchise Registration and Disclosure Law, a franchisee is permitted to enter into litigation with the franchisor in the State of Maryland, regardless of the language in the Franchise Agreement and Multi-Unit Development Agreement. 8. Sections 20.1 ("Acknowledgments") and 20.6 ("Financial Performance Representations") of the Franchise Agreement are hereby deleted. 9. Section 2.D. ("Grant of Development Rights Schedule Compliance") of the Multi-Unit Development Agreement is revised to delete the following language: "You representWe are relying on your representation that you conducted your own independent investigation and analysis of the prospects for the establishment of Businesses within the Development Area." 10. Section 29 ("No Guarantee of Success") of, approve the Multi Unit Development Schedule and its Development Deadlines as being reasonable and viable, and recognize that failure to comply with the Development Schedule will constitute a breach of this Agreement-is hereby deleted." 11. 9. No statement, questionnaire, or acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise. IN WITNESS WHEREOF, the parties hereto have duly executed, sealed and delivered this Addendum dated this day of , 20 . ATTEST SKY ZONE FRANCHISE GROUP, LLC By: Witness Name: Title: Witness FRANCHISEE:

STATE LAW ADDENDUM - MINNESOTA

Notwithstanding anything contained in the foregoing Franchise Agreement, Multi-Unit Development Agreement and Franchise Disclosure Document ("FDD") to the contrary, the following provisions of the Minnesota Franchise Act shall apply to any franchise or franchisee located in the State of Minnesota, which shall control to the extent of any inconsistency:

1. <u>The following paragraph is added to Section IX.B.7.5</u> of the Franchise Agreement and Item 13 of the FDD:

FRANCHISOR agrees to protect FRANCHISEE against claims of infringement or unfair competition with respect to FRANCHISEE'S authorized use of the Marks when, in the opinion of counsel to FRANCHISOR, FRANCHISEE'S rights granted therein warrant protection.

- 2. Section XVIII.GArticle 19 of the Franchise Agreement, IX.GSections 12 and 14 of the Multi-Unit Development Agreement, the State Cover Sheet and Item 17v and 17w of the FDD are modified to provide the following: Minn. Stat. §§80C.21 and Minn. Rule 2860.4400J prohibit FRANCHISOR from requiring litigation to be conducted outside Minnesota. In addition, nothing in the FDD or Franchise Agreement can abrogate or reduce any of FRANCHISEE'S rights as provided for in Minnesota Statutes, Chapter 80C, or FRANCHISEE'S rights to any procedure, forum or remedies provided for by the law of Minnesota.
- 3. Sections IV.BArticles 14 and XII.A15 of the Franchise Agreement, Article 10 of the Multi-Unit Development Agreement, and Item 17 of the FDD are hereby modified to provide the following: With respect to franchises governed by Minnesota law, FRANCHISOR will comply with Minn. Stat. Sect 80C.14, subds. 3, 4 and 5 which require, except in certain specified cases, that FRANCHISEE will be given 90 days' notice of termination (with 60 days to cure) and 180 days' notice for non-renewal of the Franchise Agreement.
- 4. <u>Sections IV.BArticles 13</u> and <u>XV.B14</u> of the Franchise Agreement, Article 6 of the Multi-Unit Development Agreement, and 17.b and 17.m of the FDD are hereby modified as follows: Any release executed in connection herewith will not apply to any claims that may arise under the Minnesota Franchise Act.
- 5. Sections XVIII.GSection 16.1 and QArticle 19 of the Franchise Agreement, Articles 12 and 15 of the Multi-Unit Development Agreement, and 17b, 17.v and 17.w of the FDD are hereby modified to the extent required by the follows:

"Minnesota Statues, Section 80C.21 and Minnesota Rule 2860.4400 (J) prohibit the franchisor from requiring litigation to be conducted outside Minnesota, requiring waiver of jury trailtrial, or requiring the franchisee to consent to liquidated damages, termination penalties or judgment notes. In addition, nothing in the Franchise Disclosure document or agreement(s) can abrogate or reduce any of franchisee's rights as provided for in Minnesota Statutes, Chapter 80C, or franchisee's rights to any procedure, forum, or remedies provided for by the laws of the jurisdiction."

6. Other Provisions Unaffected: Al	l other terms and provisions contained in the Franchise
Agreement, Multi-Unit Developm	ment Agreement, and FDD shall remain in full force and
effect, except to the extent specific	
effect, except to the extent specific	carry modified notoni.
IN WITNESS WHEREOF the part	ies hereto have duly executed, sealed and delivered this
, <u> </u>	,
Addendum dated this day of _	, 20
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
ATTEST	SKT ZONE TRANCHISE GROUT, ELC
	By:
Witness	
VV IUICSS	
	Name:
	Title:
Witness	
Willess	
	FRANCHISEE:

STATE LAW ADDENDUM - NEW YORK

REQUIRED BY THE DEPARTMENT OF LAW OF THE STATE OF NEW YORK

The following Items are required to be included within the Disclosure Document and shall be deemed to supersede the language in the Disclosure Document itself:

1. The following information is added to the cover page of the Franchise Disclosure Document:

INFORMATION COMPARING FRANCHISORS IS AVAILABLE. CALL THE STATE ADMINISTRATORS LISTED IN EXHIBIT G OR YOUR PUBLIC LIBRARY FOR SOURCES OF INFORMATION. REGISTRATION OF THIS FRANCHISE BY NEW YORK STATE DOES NOT MEAN THAT NEW YORK STATE RECOMMENDS IT OR HAS VERIFIED THE INFORMATION IN THIS FRANCHISE DISCLOSURE DOCUMENT. IF YOU LEARN THAT ANYTHING IN THE FRANCHISE DISCLOSURE DOCUMENT IS UNTRUE. CONTACT COMMISSION AND NEW YORK STATE FEDERAL TRADE DEPARTMENT OF LAW, INVESTOR PROTECTION BUREAU, 28 LIBERTY STREET, 21ST FLOOR, NEW YORK, NEW YORK 10005. THE FRANCHISOR MAY, IF IT CHOOSES, NEGOTIATE WITH YOU ABOUT ITEMS COVERED IN THE FRANCHISE DISCLOSURE DOCUMENT. HOWEVER, THE FRANCHISOR CANNOT USE THE NEGOTIATING PROCESS TO PREVAIL UPON A PROSPECTIVE FRANCHISEE TO ACCEPT TERMS WHICH ARE LESS FAVORABLE THAN THOSE SET FORTH IN THIS FRANCHISE DISCLOSURE DOCUMENT.

2. The following is added at the end of Item 3:

Except as provided above, with regard to the franchisor, its predecessor, a person identified in Item 2, or an affiliate offering franchises under the franchisor's principal trademark:

- A. No such party has an administrative, criminal or civil action pending against that person alleging: a felony, a violation of a franchise, antitrust, or securities law, fraud, embezzlement, fraudulent conversion, misappropriation of property, unfair or deceptive practices, or comparable civil or misdemeanor allegations.
- B. No such party has pending actions, other than routine litigation incidental to the business, which are significant in the context of the number of franchisees and the size, nature or financial condition of the franchise system or its business operations.
- C. No such party has been convicted of a felony or pleaded nolo contendere to a felony charge or, within the 10 year period immediately preceding the application for registration, has been convicted of or pleaded nolo contendere to a misdemeanor charge or has been the subject of a civil action alleging: violation of a franchise, antifraud, or securities law; fraud; embezzlement; fraudulent

conversion or misappropriation of property; or unfair or deceptive practices or comparable allegations.

- D. No such party is subject to a currently effective injunctive or restrictive order or decree relating to the franchise, or under a Federal, State, or Canadian franchise, securities, antitrust, trade regulation or trade practice law, resulting from a concluded or pending action or proceeding brought by a public agency; or is subject to any currently effective order of any national securities association or national securities exchange, as defined in the Securities and Exchange Act of 1934, suspending or expelling such person from membership in such association or exchange; or is subject to a currently effective injunctive or restrictive order relating to any other business activity as a result of an action brought by a public agency or department, including, without limitation, actions affecting a license as a real estate broker or sales agent.
- 3. The following is added to the end of Item 4:

Neither the franchisor, its affiliate, its predecessor, officers, or general partner during the 10-year period immediately before the date of the offering circular: (a) filed as debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code; (b) obtained a discharge of its debts under the bankruptcy code; or (c) was a principal officer of a company or a general partner in a partnership that either filed as a debtor (or had filed against it) a petition to start an action under the U.S. Bankruptcy Code or that obtained a discharge of its debts under the U.S. Bankruptcy Code during or within 1 year after that officer or general partner of the franchisor held this position in the company or partnership.

4. The following is added to the end of Item 5:

The initial franchise fee constitutes part of our general operating funds and will be used as such in our discretion.

5. The following is added to the end of the "Summary" sections of Item 17(c), titled "Requirements for franchisee to renew or extend," and Item 17(m), entitled "Conditions for franchisor approval of transfer":

However, to the extent required by applicable law, all rights you enjoy and any causes of action arising in your favor from the provisions of Article 33 of the General Business Law of the State of New York and the regulations issued thereunder shall remain in force; it being the intent of this proviso that the non-waiver provisions of General Business Law Sections 687.4 and 687.5 be satisfied.

6. The following language replaces the "Summary" section of Item 17(d), titled "Termination by franchisee":

You may terminate the agreement on any grounds available by law.

7. The following is added to the end of the "Summary" section of Item 17(j), titled "Assignment of contract by franchisor":

However, no assignment will be made except to an assignee who in good faith and judgment of the franchisor, is willing and financially able to assume the franchisor's obligations under the Franchise Agreement.

8. The following is added to the end of the "Summary" sections of Item 17(v), titled "Choice of forum", and Item 17(w), titled "Choice of law":

The foregoing choice of law should not be considered a waiver of any right conferred upon the franchisor or upon the franchisee by Article 33 of the General Business Law of the State of New York.

Addendum dated this day of	reto have duly executed, sealed and delivered this, 20
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	By:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE LAW ADDENDUM - NORTH DAKOTA

DISCLOSURE DOCUMENT REQUIRED BY THE STATE OF NORTH DAKOTA

This addendum to the Disclosure Document, Franchise Agreement, and Multi-Unit Development Agreement effectively amends and revises said documents as follows:

- 1. <u>Item 17(c)</u> of the Disclosure Document—and, Articles <u>III_13</u> and <u>XIV_14</u> of the Franchise <u>Agreement</u>, and Article 6 of the <u>Multi-Unit Development</u> Agreement are hereby amended to indicate that a franchisee shall not be required to sign a general release.
- 2. Covenants not to compete are generally considered unenforceable in the State of North Dakota, in accordance with Section 51-19-09 of the North Dakota Franchise Investment Law. <u>Item 17(r)</u> of the Disclosure Document—and, Article X<u>17</u> of the Franchise <u>Agreement and Articles 5 and 10 of the Multi-Unit Development Agreement are amended accordingly.</u>
- 3. <u>Item 17(u)</u> of the Disclosure Document, Article <u>XVI19</u> of the Franchise Agreement and Article <u>VIII12</u> of the Multi-Unit Development Agreement are amended to provide that arbitration shall be held at a site that is agreeable to all parties.
- 4. <u>Item 17(v)</u> of the Disclosure Document and the provisions of Article $\frac{XVII}{19}$ of the Franchise Agreement and Article $\frac{1X}{14}$ of the Multi-Unit Development Agreement which require jurisdiction of courts in the State of $\frac{1}{12}$ are deleted.
- 5. <u>Item 17(w)</u> of the Disclosure Document, Article <u>XVII19</u> of the Franchise Agreement and Article <u>IX13</u> of the Multi-Unit Development Agreement are amended to indicate that the agreements are to be construed according to the laws of the State of North Dakota.
- 6. Apart from civil liability as set forth in Section 51-19-12 N.D.C.C., which is limited to violations of the North Dakota Franchise Investment Law (registration and fraud), the liability of the franchisor to a franchisee is based largely on contract law. Despite the fact that those provisions are not contained in the franchise investment law, those provisions contain substantive rights intended to be afforded to North Dakota residents. Therefore, North Dakota franchisees will not be required to waive their rights under North Dakota law.
- 7. The provisions of Article XVII<u>19</u> of the Franchise Agreement and Article <u>IX</u><u>15</u> of the Multi-Unit Development Agreement which require a franchisee to consent to: (1) a waiver of trial by jury and (2) a waiver of exemplary and punitive damages are contrary to Section 51-19-09 of the North Dakota Franchise Investment Law and are hereby deleted.
- 8. The provisions of Article XVII<u>19</u> of the Franchise Agreement and Article <u>IX<u>16</u> of the Multi-Unit Development Agreement which require a franchisee to consent to a limitation of claims are hereby amended to state that the statute of limitations under North Dakota law applies.</u>

IN WITNESS WHEREOF, the parties hereto Addendum dated this day of	· · · · · · · · · · · · · · · · · · ·
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	By:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE LAW ADDENDUM - RHODE ISLAND

DISCLOSURE DOCUMENT REQUIRED BY THE STATE OF RHODE ISLAND

The following amends <u>Item 17</u> and is required to be included within the Disclosure Document and shall be deemed to supersede the language in the Disclosure Document itself:

Section 19-28.1-14 of the Rhode Island Franchise Investment Act provides that:

"A provision in a franchise agreement restricting jurisdiction or venue to a forum outside of this state or requiring the application of the laws of another state is void with respect to a claim otherwise enforceable under this Act."

IN WITNESS WHEREOF, the parties here Addendum dated this day of	eto have duly executed, sealed and delivered thi, 20
ATTEST	SKY ZONE FRANCHISE GROUP, LLC
	By:
Witness	
	Name:
	Title:
Witness	
	FRANCHISEE:

STATE LAW ADDENDUM TO THE DISCLOSURE DOCUMENT- WASHINGTON

1. The following language is added to the end of Item 17 of the Franchise Disclosure Document:

WASHINGTON ADDENDUM TO THE FRANCHISE DISCLOSURE DOCUMENT, THE FRANCHISE AGREEMENT, AND ALL RELATED AGREEMENTS

The provisions of this Addendum form an integral part of, are incorporated into, and modify the Franchise Disclosure Document, the franchise agreement, and all related agreements regardless of anything to the contrary contained therein. This Addendum applies if: (a) the offer to sell a franchise is accepted in Washington; (b) the purchaser of the franchise is a resident of Washington; and/or (c) the franchised business that is the subject of the sale is to be located or operated, wholly or partly, in Washington.

- <u>L. Conflict of Laws.</u> In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.
- **2. Franchisee Bill of Rights.** RCW 19.100.180 may supersede <u>provisions in</u> the franchise agreement <u>inor related agreements concerning</u> your relationship with the franchisor, including <u>in</u> the areas of termination and renewal of your franchise. There may also be court decisions <u>which maythat</u> supersede the franchise agreement <u>inor related agreements concerning</u> your relationship with the franchisor. <u>Franchise agreement provisions</u>, including <u>the areas of termination and renewal of your franchise</u> those summarized in Item 17 of the Franchise Disclosure Document, are subject to state law.
- 3. Site of Arbitration, Mediation, and/or Litigation. In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.
- 4. General Release. A release or waiver of rights executed by a franchisee may not include rights in the franchise agreement or related agreements purporting to bind the franchisee to waive compliance with any provision under the Washington Franchise Investment Protection Act or any rulerules or orderorders thereunder is void except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable., in accordance with RCW 19.100.220(2). In addition, any such release or waiver executed in connection with a renewal or transfer of a franchise is likewise void except as provided for in RCW 19.100.220(2).

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party

seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained

- 5. Statute of Limitations and Waiver of Jury Trial. Provisions contained in the franchise agreement or related agreements that unreasonably restrict or limit the statute of limitations period for claims under the Washington Franchise Investment Protection Act, or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.
- <u>6.</u> <u>Transfer Fees. Transfer fees are collectable only to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.</u>
- 7. <u>Termination by Franchisee.</u> The franchisee may terminate the franchise agreement under any grounds permitted under state law.
- 8. <u>Certain Buy-Back Provisions.</u> Provisions in franchise agreements or related agreements that permit the franchisor to repurchase the franchisee's business for any reason during the term of the franchise agreement without the franchisee's consent are unlawful pursuant to RCW 19.100.180(2)(j), unless the franchise is terminated for good cause.
- <u>9.</u> <u>Fair and Reasonable Pricing.</u> Any provision in the franchise agreement or <u>elsewhere</u>related agreements that <u>conflict</u> with these limitations are void and unenforceable in Washington.
- RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting arequires the franchisee from (i) solicitingto purchase or hiringrent any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such product or service for more than a fair and reasonable price is unlawful under RCW 19.100.180(2)(d).
- <u>Maiver of Exemplary & Punitive Damages.</u> RCW 19.100.190 permits franchisees to seek treble damages under certain circumstances. Accordingly, provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

ADDENDUM TO THE FRANCHISE AGREEMENT - WASHINGTON

	THI	SPIDE	Q is made	by and b	etween S l	VV 70NE	FFRANC	THICE	CROUP	HC
				•						
a Missouri	limited	-liability	company	whose whose whose whose whose whose whose whose with the contract of the cont	principal	business	address	is 86	N. Univ	ersitv
		_								-
Avenue,	Suite	330,	F10V0,	Utan	04001	(we,	us,	OI	our),	anu
-				a(n)						
			,	u(11) _						
("you" or ")	your").									

1. BACKGROUND. We and you are parties to that certain Franchise Agreement dated _______, 20_____ (the "Franchise Agreement"). This Rider is annexed to and forms part of the Franchise Agreement. This Rider is being signed because (a) you are domiciled in Washington; and/or (b) the Sky Zone Park you will operate under the Franchise Agreement will be located or operated in Washington; and/or (c) any of the offer or sales activity relating to the Franchise Agreement occurred in Washington.

2. WASHINGTON LAW. The following paragraphs are added to the end of the Franchise Agreement:

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the franchise agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the franchise agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under the Washington Franchise Investment Protection Act or any rule or order thereunder requiring franchisees to waive exemplary, punitive, or similar damages are void, except when executed pursuant to a negotiated settlement after the agreement is in effect and where the parties are represented by independent counsel, in accordance with RCW 19.100.220(2).

11. Franchisor's Business Judgement. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under the Act, or rights or remedies under the Act such as a right to a jury trial, in the franchise agreement or related agreements stating that the franchisor may exercise its discretion on the basis of its reasonable business judgment may not be enforceable limited or superseded by RCW 19.100.180(1), which requires the parties to deal with each other in good faith.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

- <u>12.</u> <u>Indemnification.</u> Any provision in the franchise agreement or related agreements requiring the franchisee to indemnify, reimburse, defend, or hold harmless the franchisor or other parties is hereby modified such that the franchisee has no obligation to indemnify, reimburse, defend, or hold harmless the franchisor or any other indemnified party for losses or liabilities to the extent that they are caused by the indemnified party's negligence, willful misconduct, strict liability, or fraud.
- 13. Attorneys' Fees. If the franchise agreement or related agreements require a franchisee to reimburse the franchisor for court costs or expenses, including attorneys' fees, such provision applies only if the franchisor is the prevailing party in any judicial or arbitration proceeding.
- <u>14.</u> <u>Noncompetition Covenants.</u> Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any <u>provisions provision</u> contained in the franchise agreement or elsewhere that <u>conflict onflicts</u> with these limitations <u>areis</u> void and unenforceable in Washington.
- **15.** <u>Nonsolicitation Agreements.</u> RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the franchise agreement or elsewhere are void and unenforceable in Washington.

ADDENDUM TO THE MULTI-UNIT DEVELOPMENT AGREEMENT - WASHINGTON

The State of Washington has a statute, RCW 19.100.180, which may supersede the Multi-Unit Development Agreement in your relationship with the Franchisor, including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the Multi-Unit Development Agreement in your relationship with the Franchisor, including the areas of termination and renewal of your franchise.

In the event of a conflict of laws, the provisions of the Washington Franchise Investment Protection Act, Chapter 19.100 RCW will prevail.

RCW 19.100.180 may supersede the multi-unit development agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise. There may also be court decisions which may supersede the multi-unit development agreement in your relationship with the franchisor including the areas of termination and renewal of your franchise.

In any arbitration or mediation involving a franchise purchased in Washington, the arbitration or mediation site will be either in the state of Washington, or in a place mutually agreed upon at the time of the arbitration or mediation, or as determined by the arbitrator or mediator at the time of arbitration or mediation. In addition, if litigation is not precluded by the multi-unit development agreement, a franchisee may bring an action or proceeding arising out of or in connection with the sale of franchises, or a violation of the Washington Franchise Investment Protection Act, in Washington.

A release or waiver of rights executed by a franchisee may not include rights under

- <u>acknowledgment signed or agreed to by a franchisee in connection with the commencement of the franchise relationship shall have the effect of (i) waiving any claims under any applicable state franchise law, including fraud in the inducement, or (ii) disclaiming reliance on any statement made by any franchisor, franchise seller, or other person acting on behalf of the franchisor. This provision supersedes any other term of any document executed in connection with the franchise.</u>
- <u>agreement</u> or related agreements that prohibits the franchisee from communicating with or complaining to regulators is inconsistent with the express instructions in the Franchise Disclosure Document and is unlawful under RCW 19.100.180(2)(h).
- 18. Advisory Regarding Franchise Brokers. Under the Washington Franchise Investment Protection Act-or any rule or order thereunder except when executed pursuant to a negotiated settlement after the agreement is in effect and where, a "franchise broker" is defined as a person that engages in the parties are represented by independent counsel. Provisions such as those which unreasonably restrict or limit the statute of limitations period for claims under business of the Act, offer or rights or remedies under the Act such as a right to a jury trial, may not be enforceable.

Transfer fees are collectable to the extent that they reflect the franchisor's reasonable estimated or actual costs in effecting a transfer.

Pursuant to RCW 49.62.020, a noncompetition covenant is void and unenforceable against an employee, including an employee of a franchisee, unless the employee's earnings from the party seeking enforcement, when annualized, exceed \$100,000 per year (an amount that will be adjusted annually for inflation). In addition, a noncompetition covenant is void and unenforceable against an independent contractor of a franchisee under RCW 49.62.030 unless the independent contractor's earnings from the party seeking enforcement, when annualized, exceed \$250,000 per year (an amount that will be adjusted annually for inflation). As a result, any provisions contained in the multi-unit development agreement or elsewhere that conflict with these limitations are void and unenforceable in Washington.

RCW 49.62.060 prohibits a franchisor from restricting, restraining, or prohibiting a franchisee from (i) soliciting or hiring any employee of a franchisee of the same franchisor or (ii) soliciting or hiring any employee of the franchisor. As a result, any such provisions contained in the multiunit development agreement or elsewhere are void and unenforceable in Washington.

IN WITNESS WHEREOF, the sale of franchises. A franchise broker represents the franchisor and is paid a fee for referring prospects to the franchisor and/or selling the franchise. If a franchisee is working with a franchise broker, franchisees are advised to carefully evaluate any information provided by the franchise broker about a franchise.

day of
SKY ZONE FRANCHISE GROUP, I
By:
Name:
Title:

EXHIBIT D TO THE DISCLOSURE DOCUMENT

OPERATIONS MANUAL TABLE OF CONTENTS OF MANUAL



OPERATIONS MANUAL

Table of Contents

SKY ZONE OPERATIONS MANUAL	II
Manual Purpose and Limitations	iii
Confidentiality Agreement	
Disclaimer	
TABLE OF CONTENTS	
A. INTRODUCTION	2
Welcome Letter	
Who Are We?	
History of Sky Zone	
Services Provided to Franchise Partners	
Franchise Partner Responsibilities	7
Paying Other Fees	
B. PRE-OPENING PROCEDURES	12
Pre-Opening Responsibilities	
Establishing Your Business Form	
Real Estate Site Selection	. 15
Space Planning Process	
Required Utilities and Services	
Required Licenses and Permits	
Setting Up Bank Accounts	
Procuring Required Insurance	
Meeting Your Tax Obligations	
Conducting Your Grand Opening	
Digital Setup	
Section B Forms	
C. HUMAN RESOURCES	
Federal and State Law Resources	
EEOC Guidelines	
Laws Regarding Harassment and Discrimination	
Wage and Labor Laws	
Immigration Reform/Control Act	
Profile of Ideal Sky Zone Team Member	
Job Descriptions	
Recruiting Team Members	
Hiring on a Probationary Period	
Personnel Policies	
Time-Tracking Procedures	
Uniform and Dress Code	
Performance Evaluations	
Progressive Discipline	
Termination/Separation	
Section C Forms	
D. LEARNING & DEVELOPMENT	
Tips for Training and Communication	
Park Operations Training	98



OPERATIONS MANUAL

Justie aupport	99
Sky Zone University (eLearning)	100
Ongoing Training Process	102
E. GUEST EXPERIENCE	. 103
Defining the Guest Experience	103
Managing the Guest Experience	
Answering Guest Questions	
Explaining Sky Zone Policies	
Core Menu Offerings	
Selling Open Jump Tickets	
Multi-Day Pass	
Membership	
Sky Zone Price Ranges	
Fransacting Sales	
Section E Forms	
F. BIRTHDAY PROCEDURES	
Selling Birthday Parties	
Pre-Birthday Party Procedures	
Day of Birthday Party Procedures	
G. EVENT PROCEDURES	
Selling Corporate Events	
• 1	
H. PROGRAM PROCEDURES	
Core Programs	
Glow	_
Lock-ins	
Little Leapers	
Fitness Programs	
Overview of SkyFit Classes	
. MANAGING A SKY ZONE TRAMPOLINE PARK	
Hours of Operation	
Overview of Responsibilities	
Managing Personnel	
Managing the Guest Experience	
nventory Management	
Operational and Financial Reporting	
Loss Prevention Techniques	
Required Franchise Partner Reporting	
J. COURT RULES AND RISK MANAGEMENT	. 170
Executive CircusTrix, LLC Board Corporate Safety Philosophy	170
Setting the Court Capacity	170
General Court Rules for Participants	172
Dodgeball Rules	175
SkySlam Rules	177
Foam Zone Rules	
Fhe Rules Speech	
Court Monitor Responsibilities	180
Supervision	181
Rules Delivery & Guest Engagement	
Policies	



OPERATIONS MANUAL

General Risk Management Procedures	191
Park Incidents	194
Security Issues	217
Legal Responsibilities and Issues	220
SPECIAL PROCEDURES	223
Covid-19 Incident Reporting	225
K. MAINTAINING YOUR PARK	227
Cleaning Program	227
COURT MAINTENANCE MANUAL	230
Preface	230
Courts Overview	231
Equipment Inspections and Maintenance	233
Ordering	250
Repair and Installation Procedures	259
Court Materials	279
Section K Forms	283
L. MARKETING AND PROMOTION	292
Required Advertising Expenditures	292
Brand Standards	294
Obtaining Advertising Approval	295
Digital Advertising Policy	296
Public Relations/Community Involvement	7
Outside Sales	8
Section L Forms	8
M. PARK TECHNOLOGY	11
Sky Zone Digital Support Guide	11
Technology Points of Contact	

A. INTRODUCTION-6

Welcome Letter	6
History of Sky Zone	7
Services Provided to Franchise Partners	
Franchise Partner Responsibilities	
B. PRE OPENING PROCEDURES	
Pre-Opening Responsibilities	
Establishing Your Business Form	
Real Estate Site Selection	
Space Planning Process	
Fixtures, Equipment, Inventory, and Supplies	25
Required Utilities and Services	
Required Licenses and Permits	
Setting Up Bank Accounts	
Procuring Required Insurance	28
Meeting Your Tax Obligations	29
Conducting Your Grand Opening	
Digital Setup	31
Section B Forms	32
C. HUMAN RESOURCES	34
Federal and State Law Resources	34
EEOC Guidelines	34
Laws Regarding Harassment and Discrimination	
Wage and Labor Laws	35
Immigration Reform/Control Act	35
Profile of Ideal Sky Zone Team Member	35
Job Descriptions	36
Recruiting Team Members	46
Hiring on a Probationary Period	52
Personnel Policies	53
Time Tracking Procedures	53
Uniform and Dress Code	54
Performance Evaluations	55
Progressive Discipline	57
Termination/Separation	
Section C Forms	61
D. LEARNING & DEVELOPMENT	100
Tips for Training and Communication	100
Park Operations Training	101
Onsite Support	102
Sky Zone University (eLearning)	
Ongoing Training Process	105
E. GUEST EXPERIENCE	106
Defining the Guest Experience	106

 $\frac{1_{\underline{Sky\ Zone\ Franchise\ Group,\ LLC}}}{\underline{\underline{April\ 2025\ FDD}}}_{\underline{Ex.\ E-List\ of\ Franchisees}}$

Managing the Guest Experience	110
Answering Guest Questions	111
Explaining Sky Zone Policies	114
Core Menu Offerings	115
Selling Open Jump Tickets	
Multi Day Pass	
Membership	117
Sky Zone Price Ranges	
Transacting Sales	119
Section E Forms	<u>122</u>
F. BIRTHDAY PROCEDURES	127
Selling Birthday Parties	127
Pre-Birthday Party Procedures	135
Day of Birthday Party Procedures	
G. EVENT PROCEDURES	
Selling Corporate Events	139
H. PROGRAM PROCEDURES	143
Core Programs	
Glow	144
Lock Ins	148
Toddler Time	149
Fitness Programs	
Overview of SkyFit Classes	
I. MANAGING A SKY ZONE TRAMPOLINE PARK	156
Hours of Operation	
Overview of Responsibilities	157
Managing Personnel	159
Managing the Guest Experience	
Inventory Management	165
Operational and Financial Reporting	
Loss Prevention Techniques	168
Required Franchise Partner Reporting	169
J. COURT RULES AND SAFETY MANAGEMENT	170
Executive CircusTrix, LLC Board Corporate Safety Philosophy	170
Setting the Court Capacity	-170
General Court Rules for Participants	172
General Court Safety Rules	<u>174</u>
With Restrictions/Content/Warning Box & Play at Your Own Risk	174
Guest Patron Safety Rules	175
Dodgeball Safety & Game Rules	176
SkySlam Rules	179
Foam Zone Rules	181
Toddler Time	183
The Rules Speech	184
Court Monitor Responsibilities—	184

Supervision	18€
Rules Delivery & Guest Engagement	188
Policies	100
General Safety Management Procedures	
Park Incidents	
Security Issues	221
Legal Responsibilities and Issues	224
SPECIAL PROCEDURES	
RISK/LEGAL/SAFETY TEAM RESOURCES	228
Covid-19 Incident Reporting	
K. MAINTAINING YOUR PARK	
Cleaning Program	
Court Maintenance Manual	
Preface	232
Courts Overview	
Equipment Inspections and Maintenance	
Ordering	
Repair and Installation Procedures	
Court Materials	
Section K Forms	
L. MARKETING AND PROMOTION	295
Required Advertising Expenditures	295
Brand Standards	207
Obtaining Advertising Approval	
Digital Advertising Policy	
Public Relations/Community Involvement	
Outside Sales	
Section L Forms	
M. PARK TECHNOLOGY	
Sky Zone Digital Support Guide	
Tashnalagy Points of Contact	11

TOTAL PAGES: 312

EXHIBIT E TO THE DISCLOSURE DOCUMENT

LIST OF FRANCHISEES

(as of February 29 December 31, 2024)

Franchisee	Phone	Address	City	State/ Province	Zip Code
Kim Moore	(256) 980-1700	250 Mangum St	Florence	AL	35630
Chris and Kim Fudge	(334) 239-2587	5544 Atlanta Hwy	Montgomery	AL	36117
Cheryl Farago and Katie Moore	(480) 226-1842	1095 S. Arizona Ave	Chandler	AZ	85286
Usman Rao	(602) 704-3404	3636 W Southern Avenue	Laveen	AZ	85041
Matthew Frick	(623) 979-4000	9040 W Larkspur Dr #134	Peoria	AZ	85381
Chance Holmes, Matthew Frick	(602) 493-5867	4857 E Greenway Rd Ste A	Phoenix	AZ	8525 4
Chance Holmes, Matthew Frick	(602) 493-5867	4857 E Greenway Rd Ste A	Scottsdale	AZ	85254
Joshua Rathweg, Stephan Rathweg	(858) 400-4550	6133 Innovation Way, Ste 102	Carlsbad	CA	92009
Faisal Zia, Mohammed Zia, Farhan Qadri, Irfan Ahmed	(562) 203-1333	10755 South St ,	Cerritos	CA	90703
Faisal Zia, Farhan Qadri, Irfan Ahmed	(626) 986-0106	1314 N Azusa Ave	Covina	CA	91722
Guminder Garcha	(510) 246-3098	39177 Farwell Dr	Freemont	CA	94538
Usman Rao	(916) 314-0025	3132 Dwight Rd #900	Elk Grove	CA	95758
Usman Rao		8787 San Ysidro Ave	Gilroy	CA	95020
<u>Todd Carlson</u>	(707) 492-5867	828 Alamo Dr	<u>Vacaville</u>	<u>CA</u>	<u>95688</u>
AJ Ahmad	(714) 249-7676	1411 Village Way	Santa Ana	CA	92705
Brag Bergum	(415) 948-4724	3215 Coffey Lane	Santa Rosa	CA	95403
Joshua Rathweg, Stephan Rathweg	(858) 779-2377	860 Los Vallecitos Blvd	San Marcos	CA	92069
Joshua Rathweg, Stephan Rathweg, Joseph Conway, Johnnie Hernandez	(209) 888-0900	5358 Pacific Ave	Stockton	CA	95207
Todd Carlson	(707) 492-5867	828 Alamo Dr	Vacaville	CA	95688
Mohammed Zia	(805) 804-9555	2825 Johnson Dr	Ventura	CA	93003
Jim Bellino	(714) 415-5867	1025 Westminster Mall Suite 2086A	Westminster	CA	92683
Bryan Hendershot, Chad Tinney	(719) 354-2626	1750 E Woodmen Rd	Colorado Springs	СО	80920
Edward Kim	(203) 902-5437	360 Dr Martin Luther King Jr Dr	Norwalk	CT	06854
Usman Rao	(203) 880-5577	25 Trefoil Dr	Trumbull	CT	06611
Eric & Alina Carr	(302) 449-1252	211 Executive Dr #3	Newark	DE	19702

 $\frac{1}{2} \frac{\text{Sky Zone Franchise Group, LLC}}{\text{April 2025 FDD}} \\ \frac{\text{Ex. E - List of Franchisees}}{\text{Ex. E - List of Franchisees}}$

Franchisee	Phone	Address	City	State/ Province	Zip Code
Larry Roberts	(352) 404-4134	2510 S Hwy 27	Clermont	FL	34711
Rachel Fabricant, Keith McAdams	(305) 640-5424	10200 SW 186th St	Cutler Bay	FL	33157
Rachel Fabricant, Keith McAdams	(305) 640-5424	5450 NW 82nd Ave	Doral	FL	33166
James Fiore, Christopher Fiore	(727) 900-5867	13000 66th St N	Largo (Clearwater)	FL	33773
Han Yu	(352) 304-5050	2400 SW College Rd	Ocala	FL	34471
Roger Duncan	(954) 417-3999	1834 SW 2nd St	Pompano Beach	FL	33069
Michael Todd, Michele Todd	(321) 265-5867	624 Barnes Blvd	Rockledge	FL	32955
Paul Orsino	(941) 363-6359	6180 Edgelake Drive	Sarasota	FL	34240
Husein Khimani	(678) 324-9590	3520 Gravel Springs Rd	Buford	GA	30519
Joshua Rathweg	(762) 253-3854	3716 Gentian Blvd	Columbus	GA	31907
Yong Jung, Han Yu	(478) 239-2900	245 Tom Hill Sr Blvd Suite 100B	Macon	GA	31210
Noorali Somani	(770) 914-9444	165 Mill Rd	McDonough	GA	30253
Jeff Palmer, Cody Palmer	(319) 382-2222	5515 Council St NE	Cedar Rapids	IA	52402
John Sawhill	(515) 986-2524	1300 SE Gateway Dr Suite 103	Grimes	IA	50111
Faraz Syed	(331) 207-4759	87 Executive Dr	Aurora	II.	60504
Thomas Williamson	(630) 478-8339	325 W Lake St Ste A	Elmhurst	IL.	60126
Roger Duncan	(618) 589-1111	10850 Lincoln Trail	Fairview Heights	IL	62208
Faraz Syed	(815) 314-4890	2318 Essington Rd	Joliet	IL	60435
Ram Talluri, Patrick Rexroad	(217) 331-6600	3454 Liberty Dr	Springfield	IL	62704
Craig Love	(812) 730 4759	49 N Green River Rd	Evansville	IN	47715
Gary & Joanna Kingery	(317) 759-9925	4150 Kildeer Dr	Indianapolis	IN	4 6237
Gary Kingery	(317) 268-3200	851 Columbia Rd Ste 172	Plainfield	IN	4 6168
Justin Tauber, Kyle Ropac, Dennis Caudill	(219) 227-3777	1035 Eagle Ridge Dr	Schererville	IN	4 6375
John Sawhill	(913) 213-5900	6495 Quivira Rd	Shawnee	KS	66216
L Dee Murdock	(785) 337-8400	1801 SW Wanamaker Rd Ste B20	Topeka	KS	66604
Chad Pannunzio, Michael Bonfini	(859) 629-4455	3644 Boston Rd #130	Lexington	KY	40514
Jiya Patel, Vimalkumar Patel	(501) 546-7744	4200 Outer Loop	Louisville	KY	40219
Chris and Kim Fudge	(337) 417-8000	3814 Ambassador Caffery Pkwy	Lafayette	LA	70503
Paul Facchina	(301) 683 7997	1300 Crain Hwy	Bowie	MD	20716
Vinod Patel	(301) 420 5867	12114 Insurance Way	Hagerstown	MD	21740
Craig Hansen, Vinod Patel, Roger Duncan , Ryan Williams	(410) 874-5050	8645 Pulaski Hwy	East Baltimore/Rosedale	MD	21237
Vinod Patel, Roger Duncan	(301) 420-5867	12114 Insurance Way	<u>Hagerstown</u>	<u>MD</u>	<u>21740</u>
Craig HansenRoger Duncan,	(410) 842-1661	23 W Aylesbury Rd	Timonium	MD	21093

Sky Zone Franchise Group, LLC
April 2025 FDD
Ex. E – List of Franchisees

Franchisee	Phone	Address	City	State/ Province	Zip Code
Vinod Patel, Ryan Williams					
Paul Facehina <u>Tunde</u> Ogungbade, Omololu Fatukasi, Niyi Hebert	(301) 861-2600	4390 Crain Hwy White Plains Waldorf		MD	20695
Roger Duncan	(616) 214-4221	3640 29th Street SE	Kentwood	<u>MI</u>	<u>49512</u>
Roger Duncan	<u>(586) 930-0600</u>	50810 Sabria Drive	Shelby Township	<u>MI</u>	<u>48310</u>
Roger Duncan	(734) 981-0007 (734) 981-0007	425500 <u>42550</u> Executive Dr <u>Drive</u>	Canton	MI	48188
Roger Duncan	(586) 930-0600	50810 Sabrina Dr	Shelby Township	MI	48315
Roger Duncan	(616) 214-4221	3636 29 th St. E.	Kentwood	MI	49512
Craig Love	(228) 207-8969	2600 Beach Blvd. Unit 18	Biloxi	MS	39531
Ryan & Rebecca Przysucha	(601) 899-0015	2000 County Line Rd Suite B	Ridgeland	MS	39157
Aaron Jurgensmeyer, Charles Jurgensmeyer	(573) 309-9600	1201 American Pkwy	Columbia	MO	65202
Roger Duncan	(636) 364-4444	631 Gravois Rd	Fenton	MO	63026
John Sawhill	(816) 457-0068	2801 NE McBaine Dr	Lee's Summit	MO	64064
Matt Strelo, Kim Strelo	(314) 384-4075	7375 Watson Road	<u>Shrewsbury</u>	<u>MO</u>	<u>63119</u>
Matt Strelo, Kim Strelo	(636) 321-4024	61 N Central Drive	O'Fallon	<u>MO</u>	<u>63366</u>
Roger Duncan	(636) 364-4444	631 Gravois Road	<u>Fenton</u>	<u>MO</u>	<u>63026</u>
<u>Craig Love</u>	(228) 207-8969	2600 Beach Blvd. Unit 18	<u>Biloxi</u>	<u>MS</u>	<u>39531</u>
Ryan & Rebecca Przysucha	<u>(601) 899-0015</u>	2000 County Line Rd Suite B	Ridgeland	<u>MS</u>	<u>39157</u>
Pralhad & Roopal Khatri	<u>(919) 425-0800</u>	1720 Guess Rd Ste 90	<u>Durham</u>	<u>NC</u>	<u>27701</u>
<u>Jerry & Erica Carpenter</u>	<u>(336) 550-1800</u>	1572-A, Highwoods Blvd	<u>Greensboro</u>	<u>NC</u>	<u>27410</u>
Brian Spicker, Alden Milam	<u>(704) 709-9050</u>	7325 Smith Corners Blvd	North Charlotte	<u>NC</u>	<u>28269</u>
Brian Spicker, Alden Milam	<u>(704) 208-4040</u>	10200 Centrum Pkwy	<u>Pineville</u>	<u>NC</u>	<u>28134</u>
Alan & Gina Wilson	<u>(919) 948-4450</u>	2101 Westinghouse Blvd Suite 111	<u>Raleigh</u>	<u>NC</u>	<u>27604</u>
<u>Drazen Samardzic, Dustin</u> <u>Maier</u>	<u>(701) 354-9099</u>	5003 Ottawa St	<u>Bismarck</u>	<u>ND</u>	<u>58503</u>
<u>John Sawhill</u>	<u>(701) 478-8484</u>	<u>940 40th St S</u>	<u>Fargo</u>	<u>ND</u>	<u>58103</u>
Benjamin & Tracy Babcock	(402) 401-7600	4215 S 133rd St	Omaha	NE	68137
Mohammed Zia	(702) 966-4080	7440 Dean Martin Dr, Ste 201	Las Vegas	NV	89139
John Sawhill	(603) 413-3533	365 Lincoln St.	Manchester	NH	03103
Michael Janay	(201) 574-1800	80 Commerce Dr	Allendale	NJ	07401
Usman Rao	(609) 587-5867	17 Quakerbridge Plaza Unit B		NJ	08619
Abraham Tress	(609) 398-4759	1001 New Hampshire Ave	Lakewood	NJ	08701
David Hustrulid, Andre Gomez, Jack Miller, Dan	(856) 600-4402	2834 NJ-73	Maple Shade	NJ	08052
Vrebalovieh Irfan Ahmed, Faisal Zia, Farhan Qadri, Andrew Del Duke					
Michael Janay	(973) 527-7000	61 International Drive	Mt Olive	NJ	07828

Sky Zone Franchise Group, LLC
April 2025 FDD
Ex. E – List of Franchisees

Franchisee	Phone	Address	City	State/ Province	Zip Code
		South Unit ANCB			
Mike Janay	(973) 396-4001	60 Chapin Rd	Pine Brook	NJ	07058
Elliot Field, Gary Field, Dolly Field	(908) 756-5867	600 Hadley Rd	South Plainfield	NJ	07080
Michael Janay	(973) 671-5100	25 US-22	Springfield	NJ	07081
Joshua Rathwag <u>Rathweg,</u> <u>Stephan Rathweg</u>	(575) 300-2651(575) 300-2651	805 S. Solano Driver <u>Drive</u>	Las Cruces	NM	88001
Mohammed Zia	<u>(702) 966-4080</u>	7440 Dean Martin Dr, Ste 201	<u>Las Vegas</u>	NV	<u>89139</u>
Rusty Saunders Roger Duncan	(518) 417-3838	22 Clifton Country Rd Suite 99C	Clifton Park	NY	12065
Rusty Saunders Roger <u>Duncan</u>	(518) 836-5867	235 Corinth Rd	Queensbury <u>/Glen Falls</u>	NY	12804
Steve Merwin, Pam Tuohey	(315) 928-3737	3179 Erie Blvd E4155 NY-31	Syracuse <u>Clay</u>	NY	13214 <u>13041</u>
Brian Spicker, Alden Milam	(704) 709-9050	7325 Smith Corners Blvd	Charlotte	NC	28269
Pralhad & Roopal Khatri	(919) 425-0800	1720 Guess Rd Ste 90	Durham	NC	27701
Jerry & Erica Carpenter	(336) 550-1800	1572-A, Highwoods Blvd	Greensboro	NC	27410
Brian Spicker, Alden Milam	(704) 208-4040	10200 Centrum Pkwy	Pineville	NC	28134
Alan & Gina Wilson	(919) 948-4450	2101 Westinghouse Blvd Suite 111	Raleigh	NC	27604
Drazen Samardzic, Dustin Maier	(701) 354-9099	5003 Ottawa St	Bismarck	ND	58503
John Sawhill	(701) 478 8484	940-40th St S	Fargo	ND	58103
Charles Hallis	(234) 249-3030	6217 Chittenden Rd	Boston Heights	ОН	44236
Chas Hallis	(330) 538-8300	4381 Whipple Ave NW	Canton	ОН	44718
Joel Karg	(937) 396-6600	976 Senate Dr	Dayton	OH	4545 9
Cyrus Kharche <u>Roger</u> Duncan	(440) 467-5867	750 Alpha Dr	Highland Heights	ОН	44143
Roger Duncan	(440) 467-5867	31500 Viking Pkwy	<u>Westlake</u>	<u>OH</u>	<u>44145</u>
Cyrus Kharche	(513) 342-0872	11745 Commons Drive	<u>Springdale</u>	<u>OH</u>	45246
Imed Jmiai	(419) 549-5818	1730 N. Union St	Lima	ОН	45801
Cyrus Kharche	(513) 342 0872	11745 Commons Dr.	Springdale	OH	45246
Cyrus Kharche	(440) 467-5867	31500 Viking Pkwy	Westlake	OH	44145
Bryan Hendershot, Andy Gray, Larry Haskett, Irfan Ahmed, Faisal Zia	(405) 359-1631	2525 S Broadway	Edmond	OK	73013
Bryan Hendershot	(918) 877-7700	8306 E 61st St	Tulsa	OK	74133
Sarah Spronehle	(503) 616-3986	2990 NE Hogan Dr	Gresham	OR	97030
Roger Duncan	(610) 759-9663	2285 Schoenersville Rd	Bethlehem Allentown	PA	18017
Joel Karg, David Beckett	(724) 251-6100	281 Georgetown Rd	Canonsburg	PA	15317
Richard Ferns, Philip Stoops Faraz Syed	(215) 642-8777	145 Liberty Ln	Chalfont	PA	18914
Richard Ferns, Philip Stoops Faraz Syed	(484) 418-1500	10 Conchester Rd	Glen Mills	PA	19342

Sky Zone Franchise Group, LLC
April 2025 FDD
Ex. E – List of Franchisees

Franchisee	Phone	Address	City	State/ Province	Zip Code
Vinod Patel Roger Duncan	(717) 208-8922	1701 Hempstead Rd #102	Lancaster	PA	17601
Craig Hansen, Vinod Patel, Ryan Williams Roger Duncan	(717) 208-8908	95 Gateway Dr	Mechanicsburg	PA	17050
Joel Karg, David Beckett	(412) 810-0200	160 Levin Way	Monroeville	PA	15146
Michael Todd	(215) 516-5867	9490 Blue Grass Rd	Philadelphia	PA	19114
Mike Crounse, Jeff Bowne	(570) 316-0500	195 Enterprise Way	Pittston/Scranton	PA	18640
Alan & Gina Wilson	(864) 558-2400	2465 Laurens Rd	Greenville	SC	29607
Brad Sullivan, Dennis Sullivan	(843) 588-5777	411 Wando Park Blvd	Mt. Pleasant	SC	29464
Steve Mays	(843) 242-3264	2200 N Oak St	Myrtle Beach	SC	29577
Reynaldo Gonzales	(605) 553-9910	5129 S Solberg Ave	Sioux Falls	SD	57108
Tommy Ray Carter, II	(731) 868-4418	65 E Park Square	Jackson	TN	38305
Roger Duncan	(615) 987-0102	1220 NW Broad Street	Murfreesboro	TN	37129
Bryan Hendershot	(469) 499-3100	3823 Irving Mall	<u>Irving</u>	<u>TX</u>	<u>75062</u>
Joshua Rathweg, Stephan Rathweg	(858) 779-2377	1691 N Zaragoza Rd	El Paso <u>East</u>	TX	79936
Joshua Rathweg, Stephan Rathweg	(858) 779-2377	4585 Ripley Dr Building #5	El Paso West	TX	79922
Bryan Hendershot	(469) 499-3100	3823 Irving Mall	<u>Irving</u>	TX	75062
Adebara Lawrence, Taiwo John, Omotola Montoni, Omolulu	(844) 567-6088	2055 Executive Dr	Hampton	VA	23666
Roger Duncan, Omobolai Benson, Omololu Fatukasi, Omobolaji Omotayo					
Luke Phillips, Terry Phillips, William Phillips	(804) 379-2500	1345 Carmia Way	Richmond	VA	23235
Roger Duncan	(414) 522-1945	9009 N. Deerbrook Trail	Brown Deer	<u>WI</u>	<u>53223</u>
Sheila & Blaine Nease	(304) 833-9663	500 Southridge Blvd	Charleston	₩V	25309
Mark & Barbara Glazer Roger Duncan	(414) 409-8150	4940 S 76th St	Greenfield	WI	53220
Roger Duncan	(608) 856-5867	2134 W Beltline Hwy	Madison	WI	53713
Roger Duncan	(262) 696-1600	W229 N, 1420 Westwood Dr	Waukesha	WI	53186
Sheila & Blaine Nease	(304) 833-9663	500 Southridge Blvd	Charleston	<u>WV</u>	<u>25309</u>

LIST OF FRANCHISEES WHO LEFT THE SYSTEM IN $\frac{2023}{2024}$

(as of February 29December 31, 2024)

If you buy this franchise, your contact information may be disclosed to other buyers when you leave the franchise system.

LIST OF TRANSFERS

Franchisee	Phone	City	State	Zip Code
Ryan Senters	(623) 362-8404	Laveen	AZ	85041
Mark Canessa and David Ruiz	(760 <u>706</u>) 578-8177	Covina	CA	91722
Steven Rodier	NA	Trumbull	CT	06611
Bron Launsby	(901) 590-6325	Ft. Lauderdale (Pompano Beach)	FL	33069
Kim L. Betts	<u>N/A</u>	<u>Columbus</u>	<u>GA</u>	<u>31907</u>
Mark and Barbara Glazer	NAN/A	Aurora	IL	60504
Mark and Barbara Glazer	<u>N/A</u>	<u>Joliet</u>	<u>IL</u>	<u>60435</u>
Bron Launsby	(901) 590-6325	Fairview Heights	IL	62208
Gary Kingery	(314) 407-8206	<u>Plainfield</u>	<u>IN</u>	<u>46168</u>
Gary Kingery	(314) 407-8206	Indianapolis South	<u>IN</u>	<u>46237</u>
L. Dee Murdock and Holly Murdock-Moss	(417) 483-3151	<u>Topeka</u>	<u>KS</u>	<u>66604</u>
Adam Knight/Portia Richardson	(301) 885-9593; (240) 476-4403	<u>Waldorf</u>	<u>MD</u>	<u>20695</u>
Mark and Barbara Glazer	NA	Joilet	IL	60435
Bron Launsby	(901) 590-6325	Canton	MI	48188
Bron Launsby	(901) 590-6325	Kentwood/Grand Rapids	MI	49512
Bron Launsby	(901) 590-6325	Shelby Township	MI	48315
Bron Launsby	(901) 590-6325	Fenton	MO	63026
<u>David Hustrulid</u>	<u>N/A</u>	Maple Shade	<u>NJ</u>	<u>08619</u>
Chirog Patel	(302) 229-6795	Hamilton	NJ	08052
Bryan Hendershot	(918) 607-5470	<u>Edmond</u>	<u>OK</u>	<u>73013</u>
Richard Ferns	(484) 588-6191	<u>Chalfont</u>	<u>PA</u>	<u>18914</u>
Richard Ferns	(484) 588-6191	Glen Mills	<u>PA</u>	<u>19342</u>
David Hustrulid	(952) 567-4450	Maple Shade	NJ	08619
Bron Launsby	(901) 590-6325	Murfreesboro	TN	37129
Mark and Barbara Glazer	NAN/A	Greenfield	WI	53220
Bron Launsby	(901) 590-6325	Madison	WI	53713
Bron Launsby	(901) 590-6325	Waukesha/Milwaukee	WI	53186

LIST OF TERMINATIONS, NON-RENEWALS, REACQUIRED BY FRANCHISOR AND CEASED OPERATIONS/OTHER

Franchisee	Phone	City	State	Zip Code
Matthew Frick	(602) 690-7290	<u>Peoria</u>	<u>AZ</u>	<u>85381</u>
Matthew Frick	(602) 690-7290	<u>Phoenix</u>	<u>AZ</u>	<u>85381</u>
Matthew Frick	<u>(602) 690-7290</u>	Scottsdale (S. Phoenix)	<u>AZ</u>	<u>85254</u>
<u>Jim Bellino</u>	<u>(714) 415-5867</u>	Westminster	<u>CA</u>	<u>92683</u>
Edward Kim	NA	Bethel	CT	06883
James Fiore	(610) 392-4509	North Tampa	FL	33701
Natalee Pessoa	(386) 453-0211	Douglasville	GA	30337
Yong Jung, Han Yu	(478) 239-2900	Macon	<u>GA</u>	<u>31210</u>
<u>Jeff Cole</u>	(925) 212-6179	Roswell	<u>GA</u>	<u>30075</u>
Noshella Thomas	(314) 262-6462	Saint Ann	MO	63074
Michael Janay	<u>(973) 768-5633</u>	<u>Allendale</u>	<u>NJ</u>	<u>07401</u>
Michael Janay	<u>(973) 768-5633</u>	Mt. Olive	<u>NJ</u>	<u>07828</u>
Michael Janay	(973) 768-5633	Pine Brook	<u>NJ</u>	<u>07058</u>
Elliott Field	<u>N/A</u>	South Plainfield	<u>NJ</u>	<u>07080</u>
Bryan HendershotMichael Janay	NA(973) 768-5633	Springfield	MO <u>NJ</u>	65619 <u>07081</u>
Anwar Hossain	(914) 513-9689	Queens	NY	10701
Joel Karg	<u>N/A</u>	<u>Dayton</u>	<u>OH</u>	<u>45459</u>
Joel Karg	N/A	Cannonsburg	<u>PA</u>	<u>15317</u>
Joel Karg	N/A	<u>Monroeville</u>	<u>PA</u>	<u>15146</u>
Tommy and Donessa Carter	NAN/A	Memphis	TN	38305
Adebara Lawrence	(301) 728-0977	Frisco	<u>TX</u>	<u>75034</u>
Shawn Singh	(917) 370-5185	Houston SW	TX	77096

The following Franchisees have signed a Franchise Agreement but did not have their Parks open by the end of the most recent fiscal year:

Franchisee	Phone	City	State	Email
	(310)	Pico		mo@resultexllc.com
Mohammed Zia	779-5336 <u>779-5</u>	RivieraBaker	CA	
	<u>226</u>	sfield		
Sahin Kadizade	(323) 304-6673	<u>Highland</u>	<u>CA</u>	
Mohammed Zia	(310) 779-5226	<u>LA/Culver</u> <u>City</u>	<u>CA</u>	
<u>Usman Rao</u>	(408) 595-5387	<u>Milpitas</u>	<u>CA</u>	
Murali Repakula	(408) 313-0520	Palo Alto	<u>CA</u>	
Anwar Sandhu	(510) 688-4604	<u>Pinole</u>	<u>CA</u>	
<u>Usman Rao</u>	(408) 595-5387	Simi Valley	CA	
Ifan Ahmed	(714) 612-6223	Chino Hills	CA	Ibhaghani75@gmail.com
Divyang Joshi*	(208) 608-8287	Irvine	CA	divyangjoshi@live.com
Divyang Joshi*	(208) 608-8287	Fullerton	CA	divyangjoshi@live.com

Sky Zone Franchise Group, LLC
April 2025 FDD
Exhibit E – List of Franchisees

Franchise	e	Pho	one	City		Sta	ate		En	nail
Yim Murphy		(415) 20	02-4868	Sunnyva	le	C	A			[®] gmail.com
Usman Ra	0	(408) 59	95-5387	Tracy	7	C	CA	us	man.rao(@gmail.com
Mohammed	<u>Zia</u>	(310) 77	79-5226	Walnut C	reek	<u>C</u>	CA CA			
Faisal Zia, Farhan	Qadri	(626) 29	90-9381	San Bernar	dino	E	A	fais	salzia75@	hotmail.com
Nick and Li	sa	(72	20)	Castle	•			nick@ı	nicholasc	ustomhomes.com
Nettleton David ar	nd Kristi	635-892	4 373-6	RockLitt	<u>leto</u>	C	CO			
Gilliland		92	29	<u>n</u>						
	<u>Mahi</u>	tha Yalam	anchili	(916) 281-	7742		<u>De</u>	ecatur	<u>GA</u>	
	R	Roger Dunc	<u>an</u>	(540) 239-	5474		<u>A</u>	<u>lsip</u>	<u>IL</u>	
Ayesha Afzal		(408) 89	90-0874	Hartfor	d	C	T	ť	ayafzal@	gmail.com
Keith McAdams		(305) 40	09-8533	Pembrol Pines	ке	F	L	Keith.	meadams	s@skyzone.com
David and Willian	n Milby	(478) 9:	54-0642	Warner Ro	bins	G	}A	david	l.mercylii	nk@gmail.com
Srini Kusam		(224) 23	35-2694	Schaumb	urg	Ŧ	L	Srin	i.vanukuı	ri@gmail.com
Steve Shaqra			10-9285	Willowbr			L			hotmail.com
Sarah Lotchfield			51-4674	Muncie			N			outlook.com
Adebara Lawrence		` /	28-0977	Gambril			ID			ee@skyzone.com
Abi and Victor Ak	inwale	\ /	94-3016	Sykesvil			ID			@gmail.com
Krystal Polite		(617-22	24-8402	Burlingt	on	N	IC	krysta	ıl@polite	properties.com
Arpan Shah & An Bhutani*	iit	(773) 30	68-7193	Berger	1	A	11	am	@gmail.com	
Arpan Shah & Am Bhutani*	iit	(773) 30	68-7193	Rutherfo	rd	A	11	am	@gmail.com	
Dalton and Kather	ine	(385) 3'	75-4306	Saratog		U	JT	dalton@pfstaffii		ingservices.com
Paspuel Paspuel				Spring	S					
Adebara Lawrence*To Williamson	<u>m</u>	(301) 728-0977 4) 456-77	<u>(22</u>	Arlington <u>Heights</u>	<u>IL</u>		VA	Adeba	ira.lawre i	nce@skyzone.com
	R	Roger Dunc	an	(540) 239-	5474		Be	dford	IL	
		Bolaji Bens		(301) 256-				Burnie	MD	
		oiola Ayod		(240) 413-				derick	MD	1
		Roger Dunc		(540) 239-				nsing	MI	
	_	Kang, Dust		(805) 630-				derson	NV	-
		Usman Ra		(408) 595-				oklyn	NY	-
		s Horton, l						•		-
		Leichus		(203) 414-	<u> 2295</u>		<u>B</u>	ronx	<u>NY</u>	
	Adeba	ra Lawrenc	e, Serge							-
		he, Jod Jea		(201) 720	0077		***		TX	
	Evans	S Dure, Ati	nuke O	(301) 728-	<u>·0977</u>		<u>v</u>	<u>Iaco</u>	<u>TX</u>	
		Kasumu*								
		debara	(301)		dr V	Α	Adeba	ıra.lawre	nce@sky	:
		wrence.	728-09	77 ia				zone.co	m	
	Serge 1	unde Ola, Hyacinthe,								
	Jonn (Odilli and]

Sky Zone Franchise Group, LLC
April 2025 FDD
Exhibit E – List of Franchisees

Franchise	e	Pho	one	City		State	En	nail
		<u>kunboh</u> reoje*						
Omololu Fatu	kasi	(240) 68	<u>87-0914</u>	Chantilly		VA		
Adebara Lawrence)	(301) 7	28-0977	Virginia Beac	h	VA	Adebara.lawrenc	ce@skyzone.com

^{*}Indicates the Franchisee (or an affiliate) is party to a Multi-Unit Development Agreement.

<u>The following Franchisees have signed a Multi-Unit Development Agreement but did not sign a Franchise Agreement or open a Park by the end of the most recent fiscal year:</u>

<u>Franchisee</u>	<u>Phone</u>	<u>State</u>
Pawan and Gurpreet Buttar	(480) 616-9052	<u>AZ</u>
Kamal and Pragati Sinha	(646) 522-5647	NJ
Chalmer & Karen McWilliams	(817) 821-0113	<u>OK</u>
Kinjal & Purvish Patel	(570) 574-6937	<u>PA</u>
Chalmer & Karen McWilliams	(817) 821-0113	<u>TX</u>
Adesola Anifalaje, Olusola Odumade, Adesoji Awe, Babatunde Anifalaje	(602) 743-0841	<u>TX</u>
Rodney George, Adebara Lawrence, Patrick Cherestal, Michael Pelissier, Serge Hyacinthe, Patrick Rodrigue	(240) 277-0219	<u>TX</u>
Abhishek Sinha & Pushpinder Singh	<u>(469) 888-0317</u>	<u>WA</u>

^{*}Part of a MUDA

EXHIBIT F TO THE DISCLOSURE DOCUMENT

FINANCIAL STATEMENTS

Sky Zone Franchise Group, LLC

Consolidated Financial Statements December 31, 2024, 2023 and 2022

Sky Zone Franchise Group, LLC

Master Index December 31, 2024, 2023 and 2022

Report of independent Auditor	ndependent Auditor	er	Ind	of	port	Rε	
-------------------------------	--------------------	----	-----	----	------	----	--

	Section
Operational Figure 24 Chategories As of and fourth a versus and ad Danamih at 24 2004 and	
Consolidated Financial Statements As of and for the years ended December 31, 2024 and 2023	1
Consolidated Financial Statements As of and for the years ended December 31, 2023 and 2022	II



Report of Independent Auditors

To the Management of Sky Zone Franchise Group, LLC

Opinion

We have audited the accompanying consolidated financial statements of Sky Zone Franchise Group, LLC and its subsidiaries (the "Company"), which comprise the consolidated balance sheets as of December 31 2024, 2023 and 2022, and the related consolidated statements of income, of member's equity and of cash flows for the years then ended, including the related notes (collectively referred to as the "consolidated financial statements").

In our opinion, the accompanying consolidated financial statements present fairly, in all material respects, the financial position of the Company as of December 31, 2024, 2023 and 2022, and the results of its operations and its cash flows for the years then ended in accordance with accounting principles generally accepted in the United States of America.

Basis for Opinion

We conducted our audit in accordance with auditing standards generally accepted in the United States of America (US GAAS). Our responsibilities under those standards are further described in the Auditors' Responsibilities for the Audit of the Consolidated Financial Statements section of our report. We are required to be independent of the Company and to meet our other ethical responsibilities, in accordance with the relevant ethical requirements relating to our audit. We believe that the audit evidence we have obtained is sufficient and appropriate to provide a basis for our audit opinion.

Responsibilities of Management for the Consolidated Financial Statements

Management is responsible for the preparation and fair presentation of the consolidated financial statements in accordance with accounting principles generally accepted in the United States of America, and for the design, implementation, and maintenance of internal control relevant to the preparation and fair presentation of consolidated financial statements that are free from material misstatement, whether due to fraud or error.

In preparing the consolidated financial statements, management is required to evaluate whether there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for one year after the date the consolidated financial statements are available to be issued.

Auditors' Responsibilities for the Audit of the Consolidated Financial Statements

Our objectives are to obtain reasonable assurance about whether the consolidated financial statements as a whole are free from material misstatement, whether due to fraud or error, and to issue an auditors' report that includes our opinion. Reasonable assurance is a high level of assurance but is not absolute assurance and therefore is not a guarantee that an audit conducted in accordance with US GAAS will always detect a material misstatement when it exists. The risk of not detecting a material misstatement



resulting from fraud is higher than for one resulting from error, as fraud may involve collusion, forgery, intentional omissions, misrepresentations, or the override of internal control. Misstatements are considered material if there is a substantial likelihood that, individually or in the aggregate, they would influence the judgment made by a reasonable user based on the consolidated financial statements.

In performing an audit in accordance with US GAAS, we:

- Exercise professional judgment and maintain professional skepticism throughout the audit.
- Identify and assess the risks of material misstatement of the consolidated financial statements, whether due to fraud or error, and design and perform audit procedures responsive to those risks. Such procedures include examining, on a test basis, evidence regarding the amounts and disclosures in the consolidated financial statements.
- Obtain an understanding of internal control relevant to the audit in order to design audit
 procedures that are appropriate in the circumstances, but not for the purpose of expressing an
 opinion on the effectiveness of the Company's internal control. Accordingly, no such opinion is
 expressed.
- Evaluate the appropriateness of accounting policies used and the reasonableness of significant accounting estimates made by management, as well as evaluate the overall presentation of the consolidated financial statements.
- Conclude whether, in our judgment, there are conditions or events, considered in the aggregate, that raise substantial doubt about the Company's ability to continue as a going concern for a reasonable period of time.

We are required to communicate with those charged with governance regarding, among other matters, the planned scope and timing of the audit, significant audit findings, and certain internal control-related matters that we identified during the audit.

Pricewaterhouse Coopers LLP

Salt Lake City, Utah April 22, 2025

I

Sky Zone Franchise Group, LLC

Consolidated Financial Statements December 31, 2024 and 2023

Sky Zone Franchise Group, LLC Index

December 31, 2024 and 2023

	Page(s)
Consolidated Financial Statements	
Consolidated Balance Sheets	1
Consolidated Statements of Income	2
Consolidated Statements of Member's Equity	3
Consolidated Statements of Cash Flows	4
Notes to Consolidated Financial Statements	5_11

Sky Zone Franchise Group, LLC Consolidated Balance Sheets December 31, 2024 and 2023

	2024	2023		
Assets				
Current Assets				
Cash	\$ 98,000	\$ 226,100		
Accounts receivable, net	1,810,300	2,546,600		
Related party receivable	39,979,100	67,514,600		
Inventories	45,700	46,400		
Deferred initial franchise costs - current	96,200	55,000		
Other receivables	568,900	165,500		
Deposits	7,000	44,200		
Total Current Assets	42,605,200	70,598,400		
Property, equipment, and software, net	52,700	68,300		
Right of use asset	-	121,800		
Deferred initial franchise costs - long-term	964,600	357,400		
Total Assets	\$ 43,622,500	\$ 71,145,900		
Liabilities and Member's Equity				
Current Liabilities				
Accounts payable	\$ 347,200	\$ 16,400		
Accrued expenses	911,400	886,500		
Loss fund claims reserve - current	1,670,400	2,705,400		
Deferred revenues	-	1,320,100		
Operating lease liabilities - current	-	135,200		
Deferred franchise fees - current	757,700	369,900		
Total Current Liabilities	3,686,700	5,433,500		
Long-Term Liabilities				
Loss fund claims reserve - long-term	963,000	3,850,200		
Deferred franchise fees - long-term	4,810,400	4,022,200		
Total Liabilities	9,460,100	13,305,900		
Member's Equity	34,162,400	57,840,000		
Total Liabilities and Member's Equity	\$ 43,622,500	\$ 71,145,900		

Sky Zone Franchise Group, LLC Consolidated Statements of Income Years Ended December 31, 2024 and 2023

	2024	2023
Revenues		_
Royalty fees	\$ 16,583,200	\$ 16,808,300
Merchandise sales	3,749,200	2,858,100
Franchise fees	1,913,000	854,500
Total revenues	22,245,400	20,520,900
Costs relating to Revenues		
Merchandise cost of sales	44,000	114,000
Total costs relating to revenues	44,000	114,000
Selling, General and Administrative Expenses		
Personnel - selling, general, and administrative	4,365,500	3,320,900
Travel and entertainment	459,000	530,100
Other legal and professional services	235,300	340,100
Building and information technology	77,000	482,200
Insurance	18,700	18,600
Advertising	169,800	33,000
Office	3,000	3,300
Depreciation	15,600	18,200
Other general and administrative expenses	469,300	357,300
Total selling, general, and administrative expenses	5,813,200	5,103,700
Operating income	16,388,200	15,303,200
Other Expense		
Other Income (loss)	(2,200)	12,200
Franchise tax	(63,300)	(53,600)
Total Other Expense	(65,500)	(41,400)
Net income	\$ 16,322,700	\$ 15,261,800

Sky Zone Franchise Group, LLC Consolidated Statements of Member's Equity Years Ended December 31, 2024 and 2023

Member's equity, December 31, 2022	\$ 42,578,200
Net income	15,261,800
Member's equity, December 31, 2023	\$ 57,840,000
Net income	16,322,700
Non-cash conversion of related party receivable to Member's equity	(40,000,300)
Member's equity, December 31, 2024	
	\$ 34,162,400

Sky Zone Franchise Group, LLC Consolidated Statements of Cash Flows Years Ended December 31, 2024 and 2023

	 2024	 2023
Operating activities:		
Net income	\$ 16,322,700	\$ 15,261,800
Adjustments to reconcile net income to net cash used in operating activities		
operating activities:		
Depreciation	15,600	18,200
Amortization of right of use asset	121,800	483,800
Gain on sale of property, equipment and software, net	-	(7,400)
Provision for accounts receivable losses	223,600	(55,800)
Changes in operating assets and liabilities:		
Accounts receivable, net	512,700	(1,028,800)
Related party receivable	(14,528,700)	(21,187,800)
Inventories	700	7,100
Prepaid expenses	-	25,000
Deferred initial franchise costs	(648,400)	68,000
Other receivables	(403,400)	(44,500)
Deposits	37,200	-
Accounts payable	330,800	(32,800)
Accrued expenses	768,700	383,900
Loss fund claims reserve	(3,922,200)	567,900
Deferred revenues	-	411,700
Operating lease liabilities	(135,200)	(523,000)
Deferred franchise fees	 1,176,000	 166,400
Net cash used in operating activities	 (128,100)	 (5,486,300)
Investing activities:		
Sale of property, equipment, and software	 <u> </u>	 11,900
Net cash provided by investing activities	 <u>-</u>	 11,900
Decrease in cash	(128,100)	(5,474,400)
Cash, beginning of year	 226,100	 5,700,500
Cash, end of year	\$ 98,000	\$ 226,100
Supplemental disclosure of cash flow information		
Cash paid for income taxes	63,300	65,800
Supplemental disclosure of noncash financing activities		
Conversion of related party receivable to Member's equity	40,000,300	-

1. Company Background and Description of Business

Sky Zone Franchise Group, LLC (the "Company"), a Missouri limited liability company, was formed on November 18, 2008. The Company began operations on January 1, 2009. The Company licenses and sells franchises on behalf of its affiliate, Sky Zone, LLC ("SZLLC"), which owns the patent for the world's first interlocking all trampoline-walled playing fields ("Recreation Centers").

On December 27, 2017, the Company was acquired by CircusTrix Holdings, LLC (the "Acquisition"), a Delaware limited liability company ("Parent" or "New Member"), and SZFG Merger Sub, LLC, an indirect-wholly owned subsidiary of Parent ("Merger Sub"). Merger Sub was merged with and into the Company, with the Company surviving as an indirect wholly owned subsidiary of the Parent.

During 2024, the following activity related to Sky Zone branded franchised recreation centers was noted.

Recreational Centers	Franchised
At December 31, 2023	134
Franchise recreational centers converted to Owned	(17)
Franchised recreational centers opened	6
Franchise recreational centers closed	(7)
Recreational centers rebranded	12
At December 31, 2024	128

No owned locations were converted to a franchise during the year ended December 31, 2024.

Note that these consolidated financial statements and notes include not only the results of Sky Zone Franchise Group, LLC, as mentioned above, but are consolidated to include the results of Sky Zone SIR Loss Fund, LLC, a Delaware limited liability company that was formed on April 21, 2020. The Sky Zone SIR Loss Fund provides for a self-insured retention for participating franchisees in a range from \$50,000 to \$250,000 per occurrence with franchisees being responsible to reimburse the fund for the first \$10,000 of any claim. The amount contributed by franchisees and being held by the Sky Zone SIR Loss Fund to pay out against potential injury claims is shown in the "Loss Fund Claims Reserve" lines of the consolidated balance sheets.

2. Summary of Significant Accounting Policies

Basis of Accounting and Presentation

Preparation of consolidated financial statements in accordance with generally accepted accounting principles in the United States of America ("GAAP") requires the Company to (i) adopt accounting policies within accounting rules set by the Financial Accounting Standards Board ("FASB"), and (ii) make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, expenses, and other disclosed amounts. The Company believes that the estimates used in the preparation of the consolidated financial statements are reasonable, however, actual results may differ from these estimates as new events occur, additional information is obtained and as the Company's operating environment changes.

Use of Estimates

The preparation of these consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses for the reporting period. On an on-going basis, the Company evaluates its estimates, including those related to litigation accruals, allowance for doubtful accounts, the determination of useful lives of property and equipment and intangible assets, and the recognition of revenue into deferred revenue.

Subsequent Events

The Company recognizes the effects of events or transactions that occur after the balance sheet date but before financial statements are issued ("subsequent events") if there is evidence that conditions related to the subsequent event existed at the date of the balance sheet, including the impact of such events on management's estimates and assumptions used in preparing the consolidated financial statements. Refer to Note 7 for additional details around subsequent events.

Cash and Cash Equivalents

The Company reflects cash on hand, demand deposits at financial institutions, and deposits in transit that represent a claim to cash as cash and cash equivalents. The Company's cash deposits in the United States are held at institutions insured by the Federal Deposit Insurance Corporation ("FDIC"). At times, cash held at these institutions may exceed FDIC insurance limits.

Accounts Receivable

Accounts receivable consist of (1) royalty fees; (2) merchandise sales; (3) total cost of risk billings. The Company also records allowances for current expected credit losses. These allowances are based on the Company's assessment of the collectability of accounts by considering the age of each outstanding invoice and an evaluation of expected risk of credit loss based on current conditions and reasonable and supportable forecasts of future economic conditions over the life of the receivable. The Company pools its receivables based on similar risk characteristics in estimating its expected credit losses. In situations where an accounts receivable does not share the same risk characteristics with other, the Company measures it individually. The Company also continuously evaluates such pooling decisions and adjusts as needed from period to period as risk characteristics change. The Company has an allowance for current expected credit loss activity of \$362,500 and \$138,900 as of December 31, 2024 and 2023, respectively. Account balances are charged off against the allowance after all means of collection have been exhausted and the potential for recovery is considered remote. Expected recoveries of amounts previously written off, not to exceed the aggregate of the amount previously written off, are included in determining the necessary reserve at the balance sheet date.

Inventories

Inventory consists of the costs associated with the purchase and freighting of apparel products as well as other merchandise. Inventory is stated at the lower of cost or market, using the average cost method.

Property and Equipment

Property and equipment are stated at cost, net of accumulated depreciation and amortization of the fair value adjustments associated with fixed assets. Repairs and maintenance costs are expensed as incurred as repairs and maintenance do not extend the useful life or improve the related assets. The Company capitalizes certain costs incurred in connection with developing or obtaining software. Depreciation and amortization, including amortization of leasehold improvements, are charged to operations using the straight-line method over the assets' estimated useful lives.

Impairment of Long-Lived Assets

The Company evaluates whether events and circumstances have occurred that indicate the remaining estimated useful life of long-lived assets may warrant revision or that the remaining balance of an asset may not be recoverable. The measurement of possible impairment is based on the ability to recover the balance of assets from expected future operating cash flows on an undiscounted basis.

Leases

The Company is party to a single lease relating to its office space. The Company determines if a contract is a lease at inception. A contract contains a lease if the contract conveys the right to control the use of an identified asset for a period of time in exchange for consideration. The lease term begins on the commencement date, which is the date the Company takes possession of the asset, and may include options to extend or terminate the lease when it is reasonably certain that the option will be exercised. Leases are classified as operating or finance leases based on factors such as the lease term, lease payments, and the economic life, fair value and estimated residual value of the asset. Where leases include options to purchase the leased asset at the end of the lease term, this is assessed as a part of the Company's lease classification determination. The Company's office space lease ended in 2024.

Under Accounting Standards Codification ("ASC") 842, the Company recognizes a right-of-use ("ROU") asset and lease liability to account for its leases. ROU assets represent the Company's right to use an underlying asset for the lease term and lease liabilities represent the Company's obligation to make lease payments arising from the lease. ROU assets and lease liabilities are recognized on the commencement date based on the present value of lease payments over the lease term. ROU assets are based on the lease liability and are increased by prepaid lease payments and decreased by lease incentives received. Lease incentives are amortized through the lease asset as reductions of expense over the lease term. For leases where the Company is reasonably certain to exercise a renewal option, such option periods have been included in the determination of the Company's ROU assets and lease liabilities.

Leases typically contain rent escalations over the lease term. The Company recognizes expense for these leases on a straight-line basis over the lease term. Certain leases require the Company to pay taxes, insurance, maintenance and other operating expenses associated with the leased asset. Such amounts are not included in the measurement of the ROU assets and lease liabilities to the extent they are variable in nature. These variable lease costs are recognized as a variable lease expense when incurred.

As a practical expedient, lease agreements with lease and nonlease components are accounted for as a single lease component for all asset classes. The Company estimates contingent lease incentives when it is probable that the Company is entitled to the incentive at lease commencement. The Company elected the short-term lease recognition exemption for all leases that qualify. Therefore, leases with an initial and expected renewal term of 12 months or less are not recorded on the consolidated balance sheets; instead, lease payments are recognized as lease expense on a straight-line basis over the lease term. The depreciable life of the ROU assets and leasehold improvements are limited by the expected lease term unless the Company is reasonably certain of a transfer of title or purchase option.

The Company elected to use a risk-free rate based on US treasury information as the discount rate for all leases.

Franchise Agreements

The Company has two Franchise Programs: Single Unit Franchise program and Development program. With the Single Unit Franchise, the Company receives an initial franchise fee in exchange for granting a nonexclusive right to operate one, and only one, Sky Zone Recreation Center at a specified location. The initial franchise fee for a single Recreation Center is generally paid in a lump sum upon commencement of the franchise agreement. The Development Program requires an executed Multi-Unit Development agreement which assigns a territory which requires the opening and operating of a designated number of Recreation Centers within a specified time period. A lump-sum development fee for the multiple units is paid at the time of commencement of the Multi-Unit Development agreement. A separate franchise agreement is executed for each unit. In addition to the granting an exclusive right to operate a Recreational Center, the fees also include a training program for the operation of the Recreation Center, and intellectual property of the Company for the principal owner and the managers and or assistant managers, support for site selection, architectural plans, interior and exterior design and layout, marketing and sales techniques, and other opening assistance.

During 2024, the Company recorded \$1,142,000 of Franchise fees as an out of period adjustment to correct an error in Deferred franchise fees – long term. The Company concluded that this out of period adjustment was not material to the current annual financial statements nor the previously issued annual financial statements.

Revenue Recognition

The Company's revenues consist of initial franchise fees and administrative fees received from franchisees, royalties based on a percentage of franchisee revenues, and merchandise sales to franchisees. The amount of revenue recognized reflects the consideration the Company expects to receive in exchange for the services or products. The Company excludes from revenue the sales tax it collects on behalf of government authorities. The revenue recognized in the current year that was included within the contract liability balance at the beginning of the year approximates the prior year deferred revenues and deferred franchise fees – current balances. Specifics on how the new revenue guidance provided in Accounting Standards Codification ("ASC") 606 Revenue from Contracts with Customers, is reflected across our revenue streams are discussed below:

Franchise Fee Revenue

Franchise revenues consist of the initial franchise fee and in some cases an additional administration fee for multi-unit agreements with the amounts specified as part of the franchise agreement entered into between the Company and the franchisee. The Company recognizes franchise and administrative fees as revenue ratably over the term of the park addendum contracts (e.g. from the execution date until the expiration date – typically 10 years).

Royalty Revenue

Continuing royalty fees received from franchisees are recognized in the period of the related franchisee sales.

Merchandise Revenue

Revenues from the sale of merchandise are recognized upon shipment to the franchisees, which is the point when the risk of ownership is transferred. Revenues from discounts, rebate commissions, promotional allowances, and other benefits from vendors are recognized upon shipment to franchisees.

Deferred Initial Franchise Costs

Consistent with revenue recognition relative to initial franchise fees, the Company defers all direct, incremental costs associated with initial franchise fees over the life of the franchise agreement. These costs consist mainly of commissions and international withholding expenses.

Income Taxes

As the Company is a limited liability company that is treated as a partnership for U.S. federal tax purposes, all taxable income and tax benefits are passed through to the members. Therefore, no liability for U.S. federal or state income taxes is reflected in these consolidated financial statements. However, the Company is subject to franchise taxes.

The Company had no unrecognized tax benefits at December 31, 2024 and 2023. The Company's federal and state income tax returns remain open and management continually evaluates expiring statutes of limitations, audits, proposed settlements, changes in tax law and new authoritative rulings.

If necessary, the Company recognizes interest and penalties associated with tax matters as part of operating expenses and includes accrued interest and penalties with the related tax liability in the balance sheets. No such amounts are included in the consolidated financial statements for the years ended December 31, 2024 and 2023.

Advertising

Advertising and promotional costs are expensed as they are incurred. Advertising expenses not reimbursed by the Company's unconsolidated advertising fund for the years ended December 31, 2024 and 2023 were \$169,800 and \$33,000, respectively.

Recently Adopted and Issued Accounting Standards

In June 2016, the FASB issued ASU No. 2016-13, *Financial Instruments — Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"), which changes the impairment model for most financial assets. The ASU introduces a new credit loss methodology, Current Expected Credit Losses (CECL), which requires earlier recognition of credit losses, while also providing additional transparency about credit risk. Since its original issuance in 2016, the FASB has issued several updates to the original ASU. The CECL framework utilizes a lifetime "expected credit loss" measurement objective for the recognition of credit losses for loans, held-to-maturity securities and other receivables at the time the financial asset is originated or acquired. The expected credit losses are adjusted each period for changes in expected lifetime credit losses. The methodology replaces the multiple existing impairment methods, which generally require that a loss be incurred before it is recognized.

On January 1, 2023, the Company adopted the guidance using a modified retrospective approach with a cumulative adjustment to Member's Equity. The adoption of this standard did not have a material impact on the Company's consolidated financial statements.

3. Property, Equipment, and Software

Property, equipment, and software are summarized by major classifications as of December 31, 2024 and 2023, as follows:

	Useful Life	2024	2023
Computers & software	3 years	1,090,200	1,090,200
Leasehold improvements	Lesser of ten years or lease life	155,900	155,900
	-	\$ 1,246,100	\$1,246,100
Less: Accumulated depreciation	_	1,193,400	1,177,800
	- -	\$ 52,700	\$68,300

Depreciation expense totaled \$15,600 and \$18,200 for the years ended December 31, 2024 and 2023, respectively.

4. Leases

The Company has one lease arrangement for office space that expired in 2024. The right of use asset and lease liability associated with this lease is separately stated on the face of the Company's balance sheet as of December 31, 2023. As of December 31, 2024 there were no remaining right of use asset or liability associated with this lease. The lease asset and liability were calculated utilizing the risk-free discount rate of 0.97% according to the Company's elected policy. The Company's operating lease expense for the years ended December 31, 2024 and 2023 was \$132,600 and \$526,600 respectively.

5. Related Party Transactions

The Company maintains and manages an advertising fund that has not been consolidated in these financial statements. This related party fund receives advertising fees from the franchisees for use in advertising spending to promote the Sky Zone brand. As of December 31, 2024 the Company owes the advertising fund \$116,200. As of December 31, 2023 the Company owes the advertising fund \$102,000.

The Company's sole member, CircusTrix Holdings, LLC, has paid certain costs on behalf of the Company. Additionally, cash proceeds from sales have been sent from the Company to CircusTrix Holdings, LLC. Related party receivables, current net, total \$39,979,100 and \$67,514,600 as of December 31, 2024 and 2023, respectively.

During 2024, the Company made a non-cash distribution to CircusTrix Holdings, LLC to convert \$40,000,300 of related party receivable to member's equity.

6. Retirement Plan

The Company maintains a 401(k) Savings Plan in which most employees are eligible to participate. Eligible employees may make contributions not to exceed the maximum statutory contribution amounts. The Company may match a percentage of each employee's contributions consistent with the provisions of the plan for which they are eligible. All employee and employer contributions fully vest immediately. The Company contributions to the Plan totaled \$36,600 and \$22,300 for the years ended December 31, 2024 and 2023, respectively, and are included in personnel expenses in the accompanying statements of income.

7. Subsequent Events

The Company reviewed all material events from December 31, 2024 through the date the consolidated financial statements were available to be issued, April 22, 2025. No items were identified that would materially affect the consolidated financial statements or require additional disclosure.

II

Sky Zone Franchise Group, LLC

Consolidated Financial Statements December 31, 2023 and 2022

Sky Zone Franchise Group, LLC

Index

December 31, 2023 and 2022

Page(s)

Consolidated Financial Statements

Consolidated Balance Sheets	1
Consolidated Statements of Income	2
Consolidated Statements of Member's Equity	3
Consolidated Statements of Cash Flows	4
Notes to Consolidated Financial Statements	5–11

Sky Zone Franchise Group, LLC Consolidated Balance Sheets December 31, 2023 and 2022

Assets	2023	2022
Current Assets		
Cash	\$ 226,100	\$ 5,700,500
Accounts receivable, net	2,546,600	1,462,000
Related party receivable	67,514,600	46,326,800
Inventories	46,400	53,500
Prepaid expenses	-	25,000
Deferred initial franchise costs - current	55,000	68,000
Other receivables	165,500	121,000
Deposits	 44,200	 44,200
Total Current Assets	70,598,400	 53,801,000
Property, equipment, and software, net	68,300	91,000
Right of use asset	121,800	605,600
Deferred initial franchise costs - long-term	357,400	412,400
Total Assets	\$ 71,145,900	\$ 54,910,000
Liabilities and Member's Equity		
Current Liabilities		
Accounts payable	\$ 16,400	\$ 49,200
Accrued expenses	886,500	502,600
Loss fund claims reserve - current	2,705,400	724,300
Deferred revenues	1,320,100	908,400
Operating lease liabilities - current	135,200	525,700
Deferred franchise fees - current	369,900	 516,300
Total Current Liabilities	5,433,500	 3,226,500
Long-Term Liabilities		
Loss fund claims reserve - long-term	3,850,200	5,263,400
Deferred franchise fees - long-term	4,022,200	3,709,400
Operating lease liabilities, long-term	 	 132,500
Total Liabilities	13,305,900	12,331,800
Member's Equity	 57,840,000	 42,578,200
Total Liabilities and Member's Equity	\$ 71,145,900	\$ 54,910,000

Sky Zone Franchise Group, LLC Consolidated Statements of Income Years Ended December 31, 2023 and 2022

	2023	2022
Revenues		
Royalty fees	\$ 16,808,300	\$ 17,050,100
Merchandise sales	2,858,100	2,345,600
Franchise fees	854,500	995,700
Total revenues	20,520,900	20,391,400
Costs relating to Revenues		
Merchandise cost of sales	114,000	222,600
Total costs relating to revenues	114,000	222,600
Selling, General and Administrative Expenses		
Personnel - selling, general, and administrative	3,320,900	3,474,400
Travel and entertainment	530,100	297,600
Other legal and professional services	340,100	906,100
Building and information technology	482,200	538,400
Insurance	18,600	13,400
Advertising	33,000	277,300
Outside services	-	6,600
Office	3,300	8,500
Research and development	-	6,600
Depreciation	18,200	67,600
Other general and administrative expenses	357,300	100,900
Total selling, general, and administrative expenses	5,103,700	5,697,400
Operating income	15,303,200	14,471,400
Other Income (Expense)		
Other Income (loss)	12,200	(3,500)
Franchise tax	(53,600)	(65,800)
Total Other Income (Expense)	(41,400)	(69,300)
Net income	\$ 15,261,800	\$ 14,402,100

Sky Zone Franchise Group, LLC Consolidated Statements of Member's Equity Years Ended December 31, 2023 and 2022

Member's equity, December 31, 2021	\$ 28,176,100
Net income	14,402,100
Member's equity, December 31, 2022	\$ 42,578,200
Net income	15,261,800
Member's equity, December 31, 2023	\$ 57,840,000

Sky Zone Franchise Group, LLC Consolidated Statements of Cash Flows Years Ended December 31, 2023 and 2022

	 2023	2022
Operating activities:		
Net income	\$ 15,261,800	\$ 14,402,100
Adjustments to reconcile net income to net cash provided by		
operating activities:		
Depreciation	18,200	67,600
Amortization of right of use asset	483,800	478,800
Gain on sale of property, equipment and software, net	(7,400)	-
Provision for accounts receivable losses	(55,800)	(843,100)
Changes in operating assets and liabilities:		
Accounts receivable, net	(1,028,800)	707,700
Related party receivable	(21,187,800)	(14,672,300)
Inventories	7,100	35,500
Prepaid expenses	25,000	(3,600)
Deferred initial franchise costs	68,000	26,700
Other receivables	(44,500)	127,500
Deposits	-	(44,200)
Accounts payable	(32,800)	16,000
Accrued expenses	383,900	(369,600)
Loss fund claims reserve	567,900	2,677,700
Deferred revenues	411,700	279,400
Operating lease liabilities	(523,000)	(502,600)
Deferred franchise fees	 166,400	160,500
Net cash (used in) provided by operating activities	 (5,486,300)	 2,544,100
Investing activities:		
Sale of property, equipment, and software	 11,900	 -
Net cash provided by investing activities	 11,900	
Decrease in cash	(5,474,400)	2,544,100
Cash, beginning of year	 5,700,500	3,156,400
Cash, end of year	\$ 226,100	\$ 5,700,500
Supplemental disclosure of cash flow information		
Cash paid for income taxes	52,400	65,800

1. Company Background and Description of Business

Sky Zone Franchise Group, LLC (the "Company"), a Missouri limited liability company, was formed on November 18, 2008. The Company began operations on January 1, 2009. The Company licenses and sells franchises on behalf of its affiliate, Sky Zone, LLC ("SZLLC"), which owns the patent for the world's first interlocking all trampoline-walled playing fields ("Recreation Centers").

On December 27, 2017, the Company was acquired by CircusTrix Holdings, LLC (the "Acquisition"), a Delaware limited liability company ("Parent" or "New Member"), and SZFG Merger Sub, LLC, an indirect-wholly-owned subsidiary of Parent ("Merger Sub"). Merger Sub was merged with and into the Company, with the Company surviving as an indirect wholly-owned subsidiary of the Parent.

At December 31, 2023 and 2022, there were 134 and 137 franchisee-owned Recreation Centers in operation, respectively. During the years ended December 31, 2023 and 2022, there were 7 and 5 franchise-owned Recreation Centers opened as well as 10 and 10 franchise-owned Recreation Centers closed, respectively. During the years ended December 31, 2023 and 2022, 4 and 6 franchise-owned Recreation Centers were acquired by CircusTrix Holdings, LLC, respectively. No owned locations were converted to a franchise during the year ended December 31, 2023.

Recreational Centers	Franchised
At December 31, 2022	137
Franchise recreational centers converted to Owned	(4)
Franchised recreational centers opened	7
Franchise recreational centers closed	(10)
Recreational centers rebranded	4
At December 31, 2023	134

Note that these consolidated financial statements and notes include not only the results of Sky Zone Franchise Group, LLC, as mentioned above, but are consolidated to include the results of Sky Zone SIR Loss Fund, LLC, a Delaware limited liability company that was formed on April 21, 2020. The Sky Zone SIR Loss Fund provides for a self-insured retention for participating franchisees in a range from \$50,000 to \$250,000 per occurrence with franchisees being responsible to reimburse the fund for the first \$10,000 of any claim. The amount contributed by franchisees and being held by the Sky Zone SIR Loss Fund to pay out against potential injury claims is shown in the "Loss Fund Claims Reserve" lines of the consolidated balance sheets.

2. Summary of Significant Accounting Policies

Basis of Accounting and Presentation

Preparation of consolidated financial statements in accordance with generally accepted accounting principles in the United States of America ("GAAP") requires the Company to (i) adopt accounting policies within accounting rules set by the Financial Accounting Standards Board ("FASB"), and (ii) make estimates and assumptions that affect the reported amounts of assets, liabilities, revenues, expenses, and other disclosed amounts. The Company believes that the estimates used in the preparation of the consolidated financial statements are

reasonable, however, actual results may differ from these estimates as new events occur, additional information is obtained and as the Company's operating environment changes.

Use of Estimates

The preparation of these consolidated financial statements requires management to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the date of the consolidated financial statements and the reported amounts of revenues and expenses for the reporting period. On an on-going basis, the Company evaluates its estimates, including those related to litigation accruals, allowance for doubtful accounts, the determination of useful lives of property and equipment and intangible assets, and the recognition of revenue into deferred revenue.

Subsequent Events

The Company recognizes the effects of events or transactions that occur after the balance sheet date but before financial statements are issued ("subsequent events") if there is evidence that conditions related to the subsequent event existed at the date of the balance sheet, including the impact of such events on management's estimates and assumptions used in preparing the consolidated financial statements. Refer to Note 7 for additional details around subsequent events.

Cash and Cash Equivalents

The Company reflects cash on hand, demand deposits at financial institutions, and deposits in transit that represent a claim to cash as cash and cash equivalents. The Company's cash deposits in the United States are held at institutions insured by the Federal Deposit Insurance Corporation ("FDIC"). At times, cash held at these institutions may exceed FDIC insurance limits.

Accounts Receivable

Accounts receivable consist of (1) royalty fees; (2) merchandise sales; (3) total cost of risk billings. Accounts receivable are stated at the amount management expects to collect from outstanding balances. The Company periodically evaluates the collectability of the receivables and establishes an allowance for doubtful accounts when it is probable and estimable there is a loss inherent in the accounts, based on the information available. The Company has an allowance for doubtful accounts of \$138,900 and \$194,700 as of December 31, 2023 and 2022, respectively.

Inventories

Inventory consists of the costs associated with the purchase and freighting of apparel products as well as other merchandise. Inventory is stated at the lower of cost or market, using the average cost method.

Property and Equipment

Property and equipment are stated at cost, net of accumulated depreciation and amortization of the fair value adjustments associated with fixed assets. Repairs and maintenance costs are expensed as incurred as repairs and maintenance do not extend the useful life or improve the related assets. The Company capitalizes certain costs incurred in connection with developing or obtaining software. Depreciation and amortization, including amortization of leasehold improvements, are charged to operations using the straight-line method over the assets' estimated useful lives.

Impairment of Long-Lived Assets

The Company evaluates whether events and circumstances have occurred that indicate the remaining estimated useful life of long-lived assets may warrant revision or that the remaining balance of an asset may not be recoverable. The measurement of possible impairment is based on the ability to recover the balance of assets from expected future operating cash flows on an undiscounted basis.

Leases

The Company is party to a single lease relating to its office space. The Company determines if a contract is a lease at inception. A contract contains a lease if the contract conveys the right to control the use of an identified asset for a period of time in exchange for consideration. The lease term begins on the commencement date, which is the date the Company takes possession of the asset, and may include options to extend or terminate the lease when it is reasonably certain that the option will be exercised. Leases are classified as operating or finance leases based on factors such as the lease term, lease payments, and the economic life, fair value and estimated residual value of the asset. Where leases include options to purchase the leased asset at the end of the lease term, this is assessed as a part of the Company's lease classification determination. The Company's office space lease ends in 2024.

Under Accounting Standards Codification ("ASC") 842, the Company recognizes a right-of-use ("ROU") asset and lease liability to account for its leases. ROU assets represent the Company's right to use an underlying asset for the lease term and lease liabilities represent the Company's obligation to make lease payments arising from the lease. ROU assets and lease liabilities are recognized on the commencement date based on the present value of lease payments over the lease term. ROU assets are based on the lease liability and are increased by prepaid lease payments and decreased by lease incentives received. Lease incentives are amortized through the lease asset as reductions of expense over the lease term. For leases where the Company is reasonably certain to exercise a renewal option, such option periods have been included in the determination of the Company's ROU assets and lease liabilities.

Leases typically contain rent escalations over the lease term. The Company recognizes expense for these leases on a straight-line basis over the lease term. Certain leases require the Company to pay taxes, insurance, maintenance and other operating expenses associated with the leased asset. Such amounts are not included in the measurement of the ROU assets and lease liabilities to the extent they are variable in nature. These variable lease costs are recognized as a variable lease expense when incurred.

As a practical expedient, lease agreements with lease and non-lease components are accounted for as a single lease component for all asset classes. The Company estimates contingent lease incentives when it is probable that the Company is entitled to the incentive at lease commencement. The Company elected the short-term lease recognition exemption for all leases that qualify. Therefore, leases with an initial and expected renewal term of 12 months or less are not recorded on the consolidated balance sheets; instead, lease payments are recognized as lease expense on a straight-line basis over the lease term. The depreciable life of the ROU assets and leasehold improvements are limited by the expected lease term unless the Company is reasonably certain of a transfer of title or purchase option.

The Company elected to use a risk-free rate based on US treasury information as the discount rate for all leases.

Franchise Agreements

The Company has two Franchise Programs: Single Unit Franchise program and Development program. With the Single Unit Franchise, the Company receives an initial franchise fee in exchange for granting a nonexclusive right to operate one, and only one, Sky Zone Recreation Center at a specified location. The initial franchise fee for a single Recreation Center is generally paid in a lump sum upon commencement of the franchise agreement. The Development Program requires an executed Multi-Unit Development agreement which assigns a territory which requires the opening and operating of a designated number of Recreation Centers within a specified time period. A lump-sum development fee for the multiple units is paid at the time of commencement of the Multi-Unit Development agreement. A separate franchise agreement is executed for each unit. In addition to the granting an exclusive right to operate a Recreational Center, the fees also include a training program for the operation of the Recreation Center, and intellectual property of the Company for the principal owner and the managers and or assistant managers, support for site selection, architectural plans, interior and exterior design and layout, marketing and sales techniques, and other opening assistance.

Revenue Recognition

The Company's revenues consist of initial franchise fees and administrative fees received from franchisees, royalties based on a percentage of franchisee revenues, and merchandise sales to franchisees. The amount of revenue recognized reflects the consideration the Company expects to receive in exchange for the services or products. The Company excludes from revenue the sales tax it collects on behalf of government authorities. The revenue recognized in the current year that was included within the contract liability balance at the beginning of the year approximates the prior year deferred revenues and deferred franchise fees – current balances. Specifics on how the new revenue guidance provided in Accounting Standards Codification ("ASC") 606 is reflected across our revenue streams are discussed below:

Franchise Fee Revenue

Franchise revenues consist of the initial franchise fee and in some cases an additional administration fee for multi-unit agreements with the amounts specified as part of the franchise agreement entered into between the Company and the franchisee. The Company recognizes franchise and administrative fees as revenue ratably over the term of the park addendum contracts (e.g. from the execution date until the expiration date – typically 10 years).

Royalty Revenue

Continuing royalty fees received from franchisees are recognized in the period of the related franchisee sales.

Merchandise Revenue

Revenues from the sale of merchandise are recognized upon shipment to the franchisees, which is the point when the risk of ownership is transferred. Revenues from discounts, rebate commissions, promotional allowances, and other benefits from vendors are recognized upon shipment to franchisees.

Deferred Initial Franchise Costs

Consistent with revenue recognition relative to initial franchise fees, the Company defers all direct, incremental costs associated with initial franchise fees over the life of the franchise agreement. These costs consist mainly of commissions and international withholding expenses.

Income Taxes

As the Company is a limited liability company that is treated as a partnership for U.S. federal tax purposes, all taxable income and tax benefits are passed through to the members. Therefore, no liability for U.S. federal or state income taxes is reflected in these consolidated financial statements. However, the Company is subject to franchise taxes.

The Company had no unrecognized tax benefits at December 31, 2023 and 2022. The Company's federal and state income tax returns remain open and management continually evaluates expiring statutes of limitations, audits, proposed settlements, changes in tax law and new authoritative rulings.

If necessary, the Company recognizes interest and penalties associated with tax matters as part of operating expenses and includes accrued interest and penalties with the related tax liability in the balance sheets. No such amounts are included in the consolidated financial statements for the years ended December 31, 2023 and 2022.

Advertising

Advertising and promotional costs are expensed as they are incurred. Advertising expenses not reimbursed by the Company's unconsolidated advertising fund for the years ended December 31, 2023 and 2022 were \$33,000 and \$277,300, respectively.

Recently Adopted and Issued Accounting Standards

On January 1, 2022, the Company adopted Accounting Standards Update ("ASU") 2016-02, "Leases (Topic 842)," using the modified retrospective approach. This pronouncement requires lessees to recognize a lease liability and a right-of-use asset for each lease with a term longer than twelve months and adds new presentation and disclosure requirements for both lessees and lessors. The recognized liability is measured at the present value of lease payments not yet paid, and the corresponding asset represents the lessee's right to use the underlying asset over the lease term and is based on the liability, subject to certain adjustments. For income statement and statement of cash flow purposes, the standard retains the dual model with leases classified as either operating or finance leases. Operating leases will result in straight-line expense while finance leases will result in a front-loaded expense pattern.

The Company elected the package of practical expedients permitted under the transition guidance in ASC 842 and did not reassess prior conclusions related to contracts containing leases, lease classification and initial direct costs. The Company elected the hindsight practical expedient, which permits the use of hindsight when determining lease term and impairment of operating lease assets.

The adoption of the new lease standard resulted in the recognition of operating lease ROU assets and lease liabilities for lease arrangements with an initial term greater than twelve months. The adoption of ASU 842 had a material impact on the Company's consolidated statements of income and consolidated statements of cash flows. The impact on the consolidated balance sheets as of adoption date is shown in Note 4.

In June 2016, the FASB issued ASU No. 2016-13, *Financial Instruments* — *Credit Losses* (*Topic 326*): *Measurement of Credit Losses on Financial Instruments* ("ASU 2016-13"), which changes the impairment model for most financial assets. The ASU introduces a new credit loss methodology, Current Expected Credit Losses (CECL), which requires earlier recognition of credit losses, while also providing additional transparency about credit risk. Since its original issuance in 2016, the FASB has issued several updates to the original ASU. The CECL

framework utilizes a lifetime "expected credit loss" measurement objective for the recognition of credit losses for loans, held-to-maturity securities and other receivables at the time the financial asset is originated or acquired. The expected credit losses are adjusted each period for changes in expected lifetime credit losses. The methodology replaces the multiple existing impairment methods, which generally require that a loss be incurred before it is recognized.

On January 1, 2023, the Company adopted the guidance using a modified retrospective approach with a cumulative adjustment to Member's Equity. The Company has not restated comparative information for 2022 and, therefore, the comparative information for 2022 is reported under previous guidance and is not comparable to the information presented for 2023. The adoption of this standard did not have a material impact on the Company's consolidated financial statements.

3. Property, Equipment, and Software

Property, equipment, and software are summarized by major classifications as of December 31, 2023 and 2022, as follows:

	Useful Life	2023	2022
Furniture and equipment Computers & software	1 – 7 years 3 years	\$ 1,090,200	\$ 201,300 1,090,300
Leasehold improvements Construction In Progress	Lesser of ten years or lease life	155,900 -	155,900 5,200
Ŭ	•	\$ 1,246,100	\$1,452,700
Less: Accumulated depreciation		\$ 1,177,800 68,300	1,361,700 \$ 91,000

Depreciation expense totaled \$18,200 and \$67,600 for the years ended December 31, 2023 and 2022, respectively.

4. Leases

The Company has one lease arrangement for office space that expires in 2024. The right of use asset and lease liability associated with this lease is separately stated on the face of the Company's balance sheets. The lease asset and liability were calculated utilizing the risk-free discount rate of 0.97% according to the Company's elected policy. The Company's operating lease expense for the year ended December 31, 2023 was \$526,600 and is recorded in the "Building and information technology" line of the Company's consolidated statements of income. Short-term lease costs were not significant. Variable lease costs for the year ended December 31, 2023 were 8,700.

Future minimum annual rentals, including reasonably assured options, required under leases that had terms greater than one year are as follows:

Year ended December 31,

2024	132,600
Total lease payments	132,600
Less: imputed interest	2,600
Present value of lease liabilities	\$ 135,200

In adopting ASU 2016-02, the Company elected to use the modified retrospective approach. No cumulative-effect adjustment to retained earnings in the period of adoption was required; however, a remeasurement of lease assets and lease options was required. A summary of adjustment to the January 1, 2022 consolidated balance sheets upon adoption is shown below:

	,		SU 2016-02 ljustments	J	anuary 1, 2022
Operating lease right-of-use assets	\$	-	\$ 1,084,400	*	1,084,400
Deferred rent - current		23,800	(23,800)	\$	-
Deferred rent - long-term		52,600	(52,600)	\$	-
Operating lease liabilities, current		-	\$ 526,500	\$	526,500
Operating lease liabilities, long-term		-	\$ 634,400	\$	634,400

5. Related Party Transactions

The Company maintains and manages an advertising fund that has not been consolidated in these financial statements. This related party fund receives advertising fees from the franchisees for use in advertising spending to promote the Sky Zone brand. As of December 31, 2023 the Company owes the advertising fund \$102,000. As of December 31, 2022 the advertising fund owed the Company in the amount of \$136,500.

The Company's sole member, CircusTrix Holdings, LLC, has paid certain costs on behalf of the Company. Additionally, cash proceeds from sales have been sent from the Company to CircusTrix Holdings, LLC. Related party receivables, current net, total \$67,514,600 and \$46,326,800 as of December 31, 2023 and 2022, respectively.

6. Retirement Plan

The Company maintains a 401(k) Savings Plan in which most employees are eligible to participate. Eligible employees may make contributions not to exceed the maximum statutory contribution amounts. The Company may match a percentage of each employee's contributions consistent with the provisions of the plan for which they are eligible. All employee and employer contributions fully vest immediately. The Company contributions to the Plan totaled \$22,300 and \$23,200 for the years ended December 31, 2023 and 2022, respectively, and are included in personnel expenses in the accompanying statements of income.

7. Subsequent Events

The Company reviewed all material events from December 31, 2023 through the date the consolidated financial statements were available to be issued, April 05, 2024. No items were identified that would materially affect the consolidated financial statements or require additional disclosure.

EXHIBIT G TO THE DISCLOSURE DOCUMENT

LIST OF-STATE AGENCIES/AND AGENTS FOR SERVICE OF PROCESS

STATE AGENCIES/AGENTS FOR SERVICE OF PROCESS

Listed here are the names, addresses and telephone numbers of the state agencies having responsibility for-the franchising disclosure/registration laws. We may not yet be registered to sell franchises in any or all of these states.

If a state is not listed, we have not appointed an agent for service of process in that state in connection with the requirements of the franchise laws. There may be states in addition to those listed below in which we have appointed an agent for service of process. There <u>may</u> also <u>may</u> be additional agents appointed in some of the states listed.

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF
		PROCESS
CALIFORNIA	Department of Financial Protection and	Commissioner of Department of
	<u>Innovation</u>	<u>Financial</u>
	One Sansome Street, Suite 600	<u>Protection and Innovation</u>
	San Francisco, CA 94104	320 West 4th Street, Suite 750
	(415) 972-8559	Los Angeles, CA 90013-2344
	(866) 275-2677	<u>(866) 275-2677</u>
FLORIDA	Dept. of Agriculture & Consumer Services	Same
	Division of Consumer Services	
	Mayo Building, Second Floor	
	Tallahassee, FL 32399-0900	
	(850) 245-6000	
HAWAII	Dept. of Commerce & Consumer Affairs	Commissioner of Securities of the State
	Business Registration Division	of Hawaii
	Commissioner of Securities	Dept. of Commerce & Consumer Affairs
	335 Merchant Street, Room 203	Securities Compliance Branch
	Honolulu, HI 96813	335 Merchant Street, Room 203
	(808) 586-2722	Honolulu, HI 96813
ILLINOIS	Franchise Division	Illinois Attorney General
	Office of the Attorney General	Same Address
	500 South Second Street	Suite Fiduress
	Springfield, IL 62706	
	(217) 782-4465	
INDIANA	Securities Commissioner	Indiana Secretary of State
INDIANA	Indiana Securities Division	302 West Washington Street, Room E
	302 West Washington Street, Room E 111	
		018 Indiananalia IN 46204
	<u>Indianapolis, IN 46204</u>	<u>Indianapolis, IN 46204</u>
	(317) 232-6681	(317) 232-6531
<u>KENTUCKY</u>	Kentucky Attorney General's Office	Same
	Consumer Protection Division	
	1024 Capitol Center Drive	
	Frankfort, KY 40602	
	(502) 696-5389	

 $\frac{\underline{Sky\ Zone\ Franchise\ Group,\ LLC}}{\underline{April\ 2025\ FDD}}$

Ex. G – State Agencies/Agents for Service of Process

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF
=====		PROCESS
MARYLAND	Office of the Attorney General	Maryland Securities Commissioner
	Securities Division	Same Address
	200 St. Paul Place	
	Baltimore, MD 21202-2020	
	(410) 576-6360	
MICHIGAN	Michigan Dept. of Attorney General	Michigan Dept. of Attorney General
<u> </u>	Consumer Protection Division	Same
	Attn: Franchise Section	
	525 W. Ottawa Street	
	G. Mennen Williams Bldg., 1st Floor	
	Lansing, MI 48913	
	(517) 373-7117	
MINNESOTA	Minnesota Dept. of Commerce	Minnesota Commissioner of Commerce
MINIESOTA	85 7th Place East, Suite 280	Same Address
	St. Paul, MN 55101-2198	Same Address
	(651) 539-1600	
NEBRASKA		Como
<u>NEDKASKA</u>	Dept. of Banking & Finance Bureau of Securities/Financial Institutions	Same
	Division 1526 V. Street Suite 200	
	1526 K Street, Suite 300	
	Lincoln, NE 68505-2732	
	P.O. Box 95006	
	Lincoln, NE 68509-5006	
NEW YORK	(402) 471-2171	N X 1 C . CC.
NEW YORK	New York State Dept. of Law	New York Secretary of State
	Investor Protection Bureau	New York Dept. of State
	28 Liberty Street, 21st Floor	One Commerce Plaza
	New York, NY 10005	99 Washington Avenue, 6th Floor
	Phone: (212) 416-8236	Albany, NY 12231-0001
NODEN	Fax: (212) 416-6042	(518) 473-2492
NORTH	North Dakota Securities Dept.	North Dakota Securities Commissioner
<u>DAKOTA</u>	600 East Boulevard Avenue	Same Address
	State Capitol, 14th Floor, Dept. 414	
	Bismarck, ND 58505-0510	
	<u>Phone: (701) 328-4712</u>	
RHODE	<u>Dept. of Business Regulation</u>	<u>Director, Dept. of Business Regulation,</u>
<u>ISLAND</u>	<u>Securities Division</u>	Securities Division
	1511 Pontiac Avenue	Same Address
	<u>John O. Pastore Complex – Bldg. 68-2</u>	
	Cranston, RI 02920	
	<u>(401) 222-3048</u>	
SOUTH	Department of Labor and Regulation	Director of the Department of Labor and
DAKOTA	<u>Division of Insurance – Securities</u>	Regulation
	Regulation	Division of Insurance – Securities

Sky Zone Franchise Group, LLC

<u>April 2025 FDD</u>

Ex. G – State Agencies/Agents for Service of Process

STATE	STATE ADMINISTRATOR	AGENT FOR SERVICE OF
		PROCESS
	124 S. Euclid, Suite 104	Regulation
	Pierre, SD 57501	Same Address
	(605) 773-3563	
<u>TEXAS</u>	Secretary of State	Same
	Statutory Documents Section	
	<u>P.O. Box 12887</u>	
	<u>Austin, TX 78711-2887</u>	
	<u>(512) 475-1769</u>	
<u>UTAH</u>	<u>Utah Dept. of Commerce</u>	Same
	Consumer Protection Division	
	160 East 300 South (P.O. Box 45804)	
	Salt Lake City, UT 84145-0804	
	<u>Phone: (801) 530-6601</u>	
	<u>Fax: (801) 530-6001</u>	
<u>VIRGINIA</u>	State Corporation Commission	Clerk, State Corporation Commission
	Div. of Securities & Retail Franchising	1300 E. Main Street, 1st Floor
	1300 E. Main Street, 9th Floor	Richmond, VA 23219-3630
	Richmond, VA 23219	<u>(804) 371-9672</u>
	<u>(804) 371-9051</u>	
WASHINGTON	Dept. of Financial	Director, Dept. of Financial Institutions
	<u>Institutions Securities</u>	Securities Division
	<u>Division 150 Israel Rd S.W.</u>	Same Address
	<u>Tumwater, WA 98501 (360) 902-8760</u>	
WISCONSIN	Dept. of Financial Institutions	Wisconsin Commissioner of Securities
	<u>Division of Securities</u>	Same Address
	4822 Madison Yards Way, North Tower	
	<u>Madison, WI 53705</u>	
	<u>(608) 266-3431</u>	

CALIFORNIA

Commissioner of Department of Financial Protection & Innovation
Department of Financial Protection & Innovation Toll Free: 1 (866) 275-2677

Los Angeles

Suite 750 320 West 4th-Street Los Angeles, California 90013-2344 (213) 576-7500

Sacramento

2101 Arena Boulevard Sacramento, California 95834 (866) 275-2677

San Diego

1350 Front Street, Rm. 2034 San Diego, California 92101-3697 (619) 525-4233

San Francisco

One Sansome Street, Suite 600 San Francisco, California 94105-2980 (415) 972-8559

HAWAII

(for service of process)

Commissioner of Securities
Department of Commerce
and Consumer Affairs
Business Registration Division
335 Merchant Street, Room 203
Honolulu, Hawaii 96813
(808) 586-2722

(for other matters)

Commissioner of Securities
Department of Commerce
- and Consumer Affairs
Business Registration Division
335 Merchant Street, Room 205
Honolulu, Hawaii 96813
(808) 586-2722

ILLINOIS

Illinois Attorney General 500 South Second Street Springfield, Illinois 62706 (217) 782-4465

INDIANA

(for service of process)

Indiana Secretary of State 201 State House 200 West Washington Street Indianapolis, Indiana 46204 (317) 232-6531

NEW YORK

(state agency)

Indiana Secretary of State Securities Division Room E-111 302 West Washington Street Indianapolis, Indiana 46204 (317) 232-6681

MARYLAND

(for service of process)

Maryland Securities Commissioner at the Office of Attorney General-Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

(state agency)

Office of the Attorney General Securities Division 200 St. Paul Place Baltimore, Maryland 21202-2021 (410) 576-6360

MICHIGAN

Michigan Attorney General's Office Consumer Protection Division Attn: Franchise Section G. Mennen Williams Building, 1st Floor 525 West Ottawa Street Lansing, Michigan 48933 (517) 335-7567

MINNESOTA

Commissioner of Commerce Department of Commerce 85 7th Place East, Suite 280 St. Paul, Minnesota 55101 (651) 539-1600

NEW YORK

(for service of process)

Attention: New York Secretary of State New York Department of State One Commerce Plaza, 99 Washington Avenue, 6th Floor Albany, New York 12231-0001 (518) 473-2492

(Administrator)

NYS Department of Law Investor Protection Bureau 28 Liberty Street, 21** Floor New York, New York 10005 (212) 416-8236 (Phone)

NORTH DAKOTA

(for service of process)

Securities Commissioner
North Dakota Securities Department 600 East Boulevard Avenue, Suite 414 Bismarck, North Dakota 58505
(701) 328-4712

(state agency)

North Dakota Securities Department 600 East Boulevard Avenue, Suite 414 Bismarck, North Dakota 58505 (701) 328-2910

OREGON

Oregon Division of Financial Regulation 350 Winter Street NE, Suite 410 Salem, Oregon 97301 (503) 378-4140

RHODE ISLAND

Securities Division
Department of Business Regulations
1511 Pontiac Avenue
John O. Pastore Complex-Building 69-1
Cranston, Rhode Island 02920
(401) 462-9500

SOUTH DAKOTA

Division of Insurance Securities Regulation 124 S. Euclid, Suite 104 Pierre, South Dakota 57501 (605) 773-3563

VIRGINIA

(for service of process)

Clerk, State Corporation Commission 1300 East Main Street First Floor Richmond, Virginia 23219 (804) 371-9733

(for other matters)

State Corporation Commission
Division of Securities and Retail Franchising
Tyler Building, 9th Floor
1300 East Main Street
Richmond, Virginia 23219
(804) 371-9051

WASHINGTON

(for service of process)

Director Department of Financial Institutions Securities Division 150 Israel Road SW Tumwater, Washington 98501 (360) 902-8760

(for other matters)

Department of Financial Institutions Securities Division P. O. Box 9033 Olympia, Washington 98501-9033 (360) 902-8760

WISCONSIN

(for service of process)

Administrator, Division of Securities
Department of Financial Institutions
4822 Madison Yards Way, North Tower
Madison, Wisconsin 53705
(608) 266-9555

(state administrator)

Division of Securities
Department of Financial Institutions
4822 Madison Yards Way, North Tower
Madison, Wisconsin 53705
(608) 266-9555

EXHIBIT H TO THE DISCLOSURE DOCUMENT

FORM OF PURCHASE ORDER



Sales Order

> Contact:

> Contact Email:

> Installation Address:

Sky Zone

Quote valid 15 days **Quote Date**

Scope of Work Per Equipment Approval Plan Date:

Attraction			Sqft	Capacity	Serial #	Tota
Open Jump						\$0.00
Oodgeball Court #1					ļ	\$0.00
SkySlam	, Drop Zone Trampoline Structure			-}	 	\$0.00
Orop Zone (With Air Ba	Airbag, 1 Blower, Alarm, & Pit Liner Sheets					\$0.00
	Tower- 24', Multi Attraction, Roof					\$0.00
	16' Launch Slide, 2' AFF.	1				\$0.00
	17' Race Slide Slide, 0 AFF.	1				\$0.0
	18' Family Slide Slide, 0 AFF.	1				\$0.0
lega Launch	Slider Mats	40				\$0.0
	Ground Railing (ft) + 2 Gates Airbag *DOMESTIC OPTION* (9' W X 24' L X 2.5' H), Blower, Ala	193 m				\$0.0 \$0.0
	Side Sheets- Polyester/Spandex					\$0.0
	2" Pit Liner Sheet- 2" x 48" x 108"	3 7				\$0.0
	Trapeze & Swing Structure			<u> </u>	······································	\$0.0
	Shared Tower Add On					\$0.0
	Pit Type (29'L x 16'W x 4'H) Air Bag					\$0.0
rapeze and Swing	Airbag *DOMESTIC OPTION* 16'W x 29' L x 4' T, Blower, Alarm					\$0.0
	Side Sheets- Polyester/Spandex					\$0.0
	2" Pit Liner Sheet- 2" x 48" x 108"					\$0.0 \$0.0
	2" crash Pad along exits sides of Trapeze and Swing Exit Single Zip Line Structure	Right Exit		-}	 	\$0.0 \$0.0
	Zip Line Structure Zip Line Launch Platform	ragnt Lait				\$0.0 \$0.0
	Shared Tower Add On					\$0.0
ip Line	<u> </u>	Airbag				\$0.0
ip Line	Exit Floor Pads on both sides					\$0.0
	Airbag 7'W x 50' L x 4' T,					\$0.0
	Side Sheets- Polyester/Spandex	4				\$0.0
	2" Pit Liner Sheet- 2" x 48" x 108"	14		<u></u>	ļ	\$0.0
	(Includes railing & rail padding, exit pole, graphic side & back par pad flooring)	els. Excludes runway & crash				
	Number of Lanes:	2				
	Lane Width (ft):	3				
Varped Wall	Wall Heights:					\$0.0
	Exit Side:					
	Exit Landing Height:					
	·	Sky Zone, Orange Rail				
	LED Upgrade (Optional): Boulder Ball- 1.5m- Dark Blue	res 6				
hallenge Zone	Boulder Ball- 2m- Orange	2				\$0.0
-	Pit Foam (22'L x 28'6"W x 3'H, to be confirmed by FP.) Includes (Orange covered cubes only				\$0.0
	King of the Hill	1				\$0.0
		1				\$0.0
hallenge Zone	Hanging Silks Log Roll	4 1	•			\$0.0 \$0.0
	Air Bag (53'L x 22'W x 3'H)	<u> </u>				\$0.0
wall x2	4 Screens and Mat-Subscription not included \$60 monthly per Ur	it		}	<u> </u>	\$0.0
ofy Play	Structure by GFS *Per Approved Design*	•••••				\$0.0
\irDash	Includes exit padding					\$0.00
	{·•···•·······························	IC OPTION*				
ir Court	Hoop System Type Perimeter Netting with Support Trusses	Fixed Yes				\$0.0
	Soccer Goals	Yes				
	Cells:	4				
	Raised or Ground:	Ground				
	Domestic or Overseas:	Domestic Crash padding	-			
Varrior Course	Fall Protection Type Course Structure, Platforms, Padding	Crash padding				\$0.0
	Side Netting Containment Panels	Yes				
	Overhead Tube Assemblies	Included				
	Elements Per Plan- Inflatables and Net Tunnel	Included				
			Sqft	Capacity		\$0.0
		Total	ft	0 0 Qty		\$0.0 \$0.0
it Edge Padding			11	Qty	OTHE FILE	\$0.0
Shapped Pad for Ch	allenge Zone Wall				†	\$0.0
' Perimeter Skirt					}	\$0.0
adding for Rails and	Steel					\$0.0
/all Padding 8' Tall wi						\$0.0
lall Padding 5' Tall wi	th Graphic- Above angled trampoline wall					\$0.0
nflatable Joust Sticks			-		- 4	\$0.0

8' Column Pad					\$0.00
Advance Trampoline	Layout Start Date TB	BD L	ayout End Date	TBD	\$0.00
Layout	Includes, Travel, Materials, Labor, Lodging, Per Diem and Tools				\$0.00
Install	Includes, Travel, Materials, Labor, Lodging, Per Diem and Tools/Equ	ipment Rental			\$0.00
Overhead & Manageme	ent			2.5%	\$0.00
		At	traction & Installat	ion Subtotal	\$0.00
Replacement Parts		1	QTY Item Price	Subtotal	\$0.00
***************************************		***************************************	Replaceme	nt Part Total	\$0.00
Shipping, Duties, Impor	rt Taxes & Fees (Cost plus 5%. Estimated, to be reconciled to actual.)			\$0.00
Sales Tax				0.00%	\$0.00
Engineering (Attraction re	eview to be billed if incurred)				\$0.00
Attraction Permiting As	ssistance (Additional fees, 3rd Party inspections, expenses, permits, et	tc. required by local ordinances	billed in addition if inci	urred.) See	
T&C's for a detailed bro	eakdown				\$0.00
				Grand Total	\$0.00

Payment Schedule

🗀 1 st Payment	Due when order is placed	40%	January 15, 1900	\$0.00
> 2 nd Payment	Due before shipment of materials	50%	TBD	\$0.00
> 3 rd Payment	Due upon project completion	10%	TBD	\$0.00

Please make checks payable to Sky Zone, LLC., and mail to:

Sky Zone, LLC.

Attn: Accounting Department 86 N University Ave #350 Provo. UT 84601

For Wire Transfers:

- 1. Please notify Accounting@skyzone.com when you are going to process a wire, so we can apply the funds accordingly.
- 2. Please make sure that you put the invoice number and park name somewhere so we can identify the deposit.

Direct To: Wire Routing Transit Number (RTN/ABA) 322070381

Bank Name: East West Bank

Bank Address: 9300 Flair Drive, 4th Fl. El Monte, CA 91731 BNF/Field 4200 Beneficiary Account Number: 8003096388

Equipment Approval and Sales Order General Term and Conditions

Please review each section

- > Equipment layouts are based on the provided building dimensions. A shell building may have sheetrock or other materials installed on walls or support columns prior to equipment installation. These materials will change the overall dimensions of the usable space, and should be factored in when providing dimensions to Sky Zone, LLC. for design work. Sky Zone, LLC. is not responsible for any construction problems that arise from inaccurate building dimensions provided by the Franchise Partner. Any costs or delays incurred as a result of incorrect building dimensions are the sole responsibility of the Franchise Partner.
- > The equipment area requires a level & unobstructed concrete slab. A clearance height of 19' (Foam Zone), and 17' (other equipment) to the lowest ceiling obstruction is required from the concrete surface. Lower clearance may require padding and additional review.
- > The following are concrete slab design loads (non-factored, allowable loads) for Trampoline Courts. A local engineer should always be consulted to determine if the existing elevated slab (and supporting structure) can safely support the trampoline design loads and vibrations.

 A. 10 lbs./sq. ft. D.L
 - B. 100 lbs./sq. ft. L.L.
 - C. 4,000 lbs. max L.L. Concentrated point load, up or down
- D. 1,000 lbs. max D.L. Concentrated point load, down
- > The Sky Zone trampoline system is engineered for a 4" thick, contiguous concrete slab with a compressive strength of 3,000 psi. Reinforcing should be #4 @ 18" on center (max) each way. If the concrete slab does not meet these specifications:
- A. If the minimum concrete condition above is not met, the courts may have to be customized to accommodate the engineering and installation requirements of the system. This can include but is not limited to custom base plates, posts, and anchor bolts. Any additional engineering, material, or installation costs resulting from customization are the sole responsibility of the Franchise Partner.
- B. For locations on raised slabs (those not on-grade with a cavity beneath), the Franchise Partner must provide concrete GPRS scanning in all areas where the courts will be installed prior to the start of the court approval plans. Any obstructions (electrical, plumbing, structural steel supports, etc.) less than 9" below the concrete surface need to be permanently marked on the floor prior to the start of installation. Any damage resulting from contact with an unmarked obstruction is the sole responsibility of the Franchise Partner. Any active electrical or plumbing lines within 9" of the concrete surface must be turned off during the equipment installation.
- C. Cracks may develop around anchor locations, or concrete can fail. This may require you to repair or completely replace the concrete slab. All costs associated with the repair of concrete during or after court construction are the sole responsibility of the Franchise Partner. There is also the possibility of court posts or anchor plates being located on concrete seams. Additional concrete work required to anchor these posts or anchor plates are the sole responsibility of the Franchise Partner.
- D. References to concrete, the contractors who installed the same, their services, and products are provided without warranty of any kind, either expressed or implied. In no event shall Sky Zone, LLC., their agents, employees, or affiliates, be liable for any special, incidental, direct, indirect, or consequential damages of any kind, or any damages whatsoever resulting from loss
- > The Equipment Approval Plan reflects the platform and railing adjustments required to adjoin properly to the equipment. Changes should be reviewed by your architect and communicated to your contractor and railing vendor. Equipment entrance decks will need to meet local code requirements and are to be completed by a local contractor. Please review the layout of the entrance decks and stairwells with your architect or code expert.
- > General construction in the equipment area needs to be completed before the start of equipment installation. This includes but is not limited to: painting, sprinkler work and installation of cages on sprinkler heads, lighting and lighting cages, electrical work, wiring for the sound system and speaker installation, wiring and installation of security cameras, HVAC work, and any other overhead work that requires access over the equipment areas. Floors in the equipment area must be swept clean, with all debris and construction material removed out of the
- > The advanced equipment layout, installation, and any subsequent major equipment modifications are to be performed by the Sky Zone Construction team. Assistance from the Franchise Partner's team may also be required, per the terms of the final Sales Order.
- > Advance layout and equipment installation start dates can be tentatively scheduled upon receipt of executed sales order and receipt of deposit. These dates are determined based on information and guidance provided by the Franchise Partner or their contractor, and the availability of the Sky Zone, LLC. team. Final construction dates can be confirmed upon placement of orders and availability of the Sky Zone. LLC. team. Lead time for delivery of materials is 6-8 weeks from the time a signed sales orders and deposit are received.
- > All requirements outlined above, and in the preconstruction communication from Sky Zone, LLC., must be fulfilled prior to installation. If the requirements have not been met by the installation start date, Sky Zone, LLC. reserves the right to remove the crew from the job site, and return the crew when the site is ready, at the Franchise Partner's sole expense.
- > Sky Zone, LLC. requires 3 week's advance notice, in writing, to change the installation start date. Changes less than 2 weeks before the start of scheduled installation will result in additional travel charges. A revised installation date will be subject to the Sky Zone, LLC. team's availability.
- Power of at least (3) dedicated circuits is required.
- > Sufficient lighting must be available in all equipment areas for the duration of the advance layout and construction.
- > Large construction dumpsters or other trash removal service must be provided by the contractor or Franchise Partner for the duration of the equipment installation.
- A roll up door or other large entry way leading to a loading zone is required during equipment installation. If roll up door or access entry way is not provided, additional installation costs and schedule duration will be incurred and will be the responsibility of the Franchise Partner.
- > A lift and forklift will need access to the equipment area during the entire equipment installation process. If the platforms are built prior to the Installation team's arrival, access must be left open for lift and forklift access. If access to equipment areas are blocked, additional installation cost and schedule duration will be incurred and will be the responsibility of the Franchise Partner. A paved unloading area and access pathway to the site is also required.
- It is the sole responsibility of the Franchise Partner to disclose any installation and delivery logistics restrictions prior to executing the sales order. This includes but is not limited to: safety, working hours, noise, and delivery acceptance procedures. Failure to properly disclose such information may result in additional cost or delays.

- > Epoxy flooring work, or other work which can create hazardous or unhealthy conditions in the facility, cannot be conducted while the equipment installation team is on site. Please advise your Sky Zone, LLC. project manager of any other potentially dangerous conditions that the installation team should be aware of.
- > For projects where the Sky Zone, LLC construction team is limited to less than a full team, additional laborers are required to be provided by the Franchise Partner. For workers provided by the Franchise Partner that do not have OSHA safety training necessary to conduct the court installation processes, Sky Zone LLC. team and its members are not responsible for injuries that occur from the misuse of power equipment such as drills, saws, etc., from the mishandling of ladder equipment, or from any other accidents that may occur due to insufficient training. When scissor-lift access is inaccessible to the equipment installation area, the Sky Zone LLC installer will not advise any laborers to use a ladder in order to lift heavy objects overhead without proper safety training. For that reason, it is required that a material handler, lift table or similar mobile lift device is used to lift heavy objects overhead. It is the responsibility of the Franchise Partner to acquire this mobile lift device unless included in the "Equipment Rental" cost in the Sales Order. Sky Zone LLC. team members are not responsible for lifting heavy objects by use
- of a ladder.

 The Franchise Partner is required to have a designated Maintenance Team Member hired and ready for training when equipment installation begins. The Maintenance Team Member will be required during specified hours for 40 hours each week until construction is complete. All salary and benefit costs for the Maintenance Team Member are the sole responsibility of the
- > Equipment entrance pads will only be installed if the equipment platforms are complete. If platforms are not complete, installing the entrance pads are the responsibility of the Franchise Partner
- > Franchise Partner shall pay Sky Zone, LLC. in installments specified by the sales order. Should Buyer fail to pay any sum due Seller after the due date of any payment, Buyer shall pay Seller interest on the delinquent payment from the due date until paid at the rate of eighteen percent (18%) per annum; plus, Buyer shall be liable for any and all attorneys' fees, costs and expenses incurred by Seller in attempting to collect any delinquent payment.
- > Re-design fees of \$85/hour will be charged for all requested changes after the equipment layout has been approved. Any cost increases due to revised orders are the sole responsibility of
- - A. No Liability. Neither party will be liable for performance delays nor for non-performance due to causes beyond its reasonable control, except for payment obligations.
- B. Best Efforts to Cure. In the event of a threatened default or default as a result of any cause beyond its reasonable control, the defaulting party shall nonetheless exercise its best efforts
- C. Right to Terminate. In the event such an event prevents performance thereunder for a period in excess of ninety (90) days, then the non-defaulting party may elect to terminate this Agreement and/or cancel or suspend any Sales Orders thereunder by a written notice to the defaulting party.
- Risk of loss of the Equipment shall pass to the Buyer upon completion of installation.
- Limitation of liability for erected equipment:
- Buyer hereby agrees that seller shall have no liability to buyer for any claim, demand, damage, loss or liability of any kind or nature whether or not caused in whole or in part by seller or seller's negligence, whatsoever. To the extent permitted by law, buyer shall indemnify, defend (including, but not limited to, payment of all attorneys' fees and legal costs) and hold harmless seller from and against any claim, demand, damage, loss and liability of any kind and nature, including but not limited to, personal injuries (including death) and/or property damage, relating
- - ance tely

	prient (and this sale, including, willout limitation, its manufacture, selection, purchase, delivery, maintenance, assembly, installation, ins
Additional Insured:	a officers, owners, employees and agents.
Buyer agrees to name Seller as an additional	insured and to require any of Buyer's contractors to name Seller as an additional insured on any and all Commercial General Liability insurated by Buyer, and/or its contractor for the duration of the contract between buyer and Seller and until such time as the Equipment is comple
Seller warrants title to and freedom from encu contract. Seller's only liability under this warra months from the date of substantial completion. Bosu balls for SkyJoust pedestals are not cov seller in lieu of all other warranties, including the liable to the buyer, its customers or third partied damage, liquidated damages, lost sales or lost indemnify, defend (including, but not limited to kind and nature, including but not limited to, parties of the sales of the sale	mbrances of the goods, and that the goods shall be of good quality and in conformance with the description and specifications contained inty shall be to repair or provide replacement product, at seller's option, for the goods that are proven to be defective for: 1) a period of (6) in for trampoline(s), springs and trampoline pads; and 2) a period of (1) one year from the date of substantial completion for all other hardward under this warranty, and must be returned to the manufacturer in the event of any discovered defects. This express warranty is made he warranties of merchantability and fitness for a particular purpose, which are excluded from this sale and shall not apply. Seller shall not be sof or any claim, loss or damage, general or specific, incidental or consequential, including but not limited to death, personal injury, property st profits, which in any manner arises out of or in connection with the sale or use of the goods. To the extent permitted by law, buyer shall be, payment of all attorneys' fees and legal costs) and hold harmless seller from and against any claim, demand, damage, loss and liability of personal injuries (including death) and/or property damage, relating to, in connection with, or arising from the equipment and this sale, including burchase, delivery, maintenance, assembly, possession, prior use, subsequent use, dismantling, operation, ownership, or control by buyer ager.
l,	, have reviewed and agree to the Sales Order, and the Equipment Approval and Sales Order General Term and Conditions.
Signature	Date

EXHIBIT I TO THE DISCLOSURE DOCUMENT

AGREEMENT AND CONDITIONAL CONSENT TO TRANSFER

AGREEMENT AND CONDITIONAL CONSENT TO TRANSFER

Park Address: [ADDRESS]

THIS AGREEMENT AND CONDITIONAL CONSENT TO TRANSFER ("Consent") is made by and among SKY ZONE FRANCHISE GROUP, LLC, a Missouri limited liability company ("Franchisor"); [SELLER] ("Seller"); [SELLER GUARANTOR] ("Seller Guarantor"); and [BUYER] ("Buyer"), effective as of the last signature date below (the "Effective Date"). All terms capitalized in this Consent and not otherwise defined herein shall have the meanings ascribed to them in the Seller Franchise Agreement (defined below) or the Buyer Franchise Agreement (defined below), as the case may be.

Recitals

- A. Seller is the franchisee under that certain franchise agreement dated [DATE], as it may have been amended by subsequent addendum or addenda (the "Seller Franchise Agreement"), governing the ownership and operation of the Sky Zone indoor trampoline park at [ADDRESS] (the "Park").
- B. <u>Seller Guarantor personally guaranteed all of the obligations under the Seller Franchise Agreement.</u>
- <u>C.</u> <u>Seller has notified Franchisor that it and Buyer have entered into a purchase and sale agreement dated [DATE] (the "**Purchase Agreement**"), pursuant to which Seller has agreed to sell, and Buyer has agreed to purchase, all of the rights, obligations and assets relating to the Park (the "**Interests**").</u>
- <u>D.</u> <u>Buyer has also agreed to (1) assume the lease obligations for the Park, and (2) enter into Franchisor's current form of franchise agreement to govern the ownership and operation of the Park (the "Buyer Franchise Agreement") (the transfer of Interests under the Purchase Agreement, the assumption by Buyer of the Park's lease obligations and the execution of the Buyer Franchise Agreement, collectively referred to as the "Transfer").</u>
- E. Franchisor has agreed not to exercise its right of first refusal as set forth in the Seller Franchise Agreement and has agreed to approve the Transfer of the Park in accordance with the terms, and subject to the conditions, set forth in this Consent.

Agreement

- NOW, THEREFORE, for and in consideration of the foregoing recitals, which are incorporated herein, the mutual covenants contained herein and other valuable consideration, receipt and sufficiency of which are acknowledged, the parties agree as follows:
- 1. <u>Effective Date</u>. The "<u>Effective Date</u>" will be the date on which Franchisor signs this Consent acknowledging its consent to the proposed Transfer, which date shall be consistent with the effective date of the Buyer Franchise Agreement.

- 2. <u>Purchase Agreement</u>. Seller and Buyer represent and warrant that the form of Purchase Agreement provided to Franchisor is the final version of the Purchase Agreement and is the version which has been, or will be, executed by them to effectuate the Transfer. The Purchase Agreement will not be amended, and the terms set forth in the Purchase Agreement will not be changed, except with the prior written consent of Franchisor.
- 3. <u>Conditional Consent; Release of Guaranty</u>. Notwithstanding anything in this Consent to the contrary, the consent and release set forth herein are expressly contingent upon compliance with the following terms and conditions on or before the date of the closing of the Transfer (the "Closing Date"):
 - <u>a.</u> <u>Franchise Agreement. The Seller Franchise Agreement will terminate as of the Closing Date in accordance with the terms set forth in Section 7 below, and the operation of the Park will thereafter be governed by the Buyer Franchise Agreement.</u>
 - <u>B. Payment of Amounts Due. Seller will pay to Franchisor and its affiliates all amounts due from or accrued by Seller (or any affiliate of Seller) through the Closing Date.</u>
 - <u>C.</u> Transfer Fee. Upon execution of this Consent by Seller and Buyer, a transfer fee in the amount of [] (the "Transfer Fee") shall be paid to Franchisor via cashier's check or wire transfer. Except as described in Section 5 below, Seller and Buyer acknowledge and agree that Franchisor has earned the Transfer Fee upon receipt thereof and that the Transfer Fee is not refundable.
 - deposit [] (the "Fee Deposit") with Franchisor via cashier's check or wire transfer. Franchisor will refund the Fee Deposit to Seller, less any amounts which may be due pursuant to Section 3.b, within sixty (60) days following the later of the Closing Date or the date upon which Seller and Buyer comply with all terms and conditions set forth in this Consent.
 - e. Training. Prior to the Closing Date, Buyer (or its Responsible Person) and its Management Team (as such terms are defined in the Buyer Franchise Agreement) shall satisfactorily complete Franchisor's initial training program (the "Training Program") or must be currently approved by Franchisor to operate and manage the Park.
 - f. Right to Possession. Buyer will provide satisfactory evidence to Franchisor that Buyer has the right to possession of the Park by way of lease assignment and/or assumption or otherwise (with all required landlord consents), as more fully described in Section 6 below.

- g. Seller Financing. Regardless of any provision of the Purchase Agreement (or any other agreement) to the contrary, if Seller provides financing to Buyer for any portion of the purchase price for the Park and such financing is secured by any assets of the Park, Seller acknowledges and agrees that Seller does not and will not have any interests or rights, revisionary or otherwise, to operate the Park after the Closing Date pursuant to the Seller Franchise Agreement or Buyer Franchise Agreement.
- h. Park Upgrades/Renovations. Buyer will complete the upgrades and renovations of the Park identified within the "Park Improvement Plan" document attached as Exhibit A hereto, at Buyer's expense, as required to bring the condition and appearance of the Park into compliance with Franchisor's current Methods of Operation and other System requirements. Buyer shall complete all requirements and specifications in Exhibit A within the timeframes specified therein, but in no event later than six (6) months from the Effective Date.
- 4. <u>Waiver of Right of First Refusal</u>. Franchisor hereby waives its right of first refusal to purchase the Interests, as set forth in the Seller Franchise Agreement.
- 5. <u>Contingency</u>. This Consent and the Buyer Franchise Agreement may be terminated by Franchisor if:
 - <u>a.</u> <u>The Transfer between Seller and Buyer is cancelled, or otherwise not approved by Franchisor;</u>
 - <u>Seller and/or Buyer fail to meet any of the conditions and/or requirements set forth in this Consent, the Seller Franchise Agreement, and/or the Buyer Franchise Agreement; or </u>
 - c. Seller and Buyer fail to change possession and/or ownership of the Park for any reason within ninety (90) days following the anticipated Closing Date.

In the event of such termination, Seller and Buyer will execute a termination and release agreement (in a form acceptable to Franchisor) pursuant to which Franchisor will refund the Transfer Fee, without interest; provided, however, if Buyer or its Management Team have attended any portion of the Training Program, Franchisor will only be obligated to refund fifty percent (50%) of the Transfer Fee.

Assignment/Assumption of Lease. Seller and Buyer acknowledge that one of the requirements of Franchisor's consent is that the Park lease be assigned to and/or otherwise assumed by the Buyer and that the lease for the Park may require consent of and/or notice to the landlord with respect to such assignment and/or assumption. Provided (a) Buyer takes an assignment of the existing lease for the Park; (b) the terms of such lease are not amended; and (c) the lease for the Park includes the terms of Franchisor's required lease addendum, Franchisor waives the requirement for lease review and approval set forth in the Buyer Franchise Agreement. If (i) the lease terms are amended; (ii) the lease for the Park does not include the

terms of Franchisor's required lease addendum; or (iii) Buyer enters into a new lease for the Park, all lease review and approval requirements set forth in the Buyer Franchise Agreement shall remain applicable. Buyer acknowledges and agrees that Franchisor's approval of the Park location and waiver of the lease review requirement or approval of the lease terms do not constitute a recommendation, endorsement, or guarantee by Franchisor of the suitability of the Park location or the lease, and Buyer acknowledges that it has taken all steps necessary to ascertain whether the Park location and lease are acceptable to Buyer.

- 7. Termination of Seller Franchise Agreement and Guaranties. Franchisor and Seller acknowledge and agree that, as of the Closing Date and upon the Transfer and compliance with the conditions set forth in Section 3 above, the Seller Franchise Agreement and associated guaranties will automatically terminate and neither Seller nor Seller Guarantor shall have any further rights or obligations thereunder except that neither Seller nor Seller Guarantor shall be released from:
 - <u>any obligations to pay money to Franchisor under the Seller Franchise Agreement, any related guaranty agreements, or otherwise in connection with obligations arising prior to the Closing Date (whether known or unknown as of the Closing Date);</u>
 - b. the provisions of the Seller Franchise Agreement that, either expressly or by their nature, survive termination of the Seller Franchise Agreement (including, without limitation, the provisions related to confidential information, post-termination restrictive covenants, indemnification, notice, governing law, jurisdiction and venue, and dispute resolution); or
 - any obligations to pay fund contributions, actuarial-based adjustment payments, Captive SIR Program amounts, or other required payments to Sky Zone SIR Loss Fund, LLC (the "Sky Zone Loss Fund") pursuant to the Master Insurance Program and Captive SIR Program for all incidents that arise or are related to injuries or other claims, demands, or judgments through the Closing Date and regardless of whether the Fee Deposit has been refunded, in whole or in part, in accordance with Section 3.d. above. Each Guarantor shall be liable for all costs and expenses (including reasonable attorneys' fees) incurred by Franchisor or the Sky Zone Loss Fund enforcing this provision of this Consent.
- <u>Release of Franchisor</u>. Seller, Seller Guarantor, and Buyer, and each of them, on behalf of themselves and each of their respective current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, spouses, parent companies, predecessors, affiliates, subsidiaries, successors and assigns, hereby fully and forever unconditionally release and discharge Franchisor and its current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, franchisees, area directors, parent companies, predecessors, affiliates, subsidiaries, successors, and assigns (the "Franchisor Parties"), from any and all claims, demands, obligations, actions, liabilities and damages of every kind or nature whatsoever, in law or in equity, whether known or unknown, or which may hereafter be discovered, accrued, or

sustained in connection with, as a result of, or in any way arising from, any relationship or transaction with Franchisor or the Franchisor Parties, however characterized or described, including but not limited to, any claims arising from Seller's operation of the Park, the Seller Franchise Agreement, the Buyer Franchise Agreement, the Purchase Agreement, or the transactions described in this Consent, or under any applicable state or federal franchise or other law, including the Federal Trade Commission Act, and all applicable Rules of the Federal Trade Commission promulgated pursuant to the Federal Trade Commission Act.

(If the Park is located in California or if Seller, Seller Guarantor, or Buyer (as applicable) is a resident of California, the following shall apply):

Section 1542 Acknowledgment. Seller, Seller Guarantor, and Buyer recognize that he, she, or it may have some claim, demand, obligation, action, liability, defense, or damage against Franchisor or the Franchisor Parties of which Seller, Seller Guarantor, and Buyer are totally unaware and unsuspecting, which he, she, or it is giving up by executing this Consent. Nonetheless, it is the intention of Seller, Seller Guarantor, and Buyer in executing this Consent that this instrument, (i) be and is a general release which shall be effective as a bar to each and every claim, demand, obligation, action, liability, defense, or damage released by Seller, Seller Guarantor, and Buyer, and (ii) will deprive Seller, Seller Guarantor, and Buyer of each and every such claim, demand, obligation, action, liability, defense, or damage and prevent him, her, or it from asserting it against Franchisor or the Franchisor Parties. In furtherance of this intention, Seller, Seller Guarantor, and Buyer expressly waive any rights or benefits conferred by the provisions of Section 1542 of the California Civil Code, which provides as follows:

"A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party."

Seller, Seller Guarantor, and Buyer acknowledge and represent that he, she, or it has consulted with legal counsel before executing this Consent and that Seller, Seller Guarantor, and Buyer understand its meaning, including the effect of Section 1542 of the California Civil Code, and expressly consent that this Consent shall be given full force and effect according to each and all of its express terms and provisions, including, without limitation, those relating to the release of unknown and unsuspected claims, demands, obligations, actions, liabilities, defenses or damages.

(If the Park is located in Maryland or if Seller, Seller Guarantor, or Buyer (as applicable) is a resident of Maryland, the following shall apply):

<u>Any release provided for hereunder shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.</u>

(If the Park is located in Washington or if Seller, Seller Guarantor, or Buyer (as applicable) is a resident of Washington, the following shall apply):

The General Release does not apply with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.

- 9. <u>Non-Disparagement</u>. In consideration of the accommodations provided to Seller, Seller Guarantor, and Buyer, and the concessions made by Franchisor under this Consent, Seller, Seller Guarantor, and Buyer agree not to, and to use their best efforts to cause their respective current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, spouses, parent companies, predecessors, affiliates, subsidiaries, successors and assigns not to, disparage, impugn or otherwise speak or write negatively, directly or indirectly, of Franchisor or the Franchisor Parties, the Sky Zone brand, the Sky Zone franchise system, or any other service-marked or trademarked concept of Franchisor or the Franchisor Parties, or take any other action which would subject the Sky Zone brand to ridicule, scandal, reproach, scorn, or indignity or which would negatively impact the goodwill of Franchisor, the Franchisor Parties, or the Sky Zone brand.
- affiliates, employees, officers, directors, successors, assigns, and other representatives may have been involved in Buyer's purchase of the Interests from Seller, Buyer and Seller have assumed sole and full responsibility for making the final decision to purchase and sell the Interests and each has consulted, or has had the opportunity to consult but, of its own accord, elected not to consult, with its own legal and financial advisors. Buyer further understands that as part of analyzing the purchase of the Interests from Seller, it is Buyer's responsibility to meet with or otherwise gather necessary information from the appropriate parties which may or may not affect Buyer's purchase of the Interests from Seller.
- 11. Additional Documents. Buyer and Seller agree to execute such additional documents as may be necessary to complete the Transfer as contemplated by the Purchase Agreement, the Seller Franchise Agreement, and the Buyer Franchise Agreement.

12. <u>Miscellaneous Provisions</u>.

- a. Confidentiality. Except as reasonably necessary to perform Seller's, Seller Guarantor's, or Buyer's obligations or exercise or enforce Seller's, Seller Guarantor's, or Buyer's rights under this Consent, neither Seller, Seller Guarantor, nor Buyer shall provide or disclose to any third party, or use, unless authorized in writing to do so by Franchisor or properly directed or ordered to do so by public authority or court of competent jurisdiction, any information or matter that constitutes or concerns the terms and conditions of this Consent or that regards any dealings or negotiations with Seller, Seller Guarantor, or Buyer related to this Consent.
- <u>b.</u> Governing Law. This Consent will be construed and enforced in accordance with, and governed by, the laws of the state set forth in the Buyer Franchise Agreement.

- <u>C.</u> Amendment. This Consent may not be modified or amended or any term hereof waived or discharged except in a writing signed by the party against whom such amendment, modification, waiver or discharge is sought to be enforced.
- <u>d.</u> <u>Headings. The headings of this Consent are for convenience and reference only</u> and will not limit or otherwise affect the meaning hereof.
- e. Controlling Provisions. In the event of any conflict between the terms of this Consent and the terms of the Seller Franchise Agreement or the Buyer Franchise Agreement, the terms of this Consent shall control.
- <u>f.</u> Counterpart Signatures. This Consent may be executed in any number of counterparts and sent via facsimile or other electronic transmission, each of which will be deemed an original but all of which taken together will constitute one and the same instrument.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties hereto have caused this Consent to be made effective as of the Effective Date.

FRANCHISOR: SKY ZONE FRANCHISE GROUP, LLC

By:
Name:
Title:
Date*:
*This is the Effective Date
=\frac{1\text{tits is the Effective Date}}{\text{-1}}
SELLER:
[SELLER]
D
By:
Name:
<u>Title:</u>
Date:
Forwarding Address:
Forwarding Email:
1 of warding Email.
1 or warding Emain
SELLER GUARANTOR:
SELLER GUARANTOR:
-
SELLER GUARANTOR:
SELLER GUARANTOR: [SELLER GUARANTOR]
SELLER GUARANTOR: [SELLER GUARANTOR] By:
SELLER GUARANTOR: [SELLER GUARANTOR]
SELLER GUARANTOR: [SELLER GUARANTOR] By:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name: BUYER: [BUYER]
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name: BUYER: [BUYER]
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name: BUYER: [BUYER] By: Name:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name: BUYER: [BUYER] By: Name: Title:
SELLER GUARANTOR: [SELLER GUARANTOR] By: Print Name: By: Print Name: BUYER: [BUYER] By: Name:

Sky Zone Franchise Group, LLC
April 2025 FDD
Ex. I – Agreement and Conditional Consent to Transfer

Exhibit A

Park Improvement Plan

[To be attached]

EXHIBIT J TO THE DISCLOSURE DOCUMENT

FORM OF RENEWAL ADDENDUM

RENEWAL ADDENDUM TO FRANCHISE AGREEMENT

THIS RENEWAL ADDENDUM ("Addendum") is dated as of the last signature date below (the "Effective Date") and is attached to and made a part of that certain Franchise Agreement dated as of the same date hereof (the "New Agreement"), by and between SKY ZONE FRANCHISE GROUP, LLC, a Missouri limited liability company ("SZFG," "we," "us," or "our"), and [FRANCHISEE] ("Franchisee, "you," or "your"). We and you shall collectively be referred to as the "Parties." All capitalized terms set forth in this Addendum and not otherwise defined herein shall have the meanings ascribed to them in the New Agreement.

Recitals

- A. By way of information and background, you have been operating the Sky Zone Indoor Trampoline Park at [ADDRESS] (the "Business"), pursuant to a franchise agreement entered into by the Parties dated [DATE], as it may have been subsequently amended (the "Original Agreement").
- B. The initial term of the Original Agreement is scheduled to expire on [EXPIRATION DATE] (the "Expiration Date").
- <u>C.</u> <u>The Original Agreement provides that, as of the Expiration Date, you have the option to acquire a successor franchise to operate the Business for one additional term of ten (10) years, subject to certain terms and conditions set forth therein.</u>
- <u>D.</u> You have notified us that you wish to exercise your option to acquire a successor franchise for an additional term of ten (10) years ("Renewal") and, in connection therewith, the Parties have agreed to make certain modifications with respect to the terms of the New Agreement, all as set forth in this Addendum.

Agreement

NOW, THEREFORE, for and in consideration of the mutual covenants contained herein and other valuable consideration, receipt and sufficiency of which are acknowledged, the Parties agree as follows:

Amendment of New Agreement.

- 1. Section 4.2: Purchase or Lease of the Location. We have previously approved the lease for the Location as required pursuant to Section 4.2 and therefore waive the requirement for lease review and approval; provided, however, if the lease terms are amended or you enter into a new lease for the Location during the term of the New Agreement, all lease review and approval requirements set forth in the New Agreement shall remain applicable.
- <u>2. Section 4.3: Location Development. The Parties acknowledge that the development and construction of the Business, as described in Section 4.3, has previously been completed.</u>

Sky Zone Franchise Group, LLC April 2025 FDD

- 3. <u>Section 4.8: Commencement Deadline. The Parties acknowledge that the Business has commenced operations as required pursuant to Section 4.8.</u>
 - 4. Section 4.9: Grand Opening Marketing. Section 4.9 is deleted in its entirety.
- <u>5.</u> <u>Section 4.10: Opening Assistance. The Parties acknowledge that on-site opening assistance, as described in Section 4.10, has previously been completed.</u>
- 6. Section 14.1: Upgrades and Renovations. As an inducement to SZFG entering into the New Agreement and this Addendum and as required pursuant to Section 14.1 of the Original Agreement, Franchisee hereby agrees to complete the upgrades and renovations of the Business identified within the "Park Improvement Plan" document attached as Exhibit A hereto, at Franchisee's expense, as required to bring the condition and appearance of the Business into compliance with Franchisor's current Methods of Operation and other System requirements. Franchisee shall complete all requirements and specifications in Exhibit A within the timeframes specified therein, but in no event later than six (6) months from the Effective Date.
- 7. Section 14.1: Successor Franchise Fee. Concurrently with signing the New Agreement and this Addendum, you agree to pay us a successor franchise fee of \$[FEE], in the form of a lump sum payment by EFT or wire transfer. The successor franchise fee is fully earned by us when paid by you and is not refundable under any circumstance.
- 8. Original/New Agreements. As of the Effective Date, the Original Agreement will be deemed terminated, and the operation of the Business will thereafter be governed by the New Agreement.
- Release of Franchisor. You, on behalf of yourself and your respective current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, spouses, parent companies, predecessors, affiliates, subsidiaries, successors and assigns, hereby fully and forever unconditionally release and discharge us and our current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, franchisees, area directors, parent companies, predecessors, affiliates, subsidiaries, successors, and assigns (the "Released Parties"), from any and all claims, demands, obligations, actions, liabilities and damages of every kind or nature whatsoever, in law or in equity, whether known or unknown to you, which you may have against the Released Parties as of the date of this Addendum, or which may thereafter be discovered, accrued, or sustained in connection with, as a result of, or in any way arising from, any relations or transactions with the Released Parties, however characterized or described.

(If the Business is located in California or if you are a resident of California, the following shall apply):

Section 1542 Acknowledgment. It is your intention in executing this Addendum that this instrument be and is a general release which shall be effective as a bar to each and every

claim, demand or cause of action released by you, and you recognize that you may have some claim, demand or cause of action against us or the Released Parties of which you are totally unaware and unsuspecting, which you are giving up by executing this Addendum. It is your intention in executing this instrument that it will deprive you of such claim, demand or cause of action and prevent you from asserting it against us or the Released Parties. In furtherance of this intention, you expressly waive any rights or benefits conferred by the provisions of Section 1542 of the California Civil Code, which provides as follows:

"A general release does not extend to claims that the creditor or releasing party does not know or suspect to exist in his or her favor at the time of executing the release and that, if known by him or her, would have materially affected his or her settlement with the debtor or released party."

You acknowledge and represent that you have consulted with legal counsel before executing this Addendum and that you understand its meaning, including the effect of Section 1542 of the California Civil Code, and expressly consent that this Addendum shall be given full force and effect according to each and all of its express terms and provisions, including, without limitation, those relating to the release of unknown and unsuspected claims, demands and causes of action.

(If the Business is located in Maryland or if you are a resident of Maryland, the following shall apply):

Any release provided for hereunder shall not apply to any liability under the Maryland Franchise Registration and Disclosure Law.

(If the Business is located in Washington or if you are a resident of Washington, the following shall apply):

<u>The General Release does not apply with respect to claims arising under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted thereunder.</u>

- Non-Disparagement. You agree not to, and to use your best efforts to cause your respective current and former owners, agents, principals, officers, directors, shareholders, members, partners, employees, representatives, attorneys, spouses, parent companies, predecessors, affiliates, subsidiaries, successors and assigns not to, disparage, impugn or otherwise speak or write negatively, directly or indirectly, of us or the Released Parties, the Sky Zone® brand, the Sky Zone system, or any other service-marked or trademarked concept of us or the Released Parties, or take any other action which would subject the Sky Zone brand to ridicule, scandal, reproach, scorn, or indignity or which would negatively impact the goodwill of us, the Released Parties, or the Sky Zone brand.
 - 11. <u>Miscellaneous Provisions.</u>
 - (a) Governing Law. This Addendum will be construed and enforced in accordance with, and governed by, the laws of the state set forth in the New Agreement.

Sky Zone Franchise Group, LLC

April 2025 FDD

- (b) Amendment. This Addendum may not be modified or amended or any term hereof waived or discharged except in a writing signed by the party against whom such amendment, modification, waiver or discharge is sought to be enforced.
- (c) <u>Headings. The headings of this Addendum are for convenience and reference only and will not limit or otherwise affect the meaning hereof.</u>
- Controlling Provisions. This Addendum modifies the New Agreement. In the event of any conflict between a provision of the New Agreement and this Addendum, the provisions of this Addendum shall control. Except as amended by this Addendum, the New Agreement is unmodified and in full force and effect in accordance with its terms.
- (e) Counterpart Signatures. This Addendum may be executed in any number of counterparts and sent via facsimile or other electronic transmission, each of which will be deemed an original but all of which taken together will constitute one and the same instrument.

[Signature Page Follows]

IN WITNESS WHEREOF, the parties have executed and delivered this Addendum to be effective as of the Effective Date. SKY ZONE FRANCHISE GROUP, LLC, a Missouri limited liability company By: Name: Title: **DATED*:** (*This is the Effective Date of this Addendum) **FRANCHISE OWNER FRANCHISE OWNER** (IF YOU ARE A CORPORATION, (IF YOU ARE AN INDIVIDUAL AND LIMITED LIABILITY COMPANY, OR **NOT A LEGAL ENTITY): PARTNERSHIP**): Signature **Entity Name** Print Name Signature **DATED:** Name: Title:

DATED:

Exhibit A

Park Improvement Plan

[To be attached]

EXHIBIT K TO THE DISCLOSURE DOCUMENT

REPRESENTATIONS AND ACKNOWLEGMENTS STATEMENT

REPRESENTATIONS AND ACKNOWLEDGMENT STATEMENT

(Not Applicable to Prospective Developers and Franchisees in CA, HI, IL, MD, MN, NY, ND, VA, and WA)

The purpose of this Statement is to demonstrate to Sky Zone Franchise Group, LLC ("Franchisor") that the person(s) signing below ("I," "me" or "my"), whether acting individually or on behalf of any legal entity established to acquire the Development Rights (as applicable to a Multi-Unit Development Agreement) or to acquire the rights to a single Sky Zone® franchise (as applicable to a Franchise Agreement): (a) fully understands that the purchase of franchising rights is a significant long-term commitment, complete with its associated risks, and (b) is not relying on any statements, representations, promises or assurances that are not specifically set forth in Franchisor's Franchise Disclosure Document and Exhibits (collectively, the "FDD") in deciding to purchase any franchising rights from Franchisor (the "Franchise").

In that regard, I represent to Franchisor and acknowledge that:

I understand that buying a Franchise is not a guarantee of success. Purchasing or establishing any business is risky, and the success or failure of the Franchise is subject to many variables such as my skills and abilities (and those of my partners, officers, employees), the time my associates and I devote to the business, competition, interest rates, the economy, inflation, operation costs, location, lease terms, the market place generally and other economic and business factors. I am aware of and am willing to undertake these business risks. I understand that the success or failure of my business	<u>INITIAL:</u>
will depend primarily upon my efforts and not those of Franchisor.	
I received a copy of the FDD, including the Multi-Unit Development Agreement and Franchise Agreement, at least 14 calendar days (10 business days in Michigan) before I executed the Multi-Unit Development Agreement and/or Franchise Agreement (as applicable). I understand that all of my rights and responsibilities and those of Franchisor in connection with the Franchise are set forth in these documents and only in these documents. I acknowledge that I have had the opportunity to personally and carefully review these documents and have, in fact, done so. I have been advised to have professionals (such as lawyers and accountants) review the documents for me and to have them help me understand these documents. I have also been advised to consult with other franchisees regarding the risks associated with the purchase of the Franchise.	<u>INITIAL:</u>
Neither the Franchisor nor any of its officers, employees or agents (including any franchise broker) has made a statement, promise or assurance to me concerning any matter related to the Franchise (including those regarding advertising, marketing, training, support service or assistance provided by Franchisor) that is contrary to, or different from, the information contained in the FDD or as indicated below (write "None" if none provided):	<u>INITIAL:</u>
My decision to purchase the Franchise has not been influenced by any oral representations, assurances, warranties, guarantees or promises whatsoever made by the Franchisor or any of its officers, employees or agents (including any franchise broker), including as to the likelihood of success of the Franchise.	INITIAL:
I have made my own independent determination as to whether I have the capital	<u>INITIAL:</u>

Sky Zone Franchise Group, LLC

April 2025 FDD

Ex. K – Representations and Acknowledgments Statement

I have not received any information from the employees or agents (including any franchise brokerojected or forecasted sales, revenues, income, profincluding any statement, promise or assurance concern	er) concerning actual, average, its or earnings of the Franchise ing the likelihood of my success)	INITIAL
except as contained in the FDD or as indicated below (write "None" if none provided):	
Sign here if you are taking the Franchise as an INDIVIDUAL(S) (Note: use these blocks if you are an individual or a partnership but the partnership is not a separate legal entity)	Sign here if you are taking the CORPORATION, LIMITE COMPANY OR PART	D LIABILITY
<u>Signature</u>	Print Name of Legal	Entity
Print Name: Dated:	By: Print Name: Title: Dated:	
Signature Print Name:	<u>Dated.</u>	
Print Name: Dated: Do not sign this Questionnaire if you are a Mary Maryland. California residents should not complete this Questionnaire if you are a Mary Maryland.		

thereunder.

NOTE TO WASHINGTON RESIDENTS OR FRANCHISES WITH A DEVELOPMENT AREA LOCATED IN WASHINGTON: This Questionnaire does not waive any liability the franchisor may have under the Washington Franchise Investment Protection Act, RCW 19.100, and the rules adopted

EXHIBIT L TO THE DISCLOSURE DOCUMENT

STATE EFFECTIVE DATES

State Effective Dates

The following states have franchise laws that require that the Franchise Disclosure Document be registered or filed with the state, or be exempt from registration: California, Hawaii, Illinois, Indiana, Maryland, Michigan, Minnesota, New York, North Dakota, Rhode Island, South Dakota, Virginia, Washington, and Wisconsin.

This document is effective and may be used in the following states, where the document is filed, registered or exempt from registration, as of the Effective Date stated below:

State	Effective Date						
California	April 12, 2024						
Camorina	(Exempt)Pending						
Hawaii	April 29, 2024Pending						
Illinois	April 12, 2024						
Illinois	(Exempt)Pending						
Indiana	May 23, 2024						
Indiana	(Exempt)Pending						
Maryland	June 26, 2024						
iviai yiand	(Exempt)Pending						
Michigan	April 23, 2024 Pending						
Minnesota	May 17, 2024Pending						
New York	April 12, 2024						
New Fork	(Exempt)Pending						
North Dakota	April 23, 2024						
North Dakota	(Exempt)Pending						
Rhode Island	April 24, 2024						
Kilode Island	(Exempt)Pending						
South Dakota	April 22, 2024Pending						
Virginio	April 27, 2024						
Virginia	(Exempt)Pending						
Washington	April 22, 2024						
Washington	(Exempt)Pending						
Wisconsin	April 19, 2024 Pending						

Other states may require registration, filing, or exemption of a franchise under other laws, such as those that regulate the offer and sale of business opportunities or seller-assisted marketing plans.

EXHIBIT IM TO THE DISCLOSURE DOCUMENT

RECEIPTS

RECEIPT (Your Copy)

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Sky Zone Franchise Group, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

<u>Howa and New York requires require</u> that <u>we give Sky Zone Franchise Group, LLC gives</u> you this disclosure document at the earlier of the first personal meeting or 10 business days (or 14 calendar days in Iowa) before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that <u>we give Sky Zone Franchise Group, LLC gives</u> you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.] If Sky Zone Franchise Group, LLC does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580, and the appropriate state agency identified on Exhibit G.

<u>The franchisor is Sky Zone Franchise Group, LLC located at 86 N. University Avenue, Suite 350, Provo, Utah 84601. Its telephone number is 385-482-1020.</u>

Issuance Date: April 22, 2025

The name, principal business address and telephone number of each franchise seller offering the franchise: <u>Mike Revak, President; Eric Taylor, Chief Development Officer; Karen Luey, Chief Financial Officer; Joe Tenczar Chief Information Officer;</u>
Sky Zone Franchise Group, LLC , at 86 N. University Avenue, Suite 350,
Provo, UtahUT 84601,—Principal Business Address and Phone: 385-482-1020; and————————————————————————————————————
Issuance Date: April 12, 2024

Sky Zone Franchise Group, LLC authorizes the respective state <u>agentsagencies</u> identified on Exhibit G to receive service of process for it in the <u>states listed particular state</u>.

I received a disclosure document from Sky Zone Franchise Group, LLC issued Disclosure Document dated April 1222, 2024 2025, that included the following Exhibits:

EXHIBIT A FRANCHISE AGREEMENT

EXHIBIT B MULTI UNIT DEVELOPMENT AGREEMENT

EXHIBIT C STATE LAW ADDENDA

EXHIBIT D TABLE OF CONTENTS OF MANUAL

EXHIBIT E LIST OF FRANCHISEES

EXHIBIT F FINANCIAL STATEMENTS

EXHIBIT G LIST OF STATE AGENCIES/AGENTS FOR SERVICE OF PROCESS

EXHIBIT H FORM OF PURCHASE ORDER

EXHIBIT I RECEIPTS

Exhibit A Franchise Agreement

Exhibit B Multi-Unit Development Agreement

Exhibit C State Law Addenda

Exhibit D Operations Manual Table of Contents

Exhibit E List of Franchisees

Exhibit F Financial Statements

Exhibit G List of State Agencies/Agents for Service

of Process

Exhibit I Form of Purchase Order Exhibit I Agreement and Conditional Consen Exhibit J Form of Renewal Addendum	to Transfer Exhibit K Representations and Acknowledgments Statemen Exhibit L State Effective Dates Exhibit M Receipts
Date	
Signature	Print Name

Please keep this copy of the Receipt for your records.

RECEIPT (Our Copy)

This disclosure document summarizes certain provisions of the franchise agreement and other information in plain language. Read this disclosure document and all agreements carefully.

If Sky Zone Franchise Group, LLC offers you a franchise, it must provide this disclosure document to you 14 calendar-days before you sign a binding agreement with, or make a payment to, the franchisor or an affiliate in connection with the proposed franchise sale.

<u>Howa and New York requires require</u> that <u>we give Sky Zone Franchise Group, LLC gives</u> you this disclosure document at the earlier of the first personal meeting or 10 business days (or 14 calendar days in Iowa) before the execution of the franchise or other agreement or the payment of any consideration that relates to the franchise relationship. Michigan requires that <u>we give Sky Zone Franchise Group, LLC gives</u> you this disclosure document at least 10 business days before the execution of any binding franchise or other agreement or the payment of any consideration, whichever occurs first.] If Sky Zone Franchise Group, LLC does not deliver this Disclosure Document on time or if it contains a false or misleading statement, or a material omission, a violation of federal law and state law may have occurred and should be reported to the Federal Trade Commission, Washington, D.C. 20580, and the appropriate state agency identified on Exhibit G.

<u>The franchisor is Sky Zone Franchise Group, LLC located at 86 N. University Avenue, Suite 350, Provo, Utah 84601. Its telephone number is 385-482-1020.</u>

Issuance Date: April 22, 2025

The	name,	principal	business	address	and	telephone	number	of	each	franchise	seller	offering	the	fran	chise:
-	Mike Re	vak, Presid	lent; □ <u>Eri</u>	c Taylor,	Chief	- Developme	ent Office	r; 🗖	Karen	Luey, Chie	f Finan	cial Office	er; 🗖	Joe_	Tenczar.
Chie	f Informa	ation Office	e r:												

Name of Franchise Seller:
Sky Zone Franchise Group, LLC , at
86 N. University Avenue, Suite 350 ,
Provo, Utah UT 84601, Principal Business Address and Phone:
385-482-10 20; and
Issuance Date: April 12, 2024

Sky Zone Franchise Group, LLC authorizes the respective state <u>agentsagencies</u> identified on Exhibit G to receive service of process for it in the <u>states listed</u>particular state.

I received a disclosure document from Sky Zone Franchise Group, LLC issued Disclosure Document dated April 1222, 2024 2025, that included the following Exhibits:

EXHIBIT A FRANCHISE AGREEMENT

EXHIBIT B MULTI UNIT DEVELOPMENT AGREEMENT

EXHIBIT C STATE LAW ADDENDA

EXHIBIT D TABLE OF CONTENTS OF MANUAL

EXHIBIT E LIST OF FRANCHISEES

EXHIBIT F FINANCIAL STATEMENTS

EXHIBIT G LIST OF STATE AGENCIES/AGENTS FOR SERVICE OF PROCESS

EXHIBIT H FORM OF PURCHASE ORDER

EXHIBIT I RECEIPTS

Exhibit A Franchise Agreement

Exhibit B Multi-Unit Development Agreement

Exhibit C State Law Addenda

Exhibit D Operations Manual Table of Contents

Exhibit E List of Franchisees

Exhibit F Financial Statements

Exhibit G List of State Agencies/Agents for Service

of Process

Exhibit H Form of Purchase Order	Exhibit K Representations and Acknowledgments Statement
Exhibit I Agreement and Conditional Consent to Transfer	Exhibit L State Effective Dates
Exhibit J Form of Renewal Addendum	Exhibit M Receipts
	
Date	
Dute	
Signature	
5.5	
	Print Name
	1 1111 1 191119

Please sign this copy of the Receipt, date your signature and return it to Mike Revak, President, Sky Zone Franchise Group, LLC, at 86 N. University Avenue, Suite 350, Provo, Utah 84601, mike.revak@skyzone.com.

80105761v1